McGuireWoods LLP 501 Fayetteville Street Suite 500 PO Box 27507 (27611) Raleigh, NC 27601 Phone: 919.755.6600 Fax: 919.755.6699 www.mcguirewoods.com

Mary Lynne Grigg
Direct: 919.755.6573

McGUIREWOODS

October 16, 2020

#### VIA ELECTRONIC FILING

Ms. Kimberley A. Campbell, Chief Clerk North Carolina Utilities Commission **Dobbs Building** 430 North Salisbury Street Raleigh, North Carolina 27603

> DEP Late-Filed Exhibit No. 16 Re: Docket No. E-2, Sub 1219

Dear Ms. Campbell:

Per the request of the North Carolina Utilities Commission during the Duke Energy Progress, LLC ("DEP") evidentiary hearing, enclosed for filing on behalf of DEP in the above-referenced proceeding is Late-Filed Exhibit No. 16.

Please do not hesitate to contact me should you have any questions. Thank you for your assistance with this matter.

Very truly yours,

/s/Mary Lynne Grigg

MLG:kma

**Enclosures** 

Duke Energy Progress, LLC Late-Filed Exhibit No. 16 Docket No. E-2, Sub 1219 Page 1 of 439

Duke Energy Progress, LLC Docket No. E-2, Sub 1219 Late-Filed Exhibit No. 16 October 16, 2020

**Request:** Please provide any comments the Company submitted on behalf of the CCR Rulemaking Process.

**Response:** Duke Energy Progress, LLC ("DEP") provided comments on the 2010 Proposed CCR Federal Rule on November 19, 2010. Those comments are attached. DEP was also a member of the Utility Solid Waste Activities Group ("USWAG") during this time. Accordingly, USWAG's comments on the 2010 Proposed CCR Federal Rule are also attached.



November 19, 2010

Hazardous and Solid Waste Management System
Identification and Listing of Special Wastes
Disposal of Coal Combustion Residuals from Electric Utilities Docket
Attention Docket ID No., EPA-HQ-RCRA-2009-0640
Environmental Protection Agency
1200 Pennsylvania Ave. NW
Washington, DC 20460

#### ELECTRONIC SUBMITTAL ON THE INTERNET

Re: Comments on the Hazardous and Solid Waste Management System; Identification and Listing of Special Wastes; Disposal of Coal Combustion Residuals from Electric Utilities; Proposed Rule; 75 Fed. Reg. 35128 (June 21, 2010);

#### Dear Sir or Madam:

Progress Energy, Inc. (Progress Energy, the Company) is a utility holding company with two electric utility subsidiaries: Carolina Power & LIght Company, d/b/a Progress Energy Carolinas, and Florida Power Company, d/b/a Progress Energy Florida. The two utilities supply electrical power to over 3.1 million customers in North Carolina, South Carolina and Florida.

Progress Energy appreciates the opportunity to comment on the agency's proposed rule entitled "Hazardous Waste Management System; Identification and Listing of Special Wastes; Disposal of Coal Combustion Residuals from Electric Utilities," addressing the management of coal combustion residuals (CCRs) from electric utilities. The Company's two utilities own and operate several electrical power generation plants, including coal-fired generation with a capacity of 7400 megawatts. This rule is expected to have a significant impact on their operations.

The Company appreciates the EPA's recognition of the many states that have well-established and protective regulatory controls for CCRs. Progress Energy Carolinas and Progress Energy Florida have been operating under their states' regulatory programs for a number of years, which have served to not only assure proper management of coal combustion residuals but also encourage their robust reuse. Progress Energy encourages the EPA to work with the states as the agency develops its CCR regulations in order to ease integration with existing state programs.

Progress Energy supports the development of appropriate federal rules for CCRs under the Resource Conservation and Recovery Act's (RCRA) Subtitle D non-hazardous waste program. The Company has a number of suggestions and concerns with EPA's proposal; these are detailed

Duke Energy Progress, LLC Late-Filed Exhibit No. 16 Docket No. E-2, Sub 1219 Page 3 of 439

in the attached comments. Progress Energy opposes any form of Subtitle C regulation of CCRs for a number of reasons that are spelled out in the attached comments.

Progress Energy is a member of the Utility Solid Waste Activities Group (USWAG), an association of over 110 energy industry operating companies and associations, as well as the American Coal Ash Association (ACAA), an organization of more than 150 commercial, industrial and academic members whose mission is advance the responsible management and reuse of coal combustion residuals. The Company supports the comments submitted by both organizations on this rulemaking.

Thank you for the opportunity to comment on this important rulemaking and for your careful consideration of the Company's views. If you have any questions, please do not hesitate to contact me at (919) 546-3775 or Cam Wheeler at (919) 546-4655.

Sincerely,

Caroline Choi

Executive Director, Environmental Services and Strategy

Duke Energy Progress, LLC Late-Filed Exhibit No. 16 Docket No. E-2, Sub 1219 Page 4 of 439

# Comments of Progress Energy, Inc.

Hazardous and Solid Waste Management System;
Identification and Listing of Special Wastes;
Disposal of Coal Combustion Residuals From Electric Utilities;
Proposed Rule

75 Fed. Reg. 35128 (June 21, 2010)

Submitted to
United States Environmental Protection Agency
Docket No. EPA-HQ-RCRA-2009-0640

November 19, 2010

#### **Introduction and Summary**

Progress Energy (the Company) welcomes the opportunity to contribute to the important federal rulemaking for management of Coal Combustion Residues (CCR). We have a responsibility to our customers, shareholders, and the public to reliably and safely provide electricity while meeting our environmental responsibilities and public safety obligations in the most efficient and cost-effective way. Because approximately fifty percent of the Company's electric generation capacity is fueled with coal, any new regulation of coal combustion residues will have a profound effect on our operations. Progress Energy takes our obligation very seriously and provides these comments to the EPA to inform you of our agreements and concerns with, as well as our suggestions to improve, the proposed federal rule.

### Progress Energy supports federal regulation of CCRs under the EPA's Subtitle D Prime option, with adjustments

The Company has evaluated the alternatives available to EPA and believes that the "Subtitle D Prime" option, with appropriate adjustments, offers the best choice. We believe the Subtitle C option is not authorized by RCRA, would cripple the beneficial use of coal combustion residues and impose unnecessary costs on power plants, threatening jobs and increasing electricity costs. The Subtitle D prime option, with appropriate adjustments, will enable EPA to develop an environmentally protective program for CCRs without the adverse consequences of the Subtitle C approach.

While there are several significant issues to be addressed in the Subtitle D proposal, RCRA's Subtitle D framework provides the most appropriate legal basis for development of federal regulations for CCRs. The development of Subtitle D regulations is the appropriate result from EPA's two Reports to Congress and two final Regulatory Determinations under the Bevill amendment declaring that CCRs do not warrant hazardous waste regulation under RCRA Subtitle C.

EPA can develop non-hazardous waste rules for CCRs under RCRA's Subtitle D framework that will be fully protective of human health and the environment. The proposed standards for CCR landfills and surface impoundments under the Subtitle C

and Subtitle D options are virtually identical. Both options would ensure the safe management of CCRs through, among other things, the required use of liner systems, groundwater monitoring and corrective action. A key difference is that the Subtitle D rules can be implemented in a quick, practical and cost-effective manner without the adverse impacts of the Subtitle C option, including devastating the CCR beneficial use market, imposing excessive and costly controls on ancillary power plant operations, and creating a serious shortfall in disposal capacity.

Since the TVA ash release, EPA has been conducting comprehensive structural integrity assessments of utility CCR impoundments across the country. Of the more than 120 CCR impoundments evaluated thus far, *none* of the impoundments has received an "unsatisfactory" rating, meaning that not one impoundment inspected thus far presents an imminent threat of a catastrophic release. The clear answer for EPA is that a Subtitle D program for CCRs can be fully protective of human health and the environment when some adjustments are made as described in these comments below.

Progress Energy believes that any final Subtitle D regulatory program for CCRs must include a mechanism for the administration and implementation of the rules by the states. As EPA has concluded, many states already have well-established and protective regulatory controls for CCRs. Therefore, EPA must work with the states to assure the new federal rules for CCRs are easily integrated with existing state regulations. The Subtitle D rules for CCRs would be directly enforceable by the states and the public under RCRA's citizen suit provision, and violators would be subject to significant penalties. EPA would also retain its imminent and substantial endangerment authority under RCRA. Additionally, the Agency could use its imminent and substantial endangerment authorities under CERCLA, as well as other federal authorities, including the Clean Water Act, to address circumstances where a CCR unit may pose a threat.

To the extent that EPA believes that some degree of direct federal enforceability is necessary under the Subtitle D option, there are options available to EPA to achieve this result. These include, among others, developing a federally-enforceable program

for CCRs under RCRA Subtitle D using the same statutory authorities the Agency employed to promulgate the federally enforceable Subtitle D rules for municipal solid waste landfills (MSWL). The legal, practical and policy rationales of the Subtitle D option over the Subtitle C option are far too great to allow this federal enforcement concern to dissuade EPA from pursuing the Subtitle D approach for CCRs.

Progress Energy endorses federal regulation under Subtitle D of RCRA if our suggested changes to the proposed regulations are incorporated in any final rule. The Company's comments that follow will address our concerns in detail.

### Progress Energy opposes any form of Subtitle C regulation of coal combustion residues.

Under this approach, EPA would subject CCRs destined for disposal to the full range of RCRA Subtitle C hazardous waste regulations. The Agency would regulate CCRs under the Subtitle C hazardous waste rules by listing CCRs as a "special waste" subject to full hazardous waste regulation.

As detailed below, the Subtitle C option would constitute an impermissible reversal of EPA's final 2000 Regulatory Determination that CCRs do not warrant hazardous waste regulation. It would also directly contravene RCRA's plain language establishing an unambiguous statutory process by which EPA can determine whether and how to regulate CCRs under RCRA. Having completed the Bevill determination process for CCRs in 2000 pursuant to the statutorily prescribed procedures, EPA is not at liberty to undo that process. Even assuming that the statute did allow EPA to make a new determination more than ten years after completing the Bevill process, at a minimum, EPA would be obligated to adhere to the statutory procedures that Congress prescribed for making the original regulatory determination, including providing a Report to Congress setting forth its recommendation. EPA has not followed these procedures in the Subtitle C option proposal.

There are additional reasons why EPA cannot choose the Subtitle C option for CCRs. Contrary to EPA's theory that regulating CCRs under RCRA Subtitle C will increase

Duke Energy Progress, LLC Late-Filed Exhibit No. 16 Docket No. E-2, Sub 1219 Page 8 of 439

beneficial use, the docket contains record evidence identifying specific examples of how the mere specter of Subtitle C hazardous waste regulation is already adversely impacting CCR beneficial use projects across the country.

The Subtitle C hazardous waste option would result in the imposition of unduly stringent regulations and staggering compliance costs for the management of CCRs in the range of \$55.3 to \$74.5 billion, which is nearly three to four times EPA's projected Subtitle C compliance costs of approximately \$20.3 billion. Thus, the Subtitle C option directly contravenes Congress' express purpose in enacting the Bevill amendment that EPA avoid the imposition of the onerous economic burden of stringent Subtitle C hazardous waste controls on electric utilities if at all possible.

Contrary to EPA's suggestion that electric utility CCR disposal practices would remain unchanged under the Subtitle C option, at least 15 to 21 million tons of CCRs would enter the Subtitle C hazardous waste commercial disposal market on an annual basis, in contrast to the 2 million tons of hazardous waste currently disposed of in commercial hazardous waste landfills. This influx of Subtitle C wastes into the system would quickly overwhelm the 23 to 30 million tons of total commercial hazardous waste landfill disposal capacity in the nation and create an immediate compliance dilemma for electric utilities.

The Subtitle C hazardous waste option is inconsistent with the views of the other federal and state agencies that Congress directed EPA to consult with in rendering a final regulatory determination for CCRs. All of these agencies have either strongly opposed regulating CCRs under the Subtitle C hazardous waste regulations or have questioned the appropriateness of this option.

Due to the sheer volume and the physical composition of CCRs, *de minimis* volumes of CCRs are inevitably released during normal power generation and subsequent CCR handling operations. If CCRs are regulated as a listed "special waste" subject to full hazardous waste regulation, these *de minimis* releases would constitute improper

hazardous waste disposal, subjecting electric utility coal-fired power plants to potential liability for what would possibly be a perpetual state of RCRA non-compliance.

The Subtitle C option includes the unprecedented and unlawful proposal to extend RCRA Subtitle C hazardous waste jurisdiction to previously closed and/or inactive CCR surface impoundments that do not receive CCRs after the effective date of the final Subtitle C rule. This retroactive application of RCRA violates the plain language of RCRA and legal precedent.

Progress Energy is not alone in its adamant opposition to the Subtitle C option. A bipartisan group of 165 members of Congress, including a majority of the House Energy and Commerce Committee, 45 U.S. Senators, virtually all the states, other federal agencies, municipal and local governments, CCR marketers and beneficial users, labor unions, state public utilities commissions and many other third parties have made clear during this rulemaking process that regulating CCRs under RCRA's hazardous waste program would result in excessive regulation, and, in fact, would be counter-productive because it would cripple the CCR beneficial use industry. In short, the record is clear that regulating CCRs as a listed hazardous waste under RCRA Subtitle C, the most burdensome and extreme option under federal law, is not authorized or warranted. Our comments below will address the above issues in detail.

# EPA cannot promulgate Subtitle C regulations for coal combustion residuals Requirements of the Bevill amendment

The Bevill amendment to RCRA was an outgrowth of Congress' concern that EPA might discourage the development of coal as one of the nation's primary sources of energy if it prematurely and inappropriately subjected CCRs to regulation under Subtitle C of RCRA. The enacted Bevill amendment ultimately incorporated congressional policy directives into RCRA by first suspending EPA's authority to regulate CCRs under Subtitle C of RCRA until EPA completed a study to determine whether such regulation was warranted.

To ensure that all relevant information was available to the agency and to avoid duplication of effort, Congress specifically directed EPA to "review studies and other actions of other Federal and State agencies ... and invite participation by other concerned parties, including industry and other Federal and State agencies." Within six months following completion of the study and a report to Congress on the agency's findings, EPA was required to make a final regulatory determination whether to subject to Subtitle C regulation the CCRs addressed in the study and report to Congress, or determine that Subtitle C regulation of these wastes is not warranted. The Bevill amendment required the Agency to base this final regulatory determination "on information developed or accumulated pursuant to such study, public hearings, and comment…"

It is important to appreciate that Congress was particularly concerned that EPA not make this decision in a vacuum. To that end, Congress emphasized that the Agency was to seek the assistance and cooperation of those with expertise in the field including "other agencies of Government that are aware of the role coal plays in our national energy policy, or of actual disposal and utilization practices." These included the Departments of Agriculture, Commerce, Energy and Interior, as well as the Federal Highway Administration. The input of these federal agencies to assist EPA in reaching a final regulatory determination was a critical component of the Bevill amendment; it was important to Congress that the final determination reflects "a cooperative, informed effort" that met the nation's energy goals.

Also of critical importance to Congress was the Bevill amendment's decision-making process; these provisions reflect Congress' intent that EPA reach all final decisions in a transparent manner that provides the public and Congress with an opportunity to determine whether any decision to regulate CCRs under RCRA Subtitle C was appropriate. In short, EPA could only reach the decision to regulate CCRs under Subtitle C after following a statutorily-prescribed series of steps.

The Bevill amendment reflects Congress' express intent that any regulation of CCRs under RCRA not unduly impede the nation's use of coal. It addressed this concern by cautioning EPA not to impose unnecessary and costly regulations on those industries that use coal to produce electricity for the nation. It also reflects Congress' concern that regulating CCRs under RCRA Subtitle C would discourage, and perhaps end, CCR beneficial use. EPA completed its statutory obligation under the Bevill amendment in May 2000, when the agency issued its final Regulatory Determination concluding that CCRs do not warrant regulation under Subtitle C of RCRA.

EPA published its first Report to Congress on March 8, 1988, concluding that large-volume coal combustion waste streams generally do not exhibit hazardous characteristics under current RCRA regulations, and therefore "EPA does not intend to regulate under [RCRA] Subtitle C fly ash, bottom ash, boiler slag, and flue gas desulfurization wastes." The 1988 Report to Congress also encouraged the utilization of coal combustion wastes to the extent such utilization can be done in an environmentally safe manner.

EPA published its first final Regulatory Determination for CCRs on August 9, 1993. This final Regulatory Determination set forth EPA's conclusion that RCRA Subtitle C regulation of the four large-volume CCRs – fly ash, bottom ash, boiler slag, and flue gas desulfurization waste – was not warranted. Specifically, the Agency concluded that

Regulation under Subtitle C of RCRA is inappropriate for the four waste streams that were studied [fly ash, bottom ash, boiler slag, and flue gas emission control waste] because of the limited risks posed by them and the existence of generally adequate State and Federal regulatory programs. The Agency also believes that the potential for damage from these wastes is most often determined by site- or region-specific factors and that the current State approach to regulation is thus appropriate. Therefore, the Agency will continue to exempt these wastes from regulation as hazardous wastes under RCRA Subtitle C .... EPA will consider these wastes during the Agency's ongoing assessment of industrial non-hazardous wastes under RCRA Subtitle D.

EPA determined that the four large-volume CCRs did not warrant Subtitle C hazardous waste controls, concluding that the inflexible Subtitle C hazardous waste framework was

wholly inappropriate for CCRs. The agency stated that "it is unlikely that Subtitle C would effectively address the problems associated with the four large-volume fossil-fuel combustion wastes [CCRs] without imposing unnecessary costs." EPA found that the rigid nature of the Subtitle C regulatory system would result in excess costs and unnecessary regulation as applied to CCR management units:

A Subtitle C system would require coal combustion units to obtain a [RCRA] Subtitle C permit (which would unnecessarily duplicate existing State requirements) and would establish a series of waste unit design and operating requirements for these wastes, which would generally be in excess of requirements to protect human health and the environment. For example, if such wastes were placed in the Subtitle C universe, all ash disposal units would be required to meet specific liner and monitoring requirements. Since [CCR] sites vary widely in terms of topographical, geological, climatological, and hydrological characteristics (e.g., depth to groundwater, annual rainfall, distance to drinking water sources, soil type) and the wastes potential to leach into the groundwater and travel to exposure points is linked to such factors, it is more appropriate for individual States to have the flexibility necessary to tailor specific controls to the site or region specific risks posed by these wastes.

Following issuance of the 1993 Regulatory Determination, EPA began work on its second report to Congress on the remaining fossil fuel combustion wastes, including CCRs managed with certain low-volume wastes. EPA issued its second Report to Congress in March 1999, reiterating "that [RCRA] Subtitle C is inappropriate to address any problems associated with disposal of these wastes and that the continued use of site and region specific approaches by the states is more appropriate for addressing the limited human health and environmental risks that may be associated with disposal of these wastes." EPA also reiterated its belief that regulating CCRs under RCRA's hazardous waste program would unnecessarily duplicate existing state requirements and would establish a series of waste unit design and operating requirements for these wastes that would most often be in excess of requirements to protect human health and the environment.

EPA completed the statutorily-required Bevill amendment process in May 2000 when it issued its final Regulatory Determination for all remaining fossil fuel combustion wastes, including CCRs managed with certain low-volume wastes. Based on record evidence,

hearings and public comment on the 1999 Report to Congress, EPA once again concluded that CCRs and other fossil fuel combustion wastes did not warrant regulation under RCRA's Subtitle C hazardous waste program. EPA concluded instead that the development of national non-hazardous solid waste regulations under RCRA Subtitle D is appropriate when CCRs are disposed of in landfills or surface impoundments.

In addition to the fact that CCRs rarely exhibit a hazardous waste characteristic, EPA explained that its final determination was based on the improving trends in present CCR disposal and utilization practices, the current and potential utilization of CCRs (*i.e.*, beneficial use) and Congress' admonition in the Bevill amendment against duplication of efforts by other federal and state agencies. EPA pointed to the record evidence that the utility industry has made significant improvements in its waste management practices over recent years and to similar record evidence showing that "[CCR regulatory] programs have, in fact, substantially improved over the last 15 years or so."

In fact, EPA found that the ability of most states to impose specific regulatory controls for coal combustion wastes has increased almost three-fold over the past 15 years. EPA concluded that with the exception of relatively few states, the regulatory infrastructure is generally in place at the state level to ensure adequate management of these wastes. Thus, EPA concluded that "[w]e believe that subtitle D regulations are the most appropriate mechanism for ensuring that these wastes disposed in landfills and surface impoundments are managed safely." Given existing state CCR regulatory capabilities and the evidence that basic controls are increasingly being put in by the state and facilities, EPA was mindful of Congress's directive in section 8002(n) that the agency, in reaching a final determination, consider "[the] action of state and other federal agencies with a view to avoiding duplication of effort." Consistent with this statutory dictate, EPA concluded that Subtitle D controls will provide sufficient clarity and incentive for states to close the remaining gaps in coverage, and for facilities to ensure that their wastes are managed properly.

Also critical to EPA's final determination that CCRs do not warrant regulation under Subtitle C was the Bevill amendment's direction in RCRA section 8002(n) that any final decision take into account the current and potential utilization of such materials. In particular, EPA saw a potential downside to pursuing a Subtitle C approach, citing the concerns expressed by commenters that subjecting any coal combustion wastes to a Subtitle C regime would place a significant stigma on these wastes, the most important effect being that it would adversely impact beneficial reuse. EPA acknowledged the possibility that the Subtitle C approach could have unintended consequences and that the agency was particularly concerned about any adverse effect on the beneficial reuse market for these wastes. EPA concluded that, given its conclusion that the Subtitle D approach should be fully effective in protecting human health and the environment, and given the large and salutary role that beneficial reuse plays for this waste, concern over stigma was a factor supporting the agency's decision that Subtitle C regulation is unwarranted in light of the decision to pursue a Subtitle D approach.

After EPA issued its final 2000 Regulatory Determination, the agency continued to evaluate options for implementing Subtitle D non-hazardous waste regulations for CCRs. As part of this process, in 2005 and 2006, EPA and the U.S. Department of Energy (DOE) jointly conducted a study of current and recent (1) management practices for CCR disposal by industry, (2) state regulatory requirements for CCR management and (3) implementation of state requirements by state authorities. This study examined landfills and surface impoundments that were permitted, built, or laterally expanded between January 1, 1994 and December 31, 2004 to determine what changes, if any, in CCR disposal management practices and state regulatory controls had occurred since the 1994 close of the administrative record for the Bevill Regulatory Determination and EPA's 1999 Report to Congress.

In a report titled "Coal Combustion Waste Management at Landfills and Surface Impoundments, 1994-2004" (DOE/EPA Report), DOE and EPA found that, "[b]etween 1994 and 2004, the amount and quality of environmental controls used at CCW management units appear to have increased." The DOE/EPA Report cited

improvements in the management of CCRs in new or expanded landfills and surface impoundments, including the use of liners and the monitoring of groundwater. EPA and DOE also examined whether state CCR regulatory controls had similarly improved and found that "[i]n eight areas of regulatory control reviewed for this report, more CCWs destined for landfills in the States reviewed had tightened regulatory controls than had relaxed controls between the times data were collected for the 1988 [Report to Congress] and for this report." Consistent with this continuing improvement in the administration of state CCR controls, EPA and DOE also found that the grants of variances from regulatory controls by state regulators had sound scientific support.

Based on the above, the DOE/EPA Report concluded that "[t]he data and analyses documented in this report provide new information that appears to show improved management of CCWs in both landfills and surface impoundments." The findings of the DOE/EPA Report are significant because they provide updated information supporting the conclusions in the final 2000 Regulatory Determination that Subtitle C regulation of CCRs is not warranted due, in part, to the significant and continuing improvements in CCR waste management practices and corresponding improvements in state CCR regulatory programs. Since completion of the DOE/EPA Report in 2007, several states have proposed and/or implemented further enhancements of their CCR controls.

## EPA may not reverse its final regulatory determinations that CCRs do not warrant regulation under RCRA Subtitle C

The fundamental flaw in the proposed Subtitle C option is that it is premised on the notion that EPA is at liberty to reverse its final Regulatory Determinations that CCRs do not warrant regulation under Subtitle C of RCRA. Because the statute does not authorize EPA to reverse these final Regulatory Determinations, EPA may not adopt the Subtitle C hazardous waste option for CCRs. Even assuming for purposes of argument that the statute did authorize such reversal, EPA's proposal to regulate CCRs under Subtitle C does not comply with the Bevill amendment's statutorily-prescribed decision-making process. While Administration and EPA policies may have changed over the last two years, the law governing the decision-making process by which EPA may choose to regulate CCRs under RCRA has not.

Congress established a detailed process by which EPA could determine whether and how to regulate CCRs under RCRA Subtitle C. EPA completed the Bevill process pursuant to a judicially-imposed deadline and determined, in accordance with the congressionally-mandated, decision-making process, that CCRs do *not* warrant regulation under Subtitle C and but *do* warrant regulation under Subtitle D. The statutory decision-making process is complete, and RCRA does not authorize EPA to reverse its final CCR Regulatory Determinations in a manner of its own choosing.

The statute also does not authorize EPA to redo or reverse a final regulatory determination that the Agency has already properly reached pursuant to the amendment's statutorily prescribed procedures and deadlines, or to undertake another round of Bevill decision-making for CCRs. If it did, the statutory process and its associated deadlines for reaching a final Regulatory Determination would have no meaning, and EPA could continually change its mind in an abbreviated fashion, as illustrated by the proposed Subtitle C option. If Congress had wanted to give EPA this authority, it would have done so explicitly. Its absence means that EPA does not have the authority to undertake, through a process and timing of its own choosing, a second regulatory determination.

The statute contemplates only a one-time decision, directing EPA to determine, after the statutorily-required reports to Congress on CCRs, whether Subtitle C regulation for CCRs is warranted. EPA completed this statutorily-prescribed process for CCRs over ten years ago, concluding for a multitude of reasons that CCRs do not warrant regulation under RCRA Subtitle C. The statute simply does not empower EPA to revisit its final decision and render a new determination.

#### EPA has failed to adhere to the Bevill amendment's prescribed process

Even assuming that RCRA could be read as authorizing EPA to reverse its final Regulatory Determination for CCRs, the agency would have to adhere to the statutorily-prescribed procedures that Congress demanded EPA follow in determining whether and

how to regulate CCRs. EPA has strayed beyond the plain language of the statute by proposing to reverse its final Regulatory Determinations for CCRs without following the statutorily-required process it followed for issuing those determinations in the first instance. The agency's proposed Subtitle C option does not comport with the process that Congress demanded EPA adhere to in determining whether and how to regulate CCRs under RCRA Subtitle C. At a minimum, EPA must conduct thorough studies demonstrating a clear need for regulation; provide Congress a report detailing its findings on those studies; conduct public hearings and solicit comments on *that* report to Congress; issue a final determination based on the report to Congress and provide an opportunity for public comment on that report regarding whether and how to regulate CCRs under RCRA.

Prior to making any final regulatory determination for CCRs, the statute directs EPA to fully inform Congress of the Agency's initial recommendation. This is not a mere procedural formality. As Representative Bevill emphasized when introducing his amendment, it was especially critical that EPA *not* make the final regulatory determination in isolation, but that Congress and relevant federal agencies be involved in the decision-making process. The failure of EPA to prepare a report to Congress outlining fully and comprehensively the Agency's recommendation to regulate CCRs under RCRA's hazardous waste program is at odds with the plain language of the statute. This failure deprives Congress of its statutorily-prescribed right to be fully informed of any such recommendation so that it can evaluate and respond, as appropriate, to the Agency's proposed course of action *before* EPA makes a final regulatory determination and develops federal regulations.

#### The proposed Subtitle C option is inconsistent with the Bevill amendment

Even assuming, for purposes of argument, that EPA was authorized to revisit its final Bevill Regulatory Determination for CCRs *and* that it followed the statutorily-prescribed procedures for issuing a new regulatory determination, the Subtitle C option would nonetheless be unlawful because it:

- is inconsistent with the Bevill amendment's directive that any final determination not impair CCR beneficial use;
- is at odds with Congress' express directive that any regulatory program for CCRs not discourage the use of coal by imposing undue regulatory costs on the management of CCRs;
- disregards the nearly unanimous views of the federal and state agencies whom the Bevill amendment directs EPA to consult with in reaching a final regulatory determination - that Subtitle C regulation is not warranted for CCRs; and
- 4. is inconsistent with EPA's own findings that state programs can properly address CCR disposal practices under RCRA Subtitle D.

In addition, the Subtitle C option is arbitrary and capricious because EPA has not provided a reasoned explanation for reversing its previous conclusion that application of the inflexible Subtitle C regulatory program to CCRs will result in excessive and unnecessary regulation. EPA's theory that regulating CCRs under RCRA's Subtitle C will actually *increase* CCR beneficial use is one of the most incorrect aspects of the CCR proposal. This position ignores actual marketplace evidence, which already is showing declines in CCR beneficial use just from the threat of Subtitle C regulation; EPA's own prior positions; the nearly unanimous position of other federal agencies and state environmental protection agencies; the views of standards-setting organizations that have no economic, political or other vested interests in the outcome of this rulemaking; the long-standing view of Congress; and the position of electric utilities, the entities that produce the CCRs for the beneficial use market.

EPA identifies the array of valid concerns raised by industry, utilities, states, standard setting-organizations and others that Subtitle C regulation will effectively cripple the CCR beneficial use market, but offers remarkably little, if any, response to these points. In fact, EPA expressly acknowledges that, if it pursues the Subtitle C option, some states, such as Florida, "would likely prohibit the beneficial use of CCRs." Instead of providing substantive responses to these concerns, EPA persists in its position that it

understands the economic drivers in the CCR beneficial use market better than those actually engaged in the business. EPA argues that driving up the costs of CCR disposal as high as possible (as would happen under the Subtitle C option) would force more CCRs into the beneficial use market. EPA's theory, however, is based on flawed examples, and ignores the first-hand knowledge of those involved in the CCR beneficial use market who have consistently provided EPA with information and data demonstrating that any form of Subtitle C regulation will cripple the beneficial use of CCRs. EPA's theory also fails to consider whether the marketplace can absorb the vastly increased volume of CCRs that it speculates will be beneficially used under the Subtitle C option. Both the CCR producers and the CCR beneficial use market have explained that this is not the case.

EPA also argues that any stigma resulting from Subtitle C regulation will be "significantly reduced by listing CCRs as a 'special waste,'" attempting to deflect attention from the reality that these materials would be subject to hazardous waste regulation from their point of generation until their ultimate disposal. EPA's "special waste" label will do nothing to reduce the potential legal liabilities of using a material in a commercial or residential setting that is a listed waste under Subtitle C and subject to hazardous waste regulation when disposed of. RCRA makes clear that only materials that are "hazardous" waste listed or identified" under Subtitle C are subject to the statute's hazardous waste controls, including, among other things, its permitting, substantive design and operating standards, and corrective action requirements. In evaluating the potential impacts of Subtitle C regulation on CCR beneficial use, it is important to note that the sponsors of the Bevill amendment cautioned EPA against regulation of CCRs under Subtitle C for precisely the same reasons being offered today by opponents of the Subtitle C option: subjecting CCRs to the stringent Subtitle C program is certain to have adverse impacts on the use of these materials. Representative Bevill was clear that one of the primary purposes of his amendment was to reinforce the "national commitment to encourage reuse of such materials as fly ash."

Considering the above in addition to the record evidence already assembled in this rulemaking proceeding, it is clear that regulating CCRs under any Subtitle C hazardous waste regime, even with the "special waste" label, will have a significant adverse impact on CCR beneficial use. EPA itself concluded in its 2000 Regulatory Determination that the stigma from regulating CCRs under RCRA's hazardous waste program could harm CCR beneficial use. Importantly, the Subtitle C option considered and rejected in 2000 was a "contingent Subtitle C" option, which meant that CCRs managed in accordance with Subtitle D-like standards would not be classified as a hazardous waste; only CCRs not properly managed under the Subtitle D standards would have been regulated as listed hazardous wastes. EPA recognized that, even under that approach, the mere possibility of subjecting CCRs to hazardous waste regulation would have an adverse stigmatizing effect on CCR beneficial use.

The Association of State and Territorial Solid Waste Management Officials ("ASTSWMO") has warned EPA repeatedly since initiation of this rulemaking that "[t]he most compelling reason not to impose Subtitle C regulations [on CCRs] is that the beneficial use of CCB has been very successful. The 'hazardous' label of Subtitle C would be detrimental to State CCB beneficial use programs." Just last fall, after polling every state in the nation on the potential impacts on state programs of regulating CCRs under Subtitle C, ASTSWMO wrote EPA to caution that "[r]egulation under RCRA Subtitle C has the potential to put an end to many beneficial uses for [coal combustion waste] CCW. In most States, a primary requirement for a beneficial use determination is that the waste *not* be hazardous. Labeling CCW a hazardous waste will have a detrimental [impact] on its beneficial use."

Real-world evidence of the adverse impacts on CCR beneficial use stemming from just the *prospect* of Subtitle C regulation is being reported by electric utilities. Just as CCR marketers and end-users are already feeling the adverse impacts of this proposal, so too have utilities become painfully aware of the stigma associated with the proposal.

### EPA's examples of increased beneficial use despite hazardous waste labels are without merit

Rather than respond to the overwhelming evidence from federal and state agencies and the entire spectrum of entities involved in the beneficial use market that the Subtitle C option will devastate the CCR beneficial use market, EPA simply points to certain materials for which it claims that the "hazardous waste" label has not impaired beneficial use. These examples are largely irrelevant. To the extent they bear on the current proposal, these examples actually provide further evidence that regulating CCRs under Subtitle C will have an adverse stigmatic effect on CCR beneficial use.

EPA points to several other materials, including chat, electric arc furnace dust, electroplating wastewater sludge, and spent solvents, as examples of where regulation under Subtitle C has not harmed the materials' beneficial use. However, EPA did not list used oil as a hazardous waste because of concerns that the listing label would adversely impact used oil recycling. EPA's examples do nothing to support the proposition that listing CCRs under Subtitle C would actually increase beneficial use of CCRs. The agency's examples cannot overcome the direct evidence already provided by federal and state agencies and those businesses directly involved in CCR beneficial use markets that make clear that the Subtitle C option would have a devastating impact on CCR beneficial use.

#### EPAs cost estimate excludes important and impactful considerations.

A recent economic analysis prepared by the Electric Power Research Institute (EPRI) examines the full range of Subtitle C compliance costs for electric utilities, including the costs of retrofitting upstream CCR handling units (*e.g.*, hoppers, tanks and conveyor systems) to meet EPA's hazardous waste design and operating requirements (see EPRI, 2010, "Cost Analysis of Proposed National Regulation of Coal Combustion Residuals from the Electric Generating Industry," 1022296 at 4-1; EPRI Cost Report). EPA acknowledges that this latter cost element – the need to upgrade and/or retrofit upstream CCR handling units – is a necessary cost component of the Subtitle C option,

but the agency's Subtitle C cost estimate does not include this important cost component because EPA lacked information "on baseline CCR storage practices."

The EPRI Cost Report also includes another important cost element of Subtitle C compliance not evaluated by EPA – the construction costs for new wastewater treatment facilities to replace the wastewater treatment function served by CCR surface impoundments that would have to close under the Subtitle C option. In its cost analysis, EPA fails to recognize that CCR surface impoundments often serve a dual function at power plants: to manage CCRs and to serve as a component of the plant's wastewater treatment system. Mandatory closure of the CCR surface impoundments under the Subtitle C option will require many power plants to invest substantial resources in designing and constructing new wastewater treatment facilities.

In addition, the EPRI cost analysis includes the increased disposal costs that will be incurred by electric utilities under the Subtitle C option due to the need to send CCRs to off-site commercial hazardous waste disposal facilities. As detailed below, EPA mistakenly assumes that, under the Subtitle C option, electric utility CCR disposal options will remain unchanged. However, for a multitude of reasons, many utilities will not be able to permit new Subtitle C landfills onsite and will be compelled to transport CCRs currently managed onsite to offsite commercial, hazardous waste disposal facilities.

All told, the EPRI Cost Report estimates that the full compliance costs for electric utilities will be at least \$55.3 to \$74.5 billion. These numbers are in contrast to the \$20.3 billion in compliance costs estimated by EPA in its Regulatory Impact Analysis ("RIA"). Further, even the staggeringly high compliance numbers included in the EPRI Cost Report are predicated on a number of conservative assumptions regarding the ability of power plants to meet the Subtitle C requirements in a full and timely fashion. In fact, the real world compliance costs may be far higher.

Additionally, an underlying cost analysis incorporated into the EPRI Cost Report projects that these costs would be sufficiently high to render some coal-fired power plants, especially smaller generating units, not economically viable, forcing their early closure (see EOP Group, Inc., "Cost Estimates for the Mandatory Closure of Surface Impoundments Used for the Management of Coal Combustion Byproducts at Coal-Fired Electric Utilities," (November, 2010)). Thus, instead of promoting the use of coal by selecting the most cost-effective regulatory option for CCRs, EPA's Subtitle C option will actually discourage the use of coal by rendering the use of coal at some facilities uneconomical. This result is in direct contravention of the express congressional purpose behind the Bevill amendment, and such an approach is unsupportable, given that EPA readily acknowledges that the controls for CCR landfills and surface impoundments are virtually identical.

Given that the substantive disposal controls under the Subtitle D option would be just as effective in protecting human health and the environment as those provided for under the Subtitle C option at far less cost and without the severe impacts on beneficial use, the price of power and the use of coal, the Subtitle C option is directly inconsistent with the stated purpose of the Bevill amendment.

#### EPA does not justify a change in position that would use Subtitle C.

In addition to the above, EPA's selection of the Subtitle C option would be arbitrary and capricious because the agency has provided no explanation as to why its earlier conclusions that the Subtitle C option would result in excessive and inappropriate regulatory controls for CCRs are no longer true. The EPA's 1993 final Regulatory Determination concluded that the four large volume CCRs – fly ash, bottom ash, boiler slag, and flue gas desulfurization waste – do not warrant hazardous waste regulation. EPA emphasized that the inflexible Subtitle C hazardous waste framework was inappropriate for CCRs, explaining "that it is unlikely that Subtitle C would effectively address the problems associated with the four large-volume fossil-fuel combustion wastes [CCRs] without imposing unnecessary controls."

This was an important conclusion. As the agency explained in outlining the factors that it would evaluate in making a final regulatory determination, "the special status of the [CCR] waste requires that the Agency consider the impacts to the industry that regulation under Subtitle C would create in making a decision to regulate the waste as hazardous." EPA found that the rigid nature of the Subtitle C regulatory system would result in excess costs and unnecessary regulation as applied to CCR management units. After acknowledging that CCRs rarely exhibit a hazardous waste characteristic and thus would not be subject to hazardous waste controls unless they were listed as a hazardous waste, EPA emphasized that the listing approach would be inappropriate and excessive.

Having previously concluded that the Subtitle C approach would be inappropriate because, among other reasons, it would result in the over-regulation of CCRs in contravention of the Bevill amendment, EPA provides no explanation in the proposal as to what, if anything, has changed to suggest that the imposition of these very same Subtitle C controls will no longer result in inappropriate and unnecessary costs on the utility industry. As EPA itself noted earlier, "the special status of the [CCR] waste requires that the agency consider the impacts to the industry that regulation under Subtitle C would create in making a decision to regulate the waste as hazardous." There is no reasoned explanation for EPA's about-face on this issue.

The excessive costs and unduly burdensome regulation of CCRs under the proposed Subtitle C option are at least as great today as they were ten years ago when EPA rejected application of the Subtitle C hazardous waste rules for CCRs. As EPA has failed to explain why regulation of CCRs under Subtitle C is now appropriate, having expressly found earlier that such regulation would be both "inappropriate" and in "excess of requirements" for CCRs, any final Subtitle C regulation would be arbitrary and capricious.

# The Subtitle C option is inconsistent with the views of other federal and state agencies that EPA was to consult with in reaching a Final regulatory determination

EPA's proposed Subtitle C option is opposed by virtually every federal and state agency that has weighed in on this rulemaking. This broad opposition to the Subtitle C option is significant because the Bevill amendment and its legislative history make clear that the informed views of these agencies were to be considered by EPA in the issuance of any final regulatory determination for CCRs. Given this across-the-board opposition to the Subtitle C option by the groups Congress directed EPA to consult with as part of the decision-making process for CCRs, selection of the Subtitle C option would be inconsistent with the statute.

The statute specifically directs that EPA, in preparing its study on CCRs, "shall, as [the Administrator] deems appropriate, review studies and other actions of other Federal and State agencies concerning such material and invite participation by other concerned parties, including industry and other Federal and State agencies, with a view toward avoiding duplication of effort." Representative Bevill elaborated on this collaborative process by emphasizing that "EPA's studies should not proceed in a vacuum," but rather that EPA was to reach out to those with expertise in the field.

The Association of State and Territorial Solid Waste Management Officials ("ASTSWMO") expressly rejected EPA's proposed regulation of CCRs under Subtitle C:

To artificially classify [coal combustion byproducts (CCB)] as hazardous will needlessly limit the management options for both the CCBs and other wastes legitimately classified as hazardous which will be competing with CCBs for limited hazardous waste disposal capacity, while not producing any great degree of environmental protection. Transportation, manifesting and licensing requirements for CCBs as a listed hazardous waste are excessively burdensome without sufficient evidence of a benefit. It would be more appropriate to regulate and manage CCBs using design and operation standards specified for Subtitle D programs except in cases where a particular source material is deemed hazardous upon testing for characteristics.

The most compelling reason not to impose Subtitle C regulations is that the beneficial use of CCB has been very successful. The "hazardous" label of Subtitle

C would be detrimental to State CCB beneficial use programs[.] Regulation under RCRA Subtitle C has the potential to put an end to many beneficial uses for CCB.

In short, the statute and its legislative history make clear that that EPA was to "seek the assistance and cooperation of those most expert in this field" in making a final regulatory determination. EPA's selection of the Subtitle C option would be directly contrary to the unanimous opinion of these informed parties and to Congress' directive that EPA produce a "collaborative, informed" regulatory determination for CCRs.

#### The damage cases do not support the Subtitle C option

EPA suggests that the newly identified "damage cases" may warrant reversal of the 2000 Regulatory Determination, resulting in the regulation of CCRs under RCRA Subtitle C. However, the new damage cases, which are limited almost exclusively to older, unlined units (as were the original damage cases), in no way undermine EPA's original decision in the 2000 Regulatory Determination that the best means to address the damage cases is to "develop national Subtitle D regulations."

Some of the damage cases identified by EPA have had environmental impacts. At the same time, the proposed Subtitle D requirements – including the liner, groundwater monitoring, corrective action and structural stability requirements – will be fully effective in addressing EPA's concerns about the damage case sites without the attendant adverse impacts of the Subtitle C option. Therefore, the new damage cases justify neither the reversal of EPA's 2000 final Regulatory Determination nor the resulting regulation of CCRs under Subtitle C. The proven damage cases represent only a subset of CCR management units, and EPA's concerns about this distinct subset of units will be adequately addressed under the Subtitle D option.

#### The damage cases represent a subset of older, unlined CCR disposal units

At the time of the final 2000 Regulatory Determination, EPA had identified 14 "proven" damage cases and 36 "potential" damage cases. Since that time, EPA has identified 13 additional proven damage cases and four more potential damage cases, bringing the total to 27 proven cases and 40 potential damage cases. Virtually all of these damage

cases involve unlined units that began operation prior to 1990; these cases do not involve the newer, lined units that have commenced operation since that time.

It is inappropriate for EPA to include potential damage cases in its evaluation of the appropriate regulatory option for CCRs. The Bevill amendment provides that, in considering the "documented cases" of danger from the management of CCRs, the agency was only to evaluate those instances where such danger "has been proved." EPA has acknowledged this statutory limitation, observing in its final 2000 Regulatory Determination that, although the potential damage cases may pose a "potential danger" to human health and the environment in some circumstances, they do not "satisfy the statutory criteria of documented, proven damage cases because damage to human health or the environment has not been proven..." It is inconsistent with the plain language of the statute for EPA to include the potential damage cases in its assessment of the "documented cases" of danger to human health and the environment from the management of CCRs. EPA cannot use the potential damage cases as the foundation for a reversal of its final 2000 Regulatory Determination.

It is important to note that even the proven damage cases are limited to only a subset of the universe of CCR disposal units; namely, older, unlined units. Of the 16 proven damages cases identified in the 2007 Notice of Data Availability (NODA) that involved groundwater contamination, 15 of the 16 units were unlined (with the remaining unit being a surface impoundment where it was unclear whether the unit was lined or unlined). The only other proven damage case involving groundwater contamination identified since the 2007 NODA is the BBBS Sand and Gravel Quarries in Gambrills, MD, also an unlined site.

Even assuming for purposes of argument that it is appropriate to include the potential damage cases in this evaluation, the vast majority of impacts to groundwater from the potential damage cases are similarly from unlined facilities. According to EPRI's comprehensive evaluation, of the combined total of damage cases identified in the 2007 NODA, only six (10%) had liners. Since issuance of the NODA, the only other potential

damage case identified by EPA is the Battlefield Golf Course in Chesapeake, VA, an unlined site.

In addition, virtually all of the damage cases, both proven and potential, involve older units that commenced operations prior to implementation of what is now the common practice of installing liners in CCR disposal units. Of the 63 proven and potential damage cases identified in the 2007 NODA, approximately two-thirds of the facilities for which operating periods could be established began operations prior to 1976, and all began operations prior to 1990. The data are even more striking when this analysis is properly limited to the proven damage cases; all of the 24 proven damage cases identified in the NODA commenced operations between 1948 and 1983, with "the median opening year of 1967."

As EPA itself reported in the joint DOE/EPA report, the vast majority of CCR disposal units that have been constructed since 1990 are "better designed, in that they are lined and have installed groundwater monitoring systems, and therefore the total percentages of unprotected units have been reduced." In fact, the joint DOE/EPA Report found that "[t]he vast majority (98%) of the 56 identified units (both landfills and surface impoundments)" built or expanded from 1994 to 2004 have liners." The damage cases involve older and almost exclusively unlined units and represent only a segment of the total universe of CCR disposal units.

Not only do the damage cases involve only a subset of CCR disposal units, legitimate questions exist regarding whether even this limited subset of units are truly "damage" cases. EPRI's Damage Case Report identifies a number of important flaws with the proven and potential damage cases identified in the 2007 NODA, including, among other things, that (1) characterization of at least 13 of the sites as "damage" cases was questionable because of the lack of any long-term impacts that could be attributed to CCRs; (2) nearly two-thirds of the 54 groundwater damage case facilities were located at sites where there was a low potential for downgradient receptors; and (3) off-site exceedances of maximum contaminant levels (MCLs) attributable to CCR impacts, a

fundamental criterion for being classified as a proven damage case, were observed *in* only three cases of the combined total of 54 proven and potential groundwater damage cases.

These findings underscore the fact that the damage cases, in particular the proven damage cases, represent only a fraction of CCR disposal units and that the limited number and nature of these cases do not warrant Subtitle C regulation. The lack of justification for CCR regulation under Subtitle C is underscored by the fact that the Subtitle D controls can fully address EPA's concerns with these facilities.

### The proposed Subtitle D controls will prevent future damage cases, and existing damage cases are being effectively remediated

EPA's primary concerns with the identified damage cases involve surface and groundwater contamination from CCR disposal units and the structural failure of CCR surface impoundments. These are precisely the risks that the proposed Subtitle D controls are designed to address.

EPA has stated repeatedly that the standards for CCR disposal units are identical under the Subtitle C and Subtitle D options, with both proposed regulatory programs requiring liners and leachate collection systems for new disposal units and lateral expansions of existing units, as well as the retrofitting of existing CCR impoundments. Further, under the Subtitle D option, existing CCR disposal units would be subject to groundwater monitoring standards that will allow for the early detection of any groundwater contamination from both lined and existing unlined units. The Subtitle D option would impose structural integrity requirements on all CCR surface impoundments, thus addressing EPA's concerns with the potential structural failure of such units that served as the basis for identifying the TVA Kingston facility and Martin's Creek Power Plant as proven damage cases.

While the Subtitle D regulations will be effective in preventing future damage cases, the existing damage cases already are being properly remediated under state and federal

remediation programs. As EPA acknowledged with respect to the proven damage cases identified in the 2000 Regulatory Determination, all of these cases are being addressed adequately by the states (and in two cases, by EPA under the Superfund program), which EPA explained shows "the effectiveness of states responses when damages were identified." In addition, the newly identified damage cases are being effectively remediated. The EPRI Damage Case Report includes a detailed analysis of the remediation efforts at these additional sites and found that "[a]vailable information indicates that remediation is complete or underway at *all* sites where remediation was required." Indeed, of the 16 proven cases of damages to groundwater, EPA has confirmed that "corrective actions have been completed in seven cases and are ongoing in the remaining nine cases."

In short, existing damage cases are already being adequately addressed, and the proposed Subtitle D regulations will prevent future damage cases. Therefore, there is no reason for EPA to pursue the Subtitle C option when the Subtitle D option will effectively address the issues associated with the damage cases. Given the underlying goal of the Bevill amendment that EPA avoid the Subtitle C regulation of CCRs "if at all possible" (*EDF v. EPA*, 852 F.2d at 1314), there is no basis for EPA to rely on the damage cases as the basis for pursuing the Subtitle C option.

## The alleged damage cases presented by environmental groups are not proven and cannot be relied upon by EPA in this rulemaking

EPA references the allegations of additional damage cases by certain environmental organizations, including assertions made by the Environmental Integrity Project ("EIP") and Earthjustice in a February 24, 2010 report (75 Fed. Reg. at 35148) and, more recently, an August 26, 2010 report from the same two environmental organizations and the Sierra Club. EPA acknowledges that it has not had time to review these allegations, though it promises to review this information and encourages other stakeholders to also comment on these allegations.

As an initial point, it is inappropriate for EPA to rely on the allegations of these environmental groups in any final rulemaking for CCRs before the agency has fully evaluated the allegations and provided an opportunity for public hearing and comment on EPA's initial conclusions. Indeed, this is precisely what the Bevill amendment requires. Before making any final regulatory determination, EPA is required to submit its initial recommendation, along with the reasons for this recommendation, in a report to Congress, and to provide an opportunity for public comment and hearings on these initial findings. For EPA to base a new regulatory determination on the mere allegation of additional damage cases without first providing the public an opportunity to review and comment on EPA's initial conclusions regarding the allegations would be inconsistent with the dictates of the Bevill amendment.

Past experience underscores the importance of this statutorily-prescribed vetting process, as many of the former allegations of damage cases identified by environmental groups have been shown to be without merit and have been dismissed by EPA as unsubstantiated and/or incorrect. For example, prior to initiating this rulemaking, EPA gathered or received information on 135 alleged damage cases, many of which were submitted to EPA by the same environmental organizations referenced above. After evaluating these allegations, close to half (62) were dismissed "because there was little or no information supporting the concerns identified." Another six were outside the scope of the rulemaking. Of the remaining 67 allegations, 24 were identified as proven damage cases, but this included the 14 proven cases already identified in the 2000 Regulatory Determination. Less than two-tenths of the cases "alleged" to have been damage cases have, in the end, been determined by EPA to be "proven" damage cases.

The same over-counting of damage cases exists in the more recent damage allegations submitted by EIP, Earthjustice and the Sierra Club. An initial analysis by EPRI of certain of these allegations reveals that there are similar factual errors and/or lack of a factual basis for classifying these sites as damage cases. The Bevill amendment prohibits EPA from relying on the new allegations of damage cases without first fully examining these

allegations and providing the public with an opportunity to evaluate and comment on the agency's initial findings. Given that the vast majority of similar allegations in the past have proven to be without merit, this type of careful review is necessary to ensure that unfounded and/or incorrect charges of new damage cases are not used as the basis for any final decisions.

#### The Subtitle C option will result in serious disposal capacity shortfalls

As the utility industry and others have repeatedly warned EPA throughout this rulemaking process, a combination of factors associated with the Subtitle C regulation of CCRs will create a serious shortfall of onsite CCR disposal capacity, requiring more CCRs to be shipped offsite for disposal in commercial Subtitle C facilities. If only a fraction of the estimated 76 million tons of CCRs disposed of annually (see 75 Fed. Reg. at 35158) is diverted to the commercial hazardous waste disposal market, the commercial market will be overwhelmed, leaving CCR generators as well as thousands of other generators of non-CCR hazardous waste without any legally viable disposal options for their wastes.

EPA acknowledges that only two million tons of hazardous waste are disposed of annually in hazardous waste landfills. The total current national commercial hazardous waste disposal capacity is between 23.5 and 30.3 million tons. Based on a survey of electric utilities evaluating how the industry would dispose of CCRs under the Subtitle C option, the EPRI Cost Report found that at approximately 11.5% of CCRs would be diverted to the commercial Subtitle C hazardous waste disposal market. This correlates to approximately 15 to 21 million of CCRs annually being disposed of in a Subtitle C commercial landfill, in comparison to the approximately two million tons of hazardous waste currently disposed of in commercial hazardous waste landfills. This volume would quickly overwhelm the overall nationwide Subtitle C landfill capacity of 23.5 to 30.3 million tons. The states already have raised these concerns, warning EPA that they will not have the ability to permit the necessary additional hazardous waste disposal capacity in a timely fashion. These findings disprove EPA's assumption that owners and operators of CCR landfills will be willing to, or capable of, converting all their existing

CCR disposal units to permitted Subtitle C disposal units and that there will be no shift in "disposal patterns in a way that substantially increases the disposal of CCRs off-site from generating facilities to commercial hazardous waste landfills."

The results from the EPRI survey should not be surprising, as a multitude of factors make clear that subjecting CCRs to Subtitle C regulation will quickly overwhelm Subtitle C disposal capacity. First, the Subtitle C option effectively mandates the phase-out of CCR surface impoundments, meaning that approximately thirty million tons of CCRs currently managed annually in surface impoundments will have to be converted to management in landfills. In turn, utilities must either obtain permits for this additional landfill capacity or divert the additional volumes of CCRs to the commercial market. A significant percentage of existing CCR landfills will be unable to obtain Subtitle C permits for a variety of reasons, including the intense public opposition to the siting of any new hazardous waste landfills. This strong opposition to the siting of any new CCR landfills by activist groups has already been well-documented during the EPA-sponsored public hearings on this proposal.

Additionally, some states, including Florida and Kansas, prohibit the siting of hazardous waste landfills, which would foreclose any opportunity for the continued onsite or nearby disposal of CCRs. Therefore, it is simply unrealistic for EPA to assume that all existing CCR landfill disposal capacity, or even a majority of such capacity, could be readily converted to RCRA Subtitle C units.

Further compounding the disposal capacity problem is that certain CCR beneficial uses currently employed in the market will be prohibited, creating additional volumes of CCRs that must be disposed of. For example, EPA takes the position that many large-scale structural fill projects (which would fall within EPA's broad description of "large scale fill operations") do not constitute an excluded beneficial use. As a result, the millions of tons of CCRs, which have historically been beneficially used for these fill operations, would be diverted to the Subtitle C disposal market, increasing demand on the already strained disposal market. EPA may suggest, in response, that CCR beneficial use

opportunities will increase under the Subtitle C option, but this theory has already been uniformly rejected by the states, the CCR beneficial use market, and other independent third parties. Despite any additional appeal of CCR beneficial uses that may arise in the context of a severe disposal capacity shortage, numerous and significant hurdles to CCR beneficial use, including the stigmatizing effect Subtitle C regulation will have on CCRs, will severely limit the opportunities for beneficial use of CCRs. In short, promulgation of any Subtitle C option will mean that tens of millions of tons of CCRs would have to be disposed of on annual basis; there simply will not be enough Subtitle C disposal capacity to absorb these materials.

In sum, the record evidence shows that, contrary to EPA's belief, subjecting CCRs to Subtitle C regulation will quickly overwhelm the Subtitle C disposal market. The foreseeable shortfall in offsite CCR disposal capacity compounded with the inability to manage CCRs onsite will force utilities to either store CCRs onsite, in non-compliance with RCRA's permit requirements (subjecting the utility to continuing daily penalties), or cease operations to avoid producing CCRs. These alternatives are not acceptable or warranted.

#### Other issues related to Subtitle C regulation reveal its significant burdens

Another serious problem with Subtitle C regulations is the agency's proposed elimination of management options to (1) dispose of and store CCRs in surface impoundments and (2) allow for the placement of wastewaters in impoundments that contain CCRs. EPA suggests in the preamble to the proposal that the continued wet handling of CCRs in surface impoundments can be conducted in a manner that is protective of human health and the environment provided that the proposed composite liner requirements are met. However, certain components of the land disposal restrictions that the agency intends to apply to all CCR surface impoundments will virtually eliminate the use of surface impoundments, even those that employ the requisite liner systems. The application of these restrictions has not been adequately justified, and imposes unnecessary burdens on the utility industry.

In the preamble to the CCR Subtitle C proposal, EPA suggests that the continued wet handling of CCRs in surface impoundments would be authorized and that the agency would establish composite liner systems for surface impoundments and landfills. The proposed regulatory text, however, does not include the composite liner requirements for surface impoundments. EPA explained that it would adopt such requirements because the Risk Assessment on which the agency bases much of the rationale for the CCR proposal determined that "[c]omposite liners, as modeled in this assessment, effectively reduce risks from all constituents to below the risk criteria for both landfills and surface impoundment at the 90<sup>th</sup> and 50<sup>th</sup> percentiles." Based largely on this finding. EPA proposes to establish a composite liner system for landfills and surface impoundments under the Subtitle C option because the agency believes a composite liner system would be adequately protective of human health and the environment. In making this proposal, EPA intended to replace the double liner system currently in the hazardous waste regulations because such technology would be unnecessarily burdensome. Although the preamble indicated that these alternative standards would be applicable to both surface impoundments and landfills, the proposed composite liner regulations were applicable only to CCR landfills.

### The Subtitle C proposal to regulate inactive and/or previously closed surface impoundments is unlawful

One of the most extreme examples of agency overreach in the Subtitle C option is the unprecedented and unlawful proposal to extend RCRA Subtitle C jurisdiction to previously closed and/or inactive CCR surface impoundments. Even if such impoundments do not receive CCRs after the effective date of the rule, EPA would require these impoundments to close in compliance with the rule. EPA acknowledges that it has never attempted to extend RCRA in such a far-reaching manner to any other hazardous waste in its thirty-year implementation of the statute. In fact, EPA has consistently taken the opposite position throughout its administration of the RCRA program, explaining that RCRA's Subtitle C regulations are prospective in nature and are not directed at inactive facilities for a host of practical, equitable and economic reasons.

Nonetheless, EPA is proposing to reverse this position in the case of CCRs and retroactively apply RCRA's Subtitle C hazardous waste rules to previously closed and/or inactive CCR surface impoundments on the theory that the definition of "regulatory disposal" encompasses "passive leaking" from these units, and includes "the continued release of constituents to surrounding soil and groundwater through the continued infiltration of precipitation through inappropriately closed CCR impoundments ..." Under this theory, any previously closed and/or inactive CCR surface impoundment that is no longer receiving CCRs, but which has not closed pursuant to the interim status hazardous waste closure requirements, would be subject to full Subtitle C hazardous waste controls on the effective date of the Subtitle C rules and would have to "re-close" under the hazardous waste regulations. This would mean that literally hundreds of previously closed and inactive surface impoundments - many of which were properly closed decades ago under state solid waste programs, have changed owners, and now have structures built on top of them - would be considered "active" CCR disposal units and would be subject to full Subtitle C hazardous waste regulation. EPA's position runs directly counter to existing case law on the definition of disposal, as well as Congress' RCRA and CERCLA statutory scheme, and would raise a host of practical and compliance issues that EPA has recognized.

## EPA should adopt the Subtitle D Prime option, with appropriate adjustments, and coordinate implementation of the Subtitle D regulations with the states

Progress Energy supports the development of Subtitle D non-hazardous waste regulations for CCRs under the proposed Subtitle D "Prime" option, provided that EPA first incorporates certain important modifications discussed below. The Subtitle D Prime option, with the proper adjustments, will allow EPA to establish a robust and environmentally protective program for coal ash disposal units without crippling coal ash beneficial use, imposing unnecessary regulatory costs on power plants, threatening jobs, disturbing power reliability or increasing electricity costs.

### The rulemaking record supports regulation of CCRs under Subtitle D

It is important to emphasize that the development of Subtitle D non-hazardous waste regulations for CCRs is fully consistent with the rulemaking record. In fact, it is the only

lawful option available to the Agency. The development of Subtitle D regulations would be the appropriate outgrowth of EPA's two Reports to Congress and two Regulatory Determinations under the Bevill amendment declaring that CCRs do not warrant hazardous waste regulation under RCRA Subtitle C, but rather should remain primarily the province of state regulatory programs under RCRA Subtitle D. Throughout its 20year study and issuance of the required Reports to Congress and final CCR Regulatory Determinations, EPA has consistently found that the Subtitle D approach, with active state involvement, was the appropriate regulatory course for CCRs. In its first CCR Regulatory Determination in 1993, EPA concluded that, because "the potential for damage from these wastes is most often determined by site- or region-specific factors," the "current State approach to regulation" under RCRA Subtitle D is most appropriate. EPA echoed this finding in its second Report to Congress in 1999, concluding once again "that [RCRA] Subtitle C is inappropriate to address any problems associated with disposal of these wastes and that the continued use of site and region specific approaches by the states [under RCRA Subtitle D] is more appropriate for addressing the limited human health and environmental risks that may be associated with disposal of these wastes."

EPA reaffirmed the appropriateness of the Subtitle D option in its 2000 Regulatory Determination, concluding that "the subtitle D regulations are the most appropriate mechanism for ensuring that [CCRs] disposed in landfills and surface impoundments are managed safely." In addition to the fact that CCRs rarely exhibit a hazardous waste characteristic, EPA explained that the decisive factors underlying its final determination that CCRs do not warrant regulation under RCRA's Subtitle C hazardous waste regulations were:

- improving trends in present CCR disposal and utilization practices;
- the current and potential utilization of CCRs (i.e. beneficial use); and
- Congress' admonition in the Bevill Amendment against duplication of efforts by other federal and state agencies.

In concluding that Subtitle D was the appropriate regulatory path forward, EPA also pointed to the record evidence as set forth in the 1999 Report to Congress that "the utility industry has made significant improvements in its waste management practices over recent years" and to similar record evidence that "state [CCR regulatory] programs have, in fact, substantially improved over the last 15 years or so." EPA found that "the ability for most states to impose specific regulatory controls for coal combustion wastes has increased almost three-fold over the past 15 years." EPA concluded that, "with the exception of relatively few states, the regulatory infrastructure is generally in place at the state level to ensure adequate management of these wastes."

This positive trend has continued, with EPA and DOE finding in their joint 2007 report of state CCR controls that "[t]he data and analyses documented in this report provide new information that appears to show improved management of CCWs in both landfills and surface impoundments." Since the initiation of this rulemaking effort, the record support for the Subtitle D option has become even more compelling. Over the last year, the overwhelming majority of the states, the National Governors Association, municipal and local governments, ASTSWMO, the Environmental Council of the States, over two dozen state environmental protection agencies and state departments of transpiration, federal agencies including the Department of Energy, a bi-partisan group of 165 members of Congress (including the majority of the House Energy and Commerce Committee), 45 U.S. Senators and many other third parties have expressed their support for regulation of CCRs as a non-hazardous waste under RCRA Subtitle D. Given the breadth of federal agency, state, local and congressional support for the Subtitle D option, rejection of the Subtitle D option in favor of the Subtitle C approach would fly in the face of the Bevill amendment's directive that any final regulatory program for CCRs take into account the views of other federal agencies and the states to ensure a unified decision that "avoid[s] a duplication of effort." As noted above, Representative Bevill repeatedly cautioned EPA that it was not to make a final regulatory determination for CCRs "in a vacuum," but rather was to consult and coordinate with other key agencies that have evaluated CCRs, including, the Departments of Energy, Interior and Agriculture, as well as other government agencies, to ensure a coordinated regulatory approach for CCRs

consistent with the nation's "commitment to develop a coherent and consistent policy toward the use of our coal and other energy resources." These federal, state, municipal and local agencies have spoken and are virtually unanimous in their support of the Subtitle D approach.

In short, given the overwhelming views by the very agencies that Congress directed EPA to consult with in determining how to regulate CCRs under RCRA, combined with EPA's own Reports to Congress and two final Regulatory Determinations rejecting the concept of regulating CCRs under RCRA's hazardous waste program, the Subtitle D option provides the only lawful regulatory approach for these materials under RCRA.

### The Subtitle D Prime option will be fully protective of human health and the environment

Given that Subtitle D Prime's substantive controls are patterned in large part on the proposed Subtitle C controls, the overall level of protection to human health and the environment are equivalent under both options. As EPA explains in describing the protective nature of the Subtitle D option, the agency has historically interpreted both the Subtitle C and Subtitle D statutory provisions as requiring it "to establish a comparable level of protection, corresponding to an acceptable risk level ranging between 1 x 10<sup>-4</sup> to 1 x 10<sup>-6</sup>" and that EPA can establish even more stringent Subtitle D standards where it deems such regulations are appropriate. Therefore, the Subtitle D Prime option would provide the same level of protection to human health and the environment as the Subtitle C option without the corresponding adverse impacts of the Subtitle C approach.

One of the elements of the Subtitle D Prime option that makes it the better choice is that it does not require all existing surface impoundments to close if they are operating in a manner that is fully protective of human health and the environment. This option does not include the automatic requirement to dredge and line all existing CCR surface impoundments within five years of the effective date of the regulations, or close these impoundments. Under the Subtitle D Prime option, existing CCR surface impoundments would continue to be subject to groundwater monitoring and corrective action

requirements so that any threats from these units during their operating life would be detected and appropriate corrective action would be taken. Progress Energy agrees that CCR disposal units that are not fully protective must either take appropriate corrective action or close. There are many CCR surface impoundments which are perfectly safe; there is no reason why these units should not be allowed to continue operating.

## The Subtitle D program must include a mechanism for state implementation of the CCR rules

While Progress Energy supports the Subtitle D Prime option (with the adjustments described below), a major shortcoming of this approach, as well as with the other Subtitle D alternatives, is the lack of any mechanism for qualified state programs to step in and directly administer the Subtitle D CCR regulations where the state programs meet or exceed the federal controls. Such an approach would result in parallel and redundant regulatory programs for the same materials and, more importantly, would directly contravene the Bevill amendment's directive that EPA avoid duplication of effort with pre-existing regulatory programs. EPA has previously recognized this point; one of the agency's primary reasons for concluding in its 2000 Regulatory Determination that CCRs do not warrant Subtitle C regulation was Congress' admonition in the Bevill amendment against duplication of efforts by other federal and state agencies. Unfortunately, EPA has failed to adhere to this congressional directive in the proposed Subtitle D regulatory alternatives, including the Subtitle D Prime option.

As a result of its findings in the final 2000 Regulatory Determination and the supplemental 2007 joint EPA/DOE Report of state CCR programs, EPA is well aware that there are many mature state CCR regulatory programs with permitting, liner, and groundwater monitoring controls similar to those proposed in the Subtitle D options. In fact, EPA concluded in its 2000 Regulatory Determination that state CCR controls have "substantially improved over the last 15 years," and that a "high percentage of states had authority to impose protective management standards on surface impoundments and landfills, especially for groundwater monitoring, liners, and leachate collection…"

Under this dual regulatory approach, for example, an owner/operator of a CCR disposal facility could be found to be in non-compliance with a groundwater monitoring requirement that is contained in both the Subtitle D self-implementing rule and in an independently administered state regulatory program. In these circumstances, the owner/operator could be subject to a citizen suit enforcement action in federal court for alleged violation of the self-implementing Subtitle D rule, and to a wholly separate enforcement action in state court for violation of the parallel state requirement. Apart from the waste of federal and state judicial resources, a dual regulatory approach could result in inconsistent federal and state court determinations with respect to an identical regulatory requirement. It also could result in duplicative federal and state penalties for essentially the same regulatory infraction.

It is essential that any final Subtitle D rules for CCRs include a mechanism for qualified state programs to administer the rules in lieu of the self-implementing approach set forth in the proposal. EPA could readily develop a process for evaluating state CCR controls and determining whether state controls are no less stringent than the final Subtitle D controls (much like EPA does now in the case of municipal solid waste landfills under 40 C.F.R. Part 258). In those states that have qualifying controls for CCRs, EPA would allow the states to administer the controls to avoid the duplication of regulatory programs, with appropriate EPA backup. In this way, many of the self-implementing procedures in the Subtitle D approach could be eliminated as there would be a qualified state regulatory body with responsibility for ensuring compliance with the applicable requirements. This approach would allow for the requisite degree of state involvement in administration of the CCR rules that EPA concluded was appropriate in its Reports to Congress and final Regulatory Determinations and avoid duplication of regulatory controls as directed by the Bevill Amendment.

## EPA has the authority to develop Subtitle D rules administered by the states with direct EPA enforcement authority

One of EPA's stated concerns with the Subtitle D option is that it would not allow EPA to require the states to implement the Subtitle D CCR controls through state permits, approve the adequacy of the state programs or enforce the regulations against non-

compliant facilities. This concern is misplaced. EPA is underutilizing the full scope of its authority under Subtitle D by looking solely to RCRA Section 4004 for purposes of developing Subtitle D controls for CCRs. EPA can utilize the same combination of RCRA statutory authorities (i.e., RCRA Sections 4010(c) and 4005(c)) for establishing controls for CCR disposal units that it employed in promulgating federally enforceable Subtitle D rules for municipal solid waste landfills and for non-MSWLFs that receive household hazardous waste and small quantity generator waste under 40 CFR Parts 257 and 258.

The combination of these two Subtitle D provisions enables EPA to promulgate non-hazardous waste rules for CCRs that can be directly administered through state permitting programs and backed by direct EPA enforcement powers in those states that fail to adequately implement the federal rules. Such an approach offers a win-win for EPA: it provides the agency with the enforcement authority it desires under a Subtitle D CCR regulatory program while enabling the states to have a prominent role in the administration of any Subtitle D CCR rules, thereby preventing the duplication of federal and state controls in keeping with the Bevill amendment.

Progress Energy appreciates that development of a proposed Subtitle D regime allowing for administration of the regulations by qualified state programs, whether under RCRA section 4010(c) or through some other mechanism, will necessitate issuance of a supplemental proposal. Nonetheless, this additional procedural step is necessary to ensure promulgation of an environmentally sound regulatory program for CCRs that allows for appropriate state involvement. Such an approach is consistent with the rulemaking record and the dictates of the Bevill amendment.

## The Subtitle D Prime option requires certain amendments to better reflect CCR disposal operations

While Progress Energy supports the Subtitle D Prime option and believes that it can be administered through qualified state programs with back-up EPA enforcement authority in under-performing states, there are fundamental flaws with elements of the proposed regulations in need of correction. Key problems include: unrealistically short closure

timeframes for CCR disposal units; the unwarranted application of all location restrictions to *existing* CCR surface impoundments; the inclusion of inappropriate constituents in the proposed groundwater monitoring program; the failure to allow for alternate liner systems; and an overly broad definition of "CCR Landfill" that inappropriately includes all "large scale fill operations." These and other issues are discussed below.

### Compliance with EPA's proposed timeframe for closure of surface impoundments and landfills is impossible

A significant flaw in the Subtitle D proposal is the unrealistic timetable for the closure of CCR surface impoundments and landfills that are unable to meet the proposed design and operating standards. Even under the Subtitle D Prime option, where CCR surface impoundments operating in an environmentally sound manner could continue to operate, there undoubtedly will be a significant number of impoundments that will have to close. Therefore, a more realistic and achievable closure process is necessary under this option.

The Subtitle D option would require that existing landfills and surface impoundments meet certain location restrictions or performance criteria (e.g., not be located in unstable areas or have engineering measures incorporated into the design to ensure that the integrity of the structural components will not be disrupted if located in unstable areas). Further, within five years of the effective date of the Subtitle D rules, existing surface impoundments would be required to be dredged and to install a composite liner and leachate collection system, or close. Because these obligations may be impossible to fulfill and would impose significant operational costs on owners/operators, Progress Energy agrees with EPA that "many surface impoundments may close as a result of these requirements." Likewise, a number of landfills may also close if they are unable to comply with the "unstable areas" location restriction. Many of these disposal units will need to close pursuant to the Subtitle D closure framework. The proposed timetables for actually closing these landfills and surface impoundments, however, are unrealistic, and many facilities will be physically incapable of meeting these schedules.

In particular, EPA proposes to require disposal units that cannot meet the location restrictions or associated operating standards to close within five years of the effective date of the finalized regulations, which may be extended by an additional two years if the owner or operator can make a particular demonstration that there exists a lack of alternative disposal capacity and the unit poses no immediate threats to human health or the environment. Landfills typically close each component cell when the cell reaches its disposal capacity so that many cells have already been "closed" when the landfill begins final closure. Surface impoundments, on the other hand, must be entirely dewatered before closure can begin for any portion of the impoundment.

Given the disparity in the sizes of these units, the length of time necessary to dewater impoundments and the need for many of these units to be closed when the rules come into effect, Progress Energy strongly recommends that EPA not establish a specific timeframe for closure. Instead, the Company urges EPA to require utilities to close CCR surface impoundments and landfills consistent with a closure plan approved by a state, or developed and certified by a registered professional. The establishment of a closure plan, with set schedules, is the most effective method to account for the many variables associated with the closure of these units and is the approach commonly used by utilities. A closure plan also will provide EPA and the public with certainty that closure will occur in a step-wise and timely manner, without requiring facilities to comply with wholly unrealistic closure schedules.

On a related note, EPA has made no effort in its Subtitle D proposal to consider those state programs that provide a means to convert CCR surface impoundments to landfills through dewatering and other appropriate requirements. After these impoundments are appropriately converted, EPA should specifically identify these facilities as landfills and impose the regulatory requirements for existing CCR landfills rather than require these converted units to be subject to controls for existing surface impoundments, including potentially closure requirements. This will authorize the continued operation of these existing CCR landfills and regulate these units pursuant to the appropriate applicable standards and not require utilities to acquire land for the construction of new landfills.

# EPA should not apply location restrictions to existing surface impoundments or landfills

The proposed Subtitle D standards would impose various location restrictions to new and existing CCR surface impoundments and landfills. In particular, new and existing CCR surface impoundments and landfills could not be located in floodplains (a requirement that is already in effect) or unstable areas. In addition, new surface impoundments and new landfills could not be located in wetlands, fault areas or seismic impact zones, and must be located at least two feet above the natural water table. Of significant concern, EPA states in the preamble to the proposal that it intends to subject existing surface impoundments to *all* of these new location restrictions due to the risks posed by these units. In the corresponding proposed regulatory text, however, existing surface impoundments are only subject to the floodplains and unstable area location restrictions.

For purposes of these comments, Progress Energy assumes that EPA intends to subject existing CCR surface impoundments to the additional location restrictions consistent with the proposal's preamble. Presumably, EPA copied the regulatory text for the regulations applicable to MSWLFs as codified at 40 C.F.R. Part 258. Although EPA intends to base the Subtitle D CCR disposal unit regulations in large part on the MSWLF regulations, the Agency clearly departs from those regulations in proposing to require all existing surface impoundments to comply with the full range of location restrictions, a condition that is normally applied only to new units. This is an extreme departure from the MSWLFs standards; EPA has not made a compelling justification for treating CCR impoundments more harshly than MSWLF units.

Subjecting existing surface impoundments to these location restrictions must be backed by an evaluation of the risks posed by facilities in each of these types of locations. Progress Energy is not aware of structural stability issues directly caused by the placement of surface impoundments in wetlands, fault areas, or seismic impact areas of less than two feet above the natural water table. More importantly, apparently neither is

EPA; the agency does not even begin to assess the risks associated with the placement of existing impoundments in these areas and simply asserts that the risks associated with impoundments are greater than those associated with landfills, citing the TVA spill as an example of the environmental consequences of a "catastrophic collapse" of a CCR impoundment. These general assertions do not constitute the type of record evidence necessary to impose such sweeping requirements on all existing CCR surface impoundments. The agency must conduct a detailed analysis of CCR surface impoundments located in these restricted locations and demonstrate, based on this analysis, that there are increased risks from these impoundments to justify such a significant departure from how the Agency addressed and regulated MSWLFs.

Progress Energy does not believe that EPA can compile the record evidence necessary to support such a regulation. In fact, over the last year and a half since the TVA ash release, EPA has been conducting comprehensive structural integrity assessments of utility CCR impoundments across the country. The results of EPA's investigation have not, to date, identified any CCR impoundment that presents the imminent threat of a catastrophic release similar to that which occurred at TVA's Kingston facility. Specifically, EPA's structural assessments rated the impoundments as "satisfactory," "fair," "poor," or "unsatisfactory" – terms commonly used in the field of dam safety. EPA explains that, "[e]xpert experience has shown that only impoundments rated as 'unsatisfactory' pose immediate safety threats." Of the 120 CCR impoundments evaluated thus far, *none* of the impoundments received an "unsatisfactory" rating. As EPA correctly points out, even those impoundments that received a "fair" or "poor" rating may have been found by the inspectors to be structurally sound, but did not receive a "satisfactory" rating based primarily on lack of information. Therefore, there is no factual basis to assert that existing CCR impoundments pose the type of risks of catastrophic failure justifying the imposition of all the location restrictions on existing units; indeed, the record evidence makes clear that such restrictions are not warranted.

## <u>EPA should amend its run-on control requirements for CCR surface impoundments</u>

One complication of copying the proposed CCR Subtitle D regulations from the MSWLF program is that some of the particular operating requirements for MSWLF cannot be directly applied to CCR surface impoundments. An important example of this is the proposed requirement for surface impoundments and landfills to establish specific operating controls for water run-on and run-off. The run-on portion of these controls requires a "control system to prevent flow onto the active portion of the CCR landfill or surface impoundment" during peak discharge events. While this requirement may be appropriate for CCR landfills (as it is for MSWLFs), it is wholly inappropriate for surface impoundments that have been specifically designed as holding basins for storm water or that have been formed by damming a ravine and designed with outlets that are sufficient to deal with the water flow from the natural watersheds into the impoundment during peak discharge events.

Progress Energy suggests that EPA address the inapplicability of this requirement for these types of surface impoundments by requiring that CCR impoundments be designed to appropriately address peak discharge events, rather than specifying that water run-on into the impoundment be prevented. In this manner, impoundments that have been designed as described above and have been engineered to address peak discharge events will not be confronted with a restriction that is both unnecessary and effectively impossible to meet.

## <u>EPA should allow for alternative groundwater monitoring parameters and should revise certain of the proposed parameters</u>

Progress Energy supports appropriate groundwater monitoring standards for CCR surface impoundments and landfills and the development of such standards under a RCRA Subtitle D framework, which is modeled largely after the groundwater monitoring programs for MSLWFs under 40 C.F.R. Part 258 and for non-MWSLFs that may receive Conditionally Exempt Small Quantity Generator wastes under 40 C.F.R. Part 257, Subpart B. Unfortunately, unlike the groundwater monitoring programs for these units

under Parts 258 and 257, the agency is not proposing to allow an alternate list of indicator parameters to be used in the proposed CCR groundwater monitoring program because of the lack of direct state oversight and "EPA's information on CCRs indicates that their composition would not be expected to vary such that the parameters are inappropriate." The Company believes this is a mistake.

From a technical perspective, given that site-specific factors are critical for establishing constituents and corresponding concentrations for detection and assessment monitoring, it is important that EPA authorize registered professionals to certify variances from the otherwise applicable parameters used for detecting constituents in groundwater and assessing statistically significant increases from background for these parameters. Given the extreme range of natural variations in site geology and groundwater at CCR disposal units across the country, it will often be necessary for technical experts to adjust the specific groundwater monitoring standards that EPA establishes to account for various site-specific conditions and factors. Further, any variances to the list of regulatory parameters would only be available if approved by a qualified state program (under, for example, a state regulatory regime approved by EPA under RCRA Section 4005(c)) or if certified by a registered professional and such certifications are recorded in a facility's operating record, filed with the state and posted on the facility's publicly available internet site. As other critical elements of the Subtitle D program are predicated on compliance certifications by these technical professionals, there is no reason not to authorize these professionals to identify and certify, where appropriate, necessary adjustments to the groundwater monitoring parameters based on site-specific factors.

Therefore, Progress Energy recommends that EPA include language in the Subtitle D groundwater monitoring rules for CCR disposal units similar to that in the MSWLFs rules allowing for the state or a registered professional (in the context of a self-implementing Subtitle D rule) to delete any constituents from otherwise required groundwater monitoring "if it can be shown that the removed constituents are not reasonably expected to be in or derived from the waste contained in the unit." Similarly, the state or

a registered professional should be authorized to establish an alternative list of indicator parameters in lieu of some or all of the otherwise specified indicator parameters "if the alternative parameters provide a reliable indication of inorganic releases from the [CCR] unit to the groundwater."

Progress Energy also urges EPA to limit the list of assessment monitoring constituents to the primary drinking water constituents for which an MCL has been established. These primary constituents are the only materials EPA has found to contribute to adverse health effects; all other listed constituents have not been linked to health effects, and EPA should not require groundwater remediation for these parameters. Accordingly, EPA should remove aluminum, iron, manganese and molybdenum from the list of constituents for assessment monitoring.

#### The Subtitle D rule should allow for alternate liner systems

Progress Energy believes that it is absolutely essential that the Subtitle D rule allow for the use of alternative liner systems for CCR disposal units. In some regions of the country, the proposed composite liner system is unnecessary to ensure protection of groundwater. Moreover, there are alternative liner systems that can provide equal if not greater protection of groundwater and there is no reason why these alternate, but equally protective systems, should not be available.

Under the proposal, the liner system for CCR disposal units would track the default liner requirements for MSLWFs (but critically, not the standards for obtaining variances from these standards) by requiring a composite liner consisting of two components: "[a]n upper component consisting of a minimum 30 mil-flexible membrane liner (FML), and a lower component consisting of at least a two-foot layer of compacted soil with a hydraulic conductivity of no more than 1x10<sup>-7</sup> cm/sec." While the proposal currently does not allow for any deviation from the liner requirements for CCR units, EPA appropriately asks whether the Subtitle D option "should allow facilities to use an alternative design for new disposal units" provided an independent registered professional engineer or hydrologist certifies that an alternate design system would "ensure that the appropriate

concentration values for a set of constituents typical of CCRs will not be exceeded in the uppermost aquifer at the relevant point of compliance – *i.e.*, 150 meters from the unit boundary downgradient from the unit, or the property boundary if the point of compliance (i.e., the monitoring well) is beyond the property boundary." There is no question that alternate liner systems must be allowed under the Subtitle D program.

As the Agency has consistently recognized throughout the Bevill amendment regulatory determination process, given the vastly different site-specific characteristics of CCR disposal units across the country, there is no "one-size-fits all" approach when it comes to regulatory design standards for these units. Even in locations where potential risks to groundwater are greater, there are alternate liner systems that can achieve the same level of protection as the composite liner system proposed by EPA. For example, geosynthetic clay liner ("GCL") or similar systems can meet or exceed the performance standards of clay liners and certainly are equivalent, or superior in some circumstances, to two feet of compacted soil in the lower component of the composite liner systems proposed under the Subtitle D option. In many circumstances, a liner system using the GLC in lieu of the two feet of compacted soil readily enables a CCR disposal unit to meet the applicable groundwater monitoring standards.

Certainly, decisions regarding the use of alternate liner systems under a Subtitle D program can be made in a manner that is fully transparent and subject to the appropriate oversight. We would expect any final Subtitle D regulatory program to allow for the administration of the rules by qualified state programs. In these circumstances, any alternate liner systems would be subject to review and approval by a responsible state regulatory agency.

# There is an error in the proposed leachate collection system requirements for CCR surface impoundments

Progress Energy believes that EPA has inadvertently included an error in the regulatory text requiring the installation of a leachate collection system between the upper and lower components of the composite liner system for existing CCR surface impoundments and for new impoundments. Specifically, EPA proposes in the case of

existing CCR impoundments that, no later than five years after the effective date of the Subtitle D regulations, CCR impoundments shall be constructed with a composite liner with "a leachate collection system *between* the upper and lower components of the composite liner." EPA proposes a similar condition for new impoundments (and lateral extensions to existing impoundments), directing that the leachate collection system shall be installed "between the upper and lower components of the composite liner." The Company believes that this is an error, given that the composite liner systems that CCR surface impoundments are required to employ are required to have the upper flexible membrane liner (FML) component to be "installed in direct and uniform contact with the compacted soil component." In other words, given that both components of the liner system must be in direct contact with one another, there is no way that a leachate collection system can be installed between these two components.

Even in the case of new CCR landfills, the proposed regulatory text makes clear that the leachate collection system is to be installed over the liner. Therefore, EPA should eliminate the proposed requirement directing that leachate collection systems in the case of surface impoundments be between the upper and lower components of the liner system when these components must be "in direct and uniform contact" with one another.

### <u>Definitional Issues Under the Subtitle D proposal</u>

A number of the proposed definitions in the Subtitle D option require modification and/or clarification. These issues are noted below.

### Definition of "CCR landfill"

EPA's proposed definition of "CCR landfill" is overbroad to the extent that it attempts to capture *all* operations involving the use of CCRs for "large scale fill operations." This definition would inappropriately capture the use of CCRs that are being used for a legitimate functional purpose and thus are not a "solid waste" because they have not been "discarded." In addition, the phrase "large scale fill operations" is unlawfully vague and cannot be included in a final regulatory definition without a supplemental proposal

setting forth discernible and readily understandable parameters as to what constitutes the type of "large scale fill operations" that would constitute a CCR landfill. Under the Subtitle D option, EPA would define the term "CCR Landfill" as:

A disposal facility or part of a facility where CCRs are placed in or on land and which is not a land treatment facility, a surface impoundment, an underground injection well, a salt dome formation, a salt bed formation, an underground mine, cave, or a corrective action management unit. For purposes of this subpart, landfills also include piles, sand and gravel pits, quarries, and/or large scale fill operations. Sites that are excavated so that more ash can be used as fill are also considered CCR landfills.

One of the fundamental problems with this definition is that it assumes that all CCR large-scale fill operations involve the "disposal" of CCRs and that these operations therefore constitute CCR landfills. As EPA knows, however, a series of judicial decisions have established the parameters of RCRA's statutory definition of "solid waste," which also limits the regulatory parameters of what can constitute a CCR landfill under RCRA Subtitle D. These cases make clear that the definition of "solid waste" must be interpreted in a common sense manner that extends "only to materials that are truly discarded, disposed of, thrown away, or abandoned," and that materials destined for legitimate beneficial reuse or recycling are "not part of the waste disposal problem" and thus cannot be characterized as solid waste. It is incorrect to assume that all CCR large-scale fill projects involve the "discard" of CCRs. To the contrary, CCRs have long been used for a wide range of large-scale structural fill projects where they serve the functional purpose of providing structural stability for, among other things, aboveground structures, road projects, embankments, and reconfiguration of property to meet design standards for run-off controls. When used in these applications, CCRs often are being used in lieu of competing commercial materials, such as soil or gravel that would have to excavated, mined and/or purchased by end users, including, among others, federal, state and municipal governments who routinely use CCRs in large-scale road and embankment projects. In these circumstances, the CCRs are not being "discarded" in any common-sense understanding of the term, but rather are being used for a legitimate beneficial purpose in lieu of competing materials.

Further, there is no record evidence supporting the proposition that all types of "large-scale fill operations" constitute the disposal of CCRs. In fact, the sole basis referenced in the preamble for including these beneficial use activities within the definition of a "CCR landfill" is that "[t]here has also been significant community concern with large-scale fill operations." This unsubstantiated and generalized statement of concern cannot support the whole scale classification of a long-standing beneficial use activity as somehow constituting disposal.

EPA also provides no analysis whatsoever of the criteria typically employed by the agency in determining whether a particular reuse activity involves legitimate recycling. When these criteria are applied to CCR large scale fill operations, the better argument is that these operations fall squarely with the agency's core criteria of what constitutes legitimate "beneficial use." For example, CCRs used for large scale fill operations provide a "functional benefit" in terms of structural support. In addition, the CCRs often are being used as "substitutes for the use of a virgin material, conserving natural resources that would otherwise need to be obtained through practices, such as extraction." These criteria are the hallmarks of a legitimate beneficial use when applied to CCRs used in other contexts, and they are equally applicable to CCRs when used in large-scale fill projects. For the above reasons, EPA's blanket inclusion of all CCR "large scale fill operations" in the definition of a CCR landfill is at odds with RCRA's definition of "solid waste" and is arbitrary and capricious.

EPA cannot attempt to provide the critical contours of this key regulatory concept through the subsequent issuance of "guidance." Such a back-door attempt to define this important term would be in violation of EPA's obligations under the APA to provide the public with an opportunity to review and comment on proposed regulations.

#### **Definition of CCR landfill and CCR piles**

Another flaw in the proposed definition of "CCR landfill" is that it would include all "piles" of CCRs, with no limitation of any kind. As is the case with "large scale fill operations,"

inclusion of this term in the proposed definition is overbroad and would inappropriately capture CCR management activities that do not constitute disposal.

For example, CCRs often are placed into piles for purposes of staging and/or consolidation *prior* to placement in a CCR landfill. To categorize and regulate the CCR storage pile in these circumstances in the same manner as the actual landfill in which the CCRs are disposed of is illogical and obviously inappropriate. Even in the context of EPA's hazardous waste rules, the Agency recognizes the distinction between landfills and storage piles, and defines and regulates the two as separate management units; indeed, the Subtitle C regulations expressly exclude "piles" from the definition of a "landfill." The definition of "landfill" at 40 C.F.R. § 260.10 includes, "a disposal facility or part of a facility where hazardous waste is placed in or on land and which is not a pile..." The term "pile" is defined under the Subtitle C regulations to include "any noncontainerized accumulation of solid, nonflowing hazardous waste that is used for treatment or storage and that is not a containment building."

To remedy this problem, EPA should remove the term "piles" from the definition of "CCR landfill," as piles are not landfills in any ordinary (or existing regulatory) sense of the term. Progress Energy understands that EPA may be concerned about piles being used inappropriately for sham disposal of CCRs. However, a more appropriate way to address this concern would be to set a limit on the amount of time that the CCRs are allowed to be maintained in a pile. The storage of CCRs in piles beyond the timeframe established in the regulations would create a rebuttable presumption that disposal, not storage, is taking place and that the pile should be subject to the CCR landfill requirements. A timeframe of 180 days would be an appropriate length for purposes of establishing a rebuttable presumption that the pile be viewed as a "CCR landfill" and not as a storage pile when storage exceeds this time limit. As with any rebuttable presumption, however, the facility would have the opportunity to demonstrate that storage beyond the 180 days is legitimate and necessary to facilitate disposal in a landfill or impoundment (e.g., disposal capacity is limited and/or storage beyond this time period is necessary to facilitate transportation to a disposal facility).

Also, given that CCRs in piles are often continuously removed and replaced with new CCRs, the 180-day time limit would apply to the CCRs themselves and not to the existence of the pile itself, which will be continually changing in size as CCRs are periodically added and removed from it. This is the approach EPA takes in the case of applying the 90-day accumulation time limit to continuous process flow-through tanks under the Agency's hazardous waste accumulation rules under 40 C.F.R. § 262.34; the same common-sense approach should be applied here.

#### **Definition of CCR surface impoundment**

EPA's proposed definition of "CCR surface impoundment or impoundment" is overly broad in that it would inappropriately capture impoundments that are not designed to manage CCRs. To correct this problem, there is a proposed revised definition of "CCR impoundment" at the end of this section to help ensure that the final definition includes only those impoundments that truly function as CCR surface impoundments at electric utility power plants.

As currently drafted, the definition includes:

A facility or part of a facility which is a natural topographic depression, man-made excavation, or diked area formed primarily of earthen materials (although it may be lined with man-made materials), which is *designed* to hold an accumulation of CCRs containing free liquids, and which is not an injection well. Examples of CCR surface impoundments are holding, storage, settling, and aeration pits, ponds, and lagoons. CCR surface impoundments are used to receive CCRs that have been sluiced (flushed or mixed with water to facilitate movement), or wastes from wet air pollution control devices, often in addition to other solid wastes.

A critical limitation in this definition is that it is intended to capture only impoundments "designed" to accumulate CCRs. If applied appropriately, this definition generally would only extend to those impoundments into which CCRs are initially sluiced, as these are the impoundments at power plants that are "designed" to accumulate CCRs.

Progress Energy is concerned that the definition could inappropriately be interpreted to include downstream secondary and tertiary impoundments – such as polishing, cooling,

wastewater and holding ponds – that receive waters containing only *de minimis* amounts of CCRs. For example, many power plants have cooling water ponds and/or wastewater ponds constructed and used for the management of various water streams that do not contain any CCRs. A wastewater impoundment at a facility that receives non-CCR wastewaters may, however, occasionally receive *de minimis* amounts of CCRs via storm water runoff (for example, from an FGD storage pad) or from wastewater containing *de minimis* amounts of CCRs from an upstream CCR impoundment. In these circumstances, the wastewater impoundment clearly is not "designed to hold an accumulation of CCRs" and is not in any practical or technical sense "used to receive CCRs that have been sluiced" to the impoundment. It would be illogical and unnecessary to apply the full array of proposed controls for CCR impoundments, including liner systems, groundwater monitoring, siting restrictions, and related standards, to these impoundments. The final rule should make clear that cooling, polishing, wastewater, and similar ponds are not included in this universe.

The definition also should not encompass units that, having once served as impoundments, have been dewatered and closed for purposes of serving as the base for a CCR landfill or other structures. Dewatered impoundments that subsequently serve as the base for CCR landfills are not impoundments and should not be captured by the CCR surface impoundment definition.

Given the above, Progress Energy suggests that EPA's proposed definition of CCR impoundment be modified to read as follows:

CCR surface impoundment means a facility or part of a facility which is a natural topographic depression, man-made excavation, or diked area formed primarily of earthen materials (although it may be lined with man-made materials), which has a storage volume of 10 acre-feet or more and which is specifically designed for and whose primary function is to hold an accumulation of CCRs containing free liquids and which is not an injection well. Examples of CCR surface impoundments whose primary function is to hold an accumulation of CCRs are primary holding, storage, settling, and aeration pits, ponds, and lagoons used to receive CCRs that are sluiced (flushed or mixed with water to facilitate movement), or wastes from wet air pollution control devices, often in addition to other solid wastes. This definition is not intended to encompass impoundments whose primary function is not to receive CCRs but which receive de minimis

amounts of CCRs through the receipt of waters that contain only incidental amounts of CCRs. Examples of impoundments that are not CCR impoundments include, but are not limited to, cooling, polishing, process water storage, storm water management, recycling, wastewater and holding ponds. CCR impoundments also do not include former impoundments that have been dewatered and closed and/or dewatered and subsequently used as the base for CCR landfills, buildings, or other structures.

The above modifications to the definition of CCR surface impoundment will help ensure that only those impoundments that truly function as CCR impoundments are subsumed within the "CCR impoundment" definition and subjected to the proposed CCR impoundment regulations. This is the most logical approach, as it would make no sense to subject units that contain only *de minimis* or trace amounts of CCRs from incidental sources to the full array of proposed CCR impoundment requirements.

## EPA must refine the proposed regulations to ensure that legitimate beneficial uses are excluded from regulation

Progress Energy, the electric utility industry and the coal ash association have long supported the beneficial use of CCRs and have worked with EPA, other federal agencies, states, localities and standard-setting organizations to pursue the environmentally protective and beneficial use of CCRs in a wide variety of industrial, commercial and residential products, and construction activities. The Company fully supports the exclusion of beneficial uses from any hazardous or solid waste regulation, provided that the use of CCRs is protective of human health and the environment.

EPA must revise its proposed definition of CCR landfills and the proposed exclusion for CCR beneficial uses to ensure that the proposed regulations do not sweep in environmentally protective beneficial uses that could be classified as large-scale fill operations.

## EPA must ensure that environmentally protective beneficial use projects are not subject to regulation as CCR disposal units

Both the Subtitle C and Subtitle D options would encompass CCRs that are disposed of and would not apply to CCRs that are beneficially used. To implement this regulatory

distinction, EPA would include in the definition of "disposal units" uses of CCRs that EPA believes are not legitimate "beneficial uses" and instead should be regulated as CCR disposal. In attempting to establish this distinction, however, EPA uses vague, undefined terms which are subject to a wide range of possible interpretations, potentially even covering within the concept of "disposal" those types of activities that the agency believes constitute the safe beneficial use of CCRs which it explicitly supports in the proposal's preamble.

Under the Subtitle C option, EPA would exclude CCRs that are beneficially used from Subtitle C regulation by retaining such uses within the scope of the regulatory text implementing the Bevill exclusion. EPA would define "beneficial use" to include:

The use of [Coal Combustion Products or "CCPs"] that provides a functional benefit; replaces the use of an alternative material, conserving natural resources that would otherwise need to be obtained through practices such as extraction; and meets relevant product specifications and regulatory standards (where these are available). CCPs that are used in excess quantities, placed as fill in sand and gravel pits, or used in large scale fill projects, such as for restructuring the landscape, are not considered beneficial uses.

(emphasis added). Uses of CCRs that do not fall within the proposed definition of "beneficial use" would be subject to full Subtitle C regulation.

While EPA does not specifically define what would constitute "beneficial use" under the Subtitle D option, the Agency attempts to draw the same distinction between legitimate beneficial use and disposal by defining what would constitute a "CCR landfill." Indeed, the proposed definition of CCR landfill is identical under both the Subtitle C and Subtitle D options, suggesting that beneficial uses are intended to be addressed the same way in the two proposals.

As noted above, Progress Energy believes that large-scale fill operations – if designed, engineered and executed properly – constitute a legitimate beneficial use. The Company has long supported the use of industry standards in designing engineered structural fills and other beneficial use projects. In fact, USWAG developed an industry

guidance document on how utilities and independent power producers as well as downstream coal ash users should use CCRs in engineered structural fills to minimize any potential threats to human health and the environment. Progress Energy believes that structural fill projects, if developed in accordance with appropriate and accepted industry standards and protections, should be considered a legitimate beneficial use and not be viewed as "disposal" under the proposal.

Notwithstanding the Company's support of certain environmentally protective structural fill projects, the terms included in the "CCR landfill" definition, namely "piles, sand and gravel pits, quarries and/or large scale fill operations" are not defined in the regulatory text and are only loosely discussed in the proposal's preamble. The term "large scale fill operation" is especially vague, suggesting that the critical and troublesome issue for the agency is the size of the fill rather than whether the fill project was properly designed and environmentally protective. As addressed earlier in these comments, this undefined term can also apply to above ground structures, road projects, and embankments. As EPA notes in the preamble, USWAG has worked with EPA, the Federal Highway Administration, DOE and the ACAA to provide guidance and best management practices on the use of CCRs in highway construction. However, the undefined terms "large scale fill operation" (and the similar phrase "large scale fill project" which would be excluded from the proposed definition of "beneficial use" detailed above) could easily be read to cover certain highway construction projects developed and implemented consistent with this jointly-developed guidance and best management practices. Progress Energy sincerely hopes that it was not the Agency's intent to regulate these projects as CCR landfills based on a notion of the risks that they pose, given that EPA was at the forefront in developing guidance on how to safely use CCRs for these projects.

EPA proposes to define the term "beneficial uses" under the Subtitle C option, as "the use of CCPs that provides a functional benefit; replaces the use of an alternative material, conserving natural resources that would otherwise need to be obtained through practices such as extraction; and meets relevant product specifications and

regulatory standards (where these are available)." As an alternative to these criteria, EPA requests comment on whether it should provide a formal listing of all beneficial uses. Progress Energy does not believe that EPA should attempt to specifically list all the types of qualified CCR beneficial uses as any such list will certainly be underinclusive. Equally important, a formal list will delay the implementation of new and emerging strategies and technologies for using CCRs, because a new rulemaking would be necessary to officially recognize any new and qualified CCR beneficial use. This would unnecessarily restrict the development of new CCR markets and stifle innovation. Rather, the Company supports the performance criteria as proposed in the regulatory text, but believes that structural fills and other projects – including some that may not be developed yet – can and do meet these criteria.

#### **Development of structural integrity standards for CCR surface impoundments**

Progress Energy agrees with EPA's proposal to incorporate the surface impoundment integrity and inspection obligations into the RCRA program modeled largely on the standards for coal slurry impoundments regulated by the Mine Safety and Health Administration (MSHA). While MSHA standards regulate a broader array of materials with differing physical properties than CCRs, the MSHA rules provide a good starting point for developing comparable standards for CCR impoundments.

However, it is important to ensure that the same applicability criteria triggering the obligation to record impoundment information in the company's operating record apply in triggering the related obligation to conduct surface impoundment inspections.

Comments of
The Utility Solid Waste Activities Group,
The Edison Electric Institute,
The American Public Power Association, and
The National Rural Electric Cooperative Association on
Hazardous and Solid Waste Management System;
Identification and Listing of
Special Wastes; Disposal of Coal Combustion Residuals From
Electric Utilities;
Proposed Rule

75 Fed. Reg. 35128 (June 21, 2010)

Volume I Comments

submitted to
The United States
Environmental Protection Agency
Docket No. EPA-HQ-RCRA-2009-0640

November 19, 2010

Of Counsel: Venable LLP 575 Seventh Street, N.W. Washington, D.C. 20004

### **TABLE OF CONTENTS**

|          | <u>Pag</u>  | <u>e</u> |
|----------|---|----------|
| INT      | RODUCTION AND EXECUTIVE SUMMARY1  |          |
| <u>.</u> | CHAPTER I: EPA CANNOT PROMULGATE SUBTITLE C REGULATIONS FOR COAL COMBUSTION RESIDUALS   |          |
| I.       | THE BEVILL AMENDMENT12  |          |
|          | A. THE LEGISLATIVE HISTORY OF THE BEVILL AMENDMENT  |          |
| II.      | IMPLEMENTATION OF THE BEVILL AMENDMENT19  |          |
|          | A. EPA'S MARCH 1988 REPORT TO CONGRESS ON CCRS  |          |
| III.     | EPA MAY NOT REVERSE ITS FINAL REGULATORY DETERMINATIONS THAT CCRS DO NOT WARRANT REGULATION UNDER RCRA SUBTITLE C27   |          |
|          | A. EPA COMPLETED THE BEVILL REGULATORY PROCESS FOR CCRs IN 2000 AND THE STATUTE DOES NOT AUTHORIZE EPA TO REVERSE THAT DECISION   |          |
|          | B. Assuming, For Purposes Of Argument, That EPA Is Authorized To Make A Second Regulatory Determination, EPA Has Failed To Adhere To The Bevill Amendment's Prescribed Process  |          |
| IV.      | THE PROPOSED SUBTITLE C OPTION IS INCONSISTENT WITH THE BEVILL AMENDMENT AND IS ARBITRARY AND CAPRICIOUS41  |          |
|          | A. THE SUBTITLE C OPTION IS INCONSISTENT WITH THE BEVILL AMENDMENT BECAUSE IT WOULD ADVERSELY IMPACT CCR BENEFICIAL USE AND IS BASED ON THE ARBITRARY AND CAPRICIOUS POSITION THAT SUBTITLE C REGULATION OF CCRS WILL INCREASE BENEFICIAL USE |          |
|          | Will Significantly Harm CCR Beneficial Use  |          |
|          | B. THE SUBTITLE C OPTION IS INCONSISTENT WITH THE BEVILL AMENDMENT AND ARBITRARY AND CAPRICIOUS BECAUSE IT WOULD RESULT IN UNINECESSARY AND EXCESSIVELY BURDENSOME REGULATION OF CCRS 68  |          |

|       | C.           | THE SUBTITLE C OPTION IS ARBITRARY AND CAPRICIOUS BECAUSE IT IS INCONSISTENT WITH THE VIEWS OF THE OTHER FEDERAL AND STATE AGENCIES THAT EPA WAS OBLIGATED TO CONSULT IN REACHING A FINAL REGULATORY DETERMINATION | 75  |
|-------|--------------|--|-----|
| V.    | TH           | IE DAMAGE CASES DO NOT SUPPORT THE SUBTITLE C OPTION   | 82  |
|       | A.           | THE DAMAGE CASES REPRESENT A SUBSET OF OLDER, UNLINED CCR DISPOSAL UNITS   | 83  |
|       | B.           | THE PROPOSED SUBTITLE D CONTROLS WILL PREVENT FUTURE DAMAGE CASES, AND EXISTING DAMAGE CASES ARE BEING EFFECTIVELY REMEDIATED  | 86  |
|       | C.           | THE ALLEGED DAMAGE CASES PRESENTED BY ENVIRONMENTAL GROUPS ARE NOT PROVEN AND CANNOT BE RELIED UPON BY EPA IN THIS RULEMAKING  |     |
|       | <u>c</u>     | CHAPTER TWO: THE SUBTITLE C PROPOSAL IS LEGALLY FLAWED   |     |
| VI.   |              | PA'S PROPOSED LISTING OF CCRS AS A SPECIAL WASTE IS  | 92  |
|       |              | THE OVERBROAD LISTING UNDERMINES EPA'S LISTING ANALYSIS EPA'S —PAUSIBLE MISMANAGEMENT" SCENARIO DOES NOT REFLECT   | 93  |
|       |              | UTILITY PRACTICES  | 96  |
|       | C.           | EPA Has Not Properly Evaluated Regulatory Actions of Other Governmental Agencies   | 99  |
|       | D.           | THE RISK ASSESSMENT IS FUNDAMENTALLY FLAWED AND CANNOT BE RELIED ON TO SUPPORT THE SUBTITLE C OPTION, INCLUDING  |     |
|       | E            | EVALUATING THE LISTING CRITERIA FOR CCRSEPA'S DRAFT SCREENING ASSESSMENT OF FUGITIVE DUST RISKS DOES   | 102 |
|       | <b>L</b> .   | NOT SUPPORT THE PROPOSED LISTING OF CCRs   | 110 |
| VII.  | ME           | PA'S REGULATORY IMPACT ANALYSIS CONTAINS FUNDAMENTAL ETHODOLOGICAL FLAWS THAT UNDERMINE ITS MOST BASIC DNCLUSIONS  | 112 |
|       |              | EPA FAILED TO PROPERLY VALUE THE IMPACTS OF THE PROPOSED   | 112 |
|       |              | REGULATORY OPTIONS ON CCR BENEFICIAL USE   | 113 |
|       | В.           | EPA'S ASSESSMENT OF THE BENEFITS OF AVOIDED CANCER INCIDENTS CONTAINS SERIOUS ERRORS AND OVERSTATES THE RISK REDUCTION FROM THE SUBTITLE C ALTERNATIVE   | 117 |
| VIII. |              | IE SUBTITLE C OPTION PRESENTS INSURMOUNTABLE OMPLIANCE AND DISPOSAL OBSTACLES FOR ELECTRIC UTILITIES   |     |
|       | A.           | COMPLIANCE CONCERNS THE SUBTITLE C OPTION WILL RESULT IN SERIOUS DISPOSAL CAPACITY   |     |
|       |              | Shortfalls   |     |
|       | $\mathbf{C}$ | OTHER ISSUES RELATED TO SUBTITLE C REGULATION  | 131 |

| IX.  | THE SUBTITLE C PROPOSAL TO REGULATE INACTIVE AND/OR PREVIOUSLY CLOSED SURFACE IMPOUNDMENTS IS UNLAWFUL  |         |  |  |
|------|---|---------|--|--|
|      | A. EPA'S POSITION ON PASSIVE MIGRATION IS CONTRADICTORY TO FEDERAL CASE LAW  B. EPA'S PROPOSAL RUNS AFOUL OF CONGRESS' STATUTORY SCHEME  C. REGULATING PREVIOUSLY CLOSED/INACTIVE CCR SURFACE | 138     |  |  |
| Χ.   | IMPOUNDMENTS RAISES A HOST OF LEGAL AND PRACTICAL IMPLICATION:  EPA VIOLATED THE REGULATORY FLEXIBILITY ACT   |         |  |  |
| ۸.   |   |         |  |  |
|      | A. BACKGROUND OF RFA AND SBREFA   | )       |  |  |
| CH   | HAPTER THREE: EPA SHOULD ADOPT FEDERAL NON-HAZARDOU<br>RULES FOR CCRS   | S WASTE |  |  |
| XI.  | EPA SHOULD ADOPT THE SUBTITLE D PRIME OPTION, WITH APPROPRIATE ADJUSTMENTS, AND COORDINATE IMPLEMENTATION OF THE SUBTITLE D REGULATIONS WITH THE STATES                                       |         |  |  |
|      | A. THE RULEMAKING RECORD SUPPORTS THE SUBTITLE D OPTION   |         |  |  |
|      | B. THE SUBTITLE D PRIME OPTION WILL BE FULLY PROTECTIVE OF HUMAN HEALTH AND THE ENVIRONMENT   |         |  |  |
|      | C. THE SUBTITLE D PROGRAM MUST INCLUDE A MECHANISM FOR STATE  |         |  |  |
|      | IMPLEMENTATION OF THE CCR RULES   | 155     |  |  |
|      | AUTHORITY   | 158     |  |  |
|      | RCRA Sections 4010(c) and 4005(c)      EPA's Implementation of Section 4010(c)  |         |  |  |
|      | 3. Adoption of Regulations for CCRs under Section 4010(c)   |         |  |  |
| XII. | THE SUBTITLE D PRIME OPTION REQUIRES CERTAIN AMENDME TO BETTER REFLECT CCR DISPOSAL OPERATIONS  |         |  |  |
|      | A. COMPLIANCE WITH EPA'S PROPOSED TIMEFRAME FOR CLOSURE OF  |         |  |  |
|      | SURFACE IMPOUNDMENTS AND LANDFILLS IS IMPOSSIBLE  | 165     |  |  |
|      | B. EPA SHOULD NOT APPLY LOCATION RESTRICTIONS TO EXISTING SURFACE IMPOUNDMENTS OR LANDFILLS   | 170     |  |  |
|      | C. EPA SHOULD AMEND ITS RUN-ON CONTROL REQUIREMENTS FOR CCR   | 170     |  |  |
|      | SURFACE IMPOUNDMENTS  | 173     |  |  |
|      | D. EPA SHOULD ALLOW FOR ALTERNATIVE GROUNDWATER MONITORING PARAMETERS AND SHOULD REVISE CERTAIN OF THE PROPOSED   |         |  |  |
|      | PARAMETERS AND SHOULD NEVISE CERTAIN OF THE PROPOSED  | 174     |  |  |

|       | Ε.       | EPA SHOULD ALLOW REGISTERED PROFESSIONALS TO MAKE  |     |
|-------|----------|--|-----|
|       |          | COMPLIANCE CERTIFICATIONS EVEN IF THEY ARE EMPLOYEES OF THE  |     |
|       |          | OWNER/OPERATOR OF THE DISPOSAL UNIT  | 177 |
|       | F.       | THE SUBTITLE D RULE SHOULD ALLOW FOR ALTERNATE LINER SYSTEMS   | 180 |
|       | G.       | ERROR IN PROPOSED LEACHATE COLLECTION SYSTEM REQUIREMENTS  |     |
|       |          | FOR CCR SURFACE IMPOUNDMENTS   | 185 |
|       | Н.       | OMISSION IN DEFINING SCOPE OF SUBTITLE D PROGRAM   |     |
|       | I.       | DEFINITIONAL ISSUES UNDER THE SUBTITLE D PROPOSAL  | 188 |
|       |          | 1. Definition of —CR Landfill"   | 188 |
|       |          | 2. Definition of CCR Landfill and CCR Piles  | 192 |
|       |          | 3. Definition of CCR Surface Impoundment   | 193 |
|       |          | 4. Definition of Existing CCR Surface Impoundment  | 198 |
|       |          | 5. Definition of Existing CCR Landfill   | 199 |
| AIII. | TH<br>RE | PA MUST REFINE THE PROPOSED REGULATIONS TO ENSURE NAT LEGITIMATE BENEFICIAL USES ARE EXCLUDED FROM EGULATION AND THAT CCR MINE PLACEMENT ACTIVITIES ARE DESCRIPTION BY THIS RULEMAKING | 200 |
|       |          |  |     |
|       | Α.       | EPA MUST ENSURE THAT ENVIRONMENTALLY PROTECTIVE BENEFICIAL   |     |
|       |          | USE PROJECTS ARE NOT SUBJECT TO REGULATION AS CCR DISPOSAL UNITS   | 201 |
|       | D        | EPA MUST ENSURE THAT THE CCR REGULATIONS DO NOT APPLY TO   | 201 |
|       | D.       | CCR MINE PLACEMENT   | 205 |
|       |          |  | 203 |
| XIV.  |          | SWAG SUPPORTS DEVELOPMENT OF STRUCTURAL INTEGRITY  |     |
|       | ST       | ANDARDS FOR CCR SURFACE IMPOUNDMENTS   | 206 |
|       | ١٥:      | HOLON  | 213 |
|       |          | USION  | フイス |

#### INTRODUCTION AND EXECUTIVE SUMMARY

The Utility Solid Waste Activities Group (—USWAG") submits these comments in response to the United States Environmental Protection Agency's (—EPA" or -Agency") proposed rule entitled -Hazardous Waste Management System: Identification and Listing of Special Wastes; Disposal of Coal Combustion Residuals (-GCRs") From Electric Utilities." 75 Fed. Reg. 35128 (June 21, 2010) (-GCR Proposal" or -Proposal"). USWAG is an association of over one hundred and ten utilities, utility operating companies, energy companies, and associations, including the Edison Electric Institute (-EEI"), the American Public Power Association (-APPA"), and the National Rural Electric Cooperative Association (-NRECA"). USWAG has worked closely with EPA for approximately thirty years on the Agency's implementation of the Bevill Amendment's regulatory determination process for CCRs. As part of this ongoing dialogue, USWAG has provided the Agency with technical data and related input to assist EPA in reaching the appropriate Regulatory Determination for CCRs under the Resource Conservation and Recovery Act (-RCRA").

Following issuance of EPA's final 2000 Regulatory Determination declaring that CCRs do not warrant regulation under RCRA's Subtitle C hazardous waste regulations (65 Fed. Reg. 32214 (May 29, 2000)), USWAG continued its work with EPA to ensure that CCRs are managed under RCRA's non-hazardous waste program in a manner that

<sup>&</sup>lt;sup>1</sup> EEI is the association of U.S. shareholder-owned electric companies. APPA is the national association of publicly-owned electric utilities. NRECA is the national association of rural electric cooperatives. Together, USWAG members represent more than 73 percent of the total electric generating capacity, 91% of the nation's coal-fired generation, and service more than 95 percent of the nation's consumers of electricity. Throughout these comments, we refer to our industry as the — illty" or -electric utility" industry. This term is intended to include those portions of the industry and those USWAG members that generate electricity but do not directly provide electricity to the public and are technically not -utilities."

is fully protective of human health and the environment. Among other things, USWAG initiated and developed a Voluntary Action Plan to implement groundwater monitoring and appropriate corrective action at CCR disposal units. See 72 Fed. Reg. 49714 (Aug. 29, 2007). USWAG also undertook measures to ensure that the beneficial use of CCRs, which has flourished since EPA's final 2000 Regulatory Determination, is performed in an environmentally protective manner. For example, USWAG developed and shared with EPA detailed engineering and environmental guidance on the use of CCRs in engineered structural fill projects. See Letter from USWAG to EPA transmitting USWAG's Engineering and Environmental Guidance on the Beneficial Use of Coal Combustion Products in Engineered Structural Fill Projects (May 4, 2009)(attached as Appendix 1). Thus, long before EPA's initiation of this rulemaking, USWAG has worked cooperatively with EPA to ensure that CCRs are managed in a responsible manner. USWAG looks forward to continuing its work with EPA during this rulemaking process to achieve this common objective.

As we have stated throughout the public hearing process on the CCR proposal, USWAG supports the development of federal regulations for CCRs under RCRA's Subtitle D non-hazardous waste program. Indeed, the question for USWAG is not whether EPA should develop regulations for CCRs, but rather *how* best to regulate CCRs under a federal program. As detailed in these comments, we have carefully evaluated the full range of alternatives available to EPA and believe that the -Subtitle D Prime" option, with appropriate adjustments, offers the best path forward. Unlike the Subtitle C option, which is not legally authorized, would create disposal capacity shortfalls, cripple the beneficial use of CCRs and impose unnecessary costs on power

plants (and ultimately consumers), threatening jobs and increasing electricity costs, the Subtitle D option will enable EPA to develop an environmentally protective program for CCRs without the attendant adverse consequences of the Subtitle C approach.

While USWAG has a number of significant comments on the Subtitle D proposal, RCRA's Subtitle D framework provides the appropriate legal structure under which EPA may develop federal regulations for CCRs. In fact, it is the only lawful option available to the Agency. The development of Subtitle D regulations is the appropriate outgrowth of EPA's two Reports to Congress and two final Regulatory Determinations under the Bevill Amendment declaring that CCRs do not warrant hazardous waste regulation under RCRA Subtitle C.

Further, EPA readily acknowledges that it can develop non-hazardous waste rules for CCRs under RCRA's Subtitle D framework that will be fully protective of human health and the environment. The proposed substantive standards for CCR landfills and surface impoundments under the Subtitle C and Subtitle D options are virtually identical. See 75 Fed. Reg. at 35213. Both options would ensure the safe management of CCRs through, among other things, the requirements for the use of liner systems, groundwater monitoring and corrective action. A key difference, however, is that the Subtitle D rules can be implemented in a quick, practical and cost-effective manner without the collateral adverse impacts of the Subtitle C option. Indeed, EPA has determined that the CCRs being remediated from the release at TVA's Kingston facility – one of the principle rationales for this rulemaking – can be safely disposed of in a unit operating in compliance with applicable RCRA Subtitle D requirements. See Administrative Order and Agreement on Consent at ¶ 45, In the Matter of TVA Kingston Fossil Fuel Plant Release

Site, Roane County, Tennessee, (May 6, 2009) (attached as Appendix 2). Moreover, since the TVA ash release, EPA has been conducting comprehensive structural integrity assessments of CCR impoundments across the country. Of the more than 120 CCR impoundments evaluated thus far, none of the impoundments has received an -unsatisfactory" rating, meaning that not one impoundment inspected thus far presents an imminent threat of a catastrophic release similar to that which occurred at TVA's Kingston facility. EPA Coal Combustion Residuals Impoundment Assessment Reports, http://www.epa.gov/epawaste/nonhaz/industrial/special/fossil/surveys2/index.htm (last visited Nov. 3, 2010). Thus, the question for EPA is not whether a Subtitle D program for CCRs can fully protect human health and the environment – it undoubtedly can – but rather how best to implement such a program.

In this respect, USWAG believes that any final Subtitle D regulatory program for CCRs must include a mechanism for the administration and implementation of the rules by the states. As EPA itself has concluded, many states already have well-established and protective regulatory controls for CCRs. Therefore, it makes sense to coordinate development of new federal rules for CCRs with existing state regulations. Failure to do so would result in redundant and potentially inconsistent regulations and would contravene the Bevill Amendment's directive that, in developing federal regulations for CCRs, EPA avoid duplication of effort with pre-existing regulatory programs.

As the Agency acknowledges, the final Subtitle D rules for CCRs would be directly enforceable by the states and the public under RCRA's citizen suit provision and violators would be subject to significant penalties. 75 Fed. Reg. at 35211. EPA would also retain its imminent and substantial endangerment authority under RCRA to take

action against any CCR unit that posed a risk to human health and the environment. *Id*.

Additionally, the Agency could use its imminent and substantial endangerment authorities under CERCLA, as well as other federal authorities, including the Clean Water Act, to address circumstances where a CCR unit posed a threat. *Id*.

Notwithstanding this array of enforcement authorities and EPA's earlier finding that they were adequate to ensure compliance with Subtitle D rules for CCRs, 65 Fed. Reg. at 32232, EPA is concerned with the perceived lack of *direct* federal enforceability under a Subtitle D option. In response, USWAG urges EPA to consider carefully the comments from the numerous state environmental protection agencies addressing their ability to fully and effectively enforce the Subtitle D rules for CCRs. The states share EPA's goal of protecting their citizens and their natural resources from the potential mismanagement of CCRs. The testimony and comments of the states underscores their commitment to effectively enforce the Subtitle D regulations for CCRs. For example, the Pennsylvania Department of Environmental Protection testified at the CCR public hearing in Pittsburgh that there is a 96% compliance rate by facilities in that state with Pennsylvania's Subtitle D rules. Testimony of the Pennsylvania Department of Environmental Protection during EPA's CCR Public Hearing, Pittsburgh, PA, (Sept. 21, 2010) (attached as Appendix 3). This commitment by the states should allay EPA's concerns over the adequacy of the compliance mechanisms under a Subtitle D regulatory regime for CCRs.

Nonetheless, to the extent that EPA continues to believe that some degree of direct federal enforceability is necessary under the Subtitle D option, there are options available to EPA to achieve this result. These include, among others, developing a

federally-enforceable program for CCRs under RCRA Subtitle D using the same statutory authorities the Agency employed to promulgate the federally enforceable Subtitle D rules for municipal solid waste landfills. USWAG supports EPA's development of similar federally enforceable non-hazardous waste regulations for CCRs and we have outlined the legal basis for this approach in our comments.

In the end, however, EPA must not lose sight of the fact that the legal, practical and policy rationales of the Subtitle D option over the Subtitle C option are far too great to allow this enforcement concern to dissuade EPA from pursuing the Subtitle D approach for CCRs. For this reason, the Agency must carefully evaluate whether its concern with federal enforceability under the Subtitle D option is truly warranted and, if so, employ the necessary means to address this concern and *not* allow it to serve as a barrier to pursuing the Subtitle D approach.

Indeed, while USWAG is fully supportive of a federal Subtitle D program for CCRs, it is adamantly opposed to the proposed Subtitle C option. Under this approach, EPA would subject CCRs destined for disposal to the full range of RCRA Subtitle C hazardous waste regulations. 75 Fed. Reg. at 35173. The Agency would regulate CCRs under the Subtitle C hazardous waste rules by listing CCRs as a hazardous waste under EPA's hazardous waste listing criteria. *Id.* While EPA would label CCRs subject to hazardous waste regulation as -special wastes," this would be done solely for the purpose of attempting to deflect the stigmatic effects and liability concerns on beneficial use that would result from the regulation of CCRs under EPA's hazardous waste regulations. *Id.* at 35185. As even EPA concedes, the -special waste" label does nothing to alter the fact that, under the Subtitle C option, CCRs destined for disposal

would be subject to the full array of federal hazardous waste regulations. *Id.* at 35174. In fact, under the Subtitle C option, EPA would seek to regulate CCR disposal practices *more* stringently than any other hazardous waste by proposing to take the unprecedented step of subjecting previously closed and/or inactive CCR surface impoundments to full hazardous waste regulation. *Id.* at 35174, 35177.

As discussed in detail below, the Subtitle C option would constitute an impermissible reversal of EPA's final 1993 and 2000 Regulatory Determinations that CCRs do not warrant hazardous waste regulation and would be in direct contravention of RCRA's plain language establishing an unambiguous statutory process by which EPA can determine whether and how to regulate CCRs under RCRA. Having completed the Bevill determination process for CCRs in 2000 pursuant to the statutorily prescribed procedures, EPA is not at liberty to undo that process. Even assuming that the statute did allow EPA to make a new determination more then ten years after completing the Bevill process, EPA would at a minimum be obligated to adhere to the statutory procedures that Congress prescribed for making the original regulatory determination, including providing a Report to Congress setting forth its recommendation. EPA has ignored these procedures in the Subtitle C proposal.

There are additional legal and policy reasons why EPA cannot choose the Subtitle C option for CCRs. These points are detailed in our comments, but include the following issues:

 Contrary to EPA's theory that regulating CCRs under RCRA Subtitle C will increase beneficial use, the testimony and comments from numerous persons, including CCR marketers and end users, the states, independent standard

Duke Energy Progress, LLC Late-Filed Exhibit No. 16 Docket No. E-2, Sub 1219 Page 73 of 439

setting organizations and the electric industry show that regulating CCRs under RCRA's hazardous waste regulations would cripple the beneficial use market; the docket already contains record evidence identifying specific examples of how the mere specter of Subtitle C hazardous waste regulation is already adversely affecting CCR beneficial use projects;

- The Subtitle C option would impose unduly stringent regulations and staggering compliance costs for the management of CCRs in the range of at least \$55.3 to \$74.5 billion, which is nearly three to four times EPA's projected Subtitle C compliance costs of approximately \$20.3 billion;
- The Subtitle D rules would provide equally protective controls at far lower costs
  than the Subtitle C option. Thus, the Subtitle C option directly contravenes
  Congress' express purpose in enacting the Bevill Amendment that EPA avoid the
  imposition of the onerous economic burden of stringent Subtitle C hazardous
  waste controls on electric utilities if at all possible;
- Contrary to EPA's suggestion that electric utility CCR disposal practices would remain unchanged under the Subtitle C option, at least 15 to 21 million tons of CCRs would enter the Subtitle C hazardous waste commercial disposal market on an annual basis, in contrast to the 2 million tons of hazardous waste currently disposed of in commercial hazardous waste landfills; this influx of Subtitle C wastes into the system would quickly overwhelm the approximately 23 to 30 million tons of total commercial hazardous waste landfill disposal capacity in the nation and create an immediate compliance dilemma for electric utilities. This capacity shortfall would create havoc for the entire RCRA Subtitle C hazardous

Duke Energy Progress, LLC Late-Filed Exhibit No. 16 Docket No. E-2, Sub 1219 Page 74 of 439

waste program and would, among other things, adversely impact hazardous waste remediation projects across the country;

- The Subtitle C hazardous waste option is inconsistent with the views of the other federal and state agencies that Congress directed EPA to consult with in rendering a final regulatory determination for CCRs; all of these agencies have either strongly opposed regulating CCRs under the Subtitle C hazardous waste regulations or have sharply questioned the appropriateness of this option;
- Due to the sheer volume and the physical composition of CCRs, de minimis volumes of CCRs are inevitably released during normal power generation and subsequent CCR handling operations; if CCRs are regulated as a listed -special waste" subject to full hazardous waste regulation, these de minimis releases would constitute improper hazardous waste disposal, subjecting electric utility coal-fired power plants to potential liability for what would likely be a perpetual state of RCRA non-compliance; and
- The Subtitle C option includes the unprecedented and unlawful proposal to extend RCRA Subtitle C hazardous waste jurisdiction to previously closed and/or inactive CCR surface impoundments that do not receive CCRs after the effective date of the final Subtitle C rule. This retroactive application of RCRA would not be sustained because it violates the plain language of RCRA and controlling legal precedent.

USWAG is not alone in its adamant opposition to the Subtitle C option. A bi-partisan group of 165 members of Congress, including a majority of the House Energy and Commerce Committee, 45 U.S. Senators, virtually all the states, other

federal agencies, municipal and local governments, CCR marketers and beneficial users, unions, state PUCs, and many other third-parties have made clear to EPA during this rulemaking process that regulating CCRs under RCRA's hazardous waste program would result in excessive regulation, and, in fact, would be counter-productive because it would cripple the CCR beneficial use industry. In short, the record is clear that regulating CCRs as a listed waste under RCRA Subtitle C, the most burdensome and extreme option under federal law, is not authorized or warranted.

Our comments address the above issues in detail. Chapter 1 addresses why EPA cannot pursue the Subtitle C option for CCRs. In particular, Section I discusses the legislative history and text of the Bevill Amendment, as the Amendment sets forth the statutory conditions against which to evaluate the proposed Subtitle C option.

Section II provides a chronology of EPA's implementation of the Bevill Amendment over the last 30 years, including the Agency's final Regulatory Determinations that CCRs should not be regulated under Subtitle C of RCRA. Section III details why EPA's proposed Subtitle C option is inconsistent with the text and purpose of the Bevill Amendment. Section IV explains why, even assuming EPA had the authority to reverse its final Regulatory Determinations for CCRs, the Subtitle C option would nonetheless be inconsistent with the Bevill Amendment and would be arbitrary and capricious.

Section V discusses why the CCR units that qualify as proven -damage cases" do not justify the Subtitle C option.

Chapter 2 presents additional reasons why the Subtitle C option is legally flawed and cannot be pursued. In particular, Section VI explains why EPA's proposed listing of CCR, is overly broad and legally defective. Section VII details critical flaws in EPA's

Regulatory Impact Analysis (-RIA") for the Subtitle C option, including its underestimation of the true costs of this option. Section VIII outlines some of the practical compliance and disposal dilemmas presented by the Subtitle C option and Section IX explains why EPA cannot retroactively regulate under the Subtitle C option previously closed and/or inactive CCR surface impoundments. Finally, Section X addresses EPA's failure to convene a small business advocacy review panel to assess the impact of the proposal on utilities that qualify as -small business" under the Regulatory Flexibility Act (-RFA").

Chapter 3 addresses the proposed Subtitle D option. Section XI discusses USWAG's support for the Subtitle D Prime option, while making clear that any final Subtitle D rule must include a mechanism for state implementation of the federal Subtitle D rules. This section also includes a discussion of USWAG's strategy for Agency implementation of the Subtitle D rules in a manner that would allow for federal enforceability in instances where the states do not appropriately implement the federal rules. Section XII details the amendments that we believe are necessary under the Subtitle D option, including important changes to certain proposed definitions. Section XIII outlines certain amendments to the proposed regulations to ensure that EPA does not inappropriately regulate as CCR disposal those activities that constitute legitimate CCR beneficial use. Finally, Section XIV outlines USWAG's support for EPA's proposed structural integrity requirements for CCR surface impoundments.

#### **CHAPTER ONE**

# EPA CANNOT PROMULGATE SUBTITLE C REGULATIONS FOR COAL COMBUSTION RESIDUALS

#### I. THE BEVILL AMENDMENT

## A. The Legislative History Of The Bevill Amendment

The Bevill Amendment to RCRA, enacted in 1980, was an outgrowth of Congress's concern that EPA might discourage the development of coal as one of the nation's primary sources of energy if it prematurely and inappropriately subjected CCRs to regulation under Subtitle C of RCRA. Representative Bevill of Alabama, sponsor of the Amendment in the House, feared that the cost of hazardous waste regulation of CCRs would have an immediate inflationary effect on the economy and a disastrous impact on coal utilization.

To avoid this result, Representative Bevill explained in unambiguous terms that the purpose of the Amendment was to:

encourage development of coal as a primary domestic source of energy, avoid unnecessary inflationary impact, and focus the efforts of the Environmental Protection Agency in implementing the Resource Conservation and Recovery Act toward activities truly necessary to protect public health and the environment, specifically, it would require EPA to defer imposition of regulatory requirements on the disposal of the waste by-product of fossil fuel combustion, of discarded mining materials and of cement kiln dust until after EPA has completed studies to determine whether, if at all, these materials present any hazard to human health or the environment. These studies would include evaluation of the economic and environmental aspects of existing and alternative disposal and reuse options. EPA would also be required to focus on the impact of these alternatives on the use of our coal and other natural resources.

126 Cong. Rec. 3361 (1980).<sup>2</sup> Representative Bevill estimated that compliance with the proposed Subtitle C program would cost the utility industry \$1 billion in the first three years, and that these -eosts are likely to be passed along to consumers, with an immediate inflationary impact." 126 Cong. Rec. at 3361; see also id. at 3363 (remarks of Rep. Findley); 125 Cong. Rec. at 13245 (remarks of Sen. Huddleston); id. at 13246 (remarks of Sen. Ford). As the D.C. Circuit has made clear, the emphasis on economic factors . . . is consistent with . . . . Congress's obvious goal in enacting the Bevill Amendment – to relieve [the Bevill industries] of the onerous economic burden of stringent Subtitle C controls if at all possible." Envtl. Def. Fund v. EPA, 852 F.2d 1309, 1315 (D.C. Cir. 1988) (emphasis added) (-EDF v EPA"). As discussed below, the projected compliance costs associated with EPA's proposed Subtitle C option have skyrocketed to a range of at least \$5.32 to \$7.62 billion per year. See EPRI, 2010, Cost Analysis of Proposed National Regulation of Coal Combustion Residuals from the Electric Generating Industry, EPRI, Palo Alto, CA: 2010. 1022296 at 4-1 (-EPRI Cost Report").

Additionally, other supporters of the Amendment have been acutely sensitive to the nation's dependence on foreign energy sources; they intended the Bevill Amendment to reflect congressional policy that any regulatory program for CCRs under

<sup>&</sup>lt;sup>2</sup> On the same day that the Senate passed the Huddleston Amendment (the Senate predecessor to the Bevill Amendment), it also adopted a resolution expressing concern about the national security implications of dependence on foreign oil supplies and calling on the President to propose a plan for displacing specific percentages of foreign oil with domestic coal. S. Res. 175, 96th Cong., 1st Sess. (1979), 125 Cong. Rec. 13241 (1979).

<sup>&</sup>lt;sup>3</sup> While the *EDF v. EPA* decision involved a challenge to EPA's regulatory determination not to regulate Bevill mining wastes under RCRA Subtitle C, the Court's description of the purpose of the Bevill Amendment to have EPA avoid Subtitle C regulation if at all possible for Bevill wastes obviously applies to all the Bevill wastes, including CCRs.

RCRA not create disincentives for the use of coal, including regulations that would cause utilities to switch fuels from coal to other sources. *See*, *e.g.*, 126 Cong. Rec. at 3363 (remarks of Rep. Rahall) (-[a]t a time when we are seeking to encourage electric utilities and others to switch from the burning of oil to coal, it would be highly inappropriate to place further unnecessary regulatory roadblocks in the way of increased coal usage"); 125 Cong. Rec. at 13246 (remarks of Sen. Byrd) (under this amendment, -those who produce coal will be able to do so without the fear that utility customers will be forced to use some other fuel because of environmental regulations").

The Amendment also reflected the congressional intent to promote the beneficial use of CCRs since — national commitment to encourage reuse of such materials as fly ash was a key element of RCRA," which Representative Bevill observed —seem[ed] to have not received adequate attention at EPA." 126 Cong. Rec. at 3362.4

Representative Findley of Illinois, another supporter of the Amendment, was more direct in his caution to EPA about the adverse impact on CCR beneficial use resulting from the regulation of CCRs under RCRA Subtitle C. He warned that if the Agency were to regulate CCRs under RCRA's hazardous waste program, the —flourishing industry which recycles these byproducts would be gravely disrupted and possibly closed down." *Id.* at 3363.

Representative Rahall of West Virginia was equally direct that EPA avoid any unnecessary regulation of CCRs under RCRA Subtitle C for fear that such a program

<sup>&</sup>lt;sup>4</sup> Representative Bevill emphasized that the beneficial use of CCRs —saves energy," explaining, for example, that the use of fly ash in cement results in a 15 percent savings in the amount of energy needed to that product, and is cost effective, saving tax payers 10-15 percent in federally sponsored concrete construction projects. 126 Cong. Rec. at 3362.

would devastate the CCR beneficial use market. He urged that -avoidance of unnecessary regulation of coal byproducts is particularly important because of the impact such regulation will have on reuse." *Id.* He stressed that it was critical to avoid such unnecessary regulation in light of the fact that CCRs used in the beneficial use market compete with a variety of other natural aggregates and materials:

It is obvious that the imposition of any expensive regulatory requirements on these developing industries will have a disastrous impact on them. It is thus vital that EPA fully evaluate a need for regulation, and the alternatives to various regulatory schemes, before blindly and only because coal is involved, imposing regulations on these coal byproduct materials.

Id. at 3363-64. Representative Staggers, Chairman of the Interstate and Foreign Commerce Committee, which reported the RCRA reauthorization bill, echoed the concerns about the adverse impact on CCR beneficial use if CCRs were regulated under Subtitle C:

[Subtitle C regulation of fossil fuel combustion wastes] would run counter to one of the principal designs of the Resource Conversation and Recovery Act -- conservation of valuable material ... Coal combustion products, including particularly fly ash, provide significant beneficial reuses and substitution for other more costly materials ... The act [RCRA] is intended to encourage, not discourage, such beneficial reuses.

126 Cong. Rec. at 3364-65. *See also id.* at 3362 (remarks of Rep. Bevill), 3362-63 (remarks of Rep. Horton), 3363 (remarks of Rep. Findley & Rep. Rahall); 125 Cong. Rec. at 13245 (remarks of Sen. Huddleston). Senator Huddelston summed up the purpose of the Bevill Amendment by making clear to EPA that such regulation was to be an option of last resort:

In my amendment [the Senate equivalent to the Bevill amendment] the Administrator should be extremely cautious in imposing any additional regulations on these materials, particularly in view of the resulting

discouragement of coal use and the recovery and reuse of valuable fossil fuel combustion byproducts.

125 Cong. Rec. at 13245.5

### B. The Text Of The Bevill Amendment

The enacted Amendment ultimately incorporated these congressional policy directives into RCRA by first suspending EPA's authority to regulate CCRs under Subtitle C of RCRA until EPA completed a study to determine whether such regulation was warranted. Specifically, the Bevill Amendment provides that fly ash waste, bottom ash waste, slag waste, and flue gas emission control waste generated primarily from the combustion of coal or other fossil fuels shall be subject to regulation only under other applicable provisions of federal or state law in lieu of RCRA Subtitle C until at least six months after the date of submission to Congress of a comprehensive study of those wastes. RCRA § 3001(b)(3)(A), 42 U.S.C. § 6921(b)(3)(A).

This comprehensive study was to assess the environmental risks, if any, posed by CCRs, taking into account the following eight factors:

(1) the source and volumes of such material generated per year;

<sup>&</sup>lt;sup>5</sup> Both amendments were virtually identical in their treatment of utility combustion wastes. Compare 125 Cong. Rec. 13241, 13244 (1979) with 126 Cong. Rec. 3359-61 (1980). The significant difference in the two versions was that the Senate amendment also suspended the regulation of certain oil, natural gas and geothermal energy wastes, while the House amendment suspended the regulation of mining and cement kiln dust wastes. The Conference Committee reconciled the two versions by adding the Senate's oil, natural gas and geothermal waste suspension to the House version on utility, mining and cement kiln dust wastes. See H.R. Conf. Rep. No. 96-1444 at 31-32 (1980), reprinted in 1980 U.S.C.C.A.N. 5030-31.

<sup>&</sup>lt;sup>6</sup> This section reads in pertinent part:

Notwithstanding the provisions of paragraph (1) of this subsection, each waste listed below shall, except as provided in subparagraph (B) of this paragraph, be subject only to regulation under other applicable provisions of Federal or State law in lieu of this subtitle until at least six months after the date of submission of the applicable study required to be conducted under subsection .... (n) .... of section 6982 of this title and after promulgation of regulations in accordance with subparagraph (C) of this paragraph:

<sup>(</sup>i) Fly ash waste, bottom ash waste, slag waste, and flue gas emission control waste generated primarily from the combustion of coal or other fossil fuels.

- (2) present disposal and utilization practices;
- (3) potential danger, if any, to human health and the environment from the disposal and reuse of such materials;
- (4) documented cases in which danger to human health or the environment from surface runoff or leachate has been proved;
- (5) alternatives to current disposal methods;
- (6) the costs of such alternatives;
- (7) the impact of those alternatives on the use of coal and other natural resources; and
- (8) the current and potential utilization of such materials.

RCRA § 8002(n), 42 U.S.C. § 6982(n).

To ensure that all relevant information was available to the Agency and to avoid duplication of effort, Congress specifically directed EPA to review studies and other actions of other Federal and State agencies .... and invite participation by other concerned parties, including industry and other Federal and State agencies ....." *Id.* 

Within six months following completion of the study and a report to Congress on the Agency's findings, EPA was required to make a final regulatory determination whether to (1) subject to Subtitle C regulation the CCRs addressed in the study and report to Congress, or (2) determine that Subtitle C regulation of these wastes is not warranted. The Bevill Amendment required the Agency to base this final regulatory determination -en information developed or accumulated pursuant to such study, public hearings, and comment. . . ." RCRA § 3001(b)(3)(C), 42 U.S.C. § 6921(b)(3)(C). In short, EPA could only reach the decision to regulate CCRs under Subtitle C after following a statutorily-prescribed series of steps that involved congressional oversight marked by EPA's submission of reports to Congress that would inform the relevant

congressional committees of jurisdiction of the Agency's preliminary recommendations. The final regulatory determination was to be issued according to a statutorily-imposed deadline and based on public hearings and comments responding to the report to Congress. See EDF v. EPA, 852 F.2d at 1314 (—The statute clearly states that the agency is to base its regulatory determination on the information gathered for the § 8002(p) study.").

Also, it is important to appreciate that Congress was particularly concerned that EPA not make this decision in a vacuum. To that end, Congress emphasized that the Agency was to seek the assistance and cooperation of those with expertise in the field including -ether agencies of Government that are aware of the role coal plays in our national energy policy, or of actual disposal and utilization practices." 126 Cong. Rec. at 3362. These included the Department of Energy, the Department of the Interior, the Federal Highway Administration, Department of Commerce, and the Department of Agriculture. *Id.* The input of these other federal agencies to assist EPA in reaching a final regulatory determination was a critical component of the Bevill Amendment; it was important to Congress that the final determination reflect -a cooperative, informed effort" that met the nation's energy goals. *Id.* 

Also of critical importance to Congress was the Bevill Amendment's decision-making process; these provisions reflect congressional intent that EPA reach all final decisions in a transparent manner that provides the public and Congress with an opportunity to determine whether any decision to regulate CCRs under RCRA Subtitle C was appropriate. Representative Bevill emphasized this point by stressing that, after EPA issued its statutorily-required report to Congress containing its recommendation

whether CCRs warranted regulation under RCRA Subtitle C, the Agency would be required to obtain public views on [the report] and to make known whether as a result of this process, EPA believes any regulation of these materials is necessary." *Id.*Congress viewed this requirement as especially important in view of our national commitment to develop a coherent and consistent policy toward the use of our coal and other energy resources." *Id.* The public notice and comment requirement was also intended to allow interested parties to evaluate the basis of the Agency's decision, and to address the question of what degree of regulation, if any, is appropriate" for CCRs.

Taking into consideration these congressional concerns, EPA completed its statutory obligation under the Bevill Amendment in May 2000, when it issued its final Regulatory Determination concluding that CCRs do not warrant regulation under Subtitle C of RCRA.

#### II. IMPLEMENTATION OF THE BEVILL AMENDMENT

Shortly after enactment of the Bevill Amendment in 1980, EPA began the required report to Congress (—Report to Congress") but soon realized that the scope of such a report would be so broad that the Agency would have difficulty completing the work by the statutory deadline of October 1982. As a result, EPA divided the statutorily-mandated report to Congress into two phases. The initial phase addressed the four large-volume CCRs generated by coal-fired electric utilities, including: fly ash, bottom ash, boiler slag, and flue gas emission control waste. See 1988 Report to Congress at ES-2; 58 Fed. Reg. 42466, 42467 (Aug. 9, 1993).

The second phase covered all other combustion wastes generated by utilities and non-utilities, including the above-identified CCRs when co-managed with certain low volume wastes. *See id.; see also* 1999 Report to Congress at ES1-2; 65 Fed. Reg. 32214, 32218 (May 22, 2000).

## A. EPA's March 1988 Report To Congress On CCRs

Pursuant to this two-phase approach, EPA published its first Report to Congress on March 8, 1988, concluding that the large-volume coal combustion waste streams listed above generally do not exhibit hazardous characteristics under current RCRA regulations, and therefore —EPA does not intend to regulate under [RCRA] Subtitle C fly ash, bottom ash, boiler slag, and flue gas desulfurization wastes. 1988 Report to Congress at ES-6. The 1988 Report to Congress also encouraged the use of coal combustion wastes to the extent such use can be done in an environmentally safe manner. *Id.* at ES-8.

#### B. EPA's 1993 Final Regulatory Determination For CCRs

EPA published its first final Regulatory Determination for CCRs on August 9, 1993.<sup>7</sup> This final Regulatory Determination set forth EPA's conclusion that RCRA Subtitle C regulation of the four large-volume CCRs – fly ash, bottom ash, boiler slag, and flue gas desulfurization waste – was not warranted. 58 Fed. Reg. 42466 (Aug. 9, 1993). Specifically, the Agency concluded that:

<sup>&</sup>lt;sup>7</sup> In September 1991, an Oregon-based citizens group, Citizens Interested in Bull Run, Inc., filed a RCRA citizen suit against EPA for the Agency's alleged failure to meet certain statutory deadlines, including the Bevill Amendment's October 1982 statutory deadline for issuing a final regulatory determination for CCRs. *Gearhart v. Reilly*, No. 91-2345 (D.D.C. 1991) (*–Bull Run*"). USWAG intervened in the litigation, and EPA entered into a settlement agreement in which it agreed to issue the final regulatory determination for the electric utility-CCRs addressed in the 1988 Report to Congress by August 2, 1993, while deferring a supplemental report to Congress and regulatory determination on all other fossil fuel combustion wastes, including CCRs managed with certain low volume wastes, until April 1, 1998.

Regulation under Subtitle C of RCRA is inappropriate for the four waste streams that were studied [fly ash, bottom ash, boiler slag, and flue gas emission control waste] because of the limited risks posed by them and the existence of generally adequate State and Federal regulatory programs. The Agency also believes that the potential for damage from these wastes is most often determined by site- or region-specific factors and that the current State approach to regulation is thus appropriate. Therefore, the Agency will continue to exempt these wastes from regulation as hazardous wastes under RCRA Subtitle C .... EPA will consider these wastes during the Agency's ongoing assessment of industrial non-hazardous wastes under RCRA Subtitle D.

Id. at 42466.

EPA's conclusion not to regulate CCRs under Subtitle C was also based on the inflexible Subtitle C hazardous waste framework, which the Agency found inappropriate for CCRs. The Agency stated that -it is unlikely that Subtitle C would effectively address the problems associated with the four large-volume fossil-fuel combustion wastes [CCRs] without imposing unnecessary costs." *Id.* at 42477. EPA reached this conclusion after evaluating a number of factors, including -the impacts to the industry that regulation under Subtitle C would create . . . ." *Id.* at 42471. Not surprisingly, EPA found that the rigid nature of the Subtitle C regulatory system would result in excess costs and unnecessary regulation as applied to CCR management units:

A Subtitle C system would require coal combustion units to obtain a [RCRA] Subtitle C permit (which would unnecessarily duplicate existing State requirements) and would establish a series of waste unit design and operating requirements for these wastes, which would generally be in excess of requirements to protect human health and the environment. For example, if such wastes were placed in the Subtitle C universe, all ash disposal units would be required to meet specific liner and monitoring requirements. Since [CCR] sites vary widely in terms of topographical, geological, climatological, and hydrological characteristics (e.g., depth to groundwater, annual rainfall, distance to drinking water sources, soil type) and the wastes' potential to leach into the groundwater and travel to exposure points is linked to such factors, it is more appropriate for

individual States to have the flexibility necessary to tailor specific controls to the site or region specific risks posed by these wastes.

Id. at 42477.

### C. EPA's March 1999 Report To Congress On CCRs

Following issuance of the 1993 Regulatory Determination, EPA began work on its second report to Congress on the remaining fossil fuel combustion wastes, including CCRs managed with certain low-volume wastes. EPA issued its second Report to Congress in March 1999, reiterating that that [RCRA] Subtitle C is inappropriate to address any problems associated with disposal of these wastes and that the continued use of site and region specific approaches by the states is more appropriate for addressing the limited human health and environmental risks that may be associated with disposal of these wastes." 1999 Report to Congress Vol. 1 at 3-5. EPA also reiterated its belief that regulating CCRs under RCRA's hazardous waste program would unnecessarily duplicate existing state requirements, and would establish a series of waste unit design and operating requirements for these wastes that would most often exceed the requirements necessary to protect human health and the environment. *Id.* 

## D. <u>EPA's 2000 Final Regulatory Determination For CCRs</u>

EPA completed the statutorily-required Bevill Amendment process in May 2000 when it issued its final Regulatory Determination for all remaining fossil fuel combustion wastes, including CCRs managed with certain low-volume wastes. 65 Fed. Reg. 32214 (May 22, 2000). Based on the record evidence, hearings, and public comment on the 1999 Report to Congress, EPA once again concluded that CCRs and other fossil fuel combustion wastes did not warrant regulation under RCRA's Subtitle C hazardous waste program. EPA concluded instead that the development of national non-

hazardous solid waste regulations under RCRA Subtitle D is appropriate when CCRs are disposed of in landfills or surface impoundments. *Id.* 

In addition to the fact that CCRs rarely exhibit a hazardous waste characteristic (*id.* at 32222), EPA explained that its final determination was based on the following factors: improving trends in present CCR disposal and utilization practices per the Bevill study factor in section 8002(n)(2); the current and potential utilization of CCRs (*i.e.*, beneficial use) per the Bevill study factor in section 8002(n)(8); and Congress's admonition in the Bevill Amendment against duplication of efforts by other federal and state agencies. *Id.* at 32215. EPA pointed to the record evidence, as set forth in the 1999 Report to Congress, that the utility industry had made significant improvements in its waste management practices and to similar record evidence showing that -{CCR regulatory} programs have, in fact, substantially improved over the last 15 years or so. *Id.* at 32222, 32228-29.

In fact, EPA found that the ability for most states to impose specific regulatory controls for coal combustion wastes has increased almost three-fold over the past 15 years. *Id.* at 32230. EPA concluded that with the exception of relatively few states, the regulatory infrastructure is generally in place at the state level to ensure adequate management of these wastes. *Id.* at 32217 (as discussed below, a DOE/EPA Report shows that state CCR regulatory programs have continued to improve since EPA's 2000 final Regulatory Determination). Thus, EPA concluded that -we believe that subtitle D regulations are the most appropriate mechanism for ensuring that these wastes disposed in landfills and surface impoundments are managed safely." *Id.* at 32221.

Given state law, and the evidence that CCR controls were improving, EPA was mindful of Congress's directive in section 8002(n) that the Agency, in reaching a final determination, consider [the] action of state and other federal agencies with a view to avoiding duplication of effort. *Id.* at 32217. Consistent with this statutory dictate, EPA concluded that Subtitle D controls will provide sufficient clarity and incentive for states to close the remaining gaps in coverage, and for facilities to ensure that their wastes are managed properly. *Id.* at 32217.

Also critical to EPA's final determination that CCRs do not warrant regulation under Subtitle C was the Bevill Amendment's direction in RCRA section 8002(n) that any final decision take into account the current and potential utilization of such materials. In particular, EPA saw a potential downside to pursuing a Subtitle C approach, citing the concerns of commenters that subjecting any coal combustion wastes to a Subtitle C regime would place a significant stigma on these wastes, notably that it would adversely impact beneficial reuse. *Id.* at 32217. EPA acknowledged the possibility that the [Subtitle C] approach could have unintended consequences, and that it was particularly concerned about any adverse effect on the beneficial re-use market for these wastes. *Id.* Thus, while normally concerns about stigma are not a deciding factor in EPA's decisions under RCRA, EPA concluded that, given its conclusion that the subtitle D approach here should be fully effective in protecting human health and the environment, and given the large and salutary role that beneficial reuse plays for

this waste, concern over stigma is a factor supporting our decision today that subtitle C regulation is unwarranted in light of our decision to pursue a subtitle D approach." Id.8

#### Joint DOE/EPA Report On State CCR Programs Ε.

After EPA issued its final 2000 Regulatory Determination, the Agency continued to evaluate options for implementing Subtitle D non-hazardous waste regulations for CCRs. As part of this process, in 2005 and 2006, EPA and the U.S. Department of Energy (DOE) jointly conducted a study of current and recent (1) management practices for CCR disposal by industry, (2) state regulations for CCR management, and (3) implementation of state requirements by state authorities. This study examined landfills and surface impoundments that were permitted, built, or laterally expanded between January 1, 1994 and December 31, 2004 to determine what changes, if any, in CCR disposal management practices and state regulatory controls had occurred since the 1994 close of the administrative record for the Bevill Regulatory Determination and EPA's 1999 Report to Congress.

In a report titled -Coal Combustion Waste Management at Landfills and Surface Impoundments, 1994-2004" (-DOE/EPA Report"), DOE and EPA found that, -[b]etween 1994 and 2004, the amount and quality of environmental controls used at CCW

<sup>&</sup>lt;sup>8</sup> It is also is worth noting that EPA considered, and ultimately rejected, a so-called Subtitle C -contingent management" option for CCRs. This option was similar to that proposed for cement kiln dust, and would have involved establishing federal Subtitle D non-hazardous waste rules for CCR disposal facilities. CCRs managed in accordance with the applicable Subtitle D standards would remain exempt from Subtitle C hazardous waste controls. Id. at 32232. However, those facilities that committed —egrequius or repeated violations of" the applicable Subtitle D CCR management standards would have been — woved into the subtitle C program" and the CCRs at such facilities would have been subject to tailored Subtitle C hazardous waste controls. Id. Even though the contingent management Subtitle C option evaluated and rejected by EPA in 2000 was less sweeping in scope than the current Subtitle C option, EPA's concern with the adverse stigmatizing effects of this more limited option on the beneficial use of CCRs were significant enough to influence EPA's determination that the Subtitle C contingent management option was not warranted for CCRs.

management units appear to have increased." DOE/EPA Report at S5-S7. The DOE/EPA Report cited improvements in the management of CCRs in new or expanded landfills and surface impoundments, including the use of liners and the monitoring of groundwater. *Id*.

EPA and DOE also examined whether state CCR regulatory controls had similarly improved and found that: -[i]n eight areas of regulatory control reviewed for this report, more CCWs destined for landfills in the States reviewed had tightened regulatory controls than had relaxed controls between the times data were collected for the 1988 [Report to Congress] and for this report." *Id.* at S7. Consistent with this continuing improvement in the administration of state CCR controls, EPA and DOE also found that the grants of variances from regulatory controls by state regulators had sound scientific support. *Id.* at S7-S11.

Based on the above, DOE/EPA concluded that -{t}he data and analyses documented in this report provide new information that appears to show improved management of CCWs in both landfills and surface impoundments." *Id.* at S-11. The findings of the DOE/EPA Report are significant because they provide updated information supporting the conclusions in the final 2000 Regulatory Determination that Subtitle C regulation of CCRs is not warranted due, in part, to the significant and continuing improvements in CCR waste management practices and corresponding improvements in state CCR regulatory programs. *See* 65 Fed. Reg. at 32222. Since completion of the DOE/EPA Report in 2006, several states have proposed and/or implemented further enhancements of their CCR controls.

## F. <u>Notice Of Data Availability</u>

As part of the process for following-up on its 2000 Regulatory Determination to develop Subtitle D non-hazardous waste regulations for CCRs, EPA issued a Notice of Data Availability (-NODA") in August 2007 announcing the availability of new information on the management of CCRs in landfills and surface impoundments. EPA sought comment on this new information to better assess -how, if at all, this additional information should affect the Agency's decisions as it continues to follow-up on its Regulatory Determination for [CCRs] disposed of in landfills and surface impoundments." 72 Fed. Reg. 49714 (Aug. 29, 2007). The new information included (1) the joint DOE/EPA study referenced above providing an update on the status of state CCR disposal regulations, and (2) a draft risk assessment conducted by EPA on the management of CCRs in landfills and EPA's assessment of CCR disposal units that it had identified as -damage cases." Id. EPA also sought comment on a rulemaking petition filed by citizens' groups, as well as approaches suggested by the electric utility industry, including USWAG, regarding the management of CCRs. EPA stated that it would consider the new information and the associated comments, as well as the results of the peer review of the draft risk assessment, -as it continues to follow-up on its Regulatory Determination for [CCRs] disposed of in landfills and surface impoundments." Id.

# III. EPA MAY NOT REVERSE ITS FINAL REGULATORY DETERMINATIONS THAT CCRS DO NOT WARRANT REGULATION UNDER RCRA SUBTITLE C

The fundamental flaw in the proposed Subtitle C option is that it is premised on the notion that EPA is at liberty to reverse its final Regulatory Determinations that CCRs do not warrant regulation under Subtitle C of RCRA. Because the statute does not

Duke Energy Progress, LLC Late-Filed Exhibit No. 16 Docket No. E-2, Sub 1219 Page 93 of 439

authorize EPA to reverse these final Regulatory Determinations, EPA may not adopt the Subtitle C hazardous waste option for CCRs. Even assuming for purposes of argument that the statute did authorize such reversal, EPA's proposal to regulate CCRs under Subtitle C does not comport with the Bevill Amendment's statutorily-prescribed decision-making process.

While EPA's current policy goals may favor the Subtitle C option over the Subtitle D option, policies cannot trump the law. The United States Court of Appeals for the District of Columbia Circuit cautioned against precisely such precipitous agency action by new administrations when, in the pursuit of political and policy agendas, agencies stray beyond the plain language of their authorizing statutes:

We recognize that a new administration may try to effectuate new philosophies that have been implicitly endorsed by the democratic process. Nonetheless, it is axiomatic that the leaders of every administration are required to adhere to the dictates of statutes that are also products of democratic decision-making. Unless officials of the Executive Branch can convince Congress to change the statutes they find objectionable, their duty is to implement the statutory mandates in a rational manner.

Int'l Ladies' Garment Workers' Union v. Donovan, 722 F.2d 795, 827 (D.C. Cir 1983).

The D.C. Circuit's caution rings true in this case. While Administration and EPA

<sup>&</sup>lt;sup>9</sup> The same caution was echoed by in a recent speech before the Environmental Law Institute by Judge Tatel of the D.C. Circuit:

<sup>[</sup>I]t's at times like these, when a new administration is determined to change environmental policy, that our commitment to the fundamental principles of administrative law is really tested... In both Republican and Democratic administrations, I have too often seen agencies failing to display the kind of careful and lawyerly attention one would expect from those required to obey federal statutes and to follow federal statutes and to follow principles of administrative law. In such cases, it looks to all the world that agencies choose their policy first and then later seek to defend its legality. This gets it backwards.... It's backwards because whether or not agencies value neutral principles of administrative law, courts do, and they will strike down agency action that violates those principles – whatever the president's party, however popular the administration, and no matter how advisable the initiative.

policies may have changed over the last two years, the law governing the decisionmaking process by which EPA may choose to regulate CCRs under RCRA has not.

# A. <u>EPA Completed The Bevill Regulatory Process For CCRs In 2000</u> <u>And The Statute Does Not Authorize EPA To Reverse That</u> Decision

EPA has completed the Bevill process pursuant to a judicially-imposed deadline and determined, in accordance with the congressionally-mandated decision-making process, that CCRs do *not* warrant regulation under Subtitle C. The statutory decision-making process is complete and RCRA does not authorize EPA to decide to take a second bite at the apple and reverse its final CCR Regulatory Determinations in a manner of its own choosing.

Congress established a detailed process by which EPA could determine whether and how to regulate CCRs under RCRA Subtitle C. EPA has completed that statutory process, determining that CCRs do not warrant regulation under RCRA Subtitle C, and concluding instead that CCRs warrant regulation under RCRA's Subtitle D non-hazardous waste program. The plain language of the statute does not contemplate, let alone authorize, EPA undertaking another round of the Bevill decision-making for CCRs. If it did, the statutory process and its associated deadlines for reaching a final regulatory determination would have no meaning and EPA could continually change its mind in an abbreviated fashion, as illustrated by the proposed Subtitle C option.

EPA readily concedes that the Bevill Amendment does not include a process for Agency reversal of the CCR final Regulatory Determinations. In response to an inquiry

The Administrative Process and the Rule of Environmental Law"; the Honorable David S. Tatel, at an Environmental Law Institute symposium entitled An Agenda for the New EPA" (Oct. 6, 2009), (emphasis added).

from Representative Markey following the Kingston ash release regarding whether and how EPA could reverse its final CCR Regulatory Determinations for CCRs, EPA's then-Acting Associate Administrator readily acknowledged that -RCRA does not specifically spell out the process by which we [would] revise the Regulatory Determination." *See* Letter from Joyce K. Frank, EPA Acting Associate Administrator, to Rep. Markey, Chair, House Committee on Energy and Commerce, at 3 (Jan. 30, 2009) (-Frank Letter") (attached as Appendix 4). As the D.C. Circuit has held, Congress' failure to grant an agency a given power is not an ambiguity as to whether that power has, in fact, been granted. On the contrary, and as this Court persistently has recognized, *a statutory silence on the granting of a power is a denial of that power to the agency. American Bus. Ass'n v. Slater*, 231 F.3d 1, 8 (D.C. Cir. 2000) (Sentelle, J., *concurring*) (emphasis added). Here, even EPA acknowledges that the statute does not speak to whether and how EPA can revise its final CCR Regulatory Determinations. The Agency nonetheless assumes that it possesses such power. It does not.

It is a fundamental principle of administrative law that an agency's authority to act must come directly from Congress. Absent such authority, an agency may not act. *See U.S. Telecom Ass'n v. FCC*, 359 F.3d 554, 566 (D.C. Cir. 2004) ([S]tatutory silence simply leaves th[e] lack of authority untouched. In other words, the failure of Congress to use Thou Shalt Not language doesn't create a statutory ambiguity of the sort that triggers *Chevron* deference.); *Ethyl Corp. v. EPA*, 51 F.3d 1053, 1060 (D.C. Cir. 1995) (-We refuse ... to presume a delegation of power merely because Congress has not expressly withheld such power."); *Ry. Labor Executives' Ass'n v. Nat'l Mediation Bd.*, 29 F.3d 655, 671 (D.C. Cir. 1994) (*en banc*) (-Were courts to presume a delegation of

power absent an express withholding of such power, agencies would enjoy virtually limitless hegemony, a result plainly out of keeping with *Chevron* and quite likely with the Constitution as well.") (emphasis in original).<sup>10</sup>

The Bevill Amendment is completely silent on EPA's authority to redo or reverse a final regulatory determination that the Agency has already properly reached pursuant to the Amendment's statutorily prescribed procedures and deadlines. <sup>11</sup> If Congress had wanted to give EPA this authority, it would have done so explicitly. <sup>12</sup> As the above federal decisions make clear, such silence means that EPA does not have the authority to undertake, through a process and timing of its own choosing, a second regulatory determination.

The plain language of the statute provides for only a one-time decision, directing EPA to determine, not later than six months after the statutorily-required report to Congress on CCRs, whether Subtitle C regulation for CCRs is warranted. See RCRA

See also Adams Fruit Co. v. Barrett, 494 U.S. 638, 649 (1990) (—A \_gap' is not created in a statutory scheme merely because a statute does not restate the truism that States may not preempt federal law."); Aid Ass'n for Lutherans v. USPS, 321 F.3d 1166, 1174-75 (D.C. Cir. 2003) (—T]he Postal Service's position seems to be that the disputed regulations are permissible because the statute does not expressly foreclose the construction advanced by the agency. We reject this position as entirely untenable under well-established case law.").

<sup>&</sup>lt;sup>11</sup> Though, as noted above, EPA missed the statutory deadlines set forth in the statute for reaching a final regulatory determination for CCRs; a citizen suit over this missed statutory deadline resulted in EPA being subject to court-ordered deadlines for issuing both the 1993 and 2000 CCR final Regulatory Determinations.

<sup>12</sup> Indeed, when Congress wishes to give an agency the authority to revisit prior statutory compelled actions, it knows how to do so. See e.g., Clean Air Act § 112(c)(1), 42 U.S.C. § 7412(c)(1) (1990) (-GAA") (directing EPA to re-evaluate no less often then every eight years and revise, as appropriate, based on public comment or new information, the list of categories of major sources and area sources listed by EPA under CAA Section 112(c)(3)); see also CAA § 109(d), 42 U.S.C. § 7409(d) (directing EPA to review at five year intervals the criteria for pollutants published under CAA Section 7408 and the national ambient air quality standards promulgated under Section 7409 and make such revisions and promulgate new standards as may be appropriate). As explained above, Congress has not provided EPA with similar authority to reevaluate and revise its final CCR Regulatory Determinations.

§ 3001(b)(3)(C). That determination, which was to be based on information developed or accumulated pursuant to such study, public hearings, and comment, was to be published in the Federal Register and accompanied by an explanation and justification of the reasons for it. *Id.* EPA completed this statutory process for CCRs over ten years ago, concluding for a multitude of reasons that CCRs do not warrant regulation under RCRA Subtitle C. The statute simply does not empower EPA to revisit its final decision and render a new determination.

If EPA wishes once more to undertake the statutorily-prescribed process for reaching a final regulatory determination for CCRs, it must obtain this authority from Congress. If Congress were to grant EPA such authority, it would be reasonable to expect that Congress would establish new statutory deadlines and perhaps new and different statutory study factors for EPA to evaluate in undertaking an additional regulatory determination for CCRs. Absent such authority from Congress, the Agency is not empowered to alter its final 1993 and 2000 Regulatory Determinations for CCRs. Therefore, EPA cannot pursue the Subtitle C option. <sup>13</sup>

# B. <u>Assuming, For Purposes Of Argument, That EPA Is Authorized To Make A Second Regulatory Determination, EPA Has Failed To Adhere To The Bevill Amendment's Prescribed Process</u>

Even assuming that RCRA could be read as authorizing EPA to reverse its final Regulatory Determination for CCRs, the Agency would have to adhere to the statutorily-prescribed procedures that Congress demanded EPA follow in determining whether and

While —[an] agency can normally change its position and reverse a decision ... Congress, however, undoubtedly can limit an agency's discretion to reverse itself[.]" New Jersey v. EPA, 517 F.3d 574, 582-83 (D.C. Cir. 2008). Here, Congress has limited EPA's ability to reverse its final Bevill Regulatory Determination by specifying prescribed procedures and timeframes for completing the determination and not authorizing the Agency to revisit that final determination and issuing a new determination.

how to regulate CCRs. EPA itself has recognized this; in responding to Representative Markey's inquiry to EPA about the steps it would need to take to regulate CCRs under RCRA Subtitle C, the Agency stated that it would be -advisable [for EPA] to go through the same process we followed" in issuing the 1993 and 2000 final Regulatory Determinations. Despite what it told Congress, EPA has inexplicably strayed beyond the plain language of the statute by proposing to reverse its final Regulatory Determinations for CCRs without following the statutorily-required process it followed for issuing those Determinations in the first instance.

As explained above, the Bevill Amendment requires that, before making any final regulatory determination for CCRs, EPA first must study CCRs pursuant to the eight statutory study factors and then -publish a report on such study, which shall include appropriate findings." RCRA § 8002(n). The statute then directs that -such study and findings [i.e., the Report to Congress] shall be submitted to the Committee on Environment and Public Works of the United States Senate and the Committee on Energy and Commerce of the United States House of Representatives." *Id.* Following submission of the report to the appropriate congressional committees, the statute directs EPA to conduct public hearings and provide an opportunity for comment on the Report to Congress, and either determine to promulgate regulations regulating CCRs under RCRA's Subtitle C regulations or determine that such regulations are unwarranted. RCRA § 3001(b)(3)(C). Following this process, EPA was directed to publish its regulatory determination, -which shall be based on information developed or

<sup>&</sup>lt;sup>14</sup> See Frank Letter, attached as Appendix 4.

accumulated pursuant to such study, public hearings, and comment, in the Federal Register accompanied by an explanation and justification of the reasons for it." *Id.* 

Despite these unambiguous statutory conditions for determining whether and how to regulate CCRs under RCRA, EPA has not complied with these conditions in proposing to reverse and reissue its final Regulatory Determination for CCRs. Among other things, EPA has not prepared a Report to Congress setting forth its rationale for proposing to reverse course and issuing a new regulatory determination declaring that CCRs now warrant Subtitle C regulation. This failure violates the express terms of the statute that any final regulatory determination be *based* on public hearings and comments *on* the Report to Congress. This requirement is not satisfied by the truncated and abbreviated review of the Bevill Amendment's statutory study factors presented by EPA in the instant proposal.

The language of the statute could not be clearer on this point: following submission of its Report to Congress, and after public hearings, and comment on the report, EPA was to publish its regulatory determination —which shall be based on information developed or accumulated pursuant to such study [i.e., EPA's Report to Congress], public hearings, and comment, in the Federal Register accompanied by an explanation and justification of the reasons for it." RCRA § 3001(b)(3)(C). The D.C. Circuit has confirmed that the statute unambiguously directs that the final regulatory determination is to be based on the Report to Congress. EDF v. EPA, 852 F.2d at 1314 (—The statute clearly states that the agency is to base its regulatory determination on the

information gathered for the § 8002(p) study"). <sup>15</sup> As the Supreme Court stated in *Nat'l R.R. Passenger Corp. v. Nat'l Ass'n of R.R. Passengers (Amtrak),* [w]hen a statute limits a thing to be done in a particular mode, it includes the negative of any other mode." 414 U.S. 453, 458 (1974).

Thus, even assuming EPA was authorized to render a new and revised regulatory determination, such determination must be preceded and predicated on a new Report to Congress. Even EPA has recognized that any final regulatory determination must be predicated on the Report to Congress. See 75 Fed. Reg. at 35151 (explaining that EPA addressed the Bevill statutory study factors for CCRs in the 1988 and 1999 Reports to Congress and -{t}he findings of these two Reports to Congress were the basis for our decisions in the August 1993 and May 2000 Regulatory Determinations to maintain the Bevill exemption for CCRs."); see also 65 Fed. Reg. at 32219 (detailing how EPA's 2000 final Regulatory Determination was based on public comments on the Agency's 1999 Report to Congress).

Representative Bevill underscored the point that the public hearing and comment process was intended to directly respond to EPA's Report to Congress and that any final regulatory determination was to be based on the information resulting from that process: —Finally, let me direct the House's attention to the fact that after EPA concludes these studies, it will be required to obtain public views on them and to make

The Section 8002(p) study refers to the EPA's Report to Congress on certain mining wastes that also were subject to the Bevill Amendment. Importantly though, the D.C. Circuit's decision is equally applicable to CCRs, as the statutory process for EPA issuing a final Regulatory Determination for mining wastes and CCRs is identical. See RCRA § 3001(b)(3)(C) (referencing both the CCR § 8002(n) study and mining waste § 8002(p) study).

known whether, as a result of this process, EPA believes any regulation of these materials is necessary." 126 Cong. Rec. at 3362 (emphasis added).

Here, of course, there is no Report to Congress recommending that CCRs be regulated under RCRA Subtitle C. This is especially egregious given that the Reports to Congress that EPA *did* provide to Congress made clear that CCRs do *not* warrant regulation under RCRA's Subtitle C hazardous waste program and that, therefore, EPA would not pursue this regulatory path. Now, more than ten years later, EPA is proposing to reverse that decision and issue a new regulatory determination for CCRs without explaining its reasoning to Congress as required by the statute, and without allowing for public comment and hearings on the Report to Congress.

EPA may believe that its abbreviated review of the Bevill statutory factors in the proposed rule is the functional equivalent to preparing and submitting a Report to Congress, but the Agency's view is legally flawed for at least three reasons. First, as explained above, such a position would be at direct odds with the plain language of the statute, which unambiguously commands that any final regulatory determination be based on the Report to Congress. The statute simply does not authorize the type of fast-track decision-making process that EPA is attempting to employ under the Subtitle C option.

Second, the statute does not authorize the Agency to collapse the Bevill

Amendment's statutory decision-making process into the more abbreviated exercise,
which the Agency is undertaking here, by combining a proposed reversal of its earlier
final Regulatory Determinations with the proposed development of Subtitle C
regulations for CCRs. As EPA candidly admits, the end result would be the issuance of

a new regulatory determination calling for Subtitle C regulation along with the final Subtitle C regulations. This ignores the plain language and purpose of the Bevill Amendment which requires the regulatory determination to precede the promulgation of any subsequent regulations.

Looking again at the statutory text, RCRA Section 3001(b)(3)(C) provides that after issuing and obtaining public input on its Report to Congress, EPA was to -either determine to promulgate regulations" for CCRs under RCRA Subtitle C -er determine that such regulations are unwarranted." The Agency was then directed to publish its final determination in the Federal Register. *Id.* As the D.C. Circuit has explained, the text of the Bevill Amendment juxtaposes the terms \_determination and \_egulation, . . . signifying that, consistent with the principle that effect must be given to each word of a statute, the two terms were intended to have distinct meanings." *Am. Portland Cement Alliance v. EPA*, 101 F.3d 772, 775 (D.C. Cir. 1996) (-American Portland Cement"). Given this distinction, the sequence of actions directed by Section 3001(b)(3)(C) makes clear that the regulatory determination – whether or not to regulate CCRs under Subtitle C – was to be a separate and distinct step from the subsequent promulgation of any such regulations.

This statutory step-wise process is consistent with the Bevill Amendment's legislative history that EPA employ a more deliberative, sequential procedure than that traditionally applied to other wastes under RCRA in determining whether and how EPA should regulate Bevill wastes in the first instance. See e.g., EDF v. EPA, 852 F.2d at 1314 (Indeed, our reading of the statute and the legislative history strongly suggests that Congress designed the Bevill Amendment to break with the previous approach to

regulation of hazardous industrial wastes."). Under the Bevill Amendment, only *after* the regulatory determination has been issued does the statute contemplate EPA undertaking any appropriate rulemakings based on the determination. As the court explained in *American Portland Cement:* 

Section 3001(b)(3)(C) provides the product of the notice and comment process [on the Report to Congress] will be a determination of whether regulation is warranted in the future, not regulations themselves. While Congress established a timetable for EPA's Regulatory Determination, it set no deadline for the promulgation of regulations that EPA might find necessary under § 3001(b)(3)(C). Accordingly, the notice and comment proceedings [on the Report to Congress] were aimed at informing EPA's decision at an intermediate rather than ultimate stage in the rulemaking process.

#### 101 F.3d at 777.

Therefore, the accelerated decision-making process contemplated in EPA's proposed Subtitle C option – involving the simultaneous reversal and reissuance of the final Regulatory Determination for CCRs, coupled with the issuance of Subtitle C hazardous waste regulations for CCRs – is flatly inconsistent with the statute. Indeed, the public notice and comment process on the Report to Congress is focused only on the appropriateness of EPA's evaluation of the Bevill Amendment's eight statutory study factors and EPA's recommendation in that Report on whether CCRs warrant regulation under Subtitle C. The notice and comment process on any subsequent rulemaking implementing a final regulatory determination necessarily involves a wholly separate set of legal and factual issues. Therefore, as the D.C. Circuit recognized, the statute does not authorize EPA merging these two distinct agency actions into a single step. By attempting to merge the two separate agency actions together in the instant rulemaking, EPA has not clearly delineated its rationale for proposing to reverse and issue a new

regulatory determination from the rationale underlying the proposed Subtitle C regulations. The result is a confusing and disjointed proposal that fails to comply with the sequential decision-making and implementation process required by the statute. This is why Congress established a formalized step-wise process for issuing the final regulatory determination and any subsequent regulations.

Finally, EPA's failure to issue a new Report to Congress setting forth its reasoning on why it now believes CCRs warrant hazardous waste regulation – after twice reaching just the opposite conclusion – deprives Congress of its statutorilyprescribed right to be fully informed of any such recommendation. Congress required this so that it could evaluate and respond, as appropriate, to the Agency's proposed course of action before EPA makes a final Regulatory Determination and develops federal regulations. This is not a mere procedural formality. As Representative Bevill emphasized when introducing his amendment, it was especially critical that EPA not make the final regulatory determination in isolation, but that Congress and relevant federal agencies be involved in the decision-making process. 126 Cong. Rec. at 3362. The multi-step decision-making process set forth in the Bevill Amendment, including the submission of a Report to Congress, was specifically designed to enable Congress and other federal agencies to -evaluate the basis of the Agency's decision, and to address the question of what degree of regulation, if any, is appropriate [emphasis added]" and ultimately to ensure that the final regulatory determination was consistent with the nation's -commitment to develop a coherent and consistent policy toward the use of our coal and other energy resources." *Id.* This multi-stake holder process reflects Congress' intent that it, along with other informed federal agencies and not EPA in

isolation, are responsible for striking the delicate balance between the nation's environmental and energy goals.

Indeed, a key purpose of Congress requiring EPA to submit its Report to Congress was to ensure that any final regulatory determination appropriately reflect Congress' intentions that any regulation of CCRs *not* discourage the development of coal as one of the national primary sources of energy. This is precisely the type of oversight role that Congress reserves for itself when requiring agencies to submit reports to Congress prior to taking regulatory action. *See Natural Res. Def. Council, Inc. v. Hodel,* 865 F.2d 288, 319 (D.C. Cir. 1988) (*-NRDC v. Hodel*") (in explaining Congress' ability to respond to an agency report that is required to be submitted to Congress, the Court observed that *-{ii}*t scarcely bears more than passing mention that the most representative branch is not powerless to vindicate its interests or ensure Executive fidelity to Legislative directives."). In this case, EPA's failure to prepare a Report to Congress regarding its proposal to regulate CCRs under RCRA's hazardous waste program has eviscerated the statutory oversight role provided to Congress in the Bevill Amendment's decision-making process for CCRs.

For these reasons, even assuming EPA had the authority to revisit and reverse its previous, final Regulatory Determination for CCRs, the Agency's proposed Subtitle C option does not comport with the statutory process that Congress imposed on EPA in

<sup>&</sup>lt;sup>16</sup> See also generally James R. Bowers, — Egulating the Regulators" (1990); Joel D. Aberbach, — Keeping a Watchful Eye: The Politics of Congressional Oversight 132" (1990). For example, Congress can control agency action by writing more detailed legislation, by requiring staff investigations or field studies, or through agency reports required by Congress. See also United States v. S. Ind. Gas & Elec. Co., 2002 U.S. Dist. LEXIS 20936 at \*14 (S.D. Ind. Oct. 24, 2002) (noting that the purpose of Congress requesting an agency to report on a new rule was — was to provide a check on administrative agencies 'power to set policies and essentially legislate without Congressional oversight.").

determining whether and how to regulate CCRs under RCRA Subtitle C. At a minimum, EPA must conduct thorough studies demonstrating a clear need for regulation; provide to Congress a report detailing its findings on those studies; conduct public hearings and solicit comments on the new Report to Congress; and finally base any final regulatory determination on the new Report to Congress. Congress did not painstakingly craft such mechanisms into the statute for its own amusement and EPA is not at liberty to ignore such clear congressional directions.

# IV. THE PROPOSED SUBTITLE C OPTION IS INCONSISTENT WITH THE BEVILL AMENDMENT AND IS ARBITRARY AND CAPRICIOUS

Even assuming, for purposes of argument, that EPA was authorized to revisit its final Bevill Regulatory Determination for CCRs and that it followed the appropriate statutory procedures for issuing a new regulatory determination, the Subtitle C option would nonetheless be unlawful because it (1) is inconsistent with the Bevill Amendment's directive that any final determination not impair CCR beneficial use, (2) conflicts with Congress' directive that any regulatory program for CCRs not discourage the use of coal by imposing undue regulatory costs on the management of CCRs, and (3) disregards the nearly unanimous views of the federal and state agencies (with which the Bevill Amendment directs EPA to consult in reaching a final regulatory determination) that Subtitle C regulation is not warranted for CCRs. In addition, the Subtitle C option is arbitrary and capricious because EPA has not provided a reasoned explanation for reversing its previous conclusion that application of the inflexible Subtitle C regulatory program to CCRs will result in excessive and unnecessary regulation.

A. The Subtitle C Option Is Inconsistent With The Bevill Amendment
Because It Would Adversely Impact CCR Beneficial Use And Is
Based On The Arbitrary And Capricious Position That Subtitle C
Regulation Of CCRs Will Increase Beneficial Use

EPA's theory that regulating CCRs under RCRA's Subtitle C regulations will actually *increase* CCR beneficial use is one of the most fallacious aspects of the CCR proposal. 75 Fed. Reg. at 35185-87. This position ignores (1) actual marketplace evidence, which already is showing declines in CCR beneficial use just from the threat of Subtitle C regulation, (2) EPA's prior positions, (3) the views of other federal agencies and state environmental protection agencies, (4) the views of standards setting organizations that have no economic, political or other vested interests in the outcome of this rulemaking, (5) the long-standing view of Congress, and (6) the position of electric utilities – the very entities that would introduce the CCRs into the beneficial use market in the first instance.

EPA identifies the array of valid concerns raised by industry, utilities, the states, standard setting organizations, and others that Subtitle C regulation will effectively cripple the CCR beneficial use market, but remarkably offers little if any response to these points. In fact, EPA expressly acknowledges that, if it pursues the Subtitle C option, some states, such as Florida, —would likely prohibit the beneficial use of CCRs." *Id.* at 35187. Instead of providing substantive responses to these concerns, EPA persists in its position that it understands the economic drivers in the CCR beneficial use market better than those actually engaged in the business. EPA argues that driving up the costs of CCR disposal as high as possible (as would happen under the Subtitle C option) would force more CCRs into the beneficial use market. *Id.* EPA's theory, however, is based on flawed examples, and ignores the first-hand knowledge of those

actually involved in the CCR beneficial use market who have consistently provided EPA with information and data demonstrating that any form of Subtitle C regulation will cripple the beneficial use of CCRs. EPA's theory also fails to consider whether the marketplace can in fact absorb the vastly increased volume of CCRs that it speculates will be beneficially used under the Subtitle C option.

Rather than substantiate its theory that CCR beneficial use will increase under Subtitle C, EPA places on those involved in CCR beneficial use the burden of providing -actual instances" of the adverse impact on beneficial use from a Subtitle C regulation that has yet to be promulgated. *Id.* at 35222-23. While it is impossible for anyone to prove the effects of regulations that have not yet been promulgated, ash marketers, utilities and others *have* presented EPA with real world data showing the adverse impacts on CCR beneficial use resulting from the mere threat of Subtitle C regulation. The only support EPA offers for its theory that CCR beneficial use will increase under Subtitle C are examples of certain materials for which EPA claims recycling has increased since becoming subject to hazardous waste regulation. *Id.* at 35186-87. Here too, however, even EPA admits that its examples involve materials that, unlike CCRs, -are not used in residential settings." *Id.* at 31586. Moreover, as discussed below, certain of the examples cited by EPA involve materials that are not actually regulated under RCRA Subtitle C. Another example involves used oil, which EPA specifically declined to list as a hazardous waste precisely because of its concerns that doing so would undermine the used oil recycling market.

EPA also argues that the stigma resulting from Subtitle C regulation will be -significantly reduced by listing CCRs as a \_special waste, '" (*Id*. at 35187) attempting to

deflect attention from the reality that these materials would be subject to hazardous waste regulation from their point of generation until their ultimate disposal. EPA's -special waste" label will do nothing to reduce the potential legal liabilities of using a material in a commercial or residential setting that has been determined by EPA to meet the hazardous waste listing criteria and subject to hazardous waste regulation when disposed of. RCRA makes clear that only materials that are *-hazardous waste listed or* identified" under Subtitle C are subject to the statute's hazardous waste controls. See, e.g., RCRA § 3004(a) (EPA is to establish regulations for -ewners and operators of facilities for the treatment, storage, or disposal of hazardous waste identified or listed under this subchapter") and § 3005(a) (EPA is to require permits for -the treatment, storage, or disposal of hazardous waste identified or listed under this subchapter"). Put another way, only -hazardous wastes" - not -special wastes" - can be subject to RCRA's Subtitle C controls. There is no way around the +hazardous waste" label and the associated stigma that comes with this designation if EPA subjects CCRs to Subtitle C controls.

In evaluating the potential impacts of Subtitle C regulation on CCR beneficial use, it is important to note that the sponsors of the Bevill Amendment cautioned EPA against regulation of CCRs under Subtitle C for precisely the reasons being offered today by opponents of the Subtitle C option: subjecting CCRs to the stringent Subtitle C program is certain to have adverse impacts on the use of these materials.

Representative Bevill was clear that one of the primary purposes of his Amendment was to reinforce the -national commitment to encourage reuse of such materials as fly ash."

126 Cong. Rec. at 3362. In support of this objective, Representative Findley cautioned

EPA that regulation of CCRs under RCRA Subtitle C would have an adverse impact on CCR beneficial use, warning that if the Agency were to regulate CCRs under RCRA's hazardous waste program, the -flourishing industry which recycles these byproducts would be gravely disrupted and possibly closed down." *Id.* at 3363. Representative Staggers was more direct, warning that regulating CCRs as hazardous waste – even with a -special waste" label – -would run counter to one of the principal designs of [RCRA] – conservation of valuable material." *Id.* at 3364-65. And, as the CCR beneficial use industry is arguing today, Representative Rahall recognized thirty years ago that the stringent regulation of CCRs would put these materials at a distinct disadvantage in the marketplace precisely because they must compete with a variety of other natural aggregates and materials. He stated: \_[i]t is obvious that the imposition of any expensive regulatory requirements on these developing industries will have a disastrous impact on them." *Id.* at 3363.

Considering the above in addition to the record evidence already assembled in this rulemaking proceeding, it is clear that regulating CCRs under any Subtitle C hazardous waste regime, even with the -special waste" label, will have a significant adverse impact on CCR beneficial use. These issues are discussed further below.

- The Record Makes Clear That Regulating CCRs Under RCRA Subtitle C Will Significantly Harm CCR Beneficial Use
- a. <u>DOE Ash Barriers Report</u> Following enactment of the Bevill Amendment,
   Congress remained concerned with the adverse effects on CCR beneficial use resulting
   from the stigma of adverse regulatory labels. Because of this continuing concern, in
   1992, Congress directed the U.S. Department of Energy to -eonduct a detailed and

comprehensive study on the institutional, legal, and regulatory barriers to increased utilization of coal combustion byproducts by potential governmental and commercial users." Energy Policy Act § 1334, 42 U.S.C. § 13364(b)(1). In response, the Secretary of Energy submitted a Report to Congress in July 1994 finding, among other things, that the -fd]esignation of coal byproducts as a solid waste, while fostering public misconception about the risk of these materials, stymies attempts to develop or expand markets for their use." U.S. Dept. of Energy, Report to Congress, Barriers to the Increased Utilization of Coal Combustion/Desulfurization Byproducts by Governmental and Commercial Sectors at iv (1994) (-DOE Ash Barriers Report"). The DOE Ash Barriers Report also found that -State regulation of [beneficially used] fly ash as a solid waste —was an institutional barrier to increased use of coal combustion byproducts (DOE Ash Barriers Report at 16), and concluded that -fd]esignation of coal byproducts as solid waste, even if used, creates attitudinal barriers in producers, end users, utilization technology developers, marketers, and regulators [and] . . . fosters misconceptions by the public." *Id.* at 17. In addition to the -attitudinal barriers" from the solid waste label, DOE concluded that even designating CCRs that are disposed of as a non-hazardous solid waste raised legitimate barriers to beneficial use.

b. <u>EPA's 2000 Final Regulatory Determination</u> – EPA itself concluded in its 2000 Regulatory Determination that the stigma from regulating CCRs under RCRA's hazardous waste program could harm CCR beneficial use. Importantly, the Subtitle C option considered and rejected in 2000 was a -contingent Subtitle C" option, which meant that CCRs managed in accordance with Subtitle D-like standards would not be classified as a hazardous waste; only CCRs not properly managed under the Subtitle D

standards would have been regulated as listed hazardous wastes. 65 Fed. Reg. at 32216. EPA recognized that, even under that approach, the mere possibility of subjecting CCRs to hazardous waste regulation would have an adverse stigmatizing effect on CCR beneficial use. As EPA explained in rejecting the Subtitle C contingent management approach:

We also see a potential downside to pursuing a subtitle C approach. Section 8002(n)(8) [of RCRA] directs us to consider, among other factors, the current and potential utilization of such materials.' Industry commenters have indicated that they believe subjecting any coal combustion wastes to a subtitle C regime would place a significant stigma on these wastes, the most important effect being that it would adversely impact beneficial reuse. ... EPA believes the contingent management scheme we discussed should diminish any stigma that might be associated with the subtitle C link. Nonetheless, we acknowledge the possibility that the approach could have unintended consequences. We would be particularly concerned about any adverse effect on the beneficial re-use market for these wastes ....

## EPA continued:

Normally, concerns about stigma are not a deciding factor in EPA's decisions under RCRA, given the central concern under the statute for protection of human health and the environment. However, given our conclusion that the subtitle D approach here should be fully effective in protecting human health and the environment, and given the large and salutary role that beneficial reuse plays for this waste, *concern over stigma is a factor supporting our decision today that subtitle C regulation is unwarranted in light of our decision to pursue a subtitle D approach*.

Id. at 32217 (emphasis added).

Given that EPA has expressly acknowledged that just the *possibility* of subjecting CCRs to Subtitle C regulation could adversely impact CCR beneficial use, it is even more astonishing that EPA now takes the position that the more draconian Subtitle C proposal under consideration here – which would automatically subject all CCRs destined for disposal to hazardous waste regulation – will have no stigmatic effect on

CCR beneficial use and will instead *promote* CCR beneficial use. EPA fails to provide any reasoned explanation for this about-face regarding its views on the adverse impact of Subtitle C regulations on CCR beneficial use. There is no reasoned explanation for this change of position in the rulemaking record.

c. <u>CCR Beneficial Use Market</u> – Since EPA initiated this rulemaking and advised that it was considering reversing its final Regulatory Determinations and regulating CCRs under RCRA Subtitle C, the Agency has received hundreds of letters and written testimony at public hearings and has heard congressional testimony from a wide-spectrum of third parties directly involved in the business of marketing and beneficially using CCRs. Their views reiterate what EPA itself acknowledged in 2000: just the threat of Subtitle C regulation for CCRs would adversely impact CCR beneficial use.

USWAG cannot summarize all of the individual submissions in these comments, though we have included them as part of this submission for the record. <sup>17</sup> It is useful, however, to reference key findings from just a few of these comments to underscore the overwhelming evidence that Subtitle C regulation of CCRs will adversely impact CCR beneficial use in this country.

Especially critical is the information provided to EPA showing actual evidence of adverse impacts on CCR beneficial use from even the *threat* of Subtitle C regulation.

As detailed below, CCR beneficial use projects have been canceled as a result of the proposed Subtitle C option; this means that CCRs that previously would have been beneficially used are now being disposed of. This information responds to EPA's

<sup>&</sup>lt;sup>17</sup> See attached CD-ROM entitled —Public Comments and Testimony on EPA's Regulation of Coal Combustion Residuals," (Nov. 19, 2010).

request for data -supporting the claims that stigma' will drive people away from the use of valuable products [in this case CCRs], or that states will prohibit the reuse of CCRs under their beneficial use programs if EPA regulates any aspect of CCR management under Subtitle C." 75 Fed. Reg. at 35156. The information detailed below by beneficial use market (and by electric utilities) is more than just -statements or declarations" (*Id.*); it is actual record evidence demonstrating that any Subtitle C option will have a devastating impact on the beneficial use of CCRs. Such a result would be directly counter to the plain language and purpose of the Bevill Amendment and underscores that the Subtitle C option would be inconsistent with the statute as well as arbitrary and capricious.

For example, in recent testimony before the House Subcommittee on Rural Development, Entrepreneurship and Trade on the potential impact on small businesses from regulating CCRs under RCRA Subtitle C, Tom Adams, Executive Director of the American Coal Ash Association (-ACAA"), presented real world evidence of the adverse impacts on the CCR beneficial use markets already occurring merely from EPA proposing its Subtitle C option, including the following:

- Anne Arundel County in Maryland has prohibited the use of fly ash in county construction projects pending EPA's final rule.
- The Los Angeles Unified School District has stopped allowing the use of fly ash in all LAUSD projects pending EPA's final rule.
- CalStar Products opened a plant in Wisconsin to manufacture bricks and pavers from fly ash. Their process uses fly ash as a primary ingredient and consumes 85% less energy than that used in producing traditional clay bricks. The Brick Industry Association has published comments that infer safety concerns because the brick is made with -hazardous wastes".
- Reed Minerals (a division of Harsco), a large manufacturer of residential roofing shingles made using CCRs, had to threaten legal action against a proposed

advertising campaign of a competitor. The campaign theme was —Our shingles do not contain hazardous waste. Do yours?"

Statement of Thomas H. Adams, Executive Director, American Coal Ash Association, before the House Small Business Subcommittee on Rural Development, Entrepreneurship and Trade (July 22, 2010) (attached as Appendix 5).

ACAA's testimony also effectively responds to EPA's request for comment on the type of liability issues that CCR marketers and users are concerned about if CCRs are regulated under Subtitle C, explaining that a primary concern in the market is liability exposure to toxic tort lawsuits. Mr. Adams testified that:

[i]n discussions with engineers, contractors, and concrete producers over the last several months, it is clear that the use of fly ash would be severely curtailed due to fear of tort or class action suits. Many in the concrete industry do not believe EPA's assertion that the exemption would provide all the protection needed. Many do believe that a lawyer could make a simple argument to a jury that the fly ash in the disposal facility has exactly the same physical and chemical characteristics as the fly ash in the concrete in a home, hospital, daycare center, or school. Therefore if it is hazardous in the disposal facility, it must be hazardous in those structures thereby opening the door to financial claims. Even if a claim is found to be minimal, the costs of legal defense are something firms want to avoid.

ld.

Similar concerns were expressed in testimony presented to the Subcommittee by Dr. Craig H. Benson, PhD, PE, DGE, Professor of Geological Engineering and Civil & Environmental Engineering, Chair of Geological Engineering, and Director of the Recycled Materials Resource Center at the University of Wisconsin-Madison. Mr. Benson described the liability concerns that marketers of industrial byproducts have struggled to overcome with respect to the beneficial use of these materials and cautioned that the same —psychological" barriers would arise if CCRs were regulated under RCRA's hazardous waste regulations. Dr. Benson cautioned that:

none of these science-based principles and tools will overcome the psychological impact of CCPs being deemed a hazardous waste. An exemption for beneficial use will have virtually no effect on this psychological impact. The -hazardous" designation will scare users and incite liability, and thereby decimate beneficial use of CCPs. Some have proffered that a hazardous designation coupled with a beneficial use exemption will increase the amount of CCPs that are beneficially used in a manner analogous to the reduction hazardous waste volume that occurred when RCRA hazardous waste rules were originally developed. My experience suggests that this outcome is unrealistic. ... There is no basis to believe that infrastructure owners will accept that the risks of using CCPs in infrastructure are minimal when essentially the same material is deemed a hazardous waste in a different setting.

Testimony of Craig H. Benson, PhD, PE, DGE, Wisconsin Distinguished Professor,
University of Wisconsin-Madison, before the House Small Business Subcommittee on
Rural Development, Entrepreneurship and Trade (July 22, 2010) (attached as Appendix
6).

Dr. Benson also testified to the competitive disadvantage that the CCR beneficial use industry already is confronting as a result of EPA's proposal to regulate CCRs as a hazardous waste:

Manufacturers of competing products and materials that do not include CCPs have also taken advantage of the hazardous waste stigma by advertising that their products and materials do not include hazardous waste. I surmise that beneficial use of all industrial byproducts will diminish if CCPs are deemed hazardous waste. The logical inference from the perspective of a potential user is -Will the industrial byproduct I am using today be designated as a hazardous waste tomorrow? How will this affect my long-term liability?" The logical decision from the perspective of the user is to avoid beneficial use of industrial byproducts altogether.

ld.

Echoing Dr. Benson's findings at the hearing was Bill Gehrmann, President of Headwaters Resources, Inc, the -nation's largest post-combustion coal product manager" involved in all aspects of CCR beneficial use. Testimony of William H. Gehrmann, President, Headwaters Resources, Inc., before the House Subcommittee on

Rural Development, Entrepreneurship and Trade Committee on Small Business (July 22, 2010) (attached as Appendix 7). Critically, Mr. Gehrmann also testified that due to the stigma and liability issues associated with using a material subject to hazardous waste regulation, his company is already seeing a slowdown in the financing of CCR beneficial use projects:

If coal ash is designated as hazardous waste when disposed, ash technology providers and developers will face significant new customer objections and barriers to raising capital for development activities. Even in advance of enactment of any rule, companies in this sector have reported slowdowns in financing activities and customer purchases attributed to the regulatory uncertainty presented by EPA's draft rulemaking proposal.

ld.

Mr. Gehrmann also testified about the competitive disadvantages directly attributable to the Subtitle C proposal being felt by ready mix concrete producers due to their use of fly ash in lieu of cement in the production of concrete:

If coal ash is designated a hazardous waste' when disposed, end users will likely demand products that contain no hazardous' substances. This phenomenon is already being seen even in advance of EPA enacting any new rules. The drumbeat of the phrase toxic ash' in news stories about EPA's rulemaking efforts has resulted in many ready mixed concrete producers receiving calls from customers asking for fly ash to be eliminated from their concrete.

ld.

In addition to the above testimonials, dozens of individual companies who use CCRs in their products, including small ready mix concrete companies and federal and state trade associations representing the concrete, construction and related infrastructure construction industries, have written EPA cautioning that the regulation of CCRs under Subtitle C, even with an exemption for beneficial use, will have devastating

consequences for the CCR beneficial use market. Many of these companies have told EPA that they will not take on the legal risks associated with using CCRs in their products if these materials are regulated under Subtitle C when disposed of. For example, a small, ready-mix concrete company in Milwaukee, Wisconsin, wrote to EPA stating that, even with an exemption for CCR beneficial use, it would not take the risk of using CCRs in its products if regulated as hazardous waste when disposed of. The company explained:

-do you really think that motivated reporters would even try to make that distinction, certainly attorneys won't. Not only can you not call this material a hazardous waste, as a producer of a potentially hazardous waste product, I now need to know definitively that it is not now and never will be classified as a hazardous waste to continue to use it. I cannot afford to put a potential hazardous waste into my finished product."

Letter from Schmitz Ready Mix, Inc. to EPA Administrator Lisa Jackson (2009) (attached as Appendix 8). These statements and letters are from companies directly involved in all aspects of CCR beneficial use, with first-hand knowledge of, and experience with, the economic drivers in the CCR beneficial use market. They will not be directly regulated by the CCR disposal regulations, so their views regarding the crippling effects the Subtitle C option will have on CCR beneficial use are both informed and objective. Indeed, we expect that these companies would support the Subtitle C option if this option would actually lead to increased CCR beneficial use. The record evidence, however, is just to the contrary; EPA cannot ignore this significant body of evidence and pursue the Subtitle C option.

d. <u>Standard-Setting Organizations</u> – While professional standard-setting organizations generally do not weigh in on proposed EPA regulations, in this instance the prospect of the adverse impact of Subtitle C regulation on CCR beneficial use has

caused both the American Concrete Institute (-ACI") and ASTM International (-ASTM") to take the unusual action of specifically cautioning EPA that the Subtitle C option would likely result in the removal from national and international building standard specifications any materials (*i.e.*, fly ash) that are determined to be a hazardous waste in another setting.

ACI, one of the world's leading authorities on concrete technology, develops international standards and publishes technical papers and manuscripts to -advance concrete knowledge for the benefit of the general public." Letter from ACI to EPA Administrator Lisa Jackson (September 4, 2009) (attached as Appendix 9). To underscore its purely professional and objective perspective on this issue, ACI emphasized in its letter to EPA that it -is a technical society, and unlike trade organizations does not represent any trades related to or part of the concrete industry."

Id. Instead, ACI's concerns regarding the Subtitle C option -deals with the impact that designating fly ash as a hazardous waste' will have on concrete technology, the best use of concrete, and concrete's sustainable impact on society."

Id. ACI was direct with respect to these concerns, explaining that:

[i]t is ACI's opinion that designating fly ash as a -hazardous waste" will result in little or no fly ash being used in concrete in the US. We anticipate the concrete industry will no longer specify its use; and fly ash producers would not permit its beneficial use due to liability concerns, preferring to impound fly ash rather than allow its use.

Id. In short, ACI cautioned that -[d]esignation of fly ash as a hazardous waste will likely eliminate its inclusion in future project specifications for fear of possible legal exposure and liability. Such designation would also likely lead to its removal from future national codes and standards for the same reason." Id.

ACI's professional opinion was not swayed by EPA's proposal to regulate CCRs under RCRA's hazardous waste regulations while applying the -special waste" label. In its written testimony before the recent House Subcommittee hearing on the impact of the Subtitle C option on small businesses, ACI cautioned again that the stigma resulting from regulation of CCRs under RCRA Subtitle C would adversely impact CCR beneficial use, stating that -[i]f EPA designates fly ash as special waste, but requires hazardous waste regulations, acceptance throughout the different audiences in the concrete industry will be difficult to maintain." Written Testimony of Richard Stehly, President, ACI, submitted to the House Committee on Small Business, Subcommittee on Rural Development, Entrepreneurship and Trade (July 22, 2010) (attached as Appendix 10).

ASTM has been equally direct regarding the adverse impact on CCR beneficial use resulting from the Subtitle C option. ASTM is one of the largest voluntary standards development organizations in the world and its committees -develop and oversee more than 12,000 ASTM standards that are used by individuals, companies, and agencies around the world" and are referenced by government agencies in -eodes, regulations, and laws; and many others refer to them for guidance." Letter from ASTM to EPA Administrator Jackson (Dec. 23, 2009)(attached as Appendix 11). While ASTM standards are voluntary in -the sense that their use is not mandated by ASTM . . . . government agencies often give voluntary standards the force of law by citing them in laws, regulations, and codes." *Id.* Of particular importance here, ASTM Committee C009 provides test methods and specifications for concrete making materials, which serve a variety of functions, including -the means of quality assurance by purchasers, a

Duke Energy Progress, LLC Late-Filed Exhibit No. 16 Docket No. E-2, Sub 1219 Page 121 of 439

basis for training and certification of testing personnel, and protection against liability through their use in contracts." *Id*.

ASTM has advised EPA that, should it regulate CCRs under RCRA's Subtitle C program, ASTM will need to modify its product standard specifications for the use of fly ash in concrete to properly reflect the material's new regulatory status. ASTM explains that <code>-[a]\_hazardous</code> waste' designation, even with an exclusion for beneficial use, would cause the ASTM standard for fly ash to be removed from project specifications due to concerns over legal exposure, product liability, and public perception. This will likely result in little or no fly ash being used beneficially in concrete or other applications that support sustainability objectives." *Id*.

As with the views and data provided by CCR beneficial users, the technical views of ACI and ASTM cannot be ignored by EPA. They represent another significant component of the record evidence illustrating the fallacy of EPA's position that regulating CCRs under RCRA Subtitle C will somehow increase beneficial use. The data and views of the experts in the field are unanimous that any Subtitle C option will undoubtedly cripple – if not eliminate altogether – the CCR beneficial use market.

e. <u>State Environmental and Transportation Agencies</u> – State environmental protection and transportation agencies also have advised EPA that the Subtitle C option will significantly harm CCR beneficial use. They have identified state laws that would effectively preclude CCR beneficial use if regulated under Subtitle C provided EPA with more information as to why a Subtitle C designation would effectively end the beneficial use of CCRs in highway construction projects across the country.

The Association of State and Territorial Solid Waste Management Officials (-ASTSWMO") has warned EPA repeatedly since initiation of this rulemaking that -{t}|he most compelling reason not to impose Subtitle C regulations [on CCRs] is that the beneficial use of CCB has been very successful. The hazardous label of Subtitle C would be detrimental to State CCB beneficial use programs..." Letter from ASTSWMO to EPA (April 1, 2009) (attached as Appendix 12). Just last fall, after polling every state in the nation on the potential impacts on state programs of regulating CCRs under Subtitle C, ASTSWMO wrote EPA to caution that -{t}|egulation under RCRA Subtitle C has the potential to put an end to many beneficial uses for [coal combustion waste]. In most States, a primary requirement for a beneficial use determination is that the waste not be hazardous. Labeling CCW a hazardous waste will have a detrimental [impact] on its beneficial use." Letter from ASTSWMO to EPA (Nov. 4, 2009) (attached as Appendix 13).

Environmental protection agencies, including those from Colorado, Florida, Hawaii, Iowa, Kansas, Michigan, Missouri, Ohio, South Dakota, Tennessee, Virginia, West Virginia and Wisconsin, also have specifically advised that regulating CCRs under RCRA Subtitle C would complicate and/or result in a flat-out prohibition on the beneficial use of CCRs. See attachment to ASTSWMO letter to EPA (Nov. 4, 2009) (attached as Appendix 13). Michigan was blunt and to the point: —The regulation of coal ash under full RCRA Subtitle C would end the current beneficial uses of coal ash." *Id.* And, as EPA itself recognizes, the same would be true in Florida, where —[the state's] definition of hazardous waste would likely prohibit the beneficial use of CCRs" under the Subtitle C option for CCRs. 75 Fed. Reg. at 35187. Wisconsin, which has one of the most

successful CCR beneficial use programs in the country, has specifically cautioned EPA that -[w]e remain deeply concerned that such a categorization [regulating CCRs under Subtitle C] would have a significant adverse impact to our ongoing successful efforts to beneficially reuse these materials." Letter from Wisconsin Department of Natural Resources to EPA (March 16, 2009) (attached as Appendix 14).

In addition, the American Association of State Highway and Transportation

Officials (-AASHTO") has warned EPA that -the stigma and legal ramifications
associated with using a hazardous waste material could effectively eliminate the ability
to use Fly Ash in highway construction, even if exceptions are made to allow for
beneficial applications." Letter from AASHTO to EPA Administrator Lisa Jackson

(Nov. 23, 2009) (attached as Appendix 15). This view has been echoed by numerous
individual state Departments of Transportation (-DOTs"), including, among others, those
in Arizona, Colorado, Florida, Indiana, Michigan, Minnesota, New Hampshire, North
Dakota, Texas and Utah. For example, the Indiana DOT advised EPA that it -strongly
opposes any designation of CCPs as hazardous waste. Such action would have
significant and long lasting adverse effect upon our ability to beneficially use fly ash and
other CCPs in highway construction projects." Letter from Indiana Department of
Transportation to EPA Administrator Lisa Jackson (Sept. 9, 2009) (attached as
Appendix 16).

Michigan's DOT provided a similar caution to EPA:

We believe that regulating fly ash as a hazardous waste would have significant unintended negative consequences on its beneficial reuse. Even if EPA plans only to regulate the disposal of fly ash as a hazardous waste, the stigma associated with such an approach will have a chilling effect on the use of the material for our infrastructure.

Letter from Michigan Department of Transportation to Victor Mendez, Administrator, Federal Highway Administration (Nov. 10, 2009) (attached as Appendix 17).

These views are coming from the associations of state environmental and transportation agencies as well as from the individual states themselves – parties directly involved with regulation involving the beneficial use of CCRs. Their warnings to EPA constitute compelling evidence that the Subtitle C option would have a significant adverse impact on CCR beneficial use.

f. <u>Electric Utilities</u> – Real-world evidence of the adverse impacts on CCR beneficial use stemming from just the *prospect* of Subtitle C regulation also is being reported by electric utilities. Just as CCR marketers and end-users are already feeling the adverse impacts of this proposal, so too have utilities become painfully aware of the stigma associated with the proposal.

For example, Xcel Energy, which engages in beneficial use operations in Wisconsin, a state that EPA recognizes has one of the most successful CCR beneficial use programs in the nation, has had to halt the beneficial use of CCRs in a geotechnical fill project. This is due, in part, to regulatory concerns associated with the pending Subtitle C proposal and the possibility that EPA could regulate CCRs as hazardous. This project was to have been undertaken in accordance with Wisconsin's NR-538 rule, which has been cited by EPA as an example of a successful state regulatory program for the responsible reuse of CCRs. Despite the fact that this type of use has been successfully demonstrated in the past, the project has been put on hold and the fly ash from an Xcel Energy plant that was at one time under contract to be 100% beneficially used is now being landfilled.

In addition to projects being halted by state governments due to the threat of Subtitle C regulation, liability concerns have been raised by CCR beneficial use contractors resulting in cancelled beneficial use projects. Xcel reports that it recently had an ash utilization contractor object to the environmental terms and conditions of a CCR beneficial use contract due to contractor concerns over the future regulation of CCRs and potential liability under the RCRA Subtitle C hazardous waste program. Similar to the experiences of other utilities, the parties in this matter have, thus far, been unable to reach a mutually acceptable resolution of this issue, resulting in a decision to cancel plans for CCR utilization at that facility, at least temporarily. As a result, all of the CCRs will be disposed of, as opposed to being beneficially used.

Another utility in Wisconsin, Wisconsin Power and Light (-WP&L"), reports that the use of bottom ash from one of its power plants in Wisconsin for snow and ice control has been discontinued by the town of Lodi, Wisconsin. Because this market is no longer available, the facility must now determine whether to dispose of bottom ash that was previously beneficially used. The local paper reported that this decision was based solely on public opposition to beneficially using bottom ash, even though the town of Lodi chairman stated that he did not have worries about the health risk of bottom ash... but public feedback over the use of the coal by-product prompted the ban." See Lodi Enterprise —E" News, (posted Aug. 11, 2010). Jennifer Fetterly, Town of Lodi Stops Bottom Ash Use, Lodi Enterprise E-News, (Aug. 11, 2010) (available online at http://www.lodienews.com) (current as of Sept. 27, 2010). In fact, the Lodi town chairman made it clear that —[t]his decision is being made as a request by citizens, not one that is dictated by scientific data. In fact the data supports its [bottom ash's] use on

the roads." Id. (emphasis added). The paper also reported that citizens will seek to establish similar bans on the use of bottom ash for snow and ice controls in neighboring municipalities. Id.

This example is significant because the use of bottom ash for snow and ice control is explicitly identified by EPA as one of the types of CCR beneficial use that the Agency does not want to adversely impact as result of the Subtitle C option. EPA correctly recognizes that this is an important and legitimate beneficial use, explaining that:

[u]se of CCRs in highway projects is a significant practice covering road bed and embankments. CCRs used according to FHA/DOT standards provide an important function in road building, replacing material that would otherwise need to be obtained, such as aggregate or clay. In many cases, the CCRs can lead to better road performance. For snow and ice controls, the beneficial use is limited to boiler slag and bottom ash, which replaces fine aggregate that would otherwise need to be used to prevent skidding, and amounts used are in line with the materials they replace.

75 Fed. Reg. at 35163 (emphasis added). Nonetheless, the WP&L example provides real-world evidence that the stigma associated with the Subtitle C option has prompted public opposition to and bans on the very beneficial uses of CCRs that EPA acknowledges are important and legitimate. The WP&L example is one of many pieces of record evidence directly rebutting EPA's theory that the Subtitle C option will lead to an increase in CCR beneficial use.

The adverse impacts on CCR beneficial use are being reported in other parts of the country as well. PPL Corporation (—PPL"), a global energy company that engages in significant CCR beneficial use operations in Pennsylvania, reports the cancellation of CCR beneficial use projects due to the proposed Subtitle C option. PPL has testified to EPA that:

[t]he impact to beneficial uses from the stigma of labeling CCRs a hazardous waste is real — and is already occurring. One of PPL's largest marketers of coal ash for cement products has had one of its main customers stop using coal ash. WHY? Potential product liabilities if EPA actually regulates CCRs as a hazardous waste. Furthermore, many companies have told our marketers that they will not use coal ash in their products if CCRs are classified as a hazardous waste, regardless of any use exclusion by EPA. WHY? They don't want their products to contain an ingredient that would otherwise be subject to hazardous waste regulation. Based on EPA's own economic analysis, if a subtitle C regulation eliminates beneficial uses, the financial impact on our struggling economy will be in the billions of dollars.

PPL Testimony of Craig Shamory at EPA CCR Public Hearing, Pittsburgh, PA, (Sept. 21, 2010) (attached as Appendix 18). PPL also reported that one of its key CCR marketers has already had a customer stop using bottom ash as aggregate in the manufacture of concrete blocks. Other customers have told PPL that they will not use CCRs if this material is listed as a hazardous waste, despite any exclusions EPA may provide in the Subtitle C regulation for qualified beneficial uses.

LG&E and KU Energy LLC, formerly E.ON U.S. (+LG&E/KU"), a subsidiary of PPL Corporation, reported the cancellation of existing beneficial use projects and concerns about the viability of beneficial use projects in the future due to the specter of CCR regulation under Subtitle C. LG&E/KU reported that one of its ash marketers has advised that an end use customer declined to enter into a contract until finalization of the CCR rule due to concerns that the material could be designated as a hazardous waste. In addition, one of LG&E/KU's structural fill beneficial reuse projects has been canceled due to the pending rule. Further, LG&E/KU's ash marketers have expressed grave concern that many end users, including engineers, architects, and property owners, will refuse to use material designated as a hazardous waste for a beneficial use, particularly for new construction projects in schools, hospitals, and restaurants.

In sum, electric utilities, which serve as the first-step in the CCR beneficial use chain, already are reporting cancellation of CCR beneficial use projects resulting from just the threat of CCR regulation under Subtitle C. Real-world examples such as these, along with those already presented to EPA by other third parties in the CCR beneficial use system, make it clear that the Subtitle C option would have a significant adverse impact on CCR beneficial use, in direct contravention to the plain language of the Bevill Amendment and its legislative history.

EPA's Examples of Increased Beneficial Use Despite Hazardous Waste Label Are Without Merit

Rather than respond to the overwhelming evidence from federal and state agencies and the entire spectrum of entities involved in the beneficial use market that the Subtitle C option will devastate the CCR beneficial use market, EPA simply points to certain materials for which it claims that the hazardous waste" label has not impaired beneficial use. These examples are irrelevant; to the extent they do bear on the current proposal, these examples actually provide further evidence that regulating CCRs under Subtitle C will have an adverse stigmatic effect on CCR beneficial use.

a. EPA did not list used oil as a hazardous waste because of concerns that the listing label would adversely impact used oil recycling.

It is curious that EPA points to used oil as an example of a material for which the hazardous waste label has not impaired recycling. In fact, EPA did not list used oil as a hazardous waste precisely because Agency studies showed that the stigma from being labeled a hazardous waste would cripple the used oil recycling market. When first assessing whether to regulate used oil as a listed waste under Subtitle C, EPA specifically recognized that the stigma from the hazardous waste label would cause

dislocations in the [used oil] recycling market." 51 Fed. Reg. 41900, 41902 (Nov. 19, 1986). This concern was based on a study prepared for the Agency analyzing the adverse market implications for the used oil recycling market stemming from the stigma of regulating used oil as listed hazardous waste. See Analysis of Possible Market Impacts Resulting from Stigmatizing Effects of Listing Recycled Oil," by Temple, Barker, and Sloanne, Inc. (Nov. 1986).

While the D.C. Circuit ultimately concluded that EPA's hazardous waste listing regulations did not allow the Agency to consider stigma, the Bevill Amendment (pursuant to RCRA Section 8002(n)(8)) *demands* as a statutory matter that EPA consider stigma in determining whether CCRs warrant regulation under Subtitle C in the first instance. Therefore, while EPA may not be able to consider stigma under its listing regulations, it *must* consider stigma as a threshold matter under the Bevill Amendment in determining whether CCRs warrant Subtitle C regulation. Thus, the import of EPA's decision not to regulate used oil as a listed hazardous waste is that it directly belies and exposes the fallacy of EPA's current position that regulating CCRs as a listed waste under Subtitle C will result in the increased beneficial use of these materials. Even the Court recognized that listing a waste under RCRA Subtitle C could adversely impact recycling:

The EPA's concern over the possible adverse environmental consequences of listing the used oil may well be warranted. See Katzman, From Horse Carts to Minimills, 92 The Public Interest 121, 132 (Summer 1988) (-when the EPA threatened to recognize waste oil from dismantled cars as hazardous, the highly efficient waste-oil refining industry simply closed its doors until the threat disappeared. Obviously, these rejected materials do not simply evaporate; they may merely be disposed of surreptitiously.")

Hazardous Waste Treatment Council v. EPA, 861 F.2d 270, 277 (D.C. Cir. 1988).

Similarly, -[w]hen the EPA in one region threatened to classify shredder fluff as hazardous (because of the PCB content of old appliances), scrap processors across the country refused to accept heavy appliances." Id., see also Dithiocarbamate Task Force v. EPA, 98 F.3d 1394, 1401 (D.C. Cir. 1996) (-Dithiocarbamate") (observing that -EPA itself noted . . . that RCRA listing [of used oil] might result in stigma, leading to subterfuge of regulations.").

EPA's final used oil regulations do *not* list used oil as a hazardous waste. Rather, EPA excluded virtually all forms of used oil recycling from hazardous waste regulation; only those used oils that (1) are not recycled and (2) that exhibit a hazardous waste characteristic are subject to Subtitle C regulation. See 40 C.F.R. § 279.81. As EPA implicitly recognized in the final used oil rule where it, once again, did not list used oil as a hazardous waste, there is a significant difference from the perspective of stigma in classifying a material as a listed waste under Subtitle C, versus subjecting the same material to hazardous waste regulation only if it exhibits a hazardous waste characteristic. The listing option automatically subjects the entire universe of such materials to hazardous waste regulations (as well as any materials mixed with or derived from the listed waste), while the characteristic option only captures a subset of those materials that actually exhibit a hazardous waste characteristic. Hence, the listing option is the most severe method for regulating a waste under Subtitle C and it has the most draconian regulatory consequences. Because of this, there is a recognized adverse stigma associated with the listing label under Subtitle C and not the same

stigma from being subject only to the characteristic test (which all materials are subject to when discarded).

EPA understands this distinction and also understands that CCRs rarely, if ever, exhibit a hazardous waste characteristic and thus would rarely be regulated under Subtitle C, if this were the option under which they could be subject to Subtitle C regulation. Therefore, it is both disingenuous and factually incorrect for EPA to cite to used oil as an example of a material for which recycling has increased under the hazardous waste label, when EPA's own actions and market studies show that *listing* used oil as a hazardous waste – which is precisely what the Agency proposes to do with CCRs under the Subtitle C option – would have had precisely the opposite effect.

b. The examples cited by EPA of increased recycling under hazardous waste regulation are misleading and/or irrelevant.

Other than used oil, EPA points to several other materials, including chat, electric arc furnace dust (-EAF"), electroplating wastewater sludge, and spent solvents, as examples of where regulation under Subtitle C has not harmed the materials' beneficial use. 75 Fed. Reg. at 35186. Here too, EPA's examples do nothing to support the proposition that listing CCRs under Subtitle C would actually increase beneficial use of CCRs. Even assuming, for purposes of argument, that these examples had any relevance to CCRs, they cannot overcome the direct evidence already provided by federal and state agencies and those businesses directly involved in CCR beneficial use markets that make clear that the Subtitle C option would have a devastating impact on CCR beneficial use.

In an example that is almost as inappropriate as the reference to used oil recycling, EPA points to the fact that -ehat," a CERCLA mining waste, is used in road

construction as support for its position that regulating CCRs as a listed waste under RCRA Subtitle C will not harm CCR beneficial use. *Id.* As EPA knows, however, chat is not a RCRA listed waste and therefore this example has no relevance whatsoever to the question of stigma resulting from classification of a material as a listed waste under RCRA. Indeed, virtually thousands of everyday materials can be subsumed by the extremely broad definition of a CERCLA hazardous substance," whereas only a finite number of materials have ever been listed by EPA as RCRA Subtitle C listed wastes. This is a nonsensical comparison.

With respect to the remaining materials – EAF, electroplating wastewater sludge and spent solvents – the Agency itself concedes that these comparisons are of questionable relevance as they involve products that, unlike CCRs, –are *not* used in residential settings." *Id.* (emphasis added). In fact, it is precisely because CCRs are used in such a wide variety of beneficial use applications, including in wallboard and concrete in both residential and commercial settings, that the stigma of a hazardous waste listing would have such a devastating impact on the beneficial use of these materials. As explained by Bill Gehrmann, President of Headwaters Resources, Inc., in his recent testimony before a congressional subcommittee on the adverse impact of the Subtitle C option on CCR beneficial use:

In its proposed coal ash disposal rule, the EPA cites examples of other industries in which materials designated as hazardous' have been successfully recycled. None of EPA's examples, however, are analogous to coal ash – which is used without undergoing additional processing and is placed in products that come into direct contact with end users. EPA's examples also concern materials that are sold to sophisticated users accustomed to handling hazardous materials. Coal ash users do not have this level of experience and capability.

Testimony of William H. Gehrmann, President, Headwaters Resources, Inc., before the House Subcommittee on Rural Development, Entrepreneurship and Trade Committee on Small Business (July 22, 2010) (attached as Appendix 7). As these statements make clear, those entities with the most extensive experience and informed knowledge with respect to the CCR beneficial use market have provided the Agency with legitimate reasons as to why EPA's attenuated recycling examples have no bearing on the question of stigma in this rulemaking.

B. The Subtitle C Option Is Inconsistent With The Bevill
Amendment And Arbitrary And Capricious Because It
Would Result In Unnecessary And Excessively
Burdensome Regulation Of CCRs

A fundamental purpose of the Bevill Amendment is to ensure that any final regulatory option that EPA selects for CCR uses cost-effective controls, to avoid imposing undue regulatory costs on the utility industry. Representative Bevill could not have been more direct on this point when introducing his Amendment, emphasizing that any final regulation of CCRs not result in -unnecessary inflationary impact[s]" on the use of coal from overly stringent CCR regulations. 126 Cong. Rec. 3361; see also id. at 3363 (remarks of Rep. Rahall) (-[a]t a time when we are seeking to encourage electric utilities and others to switch from the burning of oil to coal, it would be highly inappropriate to place further unnecessary regulatory roadblocks in the way of increased coal usage"); 125 Cong. Rec. at 13246 (remarks of Sen. Byrd) (under this amendment, -those who produce coal will be able to do so without the fear that utility customers will be forced to use some other fuel because of environmental regulations").

The D.C. Circuit expressly acknowledged the Bevill Amendment's directive to EPA that it consider costs in making a final regulatory determination, emphasizing that:

Congress intended attention to cost and the economic impact of regulatory controls in making a regulatory determination. The emphasis on economic factors is consistent with Congress' obvious goal in enacting the Bevill Amendment – to relieve [the Bevill industries] of the onerous economic burden of stringent Subtitle C controls *if at all possible*.

EDF v. EPA, 852 F.2d at 1314 (emphasis added). Indeed, RCRA Section 8002(n)(6) specifically lists costs as an independent study factor in determining whether CCRs warrant regulation under RCRA Subtitle C.

EPA's Subtitle C option is flatly inconsistent with Congress's -ebvious goal in enacting the Bevill Amendment" to avoid Subtitle C regulation and the associated regulatory costs of this option -if at all possible." In fact, EPA readily acknowledges that one of the express purposes of the Subtitle C option is to *increase* the costs of CCRs under the Agency's misguided impression that this will increase CCR beneficial use.

See 75 Fed. Reg. at 35185. In fact, as discussed below, EPA's proposed Subtitle C option would regulate CCR disposal facilities more stringently than *any* other hazardous waste by taking the unprecedented step of extending Subtitle C controls to previously closed and inactive CCR surface impoundments. *Id.* at 35177. Thus, not only does the Subtitle C option reflect the most extreme and costly regulatory choice available to EPA under RCRA, it would apply the hazardous waste disposal rules to CCRs in a manner more stringent than that previously applied to any other hazardous wastes.

This draconian approach to regulating CCRs under RCRA is diametrically opposed to the very purpose and objective of the Bevill Amendment and cannot be pursued. While Congress directed EPA to select the most cost-effective regulatory option for CCRs, the proposed Subtitle C option appears to be specifically designed to

have just the opposite result; namely, to make the disposal of CCRs as costly and as burdensome as possible.

The EPRI Cost Report referenced above examined the full range of Subtitle C compliance costs for utilities, including the costs of retrofitting upstream CCR handling units (e.g., hoppers, tanks and conveyor systems) to meet EPA's hazardous waste design and operating requirements. See EPRI Cost Report at 2-11 – 2-12. EPA acknowledges that this latter cost element – the need to upgrade and/or retrofit upstream CCR handling units – is a necessary cost component of the Subtitle C option, but the Agency's Subtitle C cost estimate does not include this important cost component because EPA lacked information -on baseline CCR storage practices." 75 Fed. Reg. at 35212. The EPRI Cost Report also includes another important cost element of Subtitle C compliance not evaluated by EPA; specifically, the costs to power plants of having to construct new wastewater treatment facilities to replace the wastewater treatment function served by CCR surface impoundments that would have to close under the Subtitle C option. EPRI Cost Report at 2-10 – 2-11. In its cost analysis, EPA fails to recognize that CCR surface impoundments often serve a dual function at power plants: both to manage CCRs and to serve as a component of the plant's wastewater treatment system. Mandatory closure of the CCR surface impoundments under the Subtitle C option will require power plants to invest substantial resources in designing and constructing new wastewater treatment facilities.

Finally, the EPRI Cost Report also includes the increased disposal costs that will be incurred by electric utilities under the Subtitle C option due to the need to send CCRs to off-site commercial hazardous waste disposal facilities. *Id.* at 2-14 – 2-16. As

discussed further below, EPA mistakenly assumes that, under the Subtitle C option, electric utility CCR disposal options will remain unchanged. In fact, however, for a multitude of reasons, many utilities will not be able to permit new Subtitle C landfills onsite, will be legally prohibited from using nearby existing commercial RCRA Subtitle D landfills, and will be compelled to transport CCRs currently managed on-site or locally to out-of-state commercial hazardous waste disposal facilities.

All told, the EPRI Cost Report estimates that the compliance costs for electric utilities will be, at a minimum, in the range of \$55.3 to \$74.5 billion over a 20-year horizon (at a seven percent discount rate). *Id.* at 4-1. If a three percent discount rate is used, which is often appropriate where, as here, the regulatory costs could be passed onto consumers in the form of higher electricity rates, the Subtitle C compliance costs jump to a range of \$78.9 to \$110.0 billion. *Id.* at 4-4 – 4-5. These numbers are in contrast to the \$20.1 billion in compliance costs estimated by EPA in its Regulatory Impact Analysis (-RIA").

Further, even the staggeringly high compliance numbers included in the EPRI Cost Report are predicated on a number of conservative assumptions regarding the ability of power plants to meet the Subtitle C requirements in a full and timely fashion.

Id. at 4-5 – 4-6. In fact, the real world compliance costs may be far higher. These costs would be sufficiently high to render some coal-fired power plants, particularly smaller generating units, not economically viable, forcing their early closure. See EOP Group,

Further, these estimates do not include other costs that would be incurred under the Subtitle C option, including (1) landfill construction costs and associated land acquisition costs; (2) costs of replacement power during outages or to compensate for early retirements; (3) undertaking facility-wide corrective action as a condition of obtaining a RCRA Subtitle C permit; (4) state hazardous waste generator fees; and (5) the cost of capital. *Id.* at 2-8-2-9.

Inc, -Gost Estimates for the Mandatory Closure of Surface Impoundments Used for the Management of Coal Combustion Byproducts at Coal-Fired Electric Utilities," (Nov. 2010) (-EOP Report") at 13 (attached as Appendix 19). Thus, instead of promoting the use of coal by selecting the most cost-effective regulatory option for CCRs, EPA's Subtitle C option will actually discourage the use of coal by rendering the use of coal at some facilities uneconomical. This result is in direct contravention of the express congressional purpose behind the Bevill Amendment.

Such an approach is unsupportable given the fact that EPA itself readily acknowledges that the substantive controls for CCR landfills and surface impoundments are similar. See 75 Fed. Reg. at 35213. Given that the substantive disposal controls under the Subtitle D option would be just as effective in protecting human health and the environment, as those provided for under the Subtitle C option at far less cost and without the severe collateral impacts on beneficial use, the price of power and the use of coal, the Subtitle C option is inconsistent with the stated purpose of the Bevill Amendment.

The hazardous constituents present in CCR leachate also support EPA's earlier Regulatory Determinations that these materials can appropriately be managed under a Subtitle D program. As AECOM Environment summarizes in its report, EPRI conducted a comparison between the hazardous constituents present in CCR leachate and leachate from municipal solid waste landfills (-MSWLFs"), which are, of course, subject to Subtitle D regulation. See AECOM Environment, Report on CCR Proposal (Nov. 2010) (-AECOM Report") at 4 (attached as Appendix 20). This analysis determined that 99 hazardous constituents were detected in the MSWLF leachate of which 62 were

detected at the 90<sup>th</sup> percentile, while only 14 hazardous constituents were detected in the CCR leachate all of which were detected at the 90<sup>th</sup> percentile. *Id.* Given the significantly broader range of hazardous constituents in MSWLF leachate, it appears wholly unjustified to subject CCR to more restrictive Subtitle C regulation while MSWLFs are subject to Subtitle D controls.

Further, EPA's selection of the Subtitle C option would be arbitrary and capricious because EPA has not explained why its earlier conclusions that the Subtitle C option would result in excessive and inappropriate regulatory controls for CCRs are no longer true. In its 1993 final Regulatory Determination concluding that the four large volume CCRs – fly ash, bottom ash, boiler slag, and flue gas desulfurization waste – do not warrant hazardous waste regulation, EPA emphasized that the inflexible Subtitle C hazardous waste framework was inappropriate for CCRs, explaining -that it is unlikely that Subtitle C would effectively address the problems associated with the four large-volume fossil-fuel combustion wastes [CCRs] without imposing unnecessary controls." 58 Fed. Reg. at 42477 (emphasis added).

This was an important conclusion, because as EPA explained in outlining the factors that it would evaluate in making a final regulatory determination, the special status of the [CCR] waste *requires* that the Agency consider the impacts to the industry that regulation under Subtitle C would create in making a decision to regulate the waste as hazardous." 58 Fed. Reg. at 42471(emphasis added). EPA found that the rigid nature of the Subtitle C regulatory system would result in excess costs and unnecessary regulation as applied to CCR management units. After acknowledging that CCRs rarely exhibit a hazardous waste characteristic and thus would not be subject to hazardous

waste controls unless they were listed as a hazardous waste, EPA emphasized that the listing approach would be inappropriate and excessive:

Furthermore, it was noted that even if these wastes were listed as hazardous, and therefore, regulated under Subtitle C, such an approach would be inappropriate for these wastes. A Subtitle C system would require coal combustion waste units to obtain a RCRA Subtitle C permit (which would unnecessarily duplicate existing State requirements) and would establish a series of waste unit design and operating requirements for these wastes, which would generally be in excess of requirements to protect human health and the environment.

Id. at 42477 (emphasis added).

EPA reiterated this conclusion in its 1999 Report to Congress, explaining that [RCRA] Subtitle C is inappropriate to address any problems associated with disposal of these wastes and that the continued use of site and region specific approaches by the states is more appropriate for addressing the limited human health and environmental risks that may be associated with disposal of these wastes." 1999 Report to Congress Vol. at 1 3-5. EPA emphasized that regulating CCRs under RCRA's hazardous waste program -would unnecessarily duplicate existing State requirements" and -would establish a series of waste unit design and operating requirements for these wastes, which would generally be in excess of requirements to protect human health and the environment." 58 Fed. Reg. at 42471 (emphasis added).

Having previously concluded that the Subtitle C approach would be inappropriate because, among other reasons, it would result in the over-regulation of CCRs in contravention of the Bevill Amendment, EPA provides no explanation whatsoever in the instant proposal as to what, if anything, has changed to suggest that the imposition of these very same Subtitle C controls will no longer result in inappropriate and unnecessary costs on the utility industry. As EPA itself noted earlier, -the special status

of the [CCR] waste requires that the Agency consider the impacts to the industry that regulation under Subtitle C would create in making a decision to regulate the waste as hazardous." *Id.* at 42471. There is no reasoned explanation for EPA's about-face on this issue.

The excessive costs and unduly burdensome regulation of CCRs under the instant proposed Subtitle C option are at least as great today as they were ten years ago when EPA rejected application of the Subtitle C hazardous waste rules for CCRs. As EPA has completely failed to explain why regulation of CCRs under Subtitle C is now appropriate, having expressly found earlier that such regulation would be both --inappropriate" and in -excess of requirements" for CCRs, any final Subtitle C regulation would be arbitrary and capricious. See S. Co. Servs. v. FCC, 313 F.3d 574 (D.C. Cir. 2002) (noting that, in evaluating whether an agency's changed position is arbitrary and capricious, -the issue is whether the agency furnished a reasoned explanation for its changed position.").

C. The Subtitle C Option Is Arbitrary And Capricious Because It Is
Inconsistent With The Views Of The Other Federal And State
Agencies That EPA Was Obligated To Consult In Reaching A Final
Regulatory Determination

EPA's proposed Subtitle C option is opposed by virtually every federal and state agency that has opined on this rulemaking. This broad opposition to the Subtitle C option is significant because the Bevill Amendment and its legislative history make clear that the informed views of these agencies were to be considered by EPA in the issuance of any final regulatory determination for CCRs. Given this broad opposition to the Subtitle C option by the groups Congress directed EPA to consult as part of the decision-making process for CCRs, selection of the Subtitle C option would be

Inst., Inc. v. EPA, 373 F.3d 1251, 1273 (D.C. Cir. 2004) (holding that it was arbitrary and capricious for EPA to issue a regulation directly contrary to the opinion of the National Academies of Science with which EPA was required by statute to consult).

The statute specifically directs that EPA, in preparing its study on CCRs, -shall, as [the Administrator] deems appropriate, review studies and other actions of other Federal and State agencies concerning such material and invite participation by other concerned parties, including industry and other Federal and State agencies, with a view toward avoiding duplication of effort." RCRA § 8002(n). As noted above, Representative Bevill elaborated on this collaborative process by emphasizing that -EPA's studies *should not proceed in a vacuum*, "but rather that EPA was to reach out to those with expertise in the field. As Representative Bevill emphasized:

Moreover, EPA should seek the assistance and cooperation of those most expert in this field. With regard to fossil fuel byproducts, I include in this category not only representatives of coal-burning industries, but personnel from other agencies of Government that are aware of the role coal plays in our national energy policy, or of actual disposal and utilization practices. These agencies include the Department of Energy, the Department of the Interior, the Federal Highway Administration, Department of Commerce, the Department of Agriculture, among others. In the face of our current energy crisis and the increasing costs of Government, American taxpayers cannot afford to have separate agencies of Government working without coordination. Instead, we need a cooperative, informed effort directed to the goals the President has set for us.

126 Cong. Rec. at 3362 (emphasis added).

During the inter-agency review process of EPA's draft CCR proposal by OMB's Office of Information and Regulatory Affairs, the very agencies that Congress directed EPA to consult with regarding any final determination for CCRs uniformly opposed or

sharply questioned the Subtitle C option. For example, the Department of Energy emphatically rejected EPA's proposed regulation of CCR under Subtitle C:

DOE continues to believe that the regulation of these [coal combustion] practices under RCRA subtitle C as hazardous waste is unwarranted, and supports the continued collection of information to help resolve EPA's concerns regarding CCRs.

. . .

EPA ruled in 2000 that a Subtitle C designation was unwarranted, and all available evidence supports the conclusions made at that time. Continued evidence does show that unlined units have caused environmental damage; however, the promulgation of this Subtitle C regulation to treat all CCRs as hazardous waste does not appear to be justified, especially when, as mentioned by EPA in the preamble, the disposal practices at power plants have shown a history of improvement and are continuing to improve. The benefits of CCR regulation under Subtitle C are not apparent and DOE urges EPA to fully understand the implications of such a regulation before going forward with the current proposed designation.

. . .

In the Regulatory Determination (RD) of 2000, which DOE agreed with, EPA decided to regulate CCRs as Subtitle D waste, stating that —The Agency has determined that industry practices are moving toward increased use of control measures (liners, covers, etc.) and groundwater monitoring." This statement is further reinforced by the 2006 joint DOE/EPA report on recent disposal practices. <sup>19</sup>

. . .

Despite the indication of better disposal practices, the proposed rule advocates Subtitle C regulations. This is in direct opposition to the 2000 RD that Subtitle C regulations were unwarranted, when EPA determined that it is unlikely that Subtitle C would effectively address the problems associated with the four large-volume fossil-fuel combustion wastes without imposing unnecessary controls.

Interagency Working Comments on Draft Rule under EO 12866 at p. 21-22.

<sup>&</sup>lt;sup>19</sup> Coal Combustion Waste Management at Landfills and Surface Impoundments, 1994-2004. Available at: http://www.osti.gov/bridge.

The Department of the Interior (-ĐOI") sharply questioned the Subtitle C option and echoed the concerns referenced above by the CCR beneficial use industry that -[r]egulating CCR disposal as RCRA Subtitle C, in full or by implementing a hybrid regulation, we believe that the availability to purchase fly ash for use in concrete will be eliminated." *Id.* at 26. DOI stated that it concurs -with industry leaders who feel strongly that if fly ash is designated a hazardous waste, fully or in a hybrid classification, it will no longer be used in concrete." *Id.* 

The Department of Transportation (-ĐOT") similarly cautioned EPA against regulating CCRs under Subtitle C because of the adverse impact on CCR beneficial use:

DOT is concerned about the negative impacts that may result from designating fly ash as a hazardous material .... The Rule would designate all CCRs as hazardous and then exempts beneficial uses. However, DOT is concerned that this approach is confusing, and may still have unintended consequences associated with Subtitle C. For example, many States have policies that forbid the use of any hazardous materials. Thus, if fly ash is designated as a hazardous material, it would fall under these general State prohibitions and would legally prevent the use of fly ash in public works structures and highways. Furthermore, an exemption in a Federal regulation probably would not negate State laws. If a material is in any manner considered hazardous, States will not risk future liability.

Id. at p. 28-29. The Department of Agriculture also cautioned EPA that: Listing of all CCRs as hazardous waste significantly inhibits recognized, current, and continuing beneficial uses." Id. at 41.

Comments submitted by the White House Council for Environmental Quality (–CEQ") directly challenged the Subtitle C option, explaining that –it will be very expensive to regulate CCR with likely little environmental benefit based on what has been presented thus far . . . less expensive approaches could include enforcement of

annual inspections which would provide much greater environmental benefits relative to those proposed." *Id.* at 17. CEQ also noted that approximately 2.5 million tons of materials are disposed of annually as hazardous waste and that bringing CCRs into the hazardous waste system would add the order of 130 million tons per year to this inventory," thus causing disposal capacity shortfalls and likely requiring some facilities to shut down temporarily, or permanently, due to the lack of viable hazardous waste disposal options." *Id.* Further, CEQ identified the serious compliance challenges that would be imposed on utilities under the Subtitle C option, including the fact that *de minimis* releases of CCRs -would constitute improper hazardous waste disposal and subject facilities to non-compliance." *Id.* Given these problems, CEQ recommended that the Agency -re-examine if Subtitle C regulation is warranted, in accordance with Section 3004(x) and focus in on the special characteristics of such waste, the practical difficulties associated with implementation of such requirements, and site-specific characteristics." *Id.* at 18.

In addition to the above federal agencies, Congress directed EPA to consult with state agencies in assessing whether and how to regulate CCRs under RCRA. State trade associations and individual state environmental protection agencies have uniformly advised EPA that regulating CCRs under Subtitle C would have a devastating impact on CCR beneficial use. In addition, national associations representing state governments have more broadly advised EPA of their opposition to the Subtitle C option because it would be duplicative of and disruptive to existing state controls for CCRs. It is precisely this type of duplication that the Bevill Amendment directs EPA to avoid. See RCRA § 8002(n).

For example, the National Governors Association advised EPA that the

-Governors support their state-run programs and have concerns regarding potential
federal regulation of CCW. ... In summary, Governors are very concerned about
potential federal regulation of CCW as either a hazardous or nonhazardous substance
under RCRA. While we are unaware of issues EPA may have with our state programs,
we hope to work with you and your staff to address any questions." Letter from National
Governors Association to EPA at 1 (Nov. 16, 2009) (attached as Appendix 21). The
Western Governors Association also disagreed with EPA's proposed regulation of
CCRs under Subtitle C:

The federal regulation of CCB as hazardous waste would undercut existing and effective state regulatory authority resulting in additional and unwarranted regulatory programs and would add costly burdens to states' budgets that are already significantly strained. States should review their existing regulations of CCB and make any changes necessary in light of the 2008 TVA impoundment pond failure.

Western Governors agree with US EPA's 1993 and 2000 regulatory determinations that CCB do not warrant regulation as hazardous wastes and that the western states have an effective regulatory infrastructure in place to continue as the principal regulatory authorities to ensure protection of human health and the environment through the safe and secure management of CCB under state solid waste, groundwater protection and coal mine regulatory programs.

Western Governors Association, Policy Resolution 10-1 (attached as Appendix 22).

Further, the Environmental Council of the States (—ECOS") has expressly opposed EPA's proposed regulation of CCRs under Subtitle C, stating that ECOS:

Agrees with U.S. EPA's repeated assessments in 1988, 1993, 1999, 2000, and 2005 that CCW disposal does not warrant regulation as hazardous wastes under RCRA Subtitle C:

Agrees with U.S. EPA's finding in the 2005 study previously cited that -the regulatory infrastructure is generally in place at the state level to ensure

adequate management of these wastes" and believes that states should continue to be the principal regulatory authority for regulating CCW as they are best suited to develop and implement CCW regulatory programs tailored to specific climate and geological conditions designed to protect human health and the environment:

Therefore [ECOS] calls upon U.S. EPA to conclude that additional federal CCW regulations would be duplicative of most state programs, are unnecessary, and should not be adopted, but if adopted must be developed under RCRA Subtitle D rather than RCRA Subtitle C, and in addition, urges U.S. EPA to make a timely decision, and calls upon U.S. EPA to begin a collaborative dialogue with the states to develop and promote a national framework for beneficial use of CCW including use principles and guidelines, and to accelerate the development of markets for this material.

ECOS, Revised Resolution No. 08-14 (Mar. 23, 2010) (attached as Appendix 23).

Similarly, the Association of State and Territorial Solid Waste Management

Officials (-ASTSWMO") expressly rejected EPA's proposed regulation of CCRs under

Subtitle C:

To artificially classify [coal combustion products (CCB)] as hazardous will needlessly limit the management options for both the CCBs and other wastes legitimately classified as hazardous which will be competing with CCBs for limited hazardous waste disposal capacity, while not producing any great degree of environmental protection. Transportation, manifesting and licensing requirements for CCBs as a listed hazardous waste are excessively burdensome without sufficient evidence of a benefit. It would be more appropriate to regulate and manage CCBs using design and operation standards specified for Subtitle D programs except in cases where a particular source material is deemed hazardous upon testing for characteristics.

The most compelling reason not to impose Subtitle C regulations is that the beneficial use of CCB has been very successful. The -hazardous" label of Subtitle C would be detrimental to State CCB beneficial use programs[.] Regulation under RCRA Subtitle C has the potential to put an end to many beneficial uses for CCB.

Letter from ASTSWMO to EPA, at 4-5 (Apr. 1, 2009) (attached as Appendix 12).

In short, the statute and its legislative history make clear that EPA was to -seek the assistance and cooperation of those most expert in this field," including the above-

referenced federal and state agencies, in making a final regulatory determination.

EPA's selection of the Subtitle C option would be directly contrary to the unanimous views of these informed parties and to Congress' directive that EPA produce a -eollaborative, informed" regulatory determination for CCRs.

#### V. THE DAMAGE CASES DO NOT SUPPORT THE SUBTITLE C OPTION

EPA suggests that the newly identified -damage cases" may warrant reversal of the 2000 Regulatory Determination, resulting in the regulation of CCRs under RCRA Subtitle C. 75 Fed. Reg. at 35155-56.<sup>20</sup> However, the new damage cases, which are limited almost exclusively to older, unlined units as were the original damage cases, in no way undermine EPA's original decision in the 2000 Regulatory Determination that the best means to address the damage cases is to -develop national Subtitle D regulations." 65 Fed. Reg. at 32216.

USWAG acknowledges that some of the damage cases identified by EPA have had environmental impacts. At the same time, however, the proposed Subtitle D requirements – including the liner, groundwater monitoring, corrective action and structural stability requirements – will be fully effective in addressing EPA's concerns about the damage case sites without the attendant adverse impacts of the Subtitle C option. Therefore, the new damage cases justify neither the reversal of EPA's 2000 final Regulatory Determination nor the resulting regulation of CCRs under Subtitle C. We discuss below the fact that the proven damage cases represent only a subset of

One of the eight statutory study factors that EPA must evaluate in determining whether CCRs warrant regulation under RCRA Subtitle C is -documented cases in which danger to human health or the environment from surface runoff or leachate has been proved." RCRA § 8002(n)(4). These cases are referred to by EPA as -damage cases." See 75 Fed. Reg. at 35146.

CCR management units and that EPA's concerns about this distinct subset of units will be adequately addressed under the Subtitle D option.

## A. <u>The Damage Cases Represent A Subset Of Older, Unlined CCR Disposal Units</u>

At the time of the final 2000 Regulatory Determination, EPA had identified 14

—proven" damage cases<sup>21</sup> and 36—potential" damage cases.<sup>22</sup> 75 Fed. Reg. at 35142.

Since that time, EPA has identified 13 additional proven damage cases and four more potential damage cases, bringing the total to 27 proven cases and 40 potential damage cases. *Id.* at 35143. Virtually all of these damage cases involve unlined units that began operation before 1990; these cases do not involve the newer, lined units that have commenced operation since that time. *See*—Evaluation of Coal Combustion Product Damage Cases," Electric Power Research Institute, July 2010, EPRI Report No. 1020553 (—EPRI Damage Case Report").

Before addressing the nature of the damage cases, it is important to emphasize that it is inappropriate for EPA to include potential damage cases in its evaluation of the appropriate regulatory option for CCRs. The Bevill Amendment provides that, in considering the -documented cases" of danger from the management of CCRs, the

Proven" damage cases —means those cases with (i) Documented exceedances of primary maximum contaminant levels (MCLs) or other health-based standards measured in ground water at sufficient distance from the waste management unit to indicate that hazardous constituents have migrated to the extent that they could cause human health concerns, and/or (ii) where a scientific study provides documented evidence of another type of damage to human health or the environment (e.g., ecological damage), and/or (iii) where there has been an administrative ruling or court decision with an explicit finding of specific damage to human health or the environment." 75 Fed. Reg. at 35131.

<sup>&</sup>lt;sup>22</sup> Potential damage case means those cases with documented MCL exceedances that were measured in ground water beneath or close to the waste source. In these cases, while the association with CCRs has been established, the documented exceedances had not been demonstrated at a sufficient distance from the waste management unit to indicate that waste constituents had migrated to the extent that they could cause human health concerns. *Id*.

Agency was only to evaluate those instances where such danger —has been proved."

RCRA § 8002(n)(4) (emphasis added). EPA has acknowledged this statutory limitation, observing in its final 2000 Regulatory Determination that, although the potential damage cases may pose a —potential danger" to human health and the environment in some circumstances, they do not —satisfy the statutory criteria of documented, proven damage cases because damage to human health or the environment has not been proven. . . . ."

65 Fed. Reg. at 32225. Therefore, it is inconsistent with the plain language of the statute for EPA to include the potential damage cases in its assessment of the —documented cases" of danger to human health and the environment from the management of CCRs. EPA cannot use the potential damage cases as the foundation for a reversal of its final 2000 Regulatory Determination.

It is important to note that even the proven damage cases are limited to only a subset of the universe of CCR disposal units; namely, older, unlined units. Of the 16 proven damages cases identified in the 2007 NODA that involved groundwater contamination, 15 of the 16 units were unlined (with the remaining unit being a surface impoundment where it was unclear whether the unit was lined or unlined). 75 Fed. Reg. at 35147. The only other proven damage case involving groundwater contamination identified since the 2007 NODA is the BBBS Sand and Gravel Quarries in Gambrills, MD, also an unlined site. *Id*.

Even assuming, for purposes of argument, that it is appropriate to include the potential damage cases in this evaluation, the vast majority of impacts to groundwater from the potential damage cases are similarly from unlined facilities. According to EPRI's comprehensive evaluation of the combined total of 63 proven and potential

damage cases identified in the 2007 NODA, only six (10%) had liners. EPRI Damage Case Report at 3-5. Since issuance of the NODA, the only other potential damage case identified by EPA is the Battlefield Golf Course in Chesapeake, VA, an unlined site, although EPA has recently issued a report concluding that no residential wells were impacted by CCRs at the site and that there are no adverse health effects expected from exposure to surface water or sediments at the site. See http://yosemite.epa.gov/opa/admpress.nsf/0/72D8A4102AE78DA88525770D00546146.

In addition, virtually all of the damage cases, both proven and potential, involve older landfills that began operating before it was common practice to install liners in CCR disposal units. Of the 63 proven and potential damage cases identified in the 2007 NODA, approximately two-thirds of all the facilities for which operating periods could be established began operations before 1976 and *all* began operations before 1990. *Id.* at 3-7 – 3-8. The data are even more striking when this analysis is properly limited to the proven damage cases; all of the 24 proven damage cases identified in the NODA commenced operations between 1948 and 1983, with -the median opening year of 1967." *Id.* at 3-8.

As EPA itself reported in the DOE/EPA Report, the vast majority of CCR disposal units that have been constructed since 1990 are —better designed, in that they are lined and have installed groundwater monitoring systems, and therefore the total percentages of unprotected units have been reduced." 75 Fed. Reg. at 35150. In fact, the DOE/EPA Report found that —[t]he vast majority (98%) of the 56 identified units (both landfills and surface impoundments)" built or expanded from 1994 to 2004 have liners. *Id.* at 31.

The point is that the damage cases involve older and almost exclusively unlined units and represent only a segment of the total universe of CCR disposal units.

Not only do the damage cases involve an unrepresentative subset of CCR disposal units, legitimate questions exist regarding whether even this limited subset of units are truly -damage" cases. The EPRI Damage Case report identifies a number of important flaws with the proven and potential damage cases identified in the 2007 NODA, including, among other things, that (1) characterization of at least 13 of the sites as -damage" cases was questionable because of the lack of any long-term impacts that could be attributed to CCRs; (2) nearly two-thirds of the 54 groundwater damage case facilities were located at sites where there was a low potential for downgradient receptors; and (3) off-site exceedances of MCLs attributable to CCR impacts, a fundamental criterion for being classified as a proven damage case, were observed *in only three cases* of the combined total of 54 proven and potential groundwater damage cases. *Id.* at vi of Report Summary.

These findings underscore the fact that the damage cases, including in particular the proven damage cases, represent only a fraction of CCR disposal units and that the limited number and nature of these cases do not warrant Subtitle C regulation. The lack of justification for CCR regulation under Subtitle C is underscored by the fact that the Subtitle D controls can fully address EPA's concerns with these facilities.

# B. The Proposed Subtitle D Controls Will Prevent Future Damage Cases, And Existing Damage Cases Are Being Effectively Remediated

EPA's primary concerns with the identified damage cases involve surface and groundwater contamination from CCR disposal units and the structural failure of CCR

surface impoundments. 75 Fed. Reg. at 35147. These are precisely the risks, however, that the proposed Subtitle D controls are designed to address. EPA has stated repeatedly that the substantive standards for CCR disposal units are virtually identical under both the Subtitle C and Subtitle D options (id. at 35213), with both proposed regulatory programs requiring liners and leachate collection systems for new disposal units and lateral expansions of existing units, as well as the retrofitting of existing CCR impoundments. See, e.g., id. at 35243-45 (proposed 40 C.F.R. §§ 257.70-72). Further, under the Subtitle D option, existing CCR disposal units would be subject to groundwater monitoring standards that will allow for the early detection of any groundwater contamination from both lined and existing unlined units. *Id.* at 35246 (proposed 40 C.F.R. § 257.90). Finally, the Subtitle D option would impose structural integrity requirements on all CCR surface impoundments, thus addressing EPA's concerns with the potential structural failure of such units that served as the basis for identifying the TVA Kingston facility and Martin's Creek Power Plant as proven damage cases. Id. at 35243-45 (proposed 40 C.F.R. §§ 257.71(b)-(f), 257.83).

While the Subtitle D regulations will be effective in preventing future damage cases, the existing damage cases already are being properly remediated under state and federal remediation programs. As EPA acknowledged with respect to the proven damage cases identified in the 2000 Regulatory Determination, all of these cases are being adequately addressed by the states (and in two cases, by EPA under the Superfund program), which EPA explained shows the effectiveness of states responses when damages were identified." 65 Fed. Reg. at 32216. In addition, the newly identified damage cases are being effectively remediated. The EPRI Damage

Case Report includes a detailed analysis of the remediation efforts at these additional sites and found that -[a]vailable information indicates that remediation is complete or underway at *all* sites where remediation was required." *Id.* at 8-1; *see also id.* at 7-1 – 7-2. Indeed, of the 16 proven cases of damages to groundwater, EPA has confirmed that -corrective actions have been completed in seven cases and are ongoing in the remaining nine cases." 75 Fed. Reg. at 35150 n.43.

In short, existing damage cases are already being adequately addressed, and the proposed Subtitle D regulations would prevent future damage cases. Therefore, there is no reason for EPA to pursue the Subtitle C option when the Subtitle D option would effectively address the issues associated with the damage cases. Given the underlying goal of the Bevill Amendment that EPA avoid the Subtitle C regulation of CCRs if at all possible" (*EDF v. EPA*, 852 F.2d at 1314), there is no basis for EPA to rely on the damage cases as the basis for pursuing the Subtitle C option.

## C. The Alleged Damage Cases Presented by Environmental Groups Are Not Proven and Cannot Be Relied Upon by EPA in This Rulemaking

EPA references the allegations of additional damage cases by certain environmental organizations, including in particular assertions made by the Environmental Integrity Project (—EIP") and Earthjustice in a February 24, 2010 report (75 Fed. Reg. at 35148) and, more recently, an August 26, 2010 report from the same two environmental organizations and the Sierra Club. EPA acknowledges that it has not had time to review these allegations, though it promises to review this information and encourages other stakeholders to also comment on these allegations. *Id*.

As an initial point, it is inappropriate for EPA to rely on the allegations of these environmental groups in any final rulemaking for CCRs before the Agency has fully evaluated the allegations and provided an opportunity for public hearing and comment on EPA's initial conclusions. Indeed, this is precisely what the Bevill Amendment requires. Before making any final regulatory determination, EPA is required to submit its initial recommendation, along with the reasons for this recommendation, in a Report to Congress, and to provide an opportunity for public comment and hearings on these initial findings. For EPA to base a new regulatory determination on the mere allegation of additional damage cases without first providing the public an opportunity to review and comment on EPA's initial conclusions regarding the allegations would be inconsistent with the dictates of the Bevill Amendment.

Past experience underscores the importance of this statutorily-prescribed vetting process, as many of the previous allegations of damage cases identified by environmental groups have been meritless and have been dismissed by EPA as unsubstantiated and/or incorrect. For example, prior to initiating this rulemaking, EPA gathered or received information on 135 alleged damage cases, many of which were submitted to EPA by the same environmental organizations referenced above. 75 Fed. Reg. at 35147. After evaluating these allegations, close to half (62) were dismissed -because there was little or no information supporting the concerns identified." *Id.* Another six were outside the scope of the rulemaking. *Id.* Of the remaining 67 allegations, 24 were identified as proven damage cases, but this included the 14 proven cases already identified in the 2000 Regulatory Determination. *Id.* The point is that less

than two-tenths of the cases -alleged" to have been damage cases have, in the end, been determined by EPA to be -proven" damage cases.

The same over-counting of damage cases exist in the case of the more recent damage allegations submitted by EIP, Earthjustice and the Sierra Club. An initial analysis by EPRI of certain of these allegations reveals that there are similar factual errors and/or lack of a factual basis for classifying these sites as damage cases. As EPRI reports, these groups have identified 70 new alleged damage cases, -based primarily on researching state records." EPRI Comments on CCR Proposal (Nov. 19, 2010) (—EPRComments") at 43. EPRI's summary review of these new allegations reveals the same type of factual errors and/or misinterpretations as contained in past damage case allegations presented by these groups. For example, EPRI reports that -fe]ff site migration claims are frequently not substantiated," pointing out that for 35 of the alleged cases (half of the alleged new cases), environmental groups claim that groundwater contamination is -moving off-site." *Id*. at 47. In fact, however, the EPRI Report reveals that there appears to be little or no documentation of actual off-site movement at these sites, and in fact the reports state many of these sites don't have off-site wells." Id. The lack of factual foundation and simple factual errors also undermine the veracity of the new allegations. In one case, single occurrences of elevated levels of certain metals were reported as evidence of contamination, when subsequent samples over the next 12 years were all very low, indicating that the —sigle occurrences clearly appear to be outliers." Id. at 44. In yet another case, the groundwater well that is relied upon for alleging that contamination has been observed off-site, is in fact located *on*-site. *Id*. at 48. These types of factual errors and unfounded assertions underlie many of the new allegations of damage cases made by the above organizations See id. at 43-49.

In fact, similar factual inaccuracies were identified by the Pennsylvania Department of Environmental Protection (—PADEP") with respect to the damage case allegations by EIP and Earthjustice in their February 2010 report entitled -Out of Control: Mounting Damages from Coal Ash Waste Sites." In that report, EIP and Earthjustice identified four facilities located in the southwest region of PADEP as constituting -damage cases." PADEP explains that -{s}ince the EIP report did not contain standard scientific documentation, the Department has responded based on a scientific review of each facility's data and the Department's familiarity of the sites and their groundwater data and impacts." Commonwealth of Pennsylvania Department of Environmental Protection Southwest Region, Review of the Environmental Integrity Project and Earthjustice Report: -Out of Control: Mounting Damages From Coal Ash Waste Sites," (Oct.13, 2010) (attached as Appendix 24). The PADEP -reviewed and responded to each specific allegation point by point" and concluded, after a -methodical, organized and scientific review" of its data, that -the allegations regarding groundwater and surface water contamination by Primary MCL's are unfounded." Id.

In short, the Bevill Amendment prohibits EPA from relying on the new allegations of damage cases without first fully examining these allegations and providing the public with an opportunity to evaluate and comment on the Agency's initial findings. Given that the vast majority of similar allegations in the past have been meritless, this type of careful review is necessary to ensure that unfounded and/or incorrect charges of new damage cases are not used as the basis for any final decisions.

#### **CHAPTER TWO**

### THE SUBTITLE C PROPOSAL IS LEGALLY FLAWED

### VI. EPA'S PROPOSED LISTING OF CCRS AS A SPECIAL WASTE IS LEGALLY DEFECTIVE

EPA proposes to regulate CCRs under the Subtitle C option by listing CCRs as a special" waste under RCRA's hazardous waste regulations. 75 Fed. Reg. at 35166.

The listing would apply to all CCRs destined for disposal. *Id.* In proposing to regulate CCRs, the Agency evaluated CCRs against the listing criteria in 40 C.F.R.

§ 261.11(a)(3). *Id.* at 35166. Under this provision, EPA may list a waste as hazardous if, after evaluating ten individual criteria, <sup>23</sup> the Agency concludes that -the waste is capable of posing a substantial present or potential hazard to human health or the environment when improperly treated, stored, transported or disposed of, or otherwise managed." *Id.* at 35167. As discussed above, despite referring to CCRs captured by the listing as -special waste," EPA readily acknowledges that CCRs included in the listing description would be regulated under the full scope of RCRA's Subtitle C hazardous waste regulations, including -40 C.F.R. parts 260 through 268 and 270-279 and 124, and subject to the notification requirements of section 3010 of RCRA." *Id.* at 35166.

CCRs included in the listing would be categorized as -eoal combustion residuals generated by the electric power sector (Electric Utilities and Independent Power Producers)" and assigned EPA waste code S001. *Id.* at 35254 (proposed 40 C.F.R.

While the regulation lists eleven criteria, there are really only ten, as the eleventh factor is a -eatch-all, allowing the Administrator to consider any other factor she finds relevant." *Dithiocarbamate at* 1398.

§ 261.50(a)).<sup>24</sup> For purposes of the proposed S001 listing, the term -eoal combustion residuals" is defined to include -fly ash, bottom ash, boiler slag, and flue gas desulfurization materials generated by the electric utility industry." *Id.* (proposed 40 C.F.R. § 261.50(b)). The listing would cover CCRs disposed of in any type of disposal unit, regardless of the location of the unit and/or whether it has a liner and/or groundwater monitoring system.

EPA's proposed listing of CCRs is fatally defective for a number of reasons, including (1) the scope of the proposed listing is overly broad and, as a result, EPA has not properly evaluated the individual CCRs against the listing criteria; (2) EPA has employed an unrealistic scenario in evaluating the -plausible mismanagement" scenario to which the waste could be subjected; (3) EPA has not identified, let alone evaluated, other federal and state regulatory programs applicable to CCRs as required by the listing criteria; (4) the risk assessment that serves as the cornerstone for certain elements of the proposed listing contains fundamental errors; and (5) EPA's draft screening analysis involving the potential risks for CCR fugitive dust emissions is based on inaccurate inputs and does not reflect real world conditions. As a result of these flaws, EPA may not list any of the four CCRs as a listed waste under Subtitle C.

### A. The Overbroad Listing Undermines EPA's Listing Analysis

EPA proposes to include four distinct categories of waste from the combustion of coal (fly ash, bottom ash, boiler slag, and flue gas desulfurization materials (-FGD materials")) into the single listing category of -eoal combustion residuals," implicitly

For purposes of this listing, the electric power sector would be defined -as electricity-only and combined-heat-and-power (CHP) plants whose primary business is to sell electricity, or electricity and heat, to the public; i.e. NAICS code 221112 plants." *Id.* (proposed 40 C.F.R. § 261.50(b)).

concluding that all four wastes can be viewed as single waste stream for purposes of evaluating whether they meet the ten listing criteria in 40 C.F.R. § 261.11(a)(3). This approach inappropriately combines four physically and chemically distinct classes of waste into a single waste code. Because of this error, EPA has failed to properly evaluate these distinct classes of wastes against the ten regulatory listing criteria, as is required by the plain language of the listing regulations. As a result, EPA cannot proceed with listing any one of the four types of CCRs as listed waste under Subtitle C.

The D.C. Circuit addressed a similar overly-broad listing attempt by EPA in *Dithiocarbamate*. In that case, EPA unlawfully bundled into a single listing description a group of carbamate compounds that, while having -similarities in their chemical origins and structures," were distinct enough for a variety of reasons to preclude the Agency from evaluating and subsequently classifying the entire group as single listed waste. *Dithiocarbamate*, *98 F.3d at 1394*. In its holding, the court established certain rules that EPA must adhere to when evaluating whether to list a waste under 40 C.F.R. § 261.11(a)(3). EPA has not, and cannot, meet these rules in the case of CCRs.

First, the court made clear that the Agency must ensure that the waste is evaluated under *all* ten of the listing criteria set forth in § 261.11(a)(3). *Id.* at 1398-99. A second and related listing condition is that EPA is only authorized to evaluate a group of wastes in the aggregate -if the known similarities of members of a class are such that it is reasonable to infer the presence of a disputed characteristic throughout the class (not just among members for which it has been shown)." *Id.* at 1399.

As was the case in the *Dithiocarbamate* decision, while the individual categories of wastes comprising the proposed CCR listing category – *i.e.*, fly ash, bottom ash,

boiler slag and FGD materials – may share some common characteristics, it is inappropriate for EPA to view these four wastes as sharing a common characteristic when evaluating them against *each* one of the ten listing criteria under 40 C.F.R. § 262.11(a)(3). As EPA itself acknowledges, fly ash, bottom ash, boiler slag and FGD materials are produced at different points and stages in the power production process and have unique physical and chemical characteristics. *See* 75 Fed. Reg. at 35137;<sup>25</sup> *see also id.* at 35138, Table 2. The individual CCRs are also produced in significantly different volumes, which is an individual listing criterion that must be evaluated by EPA. See 40 C.F.R. § 261.11(a)(3)(viii).<sup>26</sup>

The fundamental differences in the physical and chemical composition of these four wastes, as well as their relative volumes and the site-specific differences in management practices, preclude EPA from evaluating them under the -elass-based" listing approach suggested in the proposal. *See Dithiocarbamate*, 98 F.3d at 1400. Where a group of four distinct types of wastes have different physical and chemical properties, are produced in different volumes, and are managed in different categories of disposal units with differing types of design standards and groundwater monitoring controls, it is arbitrary and capricious to view the wastes as a single class for the

<sup>2</sup> 

<sup>&</sup>lt;sup>25</sup> The chemical distinctions between the individual CCRs are well-documented, including in various EPRI reports (1987, 2010), literature sources (Pflughoeft-Hassett, et al., 2000; Attalla, et al., 2007), and the Agency's 2010 report entitled -Characterization of Coal Combustion Residues from Electric Utilities – Leaching and Characterization Data," and EPA's -Technical Background Document for the Report to Congress on Remaining Wastes from Fossil Fuel Combustion" (March 15, 1999).

Data available for the year 2009 from the American Coal Ash Association (–ACAA") on CCR production in the U.S. reveals that, out of a total of 135 million tons produced in 2009, fly ash comprised approximately 63 million tons (or 47%), bottom ash comprised approximately 16.5 million tons (or 12%), boiler slag comprised just 2 million tons (or just under 2%), and FGD materials comprised approximately 40.4 million tons (or 30%). The survey includes an –other" category to primarily account for the production of approximately 12.5 million tons of fluidized bed combustion ash (9%).

purpose of evaluating the wastes under the Agency's listing criteria. Because EPA inappropriately viewed the four distinct types of CCRs as a single class of wastes, the Agency failed to evaluate each individual CCR under the ten listing criteria as required by the plain language of the regulations. As a result, EPA cannot proceed with listing any one of the four CCRs under the Subtitle C regulations. To do so would be inconsistent with the plain language of the regulations and arbitrary and capricious.

## B. <u>EPA's "Plausible Mismanagement" Scenario Does Not Reflect</u> <u>Utility Practices</u>

Another error in the proposed listing of CCRs is the Agency's selection of an unrealistic scenario for its assessment of the —plausible type of improper management to which the waste could be subjected." 40 C.F.R. § 261.11(a)(3)(vii). This criterion is fundamental to any listing decision.

As the *Dithiocarbamate* court emphasized, the —plausible mismanagement" criterion is directly relevant to the other listing criteria, including —the potential of the constituent . . . to migrate into the environment" (§ 261.11(a)(3)(iii)) and the —nature and severity of the human health and environmental damage . . . as a result of the improper management of the waste" (§ 261.11(3)(ix)). 98 F.3d at 1440. Indeed, —the very question that the ten [listing] factors . . . are supposed to help answer — the hazard posed by the substance — is explicitly phrased in terms of improper management." *Id*. Consistent with this point, the D.C. Circuit has —insisted that the agency provide at least some factual support for a conclusion that a particular mismanagement scenario is plausible" before concluding that the materials warrant hazardous waste regulation. *Id*., citing *Edison Elec. Inst. v. EPA*, 2 F.3d 438, 446 (D.C. Cir. 1993) (—EEI v. EPA"). Here,

the Agency cannot credibly argue that the mismanagement scenario on which its proposed listing of CCRs is based is plausible.

When discussing the -plausible mismanagement" scenario for CCRs, EPA explains that

[a]s shown in the risk assessment and damage cases, the disposal of CCRs into *unlined* landfills and surface impoundments is likely to pose significant risks to human health and the environment. Additionally, documented damage cases have helped to confirm the actuality and magnitude of risks posed by these *unlined* disposal units.

75 Fed. Reg. at 35172 (emphasis added). Thus, a central element of EPA's mismanagement scenario for CCRs involves the management of CCRs in unlined landfills and surface impoundments. The —plausible mismanagement" scenario also assumes that the constituents will migrate undetected from these unlined units for a prolonged period of time, eventually reaching receptors, with —the magnitude of risk to those receptors" consistent with those predicted in EPA's risk assessment." *Id.* at 35170. Thus, the key features of EPA's plausible mismanagement scenario for CCRs are the management of CCRs in unlined units, the long-term, undetected migration of constituents from these units, and their eventual migration to downgradient receptors at concentrations that pose a risk to human health and the environment (including the assumption that the receptors are always located downgradient of the CCR management unit). *Id.* at 35169-73.

This scenario is *not* plausible as virtually no unlined CCR management units have been constructed since at least 1994 and none would be constructed under a Subtitle D non-hazardous waste regulatory regime. First, as EPA itself explains, as of 2004, the Agency estimated that -69% of the CCR landfills and 38% of the CCR surface

impoundments had liners." Id. at 35172. However, these data are based on older units. As EPA stated in its final 2000 Regulatory Determination, -{b}]etween 1985 and 1995, 75 percent of new landfills and 60 percent of new surface impoundments within the utility sector had been lined." 65 Fed. Reg. at 32228. Even more recent data described in the DOE/EPA Report found that, since 1994, -virtually all newly constructed landfills and all newly constructed impoundments are lined, whether as a permit requirement or voluntarily." DOE/EPA Report at 32. Similarly, the percentage of CCR landfills and surface impoundments employing groundwater monitoring has increased substantially over time. The DOE/EPA Report states that the -vast majority (91%)" of the newly constructed or expanded units surveyed in the report had groundwater monitoring, and that there has been an increase in groundwater monitoring for both landfills and surface impoundments since 1994." Id. at 34-35. Given the fact that a large number of CCR disposal units employ both liner systems and groundwater monitoring, EPA cannot provide any factual support for its conclusion" that its mismanagement scenario involving the long-term, undetected migration of CCR constituents from unlined units is -plausible" for a large percentage of CCRs.

More critically, EPA will not be able to make this demonstration for *any* CCR management unit under the Subtitle D option. The substantive Subtitle D standards for CCR landfill and surface impoundment units are virtually identical to those proposed under the Subtitle C option, and include siting, design, liner, groundwater monitoring and corrective action requirements. *See* 75 Fed. Reg. at 35240 (proposed 40 C.F.R. §§ 257.40-.99). As a result of these requirements, all new units and a significant majority of existing CCR management units will have to be retrofitted with liner systems

or closed. In addition, all CCR units will have to install groundwater monitoring systems to detect off-site migration and respond appropriately with corrective action.

In other words, situations involving past management practices of CCR units where there was undetected migration of constituents from unlined CCR management units are not realistic —plausible" management scenarios under the Subtitle D option. As a result, application of a —plausible mismanagement" scenario based on these past practices is arbitrary and capricious and is a fatal flaw of EPA's proposed CCR listing. See Dithiocarbamate, 98 F.3d at 1404 (vacating a listing for a waste where past management practices that served as the basis for the —plausible mismanagement" scenario were —unlikely to be repeated" going forward).

#### C. <u>EPA Has Not Properly Evaluated Regulatory Actions of Other</u> Governmental Agencies

The tenth listing criterion directs EPA to examine the -{a}ction taken by other governmental agencies or regulatory programs based on the health or environmental hazard posed by the waste or waste constituents." 40 C.F.R. § 261.11(a)(3)(x). As the Court in *Dithiocarbamate* recognized, this criterion is -interlocked" with the plausible mismanagement criterion described above and failure to -to give serious consideration to the \_softer' variables of § 261.11(a)(3)", including this criterion, -tends to turn [EPA's] application of that section into an exercise in totally standardless discretion." 98 F.3d at 1402.

Here, EPA has failed to give *any* meaningful consideration to the actions taken by other governmental agencies or regulatory programs for regulating CCRs, including most notably state regulatory programs. Indeed, EPA does not reference a *single* state regulatory program already in place for CCRs. Instead, EPA's analysis of this listing

criterion involves only a single paragraph stating in general terms that, due to the mismanagement of CCRs, —EPA and the states have taken steps to compel cleanup in several situations." 75 Fed. Reg. at 35173. EPA then simply refers to the fact that some CCR sites have been placed on the National Priorities List (—NPL") and that certain states have issued administrative orders requiring the remediation of CCR sites. *Id*.

As explained above, EPA must give serious consideration to every one of the listing criterion in § 261.11(a)(3); failure to do so renders a listing decision arbitrary and capricious. EPA's utter failure to reference any federal or state regulatory controls already in place for CCRs renders EPA's proposed listing for CCRs legally defective. It is not the responsibility of USWAG or the regulated community to bring to EPA's attention the spectrum of existing regulatory controls for CCRs; EPA bears the legal burden of evaluating existing relevant regulatory programs. See 40 C.F.R. § 261.11(a)(3)(x). Nonetheless, even a cursory review of the rulemaking record highlights the legal flaws in EPA's listing proposal.

For example, EPA itself documented in the final 2000 Regulatory Determination the growing effectiveness of state regulatory programs for CCRs, concluding that —with the exception of relatively few states, the regulatory infrastructure is generally in place at the state level to ensure adequate management of these wastes." 65 Fed. Reg. at 32217. The subsequent DOE/EPA Report further documented the growing effectiveness of state regulatory programs for CCRs, concluding that —[i]n eight areas of regulatory control reviewed for this report, more [CCRs] destined for landfills in the States reviewed had tightened regulatory controls than had relaxed controls between

the times data were collected for the 1988 [Report to Congress] and for this report."

DOE/EPA Report at 57. Even more striking is EPA's express acknowledgement in the instant proposal that CCRs are already regulated under state solid waste management programs and that surface water discharges from CCR units may be regulated under the Clean Water Act's National Pollutant Discharge Elimination System regulations. 75 Fed. Reg. at 35142. Indeed, EPA discusses in great detail the data suggesting that state regulatory controls for CCRs have generally improved, though this is one of the key areas on which EPA seeks comment. *Id.* at 35151-52. The point here is not to respond to this issue for comment, but to highlight the absence of any analysis by EPA of existing state CCR controls that is required by this listing criterion.

As the *Dithiocarbamate* court made clear, this criterion is linked to the plausible mismanagement criterion, as EPA cannot credibly evaluate what constitutes a plausible mismanagement scenario for any given waste without first evaluating what regulatory controls are in place for the waste. For example, in its final rule determining that used oil did meet the listing criteria under 40 C.F.R. § 262.11(a)(3), EPA found that an array of other regulatory controls would adequately address—any plausible mismanagement of used oil that is destined for disposal" and that, as a result, listing such materials as a hazardous waste was not necessary to protect human health and the environment. 57 Fed. Reg. 21524, 21528 (May 20, 1992) (the used oil rule survived a subsequent legal challenge; see *Natural Res. Def. Council, Inc. v. NRDC*, 25 F.3d 1063 (D.C. Cir.)).

The plain language of this listing criterion could not be clearer: EPA must evaluate the -{a}ction taken by other governmental agencies or regulatory programs" in response to the health or environmental hazard posed by CCRs. 40 C.F.R.

§ 262.11(a)(3)(x). For purposes of its listing analysis, EPA has not even identified, let alone evaluated, the adequacy of existing state CCR regulatory controls and any other federal controls as required by this listing criterion. For this reason alone, EPA's proposed listing for CCRs is defective.

# D. The Risk Assessment Is Fundamentally Flawed And Cannot Be Relied On To Support the Subtitle C Option, Including Evaluating The Listing Criteria For CCRs

As part of its evaluation of whether to reverse its final 2000 Regulatory

Determination and to list CCRs under Subtitle C, EPA relies heavily on its CCR Risk

Assessment. 75 Fed. Reg. at 35144-46. With respect to the proposed listing, the fate and transport model in the Risk Assessment is relied on extensively to evaluate the potential for constituents to migrate from CCR disposal units under the listing criterion at § 261.11(a)(3)(iii). *Id.* at 35169. In particular, EPA used the model to predict the potential migration of CCR constituents —from different waste types through different exposure pathways, to receptors and to predict whether CCRs under different management scenarios may produce risks to human health and the environment." *Id.*The groundwater risks from CCR disposal units —were based on a groundwater fate and transport model in which constituents leached to groundwater consumed as drinking water, migrated to surface water and bioaccumulated in recreationally caught and consumer fish, and on direct ecological exposure." *Id.* 

The fate and transport model employed in the Risk Assessment contains fundamental errors and overstates the risks from CCR disposal units under all three of the exposure scenarios discussed above. For the reasons discussed in detail in the AECOM Report, the Risk Assessment's fate and transport model does not represent

Duke Energy Progress, LLC Late-Filed Exhibit No. 16 Docket No. E-2, Sub 1219 Page 168 of 439

realistic, real-world risk scenarios from CCR management units. Therefore, the Risk Assessment's modeling results cannot be relied upon by the Agency in evaluating the listing criterion regarding the potential for constituents in CCRs -to migrate from the waste into the environment." 40 C.F.R. § 261.11(a)(3)(iii).

The AECOM Report details the numerous fundamental flaws in the Risk Assessment's construction, including the fate and transport model. Key flaws in the Risk Assessment including the following:

- Inappropriate Inputs. Inappropriate and outdated inputs were used to describe the CCR management scenarios. EPA relied on pre-1995 CCR management data to use as a source term for the model and did not account for updated management practices as expressed in the DOE/EPA report. This latter report demonstrates drastic improvement since 1993 in the use of liners and the employment of groundwater monitoring by owners/operators of CCR management units see Table 1-1 of the Risk Assessment (reproduced in the AECOM Report).
- Insufficient Inputs. The flaws from the use of overly conservative results from the outdated CCR management data are compounded by the paucity of data used to develop the distributional inputs in the probabilistic risk assessment. For example, while there were locational data for 52 unlined surface impoundments where CCRs and coal refuse were co-disposed, there are only 5 leachate/pore water samples upon which the probabilisitic risk assessment is based see Table A of the AECOM report. This is especially problematic because it is this scenario the co-disposal of CCRs and coal refuse in an unlined surface

Duke Energy Progress, LLC Late-Filed Exhibit No. 16 Docket No. E-2, Sub 1219 Page 169 of 439

impoundment – upon which the risk assessment's unrealistically high cancer risk estimate of 2E-02 or 2 in 100 for Arsenic III and Arsenic V is based. It is completely unfounded from a modeling or risk evaluation perspective to base a national regulation on the risk results of a scenario that has input data from only 5 samples of pore water. More fundamentally, it is arbitrary and capricious to purportedly generate scenarios representative of the geologic conditions for the entire United States from a data set consisting of only a handful of units.

- Inappropriate Groupings. The proposed CCR listing inappropriately evaluates all CCRs as one waste stream, and the CCR Risk Assessment evaluates three groups of CCR management scenarios artificially identified for the fate and transport model. This grouping leads to a flawed risk assessment as EPA's approach does not reflect real world management scenarios.
- Unrealistic Mass Estimates. The representation of the source term as the total mass of the constituent in the waste (*i.e.*, the assumption that all constituents are depleted from the landfill) is an unrealistic representation of the leaching process from CCR management units and leads to an overestimate of the calculated risks from these units. Literature and research document that only a fraction of most constituents is available to be leached from CCRs. Therefore, by modeling leaching of these relatively immobile constituents in CCRs throughout the hundreds of years required until the total mass is removed results in unrealistic loading to groundwater and transport of these constituents to unrealistic distances.

Unrealistic 10,000-Year Model Period. The Risk Assessment is predicated on the overly conservative assumption that constituent concentrations in the leachate from CCR units would remain constant throughout the 10,000 year modeling period (or until the constituent mass is depleted from the unit). Use of a 10,000-year simulation period is wholly unrealistic in that it assumes all other conditions within the modeling are essentially unchanged during this period (for example, groundwater hydraulic gradients, locations of surface water bodies, locations of receptors, climate, hydrology, permanence of disposal area, etc.). In reality, it is unreasonable to assume that these conditions will not change over 10,000 years, just as such changes have taken place in the last 10,000 years. The most recent Ice Age ended approximately 15,000 years ago, and entire hydrologic systems have been created and modified since that time. Therefore, modeling over 10,000 years results in predictions that cannot be correct; reliance on such attenuated modeling scenarios is arbitrary and capricious. For example, Table 4-7 of the Risk Assessment provides the time to peak concentration for a set of constituents for landfills and surface impoundments at the 10<sup>th</sup> to the 90<sup>th</sup> percentiles. For all landfill scenarios modeled, the shortest time to peak concentration at the 10<sup>th</sup> percentile is 300 years (boron, unlined). This result clearly does not warrant regulation of landfills under any configuration. For the surface impoundments, the shortest time to peak concentration at the 10<sup>th</sup> percentile is 70 years (Arsenic III, boron, and Selenium VI; unlined). All results at the 90<sup>th</sup> percentile for unlined surface impoundments are greater than 100 years. Therefore, the use of the 10,000-year timeframe in estimating potential risks

Duke Energy Progress, LLC Late-Filed Exhibit No. 16 Docket No. E-2, Sub 1219 Page 171 of 439

centuries into the future is not a credible modeling scenario and cannot serve as the basis for any final regulation.

- Inappropriate Attenuation Modeling. The groundwater modeling of constituent migration does not adequately reflect the attenuation processes that occur in groundwater, especially with regards to the mobility of Arsenic, which is controlled by redox conditions. In the current version of the Risk Assessment, both Arsenic III and Arsenic V are modeled. For Arsenic V to be present and stable, the groundwater must be aerobic. Under these aerobic conditions, however, Arsenic is relatively insoluble and would not be mobile in groundwater. As Arsenic is the risk-driver in the human health risk assessment results, this is a serious flaw in the analysis, resulting in overstating the risks associated with the migration of constituents from CCR disposal units.
- Lack of Model Validation. While the models used in the fate and transport analysis were peer-reviewed when they were developed (*i.e.*, a before-the-fact review of the underlying principles and mathematics), there has been no field verification of the model-predicted concentrations. That is, even if the underlying principles and mathematics were considered to be appropriate, there has been no evaluation of the model accuracy in this particular case, specifically, how the scenario-specific inputs may or may not result in realistic or believable outcomes. Use of real world, observed concentrations of the various constituents in groundwater in the vicinity of CCR units is a critical step in evaluating the reliability and confidence in the Risk Assessment's modeling predictions. The critical step of model validation is well-documented in the scientific literature and

recognized and endorsed by EPA (*see e.g.*, Resolution on the Use of Mathematical Models by EPA for Regulatory Assessment and Decision-Making," at 3 (—There is a need for models used in regulatory application to be confirmed with laboratory and field data."). Groundwater data are available for waste management units from a variety of sources, including EPRI, and a comparison of these real, observed field data to the interim predictions by the model is a critical step in validating the model predictions. This is missing in the Risk Assessment.

• Unrealistic Receptor Location and Frequency. For the drinking water scenario, the model evaluates a downgradient drinking water receptor in each and every one of the 10,000 model runs. This assumption does not account for the real-world scenarios of potential receptors being cross-gradient or upgradient of the waste management unit and grossly overestimates the probability of an exposure occurring. The model also assumes that impacted groundwater travels to every downgradient receptor location and is available for exposure; this assumption does not take into account cases where there are no downgradient receptor locations, or where receptors (that is, residents) in the downgradient locations are serviced by municipal water (thus, do not have drinking water wells), or cases where intervening water bodies intercept the further downgradient transport of constituents. By not accounting for any of these real-world scenarios, the Risk Assessment grossly overestimates the level of potential exposure via the drinking water pathway and presents an unrealistic mismanagement scenario.

Duke Energy Progress, LLC Late-Filed Exhibit No. 16 Docket No. E-2, Sub 1219 Page 173 of 439

- Incorrect Aquatic Benchmark for Boron. EPA has continued to use an incorrect aquatic benchmark for boron, such that the predicted ecological risks for boron are over-estimated by three orders of magnitude. When the correct benchmark is applied, the boron results are below levels of concern.
- Inappropriate Fish Ingestion Evaluation. EPA points to the risks for Arsenic III and Arsenic V from the ingestion of recreationally caught fish scenario as additional evidence of risks due to CCR mismanagement in surface impoundments. However, EPA has not factored into these risk estimates the fact that arsenic in organic tissues is predominantly present in a non-toxic organic form. Less than 1% of arsenic in fish tissue has been shown to be in the potentially toxic inorganic form. If EPA had accounted for this in the CCW Risk Assessment, the risks for this pathway would be below concern.
- Unfounded Use of Risk Attenuation Factor. EPA used a screening analysis to identify constituents for the full-scale risk model, however, EPA did not carry through all of the identified constituents into the full-scale model, which is not standard practice. Instead, EPA accounted for these constituents at the end of the modeling by using a contrived -risk attenuation factor," which is the ratio of the full-scale risk results to the screening level risk results for the constituents used in the full-scale model. This approach ignores the unique fate and transport properties of these omitted constituents, such that the scaled risks are not correct.
- Lack of Population Context in Rulemaking. Even accepting the fundamental flaws in EPA's risk assessment that lead to gross overestimates of potential risk

to human health, evaluating the hypothetical impact of these overly conservative risks on the U.S. population is instructive. Based on an EPRI evaluation of the potential for the occurrence of receptors downgradient of a CCR management unit, and information on the U.S population (which is currently 307,006,550), it is estimated that there could be hypothetically up to 7,770 users of downgradient groundwater as drinking water for the 508 CCR management facilities evaluated by EPA in the RIA. Further, assuming a worst-case scenario that all management units are unlined surface impoundments that contain CCRs codisposed with coal refuse (which is not the case), the predicted excess lifetime cancer risk of 2x10<sup>-2</sup> or 2 in 100 results in a hypothetical 155 lifetime cancer cases out of the total U.S. population of 307,006,550 (to help put this into perspective, the measured background cancer incidence in the U.S. population is between 1 in 2 to 1 in 3). However, because co-disposal of CCRs with coal refuse in unlined surface impoundments account for only a fraction of the CCR management scenarios in the U.S., and as the assumptions used in the Risk Assessment result in gross over-estimates of risk, the hypothetical number of cancer cases that could result from the current management practices for CCRs would be *much* less than the hypothetical 155 cases modeled under the overly conservatives assumptions in the Risk Assessment. For example, assuming that the entire receptor population was downgradient of an unlined CCR surface impoundment that did involve co-disposal with coal refuse, the number of hypothetical, estimated lifetime cancer cases would be 15, based on a cancer risk estimate of 2x10<sup>-3</sup> for Arsenic III. Similarly, the worst-case landfill risk

estimate is 5E-04 for Arsenic III in an unlined co-disposed CCR and coal refuse landfill; the number of hypothetical estimated lifetime cancer cases if all receptors are assumed to be downgradient of this type of waste management unit is 4 out of the total U.S. population of 307,006,550. Again, even these numbers of hypothetical estimated lifetime cancer cases are based on the results of an extremely flawed risk assessment.

## E. <u>EPA's Draft Screening Assessment Of Fugitive Dust Risks Does</u> Not Support the Proposed Listing Of CCRs

As part of its listing evaluation, EPA theorizes that air emissions from CCR disposal and storage facilities -may cause adverse human health effects" due to the inhalation of small-diameter -respirable" particulate matters from these sources. 75 Fed. Reg. at 35171. EPA did not provide data to support this position. Instead, EPA performed a screening-level analysis to evaluate the potential exposure to particulate matter by residents who live near CCR landfills. To do this, EPA used the SCREEN3 model and produced a draft report entitled *Inhalation of Fugitive Dust: A Screening Assessment of the Risks Posed by Coal Combustion Waste Landfills* – DRAFT (U.S. EPA 2010b). *Id.* EPA requests comment on the draft screening analysis and the results of any real-world data that may be available regarding the potential for residents to be exposed to CCR constituents by the inhalation pathway. *Id.* 

As discussed below, the draft screening model has significant flaws that preclude it from being used to support the CCR listing proposal. In addition, real-world air monitoring data from the remedial activities associated with the TVA ash release make clear that the potential inhalation exposure pathway does not present the level of risk sufficient to support a hazardous waste listing.

The attached AECOM technical report also includes a critical analysis of EPA's draft SCREEN3 assessment. The AECOM report makes clear that EPA's draft screening assessment uses a series of compounding conservative assumptions that results in a modeled concentration for 24-hour PM<sub>10</sub> of 13,390 µg/m<sup>3</sup> in comparison to the 24-hour PM<sub>10</sub> NAAQS of 150 µg/m<sup>3</sup>. To illustrate the overly conservative nature of EPA's draft model, the modeled result of 13,390 μg/m<sup>3</sup>, PM<sub>10</sub> concentrations is similar to concentrations observed in the vicinity of a volcano after eruption. AECOM Report at Appendix 20. Some of the unrealistic, conservative elements of the draft screening assessment are EPA's (1) use of the wrong emission factors, (2) use of overly conservative parameter selections, and (3) failure to take into account the real-world moisture content of CCRs. Given that the emissions estimator that EPA is using cannot take into account the moisture content of the CCRs (or any material for that matter), when other more appropriate, yet still conservative, refinements are made to the draft screening assessment, AECOM reports that many of the modeled concentrations are at or below the NAAQS for 24-hour  $PM_{10}$  (150  $\mu g/m^3$ ).

Indeed, real-world data underscore the overly conservative nature of EPA's draft assessment. A comprehensive public health assessment of the TVA coal ash release was conducted by the Tennessee Department of Health, under a cooperative agreement with the U.S. Department of Health, Agency for Toxic Substances and Disease Registry. See Public Health Assessment Final Release: Tennessee Valley Authority (TVA) Kingston Fossil Plant, Tennessee Department of Health (Sept. 7, 2010) (available online at http://health.state.tn.us/Environmental/PDFs/pha-e-TVA\_Kingston\_Fossil\_Plant\_Final.pdf (—TVA Health Assessment"). Thousands of air

measurements have been collected by TVA, EPA and the Tennessee Department of Environment and Conservation (¬TDEC") since December 2008. See id. at 45-53.

Among other things, the final assessment found that, with respect to risk from fugitive emissions associated with the release, -sampling and analysis of particulate matter by all agencies indicated that particulate matter, less than or equal to 2.5 microns in diameter (PM2.5) and less than or equal to 10 microns in diameter (PM10), in ambient air surrounding the coal ash release met all National Ambient Air Quality Standards."

Id. at xxxi. In addition, measurements of metals concentrations in air have -eonsistently been within background levels of metals in the U.S. or below any health comparison values." Id. at 64.

The real-world data from the TVA Kingston site compared to the overly conservative estimates modeled in EPA's draft screening assessment makes clear that EPA has not established a record supporting the listing of CCRs based on risks from the inhalation exposure pathway.

## VII. EPA'S REGULATORY IMPACT ANALYSIS CONTAINS FUNDAMENTAL METHODOLOGICAL FLAWS THAT UNDERMINE ITS MOST BASIC CONCLUSIONS

To support the CCR proposal, EPA conducted a Regulatory Impact Analysis (RIA) of the various regulatory options being considered in the rulemaking. Regulatory Impact Analysis For EPA's Proposed RCRA Regulation of Coal Combustion Residues (CCR) Generated by the Electric Utility Industry, Environmental Protection Agency (April 30, 2010). EPA's analysis examined the costs and benefits for each of the regulatory options related to four principle areas: (i) the benefits of avoided future groundwater contamination, including the remediation costs avoided and the value of the avoided

cancer risks; (ii) avoided cleanup costs of future surface impoundment structural failures; (iii) regulatory costs of compliance; and (iv) economic and environmental benefits from future increases in CCR beneficial use by other industries. EPA uses these data to demonstrate that if, as the Agency maintains, beneficial use of CCR increases due to the significantly higher disposal costs of CCRs under a Subtitle C program, the Subtitle C option will result in the greatest net benefits, \$66.8 to \$81.8 billion at a 7% discount rate over a 50-year period. 75 Fed. Reg. at 35215-16.

USWAG engaged EOP Group, Inc. to review the RIA to assess whether its conclusions with respect to the overall costs/benefits of the proposed regulatory options are accurate. EOP RIA Report, (Nov. 2010), attached as Appendix 25. While there are numerous problems with the RIA, we summarize here EOP's main findings that render the RIA invalid, namely EPA's failure to properly value the increase of CCRs for beneficial use and accurately assess the reduction of cancer incidents resulting from the three alternative regulatory options. We discuss below why these fundamental errors impact the RIA's conclusions.

## A. <u>EPA Failed To Properly Value The Impacts Of The Proposed</u> Regulatory Options On CCR Beneficial Use

The primary economic component of the RIA's conclusions regarding the costs and benefits of each of the three regulatory options are the benefits associated with increased beneficial use of CCRs. In the analysis of the Subtitle C option, the benefit associated with future increase in CCR beneficial use is \$84.5 billion, while the total costs of the entire regulatory scheme are \$20.3 billion and all other benefits total \$17.6 billion. The increased CCR beneficial use benefits under the other regulatory options are a similar percentage of the total cost/benefit of the options, although the expected

value is significantly lower. We address above why USWAG echoes the concerns of CCR beneficial users and marketers that subjecting CCRs to Subtitle C regulation will dramatically reduce beneficial use. Notwithstanding the potential stigmatizing effects associated with labeling CCRs as a Subtitle C waste (even if only in the context of disposal), there are other clear errors in EPA's method for valuing the assumed increase in CCR beneficial use.

The first critical error is EPA's use of the wrong baseline for assessing the costs and benefits of its proposed regulatory options. As detailed in EOP's report, the Agency includes the avoided costs of disposal as a significant benefit that will accrue if the beneficial use of CCRs increases. That is, the costs of disposing of CCRs under the proposed Subtitle C regime are considered a baseline cost, which is in essence recovered when CCRs are beneficially used. Assessing the costs and benefits of a proposed regulatory approach by including the worst-case regulatory costs of compliance as a benefit fundamentally fails to assess the actual costs and benefits of the regulatory options. The purpose of a regulatory impact assessment is to evaluate the costs and benefits of proposed regulatory options by comparison with a baseline *pre-rule* state. This assessment should consider the costs and benefits of imposing each of the regulatory options on the status quo (*i.e.*, the present state of affairs) and not assume that management options not otherwise selected can yield a benefit.

The EOP Report highlights perverse regulatory outcomes that arise when avoided costs are considered part of the baseline as EPA has done in the RIA. These include the possibility of generating even higher expected benefits by dramatically increasing the costs of compliance with an alternative. EPA could make the costs of

disposing of CCRs absurdly high (EOP suggests mixing CCRs with rose petals and diamond dust in landfills lined with \$100 bills) which would yield an expected benefit to beneficial reuse that would reflect these absurdly high costs. This would allow EPA to find a net benefit for any regulatory proposal as long as the costs of compliance with an alternative were sufficiently high. In fact, the total benefits to any potential CCR proposal would be directly proportional to the costs that the Agency could impose on CCR disposal. All else being equal, the cheaper option to comply with would always have the least net benefits, turning regulatory decision-making on its head. EPA's RIA conclusions directly reflect this approach. This is a patently absurd way of assessing the benefits associated with the potential increase in CCR beneficial use.

Another fundamental error in EPA's valuation of the benefits associated with increased beneficial use in the RIA is the Agency's faulty assumption that there can be, in the first instance, any net benefit associated with increases in the beneficial use of CCRs. As EOP explains, one must assume that there currently exists equilibrium in the marketplace for CCRs (*i.e.*, market demand meets market supply at the optimum price). Currently, utility generators of CCRs have every incentive to subsidize the use of CCRs up to their costs of disposal, and CCR marketers and users are willing to pay for or accept a subsidy for these materials up to the price/benefit associated with alternative materials. Since the market is in equilibrium, a socially optimal level of CCRs is being beneficially used.

For purposes of this rulemaking, EPA assumes that CCR beneficial use will increase due to the increased costs of disposal of CCRs. While we disagree with this assumption, even if we assume it to be correct it does not logically follow that this

increased use constitutes a societal benefit. EPA believes that increasing the cost of disposal will increase the subsidy (e.g., payment to a marketer or including transportation costs in a transaction) utilities are willing to pay ash users/marketers to accept CCRs for beneficial use. This would change the supply of CCRs in the marketplace by making more materials available at a cheaper price because utilities would pay more for these materials to be used. In the RIA, EPA assumes that this increased beneficial use of CCRs is a benefit of the regulatory options under consideration, when in fact it is merely a transfer of resources from utilities to ash marketers/users. EPA should realize that a transfer (in the form of the subsidy) from one party to another is not a net benefit overall.

Because the marketplace must be assumed to be in equilibrium, any impact that EPA's regulations make to that marketplace will distort incentives and result in a net cost to the CCR market. By increasing the costs of disposal, EPA encourages utilities to subsidize CCR beneficial uses resulting in overuse by marketers/users because they would not have accepted the same volume of CCRs at a lower price/subsidy.<sup>27</sup> If the rulemaking stigmatizes the beneficial use of CCRs (which USWAG believes would happen under a Subtitle C rule), transfers of CCRs that were producing net benefits to both parties would be abandoned, resulting in a net societal cost. The best outcome of this rulemaking on the CCR beneficial use market — setting aside other types of benefits of the rulemaking such as lower incidence of surface impoundment structural

<sup>-</sup>

<sup>&</sup>lt;sup>27</sup> USWAG is not suggesting that EPA should not adopt a federal rule that will increase the costs of disposal for utilities. What we are suggesting, however, is that increasing the costs of disposal will be suboptimal for the beneficial use marketplace. EPA can and has elsewhere in the RIA assessed the benefits to human health of these regulatory options. We simply believe that any regulatory option that increases or decreases the use of CCRs does not provide a net benefit to the beneficial use marketplace for CCRs.

failures or risks to human health—is for there to be no impact on either the supply or demand of CCRs for beneficial use.

# B. <u>EPA's Assessment Of The Benefits Of Avoided Cancer Incidents</u> <u>Contains Serious Errors and Overstates The Risk Reduction from</u> The Subtitle C Alternative

In the RIA, EPA also assesses the benefits of each of the three regulatory alternatives on reducing potential cancer risks associated with drinking groundwater contaminated with arsenic from CCR surface impoundments. The Agency's analysis includes a number of methodological errors that overstate the value of the human cancer cases avoided by the Subtitle C regulatory option.

One critical error included in the analysis relates to EPA's use of data on current state regulatory programs and its determination of expected compliance rates with federal Subtitle D programs. To assess the cost of complying with the regulatory alternatives, the Agency established a baseline of the percentage of facilities that are already performing certain activities (*e.g.*, conducting groundwater monitoring) that will be required by the regulatory options. RIA at 122-124. While EPA's estimates of the costs associated with the regulatory alternatives are reduced by the number of facilities already performing these activities, the Agency does not similarly discount the avoided cancer risks by the number of facilities already in operating in accordance with the proposed federal controls. By making this error, EPA is assuming that imposing federal requirements for activities that are already being conducted will impose no additional costs, but will increase the effectiveness of these activities in reducing cancer risks from 0% to 100% effective. This error either greatly inflates the benefits of the regulatory options or improperly reduces the costs of these options. USWAG maintains that there

are many effective state programs that properly control CCR surface impoundments and landfills. In the RIA, EPA completely disregards the effectiveness of these programs in reducing cancer risks.

Additionally, EPA's assessment of the effectiveness of the Subtitle D and Subtitle D Prime options is based directly on the percentage of states that impose groundwater monitoring requirements on new units and on existing units respectively. EPA uses this groundwater monitoring rate as a proxy for determining whether states will adopt federal Subtitle D requirements for CCR management units. EPA assumes that the Subtitle D program will only be adopted in those states that impose groundwater monitoring on new surface impoundments. Therefore, because only 48% of CCR is disposed of in states that already have such requirements, the Subtitle D regulatory option will only be 48% as effective as the Subtitle C option (which is assumed to be 100% due to the direct federal enforceability that this option includes). The same methodology is used for the Subtitle D Prime option where a 12% rate of compliance is imputed due to the rate of groundwater monitoring requirements for existing CCR surface impoundments. Based on these proxies, EPA's analysis suggests that the Subtitle D regulatory options will *only* be effective in those states that *already* impose groundwater monitoring on CCR surface impoundments.

By establishing this methodology, EPA is assuming that states will only adopt a federal rule if they already have similar controls and that no additional states (those without groundwater monitoring requirements for new or existing surface impoundments) will adopt the rule. As described above, however, even in those states that already have such requirements, the groundwater monitoring controls are assumed

to be completely ineffective at reducing cancer risks, but will be completely effective when the federal requirements are adopted in these states. Even more critically, the Agency assumes that no utility disposal facilities will comply with the Subtitle D rules unless a state regulatory program directly adopts the federal rules, notwithstanding the self-implementing nature of the regulations and the threat of enforcement by citizens and/or states (even if the rules are not adopted by the states). EPA also assumes that a Subtitle C alternative will provide greater reduction in cancer risks than a Subtitle D rule because all states will adopt these requirements and they will be 100% effective from the existing baseline controls that are 0% effective. This analysis completely fails to assess what likely improvements these regulatory options will provide, does not address the effectiveness of existing requirements, and does not assess other regulatory controls (e.g., liners, siting, construction standards) that already exist at the state level that reduce cancer risks.

Another critical error in this section of the RIA is that EPA relies on outdated information on the controls at CCR disposal units and does not take into consideration the more protective controls that have been installed at recently constructed units. See, e.g., DOE/EPA Report. Failure to use newer information understates the presence of better controls at facilities and includes this already occurring baseline trend as an expected benefit of the proposed regulatory options.

The assessment of the avoided cancer benefits also does not properly include the appropriate timeframes for discounting the value associated with reducing cancer risks. The RIA does not account for the amount of time arsenic takes to reach a drinking water well and does not discount for the time between exposure to the

contamination and the onset of cancer. Both of these errors overstate the present value of the avoided cancer risks.

As we address below, EPA includes in this rulemaking (including in the RIA's analysis of the benefits of reduced cancer risks), a number of impoundments that were included as CCR surface impoundments based on the Agency's recent survey of these units. As we described below, however, EPA defined units for the survey as CCR impoundments if they contained even *de minimis* amounts of CCRs. This inappropriately included wastewater ponds that contained little (if any) CCRs into the total number of surveyed and reported units. In the RIA, EPA uses information on the number and size of surface impoundments in 1995 to extrapolate to the much higher number of surface impoundments reported in the recent survey. RIA at 41. The Agency assumed that the large number of recently reported impoundments were the same size as those reported in 1995 which were reported based on a more precise definition of surface impoundment. *Id.* This extrapolation seriously overestimates the size of existing surface impoundments as well as the concentrations and risks associated with managed CCRs in these impoundments.

In addition to the issues identified above, regarding the errors in EPA's methodology for evaluating impacts to CCR beneficial use and reductions in cancer risk, EOP's report evaluates other errors in the RIA including (i) the double counting of avoided groundwater remediation costs and cancer risks, (ii) the failure to address alternative approaches to avoiding structural impoundment collapses and (iii) the Agency's lack of justification for its proposal and failure to properly consider regulatory alternatives in accordance with OMB guidance and Executive Order Number 12866.

Even a cursory review of the principal components of the RIA yields fundamental methodological errors that undermine even its most basic conclusions. EPA's reliance on these unfounded conclusions in this rulemaking will constitute arbitrary and capricious agency action.

### VIII. THE SUBTITLE C OPTION PRESENTS INSURMOUNTABLE COMPLIANCE AND DISPOSAL OBSTACLES FOR ELECTRIC UTILITIES

Not only does the Subtitle C option present serious legal hurdles, it would also impose serious and often intractable compliance problems for electric utilities given the physical nature of CCRs and the volume of CCRs generated by the industry in the course of producing electricity. Also, EPA is wrong in its position that, under the Subtitle C option, -disposal patterns will remain generally the same" and that Subtitle C commercial disposal capacity will not be overwhelmed by introducing at least 76 million tons of CCRs (the approximate amount of CCRs currently disposed of on an annual basis) per year to this disposal market. 75 Fed. Reg. at 35158. As discussed below, regulation of CCRs under Subtitle C would create significant disposal capacity shortfalls for electric utilities as well as other generators of non-CCR hazardous wastes.

### A. <u>Compliance Concerns</u>

One of the fundamental reasons EPA cited for deciding not to regulate CCRs in its final Regulatory Determination for the four large-volume CCRs was the recognition that the inflexible, one-size-fits-all nature of the Subtitle C hazardous waste program simply was not appropriate for the diverse nature of CCR disposal facilities. As EPA explained:

if such wastes were placed in the Subtitle C universe, all ash disposal units would be required to meet specific liner and monitoring requirements. Since [CCR] sites vary widely in terms of topographical,

geological, climatological, and hydrological characteristics (e.g., depth to groundwater, annual rainfall, distance to drinking water sources, soil type) and the wastes potential to leach into the groundwater and travel to exposure points is linked to such factors, it is more appropriate for individual States to have the flexibility necessary to tailor specific controls to the site or region specific risks posed by these wastes.

58 Fed. Reg. at 41477.

This conclusion remains true today; the Subtitle C hazardous waste program remains inappropriate for CCRs as this option will present serious practical and compliance concerns for coal-fired power plant operations. Most significantly, due to the sheer volume and the physical composition of CCRs, de minimis volumes of CCRs are inevitably released during normal power generation and subsequent CCR handling operations (e.g., fugitive and de minimis emissions from ash conveyor equipment or loading equipment and during the transport/handling of CCRs for beneficial use). This holds true notwithstanding utilization of even the most sophisticated CCR handling, conveyance, and storage equipment; de minimis releases of these materials are impossible to prevent. Nonetheless, if CCRs are regulated as a listed -special waste" subject to full hazardous waste regulation, even these de minimis releases would constitute improper hazardous waste disposal, subjecting electric utility coal-fired power plants to liability for what would likely be a perpetual state of RCRA non-compliance. This is because RCRA Subtitle C contains no de minimis exemption for listed wastes; any material, no matter how small or inconsequential, derived from a listed hazardous waste remains subject to full Subtitle C regulation. See 40 C.F.R. § 261.3(c)(2).

Clearly, the prospect of Subtitle C liability for any *de minimis* release or spill of CCRs constituting improper hazardous waste disposal is a significant compliance concern, even though there is very little risk associated with such *de minimis* releases.

Electric utilities strive for full compliance with all applicable environmental laws and regulations, but the Subtitle C option would make this objective virtually impossible to achieve. Moreover, the fact that such inevitable releases would constitute improper disposal would expose facilities to continual and harassing citizen suits based solely on these *de minimis* and inconsequential releases. This illogical result, flowing from overly stringent and unnecessary regulation of precisely the type that Congress directed EPA to avoid when issuing any final CCR regulations under RCRA, would be both an unjustifiable burden on the regulated community and wasteful of agency and judicial resources. *See* 126 Cong. Rec. at 3361.

CEQ's comments on the draft Subtitle C option underscore the seriousness of this issue, as does a letter written to EPA Administrator Lisa Jackson last spring by members of the House Energy and Commerce Committee. Interagency Working Comments on Draft Rule under EO 12866 at 17-18; Letter from Reps. Upton, Boucher, Space and Rogers to EPA Administrator Lisa Jackson (April 27, 2010) (-Upton Letter") (attached as Appendix 26). The Representatives writing the letter asked EPA whether, if CCRs were subject to Subtitle C hazardous waste regulation, the inevitable release of *de minimis* amounts of CCRs during normal power generation activities would -eonstitute improper disposal of a hazardous waste." *Id*. Understandably, these members of Congress shared the concern that, given the language and purpose of the Bevill Amendment, the inflexible nature of the Subtitle C hazardous waste program would thrust hundreds of facilities into a regulatory scenario where full compliance with the law is unachievable. Remarkably, EPA failed to provide any credible response to this legitimate concern.

In its reply to the Upton Letter, the Agency confirmed that such *de minimis* releases would be subject to full hazardous waste regulation, stating -the management of this special waste will generally be addressed in the same way that management of hazardous waste is addressed at any other production facility that generates hazardous waste." EPA Response to Upton Letter (July 15, 2010) (attached as Appendix 27). Significantly, EPA stated that the -term \_de minimis' is not used in the [hazardous waste] regulations except in one provision applicable to commercial chemical products or intermediates," which is not relevant to the regulation of CCRs under the Subtitle C hazardous waste program. *Id.* In conclusion, EPA stated simply that such *de minimis* releases would be addressed through the Subtitle C hazardous waste regulations that -address responses to unplanned releases, such as accidental spills or leaks." *Id.* 

Having confirmed that the inevitable *de minimis* releases of CCRs during normal power plant operations would be subject to full hazardous waste regulation, and understanding that such releases would therefore constitute improper disposal of CCRs, EPA apparently also recognizes that the Subtitle C option would thrust utilities into a regulatory regime where full compliance is impossible *See* 40 C.F.R. § 260.10 defining -disposal" under the Subtitle C regulations to include, among other things, the -spilling, leaking or placing of any solid waste or hazardous waste onto or on any land." Though this eventuality would seem to be of significant concern to regulators and utilities alike, EPA provides no thoughtful discussion of or practical solution to the problem.

This foreseeable consequence of the Subtitle C option is directly contrary to the Bevill Amendment's directive that EPA avoid imposition of the Subtitle C regulations on CCRs to the greatest extent possible so as not to impose undue regulatory burdens on

utilities and discourage the use of coal. See 126 Cong. Rec. at 3361. Further, such an impractical regulatory program constitutes unreasoned agency decision-making; it is a fundamental principle of administrative law that it is arbitrary and capricious to establish a regulatory program that makes full compliance impossible. See Alliance for Cannabis Therapeutics v. DEA, 930 F.2d 936, 940 (D.C. Cir. 1991) (impossible requirements imposed by the Agency are per force unreasonable, citing D.C. Transit Sys., Inc. v. Wash. Metro. Area Transit Comm., 466 F.2d 394, 402 (D.C. Cir. 1972)).

On a related note, USWAG does not support EPAs effort to set the CERCLA hazardous substance reportable quantity (RQ) for CCRs based on the RQ of the most toxic substance present in the waste (i.e., arsenic and mercury RQs of 1 pound). 75 Fed. Reg. at 35183-84. This standard would require the reporting to the National Response Center and local emergency response organizations for releases of 1 pound of CCRs, even though the maximum concentration of arsenic in the waste is 773 parts per million (ppm) and the maximum concentration of mercury is 384 ppm. *Id.* at 35185. Establishing an RQ for CCRs that is based on the thresholds for arsenic and mercury even though these substances may be present at exceedingly low levels would trigger reporting for releases that contain only minimal levels of hazardous constituents. This would divert utility personnel as well as federal and local regulators and emergency response personnel to these spills and away from other, potentially more harmful, releases. Instead, USWAG supports EPA's alternative proposal whereby the Agency would set the RQ for CCRs based on the maximum observed concentration (or EPA identified maximum concentrations) of each hazardous substance constituent such that CCRs would only trigger its RQ when a constituent triggers its individual RQ. Id. at

35185. Based on EPA's data, this would mean that CCRs would not trigger an RQ threshold until 1,294 pounds had been released based on the maximum concentration of arsenic in CCRs.

# B. <u>The Subtitle C Option Will Result In Serious Disposal Capacity</u> <u>Shortfalls</u>

As USWAG and others have repeatedly warned EPA throughout this rulemaking process, a combination of factors associated with the Subtitle C regulation of CCRs will create a serious shortfall of on-site CCR disposal capacity, requiring more CCRs to be shipped off-site for disposal in commercial Subtitle C facilities. If only a fraction of the estimated 76 million tons of CCRs disposed of annually (see 75 Fed. Reg. at 35158) is diverted to the commercial hazardous waste disposal market, the commercial market will be overwhelmed, leaving CCR generators as well as thousands of other generators of non-CCR hazardous waste without any legally viable disposal options for their wastes.

As EPA acknowledges, only two million tons of hazardous waste are disposed of annually in hazardous waste landfills; the total current national commercial hazardous waste disposal capacity is between 23.5 and 30.3 million tons. *Id.* However, based on a survey of electric utilities evaluating how the industry would dispose of CCRs under the Subtitle C option, the EPRI Cost Report found that at least 12% of CCRs would be diverted to the commercial Subtitle C hazardous waste disposal market. EPRI Cost Report at 3-6. This correlates to approximately 15 million to 21 million of CCRs annually having to be disposed of in a Subtitle C commercial landfill, in comparison to the approximately two million tons of hazardous waste currently disposed of in commercial hazardous waste landfills. *Id.* at 4-5. This would be approximately an order

of magnitude increase, on an annual basis, of the amount of wastes having to be disposed of in the Subtitle C commercial market and would, within two years, overwhelm the overall nationwide Subtitle C landfill capacity of 23.5 to 30.3 million tons. *Id.* 

As discussed below, the states have already raised these concerns, warning EPA that they will not have the ability to permit the necessary additional hazardous waste disposal capacity in a timely fashion. These findings disprove EPA's assumption that owners and operators of CCR landfills will be willing to or be capable of converting all their existing CCR disposal units to permitted Subtitle C disposal units and that there will be no shift in -disposal patterns in a way that substantially increases the disposal of CCRs off-site from generating facilities to commercial hazardous waste landfills." *Id.* at 35158.

The results from the EPRI survey should not be surprising as a multitude of factors make clear that subjecting CCRs to Subtitle C regulation will quickly overwhelm Subtitle C disposal capacity. First, the Subtitle C option effectively mandates the phase-out of CCR surface impoundments, meaning that approximately thirty million tons of CCRs currently managed annually in surface impoundments will have to be converted to management in landfills. In turn, utilities must either obtain permits for this additional landfill capacity or divert the additional volumes of CCRs to the commercial market. A significant percentage of existing CCR landfills, including commercial RCRA Subtitle D landfills that receive CCRs today, will not be able to obtain Subtitle C permits or will decline to attempt to do so for a variety of reasons, including the length and uncertainty of the Subtitle C permitting process and the intense public opposition to the citing of any

new hazardous waste landfills, especially for CCRs. See EPRI Cost Report at 3-6, 4-5 and 4-6. This strong opposition to the siting of any new CCR landfills by activist groups has already been well-documented during the EPA-sponsored public hearings on this proposal.

Additionally, some states, including Florida and Kansas, prohibit the siting of hazardous waste landfills, which would foreclose any opportunity for the continued onsite or nearby disposal of CCRs. See Letter from Kansas Department of Health and Environment to EPA (Sept. 21, 2009) (attached as Appendix 28), (explaining that Kansas –state law prohibits the land disposal of any RCRA hazardous waste. If [CCR] is declared hazardous, all current permitted disposal activities would become prohibited and these wastes would need to be transported out of state for disposal). Florida has cautioned EPA of the same result in that state. See Letter from Florida Department of Environmental Protection to EPA, (April 27, 2009) (attached as Appendix 29). Therefore, it is simply unrealistic for EPA to assume that all existing CCR landfill disposal capacity, or even a majority of such capacity, could be readily converted to RCRA Subtitle C units.

Further compounding the disposal capacity problem is that certain CCR beneficial uses currently employed in the market will be prohibited, creating additional volumes of CCRs that must be disposed of. For example, EPA takes the position that many large-scale structural fill projects (which would fall within EPA's broad description of large scale fill operations") do not constitute an excluded beneficial use. As a result, the millions of tons of CCRs, which have historically been beneficially used for these fill operations, would be diverted to the Subtitle C disposal market, increasing demand on

the already strained disposal market. EPA may suggest, in response, that CCR beneficial use opportunities will increase under the Subtitle C option, but this theory has already been uniformly rejected by the states, the CCR beneficial use market, and other independent third parties. Despite any additional appeal of CCR beneficial uses that may arise in the context of a severe disposal capacity shortage, numerous and significant hurdles to CCR beneficial use including, but not limited to, the stigmatizing effect Subtitle C regulation will have on CCRs will severely limit the opportunities for beneficial use of CCRs. In short, there is no disputing the fact that promulgation of any Subtitle C option will mean that tens of millions of tons of CCRs would have to be disposed of on an annual basis; there simply will not be enough Subtitle C disposal capacity to absorb these materials.

USWAG is not alone in cautioning EPA about the short-fall in disposal capacity that will arise under the Subtitle C option. Also vocal in their opposition to the Subtitle C approach are the states, the entities that are closest to this issue, as they have the greatest wealth of information regarding existing disposal capacity, and who ultimately would be responsible for permitting the necessary new landfills. ECOS has specifically cautioned EPA that the current Subtitle C commercial disposal capacity will not be able to accommodate the influx of CCRs into the system if CCRs are regulated under Subtitle C. Like USWAG, ECOS pointed out that, if just a fraction of CCRs currently being disposed of in on-site utility landfills are diverted into the Subtitle C commercial disposal market, the commercial market will be quickly overwhelmed.

On this point, ECOS states that -[i]n the unlikely event that beneficial use continues at its current rate *and* half of the coal fired utilities seek Subtitle C permits for

the disposal facilities that they manage, the 2013 capacity [for commercial Subtitle C facilities] will be consumed in less than one year." See Letter from ECOS to Mr. Mathy Stanislaus, EPA, at 5. (Oct. 15, 2009) (emphasis added) (attached as Appendix 30). ECOS also cautions that -[e]onsuming the commercial hazardous waste landfill capacity not only means that [CCRs] will begin to pile up unmanaged at utilities, but that the current 2 million tons of hazardous waste generated by industry and hazardous waste site remedial activities will also begin to accumulate on-site," bringing to a halt Superfund clean-ups that require off-site disposal capacity. *Id.* at 5-6. As ECOS points out, it -ean take years to permit a new hazardous waste landfill." *Id*.

ASTSWMO has echoed this same concern to EPA. Letter from ASTSWMO to Matt Hale, EPA (Nov. 4, 2009) (attached Appendix 13). Among other points, ASTSWMO observed that the state resources required to manage up to an additional 134 million tons of CCRs will divert resources from the proper management of existing hazardous wastes (which are likely far more hazardous).

In sum, the record evidence shows that, contrary to EPA's belief, subjecting CCRs to Subtitle C regulation will quickly overwhelm the Subtitle C disposal market. The foreseeable shortfall in off-site CCR disposal capacity compounded with the inability to manage CCRs on-site will force utilities to either store CCRs on-site, in non-compliance with RCRA's permit requirements (subjecting the utility to continuing daily penalties), or cease operations altogether to avoid producing CCRs. These alternatives are not acceptable or warranted.

### C. Other Issues Related To Subtitle C Regulation

Another serious problem with the proposed Subtitle C regulations is the elimination of the management options to (1) dispose of and store CCRs in surface impoundments, and (2) allow for the placement of wastewaters in impoundments that contain CCRs. EPA suggests in the preamble to the proposal that the continued wet handling of CCRs in surface impoundments can be conducted in a manner that is protective of human health and the environment provided that the proposed composite liner requirements are met. 75 Fed. Reg. at 35174. However, certain components of the land disposal restrictions (-LDRs") that the Agency proposes would eliminate the use of surface impoundments, even those that employ the requisite liner systems. The application of these restrictions has not been adequately justified, imposes unnecessary burdens on the utility industry, and constitutes arbitrary and capricious rulemaking.

In the preamble to the CCR Subtitle C proposal, EPA suggests that the continued wet handling of CCRs in surface impoundments would be authorized and that the Agency would establish composite liner systems for surface impoundments and landfills. *Id.* The proposed regulatory text, however, does not include the composite liner requirements for surface impoundments. EPA explained that it would adopt such requirements because the Risk Assessment on which the Agency bases much of the rationale for the CCR proposal determined that -{e}omposite liners, as modeled in this assessment, effectively reduce risks from all constituents to below the risk criteria for both landfills and surface impoundment at the 90<sup>th</sup> and 50<sup>th</sup> percentiles." *Id.* at 35145. Based largely on this finding, EPA proposes to establish a composite liner system for landfills and surface impoundments under the Subtitle C option because the Agency

believes a composite liner system would be adequately protective of human health and the environment. *Id.* at 35174. In making this proposal, EPA intended to replace the double liner system currently in the hazardous waste regulations because such technology would be unnecessarily burdensome. *Id.* Although the preamble indicated that these alternative standards would be applicable to both surface impoundments and landfills, the proposed composite liner regulations were only applicable to CCR landfills.

It seems that EPA did not establish composite liner requirements for surface impoundments because other portions of EPAs proposed regulatory scheme would effectively eliminate CCR impoundments. As EPA well knows, the addition of the 100 mg/l total suspended solids (—TSS") LDR treatment standard for CCRs in surface impoundments will have the practical effect of eliminating the use of surface impoundments for both the disposal of CCRs as well as the management of wastewater in impoundments that contain CCRs. See 75 Fed. Reg. at 35262 (proposed 40 C.F.R. § 268.40). The TSS standard that EPA proposes to impose on CCR-containing wastewater has never previously been applied to other types of hazardous wastes and the Agency has barely attempted to provide any justification for virtually eliminating the wet handling of CCRs in surface impoundments that would otherwise meet all other proposed management standards and all other applicable LDR treatment standards. The other applicable LDR standard is that the CCR wastewater must meet the Universal Treatment Standards (-UTS") at 40 C.F.R. Part 268. The TSS treatment standard would be in addition to the UTS LDR treatment standard. EPA establishes this standard because it believes that dry disposal of the CCR solids will protect human health and the environment which is borne out by the results of the Agency's risk assessment and

damage case assessments, which show that wet disposal poses the greatest risks of contaminant release." *Id.* at 35180. This rationale is wholly inadequate given that only a few pages earlier in the preamble EPA maintains that composite liners are adequately protective of human health and the environment for surface impoundments. *Id.* at 35174.

EPA's incorporation of the TSS LDR standard to eliminate the wet handling of CCRs in surface impoundments is unwarranted and constitutes arbitrary and capricious rulemaking. These surface impoundments would need to be constructed with composite liners which the Agency has explicitly determined are sufficiently protective of human health and the environment and would need to meet groundwater monitoring and location restrictions, as well as the LDR standards normally applicable to all other hazardous waste units, all of which, based on EPA's own determination, would result in the safe management of CCRs. Adding the TSS LDR standard to an already protective regime is an abuse of Agency rulemaking discretion.

### IX. THE SUBTITLE C PROPOSAL TO REGULATE INACTIVE AND/OR PREVIOUSLY CLOSED SURFACE IMPOUNDMENTS IS UNLAWFUL

One of the most extreme examples of Agency overreach in the Subtitle C option is the unprecedented and unlawful proposal to extend RCRA Subtitle C jurisdiction to previously closed and/or inactive CCR surface impoundments. *Id.* at 35177. Even if such impoundments did not receive CCRs after the effective date of the rule, EPA would require these impoundments to close in compliance with the rule. EPA acknowledges that it has never attempted to extend RCRA in such a far-reaching manner to any other hazardous waste in its thirty-year implementation of the statute. In fact, EPA has consistently taken the opposite position throughout its administration of

the RCRA program, explaining that RCRA's Subtitle C regulations are prospective in nature and are not directed at inactive facilities for a host of practical, equitable and economic reasons. *See e.g.*, 43 Fed. Reg. 58984 (Dec. 18, 1978); 45 Fed. Reg. 33074 (May 19, 1980).

Nonetheless, EPA is proposing to reverse this position in the case of CCRs and retroactively apply RCRA's Subtitle C hazardous waste rules to previously closed and/or inactive CCR surface impoundments on the theory that the definition of regulatory disposal" encompasses -passive leaking" from these units, and includes -the continued release of constituents to surrounding soil and groundwater through the continued infiltration of precipitation through inappropriately closed CCR impoundments ...." Id. at 35177. Under this theory, any previously closed and/or inactive CCR surface impoundment that is no longer receiving CCRs, but which has not closed pursuant to the interim status hazardous waste closure requirements, would be subject to full Subtitle C hazardous waste controls on the effective date of the Subtitle C rules and would have to <u>re-close</u>" under the hazardous waste regulations. See id. at 35258 (proposed 40 C.F.R. § 265.1300). This would mean that literally hundreds of previously closed and inactive surface impoundments - many of which were properly closed decades ago under state solid waste programs, have changed owners, and now have structures built on top of them - would be considered -active" CCR disposal units and would be subject to full Subtitle C hazardous waste regulation. EPA's position runs directly counter to existing case law on the definition of disposal, as well as Congress' RCRA and CERCLA statutory scheme, and would raise a host of practical and compliance issues that EPA has, itself, recognized.

## A. <u>EPA's Position On Passive Migration Is Contradictory To Federal Case Law</u>

EPA's current position that RCRA's definition of -disposal" applies to closed and/or inactive units has been flatly rejected by the courts. Five United States Courts of Appeals have looked directly at the question of whether the definition of -disposal" includes the type of passive migration that EPA is proposing to force into the scope of these regulations. All five courts have soundly rejected this interpretation. Although these cases have considered the definition of disposal in the context of CERCLA, 42 U.S.C. § 9601 et seq., the distinction is of no moment" as CERCLA specifically incorporates by reference RCRA's statutory definition of -disposal." Delaney v. Town of Carmel, 55 F. Supp. 2d 237, 256 (S.D.N.Y. 1999); see also Interfaith Cmty. Org. v. Honey-Well Intl Inc., 263 F. Supp. 2d 796, 846 n.10 (D.N.J. 2003) (finding that although the Third Circuit's opinion in United States v. CMDG Realty Co., 96 F.3d 706 (3rd Cir. 1996), -addressed CERCLA liability, the Court was interpreting the RCRA definition of disposal" and the Third Circuit's holding was therefore controlling in an assessment of RCRA liability as well). 29

Under RCRA, -disposal" means:

[t]he discharge, deposit, injection, dumping, spilling, leaking, or placing of any solid waste or hazardous waste into or on any land or water so that such solid waste or hazardous waste or any constituent thereof may enter

Notably, the only Administrative Law Judge (-ALJ") to have considered this issue also concluded that the term -disposal" does not include the passive migration of hazardous waste. *In re Globe Aero Ltd., Inc. and the City of Lakeland*, 1996 EPA ALJ Lexis 47 (June 4, 1996). The ALJ held that -[d]isposal is an act rather than a state of continued noncompliance. The fact that the waste or waste constituents may migrate does not provide a basis for finding a continued disposal violation." *Id.* (internal citations omitted)."

<sup>&</sup>lt;sup>29</sup> See CERCLA § 101(29), 42 U.S.C. § 9601(29) (—The term[] disposal shall have the same meaning provided in section 1004 of the Solid Waste Disposal Act . . . .)"

the environment or be emitted into the air or discharged into any waters, including ground waters.

RCRA § 1004(3), 42 U.S.C. § 6903(3). Most recently, the Court of Appeals for the Ninth Circuit joined four other circuit courts in rejecting the notion that the passive migration of contaminants through the soil falls within this statutory definition of -disposal." See Carson Harbor Vill. v. Unocal Corp., 270 F.3d 863 (9th Cir. 2001). Adhering to the Supreme Court's admonition that <del>[i]</del>n examining the statutory language, we . . . adhere to the Plain Meaning Rule, "the court reasoned that -[i]f we try to characterize this passive soil migration in plain English, a number of words come to mind, including gradual spreading, migration, seeping, oozing, and possibly leaching.' But certainly none of those words fits within the plain and common meaning of discharge, ... injection, dumping, ... or placing." Id. at 879 (ellipses in original). The Ninth Circuit's holding in Carson Harbor Vill.. therefore is directly at odds with EPA's proposed interpretation that the passive migration of constituents from previously closed CCR surface impoundments constitutes -disposal" under RCRA. See also Delaney, 55 F. Supp. 2d at 256 (the alleged passive migration of hazardous waste, through the groundwater or otherwise[,] ... does not constitute \_disposal' of hazardous waste as defined in RCRA").

In addition, the Sixth Circuit has held that <u>because disposal</u> is defined primarily in terms of active words such as injection, deposit, and placing, the potentially passive words <u>spilling</u> and <u>leaking</u> should be interpreted actively[.]" *United States v. 150*Acres of Land, 204 F.3d 698, 706 (2000). As there would be no human activity involved in the movement of water through closed and/or inactive CCR impoundments, EPA's

proposed expansion of the definition of -disposal" to include passive migration directly contradicts the Sixth Circuit's holding.

The Second Circuit has similarly agreed that -[n]one of these terms [i.e., discharge, deposit, injection, dumping, spilling, leaking, or placing] is commonly used to refer to the gradual spreading of hazardous chemicals already in the ground." *ABB Industrial Systems v. Prime Technology*, 120 F.3d 351, 358 (2d Cir. 1997). Therefore, this Circuit's holding would also not permit EPA's proposed expansion of -disposal" to include the gradual spreading of chemicals already in the ground.

In addition, the Third Circuit has held that, based on the plain meaning of the words used to define disposal, -the passive migration of contamination dumped in the land prior to ownership does not constitute disposal." *United States v. CDMG*, 96 F.3d 706, 711 (3rd Cir. 1996). The Third Circuit noted that its conclusion was supported by the fact that the term -leaching" is not included in the statutory definition of -disposal."

Id. at 715. The Third Circuit's holding in *CDMG* rejects EPA's efforts to force the -passive migration" of wastes into the definition of -disposal."

The Fifth Circuit has also agreed that where the plaintiff -failed to show that any hazardous waste \_leaked' or \_spilled' during [the defendant's] ownership of the property[,] a disposal did not occur during [the defendant's] ownership." *Joslyn Mfg. Co. v. Koppers Co.*, 40 F.3d 750, 762 (5th Cir. 1994). This position is consistent with the holdings in the Second, Third, Sixth and Ninth Circuits, and similarly demonstrates that EPA's proposed passive disposal theory violates RCRA's clear statutory language.

### B. EPA's Proposal Runs Afoul Of Congress' Statutory Scheme

As EPA knows, it is elementary that RCRA's Subtitle C hazardous waste rules have properly and consistently been construed as being *prospective* in nature. Remedial statutes such as CERCLA, as well as RCRA's imminent and substantial endangerment provision, are the congressionally-provided statutory mechanisms that EPA has used to address historic waste management operations, including previously closed CCR surface impoundments. *See Gwaltney of Smithfield v. Chesapeake Bay Found.*, 484 U.S. 49, 57 (U.S. 1987) (noting that Congress has demonstrated ... [through amendment of RCRA's citizen suit provision in 1984 to include Section 7002(a)(1)(B)] that it knows how to avoid th[e] prospective implication [of the RCRA statute] by using language that explicitly targets wholly past violations.").

The prospective nature of RCRA's Subtitle C regulatory program derives from the plain language of RCRA Section 3004(a), under which EPA is authorized to promulgate Subtitle C regulations, including the proposed regulations at issue here. That language provides, in pertinent part, that EPA is to promulgate regulations—applicable to owners and operators of facilities for the treatment, storage, or disposal of hazardous waste."

42 U.S.C. § 6924(a). In contrast to RCRA's imminent hazard and citizen suit provisions where Congress specifically referenced—past" disposal practices, the language of Section 3004(a) makes clear that the Subtitle C regulations are to apply only to active, ongoing treatment, storage, and disposal operations—not to closed and/or inactive facilities. See, e.g., RCRA § 7002 (a)(1)(B)(authorizing a citizen suit against any—past or present" generator, transporter, owner, or operator of a treatment, storage, or

disposal facility —who has contributed or who is contributing to the past or present handling, storage, treatment, transportation, or disposal of certain hazardous waste").

Additionally, EPA's proposed interpretation of -disposal" renders the statutory phrase -past . . . disposal . . . of any solid or hazardous waste" in RCRA's imminent and substantial endangerment provision meaningless because, under the Agency's interpretation, as applied to CCR impoundments, there can be no -past" disposal of CCRs in impoundments, only present, *ongoing* disposal. This interpretation of the statute defies both logic and the plain language of the statute. *See Allen Oil Co., Inc. v. Comm'r of Internal Revenue*, 614 F.2d 336, 339 (2d Cir.1980) (-[A] statute must, if reasonably possible, be construed in a way that will give force and effect to each of its provisions rather than render some of them meaningless.").

Furthermore, as the Third Circuit in *United States v. CDMG* noted in finding that the definition of disposal does not include passive migration, the term -leaching" is not included within the scope of the activities that constitute disposal. 96 F.3d at 715.

-Leaching" is not an obscure term that Congress would have overlooked in drafting the definition of -disposal." Rather, inclusion of the term -leaching" in the definition of -release" under other provisions of the CERCLA statute demonstrates that Congress was aware of this process, understood it was conceptually different than -leaking" or

As EPA well knows, —aching" is the —process or an instance of separating the soluble components from some material by percolation," see Webster's Third New International Dictionary, Unabridged 1285 (Philip Babcock Gove & the Mirriam-Webster Editorial Staff eds., 1986), and is commonly used in the environmental context to describe the type of migration of contaminants (*i.e.*, -the continued release of constituents to surrounding soil and groundwater through the continued infiltration of precipitation") that EPA now proposes include in its overly expansive interpretation of the definition of -disposal." *Id.* (-Leaching of contaminants from rain and groundwater movement is a principal cause of contaminant movement in landfills, see Superfund [Exposure Assessment Manual 8 (1988)], and is the predominant cause of groundwater contamination from landfills, Edward Repa & Charles Kufs, Leachate Plume Management 2 (1985).")

-spilling," and chose not to include it in RCRA's definition of -disposal." EPA's proposal, which would essentially read a term into RCRA's definition of disposal that Congress had included in similar statutes, but chose not to include here, impermissibly violates fundamental canons of statutory construction.

## C. Regulating Previously Closed/Inactive CCR Surface Impoundments Raises A Host Of Legal And Practical Implications

Even EPA has recognized the illogical implications of interpreting -disposal" to apply to closed and/or inactive facilities. When the Agency first promulgated the Subtitle C hazardous waste rules, it candidly and correctly acknowledged that the hazardous waste regulations, for a multitude of legal, practical, economic, equitable, and technical reasons, should not be applied to units that have ceased operation before the effective date of an otherwise applicable Subtitle C rule. The Agency explained:

RCRA is written in the present tense and its regulatory scheme is organized in a way which seems to contemplate coverage *only of those facilities which continue to operate after the effective date of the regulations*. The Subpart D standards and Subpart E permitting procedures are not directed at inactive facilities. *Enormous technical, legal, and economic problems would arise if these standards were to be directly applied to inactive facilities [i.e., those facilities which have ceased receiving, treating, storing and disposing of wastes prior to the effective date of the Subtitle C regulations] and all such facilities were required to upgrade. Such an approach also does not seem equitable because of the enormous difficulty of bringing a closed facility into compliance, and because the present owner of land on which an inactive site is located might have no connection (other than present ownership of the land) with the prior disposal activities.* 

For those reasons, EPA does not plan to apply Subpart D standards to inactive facilities. The Agency believes that it can more equitably use Section 7003 (Imminent Hazard) of RCRA to bring suit against inactive facilities which pose human health and environmental problems[.]

43 Fed. Reg. at 58984 (emphasis added). EPA's proposal offers no legal basis or other rationale for its position as to why RCRA is no longer -written in the present tense" or

why its regulatory scheme is no longer -erganized in a way which seems to contemplate coverage only of those facilities which continue to operate after the effective date of the rules." Nor does EPA provide any reason why all the -enormous technical, legal, and economic problems" that would arise from applying RCRA's Subtitle C regulations to inactive units would not apply to closed and/or inactive CCR surface impoundments. Similarly, EPA does not explain why an approach that was not equitable thirty years ago -because of the enormous difficulty of bringing a closed facility into compliance" with the Subtitle C hazardous waste regulations is somehow equitable in the case of closed CCR surface impoundments today.

Rather than provide a justification for its abrupt and complete reversal of the statutory scope of RCRA, the Agency simply asserts that this new position is appropriate because of the size of previously closed CCR surface impoundments and the volumes of CCRs that remain in these units. The same could be said, however, for the literally thousands of *non*-utility landfills and surface impoundments that had closed prior to the enactment of RCRA's Subtitle C regulations over thirty years ago when EPA correctly reasoned that RCRA was not designed to address such units. EPA has not attempted to reach these other disposal units under RCRA's Subtitle C regulations, however; the preposterous theory that the Agency now advances has been reserved for previously closed CCR surface impoundments. There is no basis for EPA's attempt to reverse course solely in the case of CCR impoundments. The lack of any reasoned analysis for this about-face is the hallmark of arbitrary and capricious agency action.

AT&T v. FCC, 236 F.3d 729 (D.C. Cir. 2001) (Agency -eannot silently depart from

previous policies or ignore precedent"), citing *Greater Boston v. FCC*, 444 F.2d 841, 852 (D.C. Cir. 1970).

EPA's proposal also assumes, without any record evidence, that *every* former CCR surface impoundment, including those that closed in full conformance with applicable state laws and regulation, are now experiencing passive disposal from those units (*i.e.*, the leaching of CCR constituents from the impoundment). Put another way, under the proposal, any former surface impoundment that did not close in accordance with the hazardous waste closure requirements, even though the hazardous waste closure rules did not apply at the time of the unit's closure, was closed improperly and is releasing constituents to the environment. EPA has no knowledge that this is in fact the case. The Agency also does not know how many such units exist, where they are located, or how they were closed. In fact, units that closed in a fully protective manner and pose no threat to human health and the environment would be subject to regulation because the Agency assumes without any evidence that constituents are leaching from the unit. Any final regulation based on such an ill-informed and wholly unsubstantiated assumption constitutes arbitrary and capricious agency action and cannot be sustained.

Notably, EPA's proposal would also require -disposal" to be interpreted as a continued state of noncompliance without any discrete ending, for however many years and/or however many miles wastes continued to migrate. This entire time, a current, former, or even adjoining landowner could find themselves liable under RCRA for this so-called -disposal." Such unlimited and unending RCRA liability clearly violates both the letter and the spirit of the statute.

EPA's proposal would have absurd practical implications and contravene

Congress' directive that, in determining whether and how to regulate CCRs under

RCRA, EPA was to avoid Subtitle C regulation if at all possible – not subject CCRs to

the most stringent disposal regulations applicable to any hazardous waste regulated

under Subtitle C. The proposal cannot be sustained because it violates the plain

language of RCRA and, for this reason, has been flatly rejected by five United States

Courts of Appeals that, upon examining the issue have determined that the inclusion of

passive migration in RCRA's definition of -disposal" is at odds with the text and purpose

of RCRA.

### X. <u>EPA VIOLATED THE REGULATORY FLEXIBILITY ACT</u>

#### A. Background of RFA and SBREFA

In addition to the points discussed above, the Subtitle C option is fatally flawed due to EPA's failure to adhere to the requirements of the Regulatory Flexibility Act (-RFA"), 5 U.S.C. §§ 601-612, as amended by the Small Business Regulatory Enforcement Fairness Act (-SBREFA"), Pub. L. No. 104-121, tit. II, 110 Stat. 857, 864-68 (1996). Congress enacted the RFA to address the often disproportionate adverse economic effect that many federal regulations had on small entities, even when those small entities had done little to create the problems which the blanket regulations sought to correct. See Pub. L. No. 96-354, § 2(a), 94 Stat. 1164. As originally enacted, the RFA required that federal agencies solicit comments from small entities; examine the impact of proposed, newly promulgated, and existing rules on small entities; examine

The term -small" as used in this section to modify the words -business," -businesses," -entity," -entities," -government," "governments," -utility," and -utilities" refers to the small business size standards established by the Small Business Administration and located at 13 C.F.R. § 121.201 (Sector 22).

regulatory alternatives that would achieve the same purposes while minimizing adverse impacts on small businesses; and review the continued need for existing rules. Pub. L. No. 96-354.

Under the RFA, agencies must prepare and publish a preliminary regulatory flexibility analysis when proposing a rule, and a final regulatory flexibility analysis when issuing a final rule. 5 U.S.C. §§ 603-604. The RFA creates an exception from these requirements where the administrator of an agency —certites that the rule will not, if promulgated, have a significant economic impact on a substantial number of small entities [—\$\square\$NOSE]." Id. § 605(b). Congress raised the bar for these no-SISNOSE certifications in 1996 with passage of SBREFA, removing the language allowing agency administrators to —plolish such certification ... along with a *succinct* statement explaining the reasons for such certification" and instead requiring agencies to publish the certification —alog with a statement explaining the reasons for such certification." See Pub. L. No. 96-354; Pub. L. No. 104-121. The removal of the word —sccinct" from this provision signals Congress' intention that agencies provide a more thorough justification before making a no-SISNOSE certification and thereby avoid the requirement of examining the rule's impact on small entities through preparation of a regulatory flexibility analysis.

SBREFA also amended the RFA to require that EPA receive input from affected small businesses through the Office of Advocacy of the Small Business Administration (–SBA") before the required initial regulatory flexibility analysis is published. See 5 U.S.C. § 609(b). Additionally, when an agency's proposal is expected to have a SISNOSE, the agency must convene a panel of employees from the agency, the Office

of Advocacy, and the Office of Management and Budget (-OMB") to review a copy of the draft proposed rule and related agency analyses under RFA. *Id.* This Small Business Advocacy Review (-SBAR") Panel must also collect advice and recommendations of each individual small entity representative identified by the agency (through consultation with the SBA's Chief Counsel for Advocacy) and report on the panel's findings with respect to this input from small entities. *Id.* § 609(b)(4)-(5). Following the report of the SBAR Panel, where appropriate the agency must modify its proposed rule, initial regulatory flexibility analysis, and/or its decision on whether an initial regulatory flexibility analysis is required. *Id.* § 609(b)(6).

### B. The Agency's Determination Under § 605(b) That The Rule Would Not Have A SISNOSE Was Arbitrary And Capricious.

In evaluating the small business impact of the Subtitle C option, EPA failed to account for a number of potentially affected small entities in Step 1. This error was compounded by EPA's omission of a number of significant economic impacts associated with the Subtitle C Option in Step 2. As a result, EPA's determination in Step 3 that the rule was appropriate for § 605(b) —N SISNOSE" certification was fatally flawed.

As a threshold matter, USWAG believes that EPA grossly undercounted the number of small power producers and small governments that will be directly and adversely affected by the Subtitle C Option, if promulgated. This issue is discussed in greater detail in the comments of USWAG members APPA and NRECA, the relevant portions of which are herein incorporated by reference.

In addition, EPA failed to consider the full range of cost impacts associated with the Subtitle C option. For example, as discussed above, compliance with the Subtitle C

option would require utilities to convert from the wet to dry handling of CCRs, effectively requiring the closure of CCR surface impoundments and forcing the construction of new wastewater treatment facilities, a function previously served by the CCR impoundments. Promulgation of the Subtitle C option would also compel many utilities to obtain RCRA Subtitle C operating permits, which in turn would require these facilities to undertake facility-wide Subtitle C corrective action. While it is difficult to provide an industry wide number for these costs, they can be significant depending on plant specific characteristics. Utilities also would incur additional costs required to retrofit upstream CCR handling units (e.g., hoppers, tanks, conveyor systems) as well as implement additional operating and management measures to meet RCRA's hazardous waste design and operating requirements. In addition, regulation of CCRs under Subtitle C would force utilities to retrofit and/or re-close previously closed or inactive impoundments. As explained above, EPRI projects that compliance costs for the utility sector under the Subtitle C option would be in the range of at least \$55.3 to \$74.5 billion. EPRI Cost Report at 4-1.

The magnitude of these costs will prove devastating for small electric cooperatives and small governments, including many members of USWAG, members APPA and NRECA. Anticipating this foreseeable result, prior to the proposal's publication both APPA and NRECA made a formal request of EPA that the Agency convene an SBAR Panel. Letter from APPA and NRECA to EPA Administrator Jackson (Nov. 3, 2009) (attached as Appendix 32). The information provided to EPA by APPA and NRECA made clear that the Subtitle C option would have an enormous impact on a number of small utilities, and certainly demonstrated that, at the very least, the RFA and

Duke Energy Progress, LLC Late-Filed Exhibit No. 16 Docket No. E-2, Sub 1219 Page 212 of 439

SBREFA required EPA to conduct a thorough analysis of the extent of these impacts before deciding how to proceed with respect to the subsequent statutorily-prescribed process. Nonetheless, the Agency declined to convene an SBAR Panel, and subsequently proceeded to issue its ill-informed and short-sighted no-SISNOSE certification. These actions were in direct contravention of the RFA, as amended by SBREFA, and were arbitrary and capricious.

#### CHAPTER THREE

### EPA SHOULD ADOPT FEDERAL NON-HAZARDOUS WASTE RULES FOR CCRS

# XI. EPA SHOULD ADOPT THE SUBTITLE D PRIME OPTION, WITH APPROPRIATE ADJUSTMENTS, AND COORDINATE IMPLEMENTATION OF THE SUBTITLE D REGULATIONS WITH THE STATES

As USWAG has testified at the EPA public hearings on this proposal, we support the development of Subtitle D non-hazardous waste regulations for CCRs under the proposed Subtitle D —Prime" option, provided that EPA first incorporates certain important modifications discussed below. See e.g., Testimony of USWAG Executive Director Jim Roewer at the EPA CCR Proposal hearing, Denver, CO (September 2, 2010) (attached as Appendix 31). The question for USWAG is not whether to regulate CCRs under a federal program, but *how* to regulate CCRs. The Subtitle D Prime option, with the proper adjustments, will allow EPA to establish a robust and environmentally protective program for coal ash disposal units without crippling coal ash beneficial use and imposing unnecessary regulatory costs on power plants, threatening jobs, power reliability, and increasing electricity costs. While some of our suggested modifications to the Subtitle D Prime proposal will likely necessitate issuance of a supplemental proposal, this additional procedural step is worthwhile because it will help ensure promulgation of an environmentally protective regulatory program for CCRs consistent with the rulemaking record and the dictates of the Bevill Amendment.

#### A. The Rulemaking Record Supports The Subtitle D Option

At the outset, it is important to reiterate that the development of Subtitle D non-hazardous waste regulations for CCRs is consistent with the rulemaking record; in fact, as discussed above, it is the only lawful option available to the Agency. The

development of Subtitle D regulations would be the appropriate outgrowth of EPA's two Reports to Congress and two Regulatory Determinations under the Bevill Amendment declaring that CCRs do not warrant hazardous waste regulation under RCRA Subtitle C, but rather should remain primarily the province of state regulatory programs under RCRA Subtitle D.

Throughout its 20-year study and issuance of the required Reports to Congress and final CCR Regulatory Determinations, EPA has consistently found that the Subtitle D approach, with active state involvement, was the appropriate regulatory course for CCRs. In its first CCR Regulatory Determination in 1993, EPA concluded that, because the potential for damage from these wastes is most often determined by site- or region-specific factors," the -eurrent State approach to regulation" under RCRA Subtitle D is most appropriate. 58 Fed. Reg. at 42466. EPA echoed this finding in its second Report to Congress in 1999, concluding once again that [RCRA] Subtitle C is inappropriate to address any problems associated with disposal of these wastes and that the continued use of site and region specific approaches by the states [under RCRA Subtitle D] is more appropriate for addressing the limited human health and environmental risks that may be associated with disposal of these wastes." 1999 Report to Congress at 3-5.

EPA reaffirmed the appropriateness of the Subtitle D option in its 2000

Regulatory Determination, concluding that the subtitle D regulations are the most appropriate mechanism for ensuring that [CCRs] disposed in landfills and surface impoundments are managed safely." 65 Fed. Reg. 32221. In addition to the fact that CCRs rarely exhibit a hazardous waste characteristic, EPA explained that the decisive factors underlying its final determination that CCRs do not warrant regulation under

RCRA's Subtitle C hazardous waste regulations were: (1) improving trends in present CCR disposal and utilization practices (per the Bevill study factor in 8002(n)(2)); (2) the current and potential utilization of CCRs (*i.e.*, beneficial use), (per the Bevill study factor in 8002(n)(8)); and (3) Congress's admonition in the Bevill Amendment against duplication of efforts by other federal and state agencies. *Id.* at 32222, 32215.

In concluding that Subtitle D was the appropriate regulatory path forward, EPA also pointed to the record evidence as set forth in the 1999 Report to Congress that the utility industry has made significant improvements in its waste management practices over recent years" and to similar record evidence that the ECCR regulatory] programs have, in fact, substantially improved over the last 15 years or so." *Id.* at 32215, 32228-29. EPA found that the ability for most states to impose specific regulatory controls for coal combustion wastes has increased almost three-fold over the past 15 years." *Id.* at 32230. EPA concluded that, with the exception of relatively few states, the regulatory infrastructure is generally in place at the state level to ensure adequate management of these wastes." *Id.* at 32217. Critically, this positive trend has continued, with EPA and DOE finding in their joint report of state CCR controls that the theorem that an analyses documented in this report provide new information that appears to show improved management of CCWs in both landfills and surface impoundments." DOE/EPA Report at S-11.

Since the initiation of this rulemaking effort, the record support for the Subtitle D option has become even more compelling. Over the last year, the overwhelming majority of the states, the National Governors' Association, municipal and local governments, ASTSWMO, ECOS, over two dozen state environmental protection

agencies and state departments of transportation, federal agencies including the Department of Energy, a bi-partisan group of 165 members of Congress, including the majority of the House Energy and Commerce Committee, 45 U.S. Senators, and many other third-parties have expressed their support for regulation of CCRs as a non-hazardous waste under RCRA Subtitle D.<sup>32</sup> Given the breadth of federal agency, state, and local government and congressional support for the Subtitle D option, rejection of the Subtitle D option in favor of the Subtitle C approach would fly in the face of the Bevill Amendment's directive that any final regulatory program for CCRs take into account the views of other federal agencies and the states to ensure a unified decision that -avoid[s] a duplication of effort." See RCRA § 8002(n).

As discussed above, Representative Bevill repeatedly cautioned EPA that it was not to make a final regulatory determination for CCRs in a vacuum," but rather was to consult and coordinate with other key agencies that also have evaluated CCRs, including, among others, the Departments of Energy, the Interior, and Agriculture, and other government agencies to ensure a coordinated regulatory approach for CCRs consistent with the nation's -commitment to develop a coherent and consistent policy toward the use of our coal and other energy resources." 126 Cong. Rec. at 3362. These federal, state, municipal and local agencies have spoken and they are overwhelmingly in support of the Subtitle D option over the Subtitle C approach.

Even EPA has found that the coal ash being recovered from TVA Kingston's CCR release can be safely disposed of in a RCRA Subtitle D non-hazardous waste

<sup>&</sup>lt;sup>32</sup> See attached CD-ROM entitled —Public Comments and Testimony on EPA's Regulation of Coal Combustion Rssiduals," (Nov. 19, 2010).

facility under the federal Superfund program. In other words, the CCRs from the very event that served as a primary impetus for this rulemaking proceeding are being safely disposed of in a Subtitle D non-hazardous waste facility with EPA's explicit approval. And just recently, the Tennessee Department of Health, in conjunction with the federal Agency for Toxic Substances and, Disease Registry, issued a final Public Health Assessment concluding that the coal ash from the TVA release did not result in groundwater contamination nor result in ambient releases of ash above levels of concern. Public Health Assessment, Tennessee Valley Authority (TVA) Kingston Fossil Plant, Tennessee Department of Health (Sept. 7, 2010).

In short, given the overwhelming views by the very agencies that Congress directed EPA to consult in determining how to regulate CCRs under RCRA, combined with EPA's own Reports to Congress and two final Regulatory Determinations rejecting the concept of regulating CCRs under RCRA's hazardous waste program, the Subtitle D option provides the only lawful regulatory approach for these materials under RCRA.

### B. The Subtitle D Prime Option Will Be Fully Protective Of Human Health And The Environment

USWAG supports the Subtitle D Prime option because it would ensure the development of fully protective controls for CCRs in an effective and practical manner. It would include all the elements of the primary —Subtitle D" option with the exception that it —would not require the closure or installation of composite liners in existing surface impoundments; rather, these surface impoundments could continue to operate for the remainder of their useful life." 75 Fed. Reg. at 35210.

As EPA itself acknowledges, many of the Subtitle D option's substantive requirements, including those under Subtitle D Prime, are the same as those under the

Subtitle C option, including the technical and design standards for CCR liner systems, groundwater monitoring and associated corrective action, and surface impoundment integrity standards. 75 Fed. Reg. at 35193. In fact, many of the provisions under the Subtitle D option –either correspond to the provisions EPA is proposing to establish for RCRA Subtitle C, or are modeled after existing Subtitle C requirements." *Id.* In addition to borrowing from key elements of RCRA's existing standards for municipal solid waste landfills under 40 C.F.R. Part 258 – which the Agency found –would be expected to address the risks presented by the constituents in CCR wastes" – the proposed Subtitle D rules also considered and incorporated some of the self-implementing elements of RCRA's hazardous waste interim status requirements into the Subtitle D proposal. *Id.* 

Given that the Subtitle D Prime's substantive controls are patterned in large part on the proposed Subtitle C controls, the overall level of protection to human health and the environment are equivalent under both options. As EPA explains in describing the protective nature of the Subtitle D option, the Agency has historically interpreted both the Subtitle C and Subtitle D statutory provisions as requiring it -to establish a comparable level of protection, corresponding to an acceptable risk level ranging between 1 x 10<sup>-4</sup> to 1 x 10<sup>-6</sup> and that EPA can establish even more stringent Subtitle D standards where it deems such regulations are appropriate. *Id.* Therefore, the Subtitle D Prime option would provide the same level of protection to human health and the environment as the Subtitle C option without the corresponding adverse impacts of the Subtitle C approach.

One of the elements of the Subtitle D Prime option that makes it the better choice is that it does not require all existing surface impoundments to close if they are

operating in a manner that is fully protective of human health and the environment. This option does *not* include the automatic requirement to dredge and line all existing CCR surface impoundments within five years of the effective date of the regulations, or close these impoundments. *Id.* at 35244 (proposed 40 C.F.R. § 257.71(g)). Under the Subtitle D Prime option, existing CCR surface impoundments would continue to be subject to groundwater monitoring and corrective action requirements so that any threats from these units during their operating life would be detected and appropriate corrective action would be taken. USWAG agrees that CCR disposal units that are not fully protective must either take appropriate corrective action or close; however, there are many CCR surface impoundments that are perfectly safe. There is no reason why these units should not be allowed to continue operating.

Indeed, one of the largest cost elements of the primary Subtitle D option is the mandatory upgrading and/or closure of all CCR surface impoundments, irrespective of their environmental performance. According to the EOP Group's analysis, the costs to the utility industry of converting CCR wet handling systems to dry CCR handling systems under the Subtitle D option would be approximately \$34 billion dollars. See EOP Report at 16 (attached as Appendix 19).<sup>33</sup> There is no basis for the blanket imposition of this requirement and the associated staggering costs on existing units that

\_

From a strict analytical perspective, EOP also evaluated compliance costs under the Subtitle D option assuming no facilities converted from wet to dry handling, but instead constructed new impoundments meeting the Subtitle D composite liner and groundwater monitoring requirements. Under this scenario, Subtitle D compliance costs were estimated to be approximately \$7 billion. *Id.* at 15. However, as EOP points out, this scenario is not realistic for a number of reasons including lack of land availability for constructing new impoundments, permitting concerns, local opposition and long-term liability. *Id.* at 15-16. In fact, EOP reports that, for these reasons, many utilities would undertake wet-to-dry conversions for all fly ash ponds and at least half of all bottom ash and FGD impoundments. *Id.* at 16. Indeed, even the \$34 billion Subtitle D cost estimate does not include the fully loaded costs of converting from wet-to-dry CCR management.

are operating in a safe and effective manner. USWAG therefore supports the Subtitle D Prime option because it will ensure that CCR disposal practices are protective of human health and the environment without requiring the needless closure of environmentally sound CCR impoundments.

#### C. <u>The Subtitle D Program Must Include A Mechanism For State</u> Implementation Of The CCR Rules

While USWAG supports the Subtitle D Prime option (with the adjustments discussed below), a major shortcoming of this approach, as well as with the other Subtitle D alternatives, is the lack of any mechanism for qualified state programs to step in and directly administer the Subtitle D regulations, where the state programs meet or exceed the federal controls. Such an approach makes no practical sense, and would result in parallel and redundant regulatory programs for the same materials and, more importantly, would directly contravene the Bevill Amendment's directive that EPA avoid duplication of effort with pre-existing regulatory programs. See RCRA 8002(n). EPA has previously recognized this point. One of the Agency's primary reasons for concluding in its 2000 Regulatory Determination that CCRs do not warrant Subtitle C regulation was Congress's admonition in the Bevill Amendment against duplication of efforts by other federal and state agencies. 65 Fed. Reg. at 32215. Unfortunately, EPA has failed to adhere to this congressional directive in the proposed Subtitle D regulatory alternatives, including the Subtitle D Prime option.

This section provides, in relevant part, that the Administrator shall, as he deems appropriate, review studies and other actions of other Federal and State agencies concerning [CCRs] and invite participation by other concerned parties, including industry and other Federal and State agencies, with a view toward avoiding duplication of effort." 42 U.S.C. § 6982(n).

As a result of its findings in the final 2000 Regulatory Determination and the supplemental DOE/EPA Report of state CCR programs, EPA is well aware that there are many mature state CCR regulatory programs with permitting, liner, and groundwater monitoring controls similar to those proposed in the Subtitle D options. In fact, EPA concluded in its 2000 Regulatory Determination that state CCR controls have—substantially improved over the last 15 years," and that a High percentage of states had authority to impose protective management standards on surface impoundments and landfills, especially for groundwater monitoring, liners, and leachate collection . . . ."

Id. at 32229. The DOE/EPA Report, which evaluated changes in state CCR controls that occurred after 1995 (the cut-off date for the data used in the 1999 Report to Congress), found even further improvements in state CCR controls. Based on the states surveyed, the report found more stringent state CCR controls with respect to permitting, liner, groundwater monitoring, leachate collection, closure and post-closure care, and siting requirements. DOE/EPA Report at S-8.

In addition, the most recent survey of state CCR controls compiled in 2009 by ASTSWMO confirms the increasing level of protection provided by state controls, finding, among other things, that 86% of the states with CCR landfills and 69% of states with CCR surface impoundments have permitting requirements for these disposal units. Letter from ASTSWMO to Matt Hale, EPA (April 1, 2009) (attached as Appendix 12). Nonetheless, under all the proposed Subtitle D options, these state controls would be administered in addition to and without coordination with nearly identical, self-implementing federal controls. Such duplication is both unnecessary and unduly burdensome.

Under this dual regulatory approach, for example, an owner/operator of a CCR disposal facility could be found to be in non-compliance with a groundwater monitoring requirement that is contained in both the Subtitle D self-implementing rule and in an independently administered state regulatory program. In these circumstances, the owner/operator could be subject to a citizen suit enforcement action in federal court for alleged violation of the self-implementing Subtitle D rule, and to a wholly separate enforcement action in state court for violation of the parallel state requirement. This makes no sense; apart from the waste of federal and state judicial resources, it could result in inconsistent federal and state court determinations with respect to an identical regulatory requirement.

It is essential that any final Subtitle D rules for CCRs include a mechanism for qualified state programs to administer the rules in lieu of the self-implementing approach set forth in the proposal. As discussed below, EPA could readily develop a process for evaluating state CCR controls and determining whether state controls are no less stringent than the final Subtitle D controls (much like EPA does now in the case of municipal solid waste landfills under 40 C.F.R. Part 258). In those states that have qualifying controls for CCRs, EPA would allow the states to administer the controls to avoid the duplication of regulatory programs, with appropriate EPA backup. In this way, many of the self-implementing procedures in the Subtitle D approach could be eliminated as there would be a qualified state regulatory body with responsibility for ensuring compliance with the applicable requirements. This approach would allow for the requisite degree of state involvement in administration of the CCR rules that EPA concluded was appropriate in its Report to Congress and final Regulatory

Determinations, and avoid duplication of regulatory controls as directed by the Bevill Amendment.

### D. <u>EPA Has The Authority To Develop Subtitle D Rules Administered</u> <u>By The States With Direct EPA Enforcement Authority</u>

One of EPA's stated concerns with the Subtitle D option is that it would not allow EPA to require the states to implement the Subtitle D CCR controls through state permits, approve the adequacy of the state programs, or enforce the regulations against non-compliant facilities. 75 Fed. Reg. at 35194. This concern is misplaced. EPA is underutilizing the full scope of its authority under Subtitle D by looking solely to RCRA Section 4004 for purposes of developing Subtitle D controls for CCRs. 75 Fed. Reg. at 35193-35194. EPA can utilize the same combination of RCRA statutory authorities – *i.e.*, RCRA Sections 4010(c) and 4005(c) – for establishing controls for CCR disposal units that it employed in promulgating federally enforceable Subtitle D rules for municipal solid waste landfills (-MSWLFs") and for non-MSWLFs that receive household hazardous waste and small quantity generator waste under 40 C.F.R. Parts 257 and 258.

The combination of these two Subtitle D provisions enables EPA to promulgate non-hazardous waste rules for CCRs that can be directly administered through state permitting programs and backed up by direct EPA enforcement powers in those states that fail to adequately implement the federal rules. Such an approach offers a win-win for EPA: it provides the Agency with the enforcement authority it desires under a Subtitle D CCR regulatory program while enabling the states to have a prominent role in the administration of any Subtitle D CCR rules, thereby preventing the duplication of federal and state controls in keeping with the Bevill Amendment.

#### 1. RCRA Sections 4010(c) and 4005(c)

By way of background, RCRA Section 4010(c) was added to RCRA as part of the 1984 Hazardous and Solid Waste Amendments for purposes of having EPA upgrade its regulations with respect to those Subtitle D facilities -that may receive hazardous household wastes or hazardous wastes from small quantity generators" (referred to by EPA as -GESQG wastes"), including -facilities potentially receiving such wastes." 42 U.S.C. § 6949a(c). As EPA knows, the standards developed under this provision must be those necessary to -protect human health and the environment" and -[a]t a minimum [the criteria] for facilities potentially receiving such wastes should require groundwater monitoring as necessary to detect contamination, establish criteria for the acceptable location of new or existing facilities, and provide for corrective action as appropriate."

Section 4005(c), which was enacted as a companion provision to section 4010(c), directs the states to adopt the new regulations for Subtitle D facilities that may receive household or CESQG wastes and, of critical importance here, specifically authorizes EPA to *enforce* such regulations in those states that fail to adequately adopt the federal rules. Specifically, that section provides in pertinent part that, after EPA has promulgated the revised criteria under section 4010(c) for the specified Subtitle D facilities, —each State shall adopt and implement a permit program or other system o[f] prior approval and conditions to assure that each solid waste management facility within such State which may receive" household or CESQG hazardous waste—will comply with" the revised criteria promulgated by EPA under RCRA § 4010(c) for such facilities. RCRA § 4005(c)(1)(B), 42 U.S.C. § 6945(c)(1)(B).

EPA is then required to determine whether each state -has developed an adequate program" to implement the revised criteria. RCRA § 4005(c)(1)(C), 42 U.S.C. § 6945(c)(1)(C). If EPA determines that a state has not done so, the Agency -may use the authorities available under sections 6927 and 6928 of this title [i.e., RCRA Subtitle C enforcement and inspection authorities]" to enforce such revised criteria. *Id.* EPA has correctly described this authority as -significant in that *it represents the first authority for EPA enforcement of regulatory requirements under Subtitle D.*" 53 Fed. Reg. 33314, 33383 (Aug. 30, 1988) (emphasis added). Therefore, the Agency has already acknowledged that Subtitle D regulations promulgated under these provisions of the statute are enforceable by EPA.

#### 2. EPA's Implementation of Section 4010(c)

EPA's history in developing regulations under Section 4010(c) is informative in underscoring why the Agency can use this authority, in conjunction with RCRA section 4005(c), in establishing federally enforceable Subtitle D rules for CCRs. The Agency implemented the Section 4010(c) regulations in phases. In 1991, the Agency issued upgraded Subtitle D standards for all MSWLFs. These regulations (codified in the Subtitle D regulations at 40 C.F.R. Part 258) establish minimum federal criteria for MSWLFs, including location restrictions, facility design and operating criteria, liner requirements (for new MSWLFs), groundwater monitoring requirements, corrective action requirements, financial assurance requirements, and closure and post-closure care requirements. These criteria apply to any MSWLF that accepts any household waste, whether or not such waste would be considered a household *hazardous* waste. See definition of -municipal solid waste landfill unit" at 40 C.F.R. § 258.2. In other

words, the Subtitle D regulations for MSWLFs promulgated under Section 4010(c) apply to all MSLWFs, even though such facilities may not actually *receive* household hazardous waste or CESQG hazardous wastes.<sup>35</sup>

EPA recognized when developing the MSWLF rules that the scope of Section 4010(c) extends beyond MSWLFs to -the subset of solid waste disposal facilities that may receive hazardous household wastes or hazardous wastes from small quantity generators." See 53 Fed. Reg. at 33322 (emphasis added). For example, EPA stated that it was -eoncerned about the estimated 28,000 industrial solid waste disposal facilities and 2,600 construction/demolition landfills," but was limiting the scope of this initial rulemaking -to MSWLFs because there are insufficient data currently available to develop requirements for these other facilities." *Id.* at 33326.

In 1996, EPA issued its second round of Section 4010(c) regulations for non-municipal non-hazardous waste disposal units that receive CESQG hazardous waste.<sup>36</sup> These upgraded Subtitle D criteria establish standards pertaining to location restrictions, groundwater monitoring and corrective action. *See* 40 C.F.R. Part 257, Subpart B. Like the upgraded MSWLF Subtitle D rules, EPA has direct regulatory enforcement authority for these rules in states that the Agency determines have not properly implemented the revised criteria. *See* 61 Fed. Reg. 34252, 34266 (July 1, 1996).

<sup>&</sup>lt;sup>35</sup> This is consistent with the statutory language in section 4010(c) directing EPA to upgrade the Subtitle D standards for any Subtitle D facilities that -may" receive household hazardous waste or CESQG hazardous waste or may -potentially receive" such wastes.

<sup>&</sup>lt;sup>36</sup> The timeframe for issuance of these second round of regulations was dictated by a lawsuit brought by Sierra Club alleging that EPA was required to issue its Section 4010(c) rulemakings by the statutory deadline in Section 4010(c). Without any adjudication of law or fact, EPA entered into a Consent Decree with Sierra Club agreeing to issue the second round of Section 4010(c) regulations by June 1, 1996. See Consent Decree in No. 93-2167 (D.D.C.).

During this second rulemaking, EPA once again addressed the scope of Section 4010(c). A commentor pointed out that the statutory language in Section 4010(c) directing EPA to upgrade the Subtitle D criteria for facilities that *-may receive*" CESQG hazardous waste and for *-*facilities *potentially* receiving" CESQG hazardous waste required the rules to extend to all Subtitle D facilities that could accept such wastes (even if they in fact did not receive CESQG hazardous wastes). *Id.* at 34254. At that time, however, EPA took the position that section 4010(c) only was intended to reach facilities that *actually* received household hazardous or CESQG wastes. *Id.*<sup>37</sup> In fact, however, Section 4010(c) does not contain this limitation, but applies to *any* Subtitle D facility that *-may*" or has the *-potential*" to receive household or CESQG hazardous wastes. The Agency itself recognized this broad statutory reach when it issued the Section 4010(c) regulations for MSWLFs, which apply to any MSWLF that *-may*" receive household *hazardous* waste or CESQG hazardous wastes, regardless of the fact of whether the MSWLF unit actually received such wastes.

#### 3. Adoption of Regulations for CCRs under Section 4010(c)

The plain language of Section 4010(c) directs EPA to promulgate upgraded Subtitle D standards for -facilities that *may receive*" CESQG hazardous wastes and -for facilities *potentially* receiving such wastes." 42 U.S.C. § 6949a(c) (emphasis added). This language applies on its face to CCR disposal units, as they are Subtitle D solid

statutory study schedule than most other of the solid waste disposal facilities subject to the Section 4010(a) study schedule. USWAG agreed with EPA at the time that it would be premature to include CCRs in this rulemaking until the Bevill Amendment study process for CCR wastes was complete. Now that the Bevill study process is complete and EPA has determined that CCRs do not warrant hazardous waste regulation under RCRA Subtitle C, Section 4010(c) provides EPA with the appropriate statutory authority for adopting non-hazardous Subtitle D rules for CCRs.

waste management units that, under RCRA's statutory scheme, *may receive* CESQG hazardous waste and clearly have the potential to receive such wastes. Indeed, the legislative history to Section 4010(c) underscores that Congress did not intend EPA to construe the statute in such a narrow manner, as -{t}]he impetus for requiring these revisions is primarily the concern for potential disposal of hazardous materials with nonhazardous wastes. S. Rep. No. 98-284, at 50 (1983), reprinted in -A Legislative History of the Solid Waste Disposal Act, As Amended," at 2076 (1991). In short, nothing in the plain language of the statute limits these provisions to only those facilities that -actually receive" household hazardous wastes or CESQG wastes. If Congress had intended to limit the statute in this manner, it would have written the law to say just that; it did not.

In light of the above, USWAG urges EPA to re-evaluate its statutory authorities under RCRA Sections 4010(c) and 4005(c) as applied to CCR disposal units. Section 4010(c) directs EPA to establish fully protective Subtitle D disposal standards for CCRs, including: groundwater monitoring as necessary to detect and respond to groundwater contamination, corrective action, and criteria for the acceptable location of new or existing facilities. These criteria are fully consistent with the proposed Subtitle D controls in the instant proposal.

Equally important, development of CCR regulations under Section 4010(c) would allow for a meaningful and coordinated state role in the administration of the CCR rules, while providing EPA with direct enforcement authority in those states that fail to properly implement the federal rules. This balancing of the federal/state roles will ensure that there is a full and uniform application of the CCR Subtitle D rules in all the states and

would avoid the duplication of federal/state CCR controls as directed by the Bevill Amendment.

As noted above, USWAG appreciates that development of a proposed Subtitle D regime allowing for administration of the regulations by qualified state programs, whether under RCRA Section 4010(c) or through some other mechanism, will necessitate issuance of a supplemental proposal. Nonetheless, this additional procedural step is necessary to ensure promulgation of an environmentally sound regulatory program for CCRs that allows for appropriate state involvement. Such an approach is consistent with the rulemaking record and the dictates of the Bevill Amendment.

#### XII. THE SUBTITLE D PRIME OPTION REQUIRES CERTAIN AMENDMENTS TO BETTER REFLECT CCR DISPOSAL OPERATIONS

While USWAG supports the Subtitle D Prime option, and believes that it can be administered through qualified state programs with back-up EPA enforcement authority in under-performing states, there are fundamental flaws with elements of the proposed regulations in need of correction. Key problems include: unrealistically short closure timeframes for CCR disposal units; the unwarranted application of all location restrictions to existing CCR surface impoundments; the inclusion of inappropriate constituents in the proposed groundwater monitoring program, the failure to allow for alternate liner systems, and an overly broad definition of -GCR Landfill" that inappropriately includes all -large scale fill operations." These and other issues are discussed below.

## A. <u>Compliance With EPA's Proposed Timeframe For Closure Of Surface Impoundments And Landfills Is Impossible</u>

A significant flaw in the Subtitle D proposal is the unrealistic timetable for the closure of CCR surface impoundments and landfills that are unable to meet the proposed design and operating standards. Even under the Subtitle D Prime option, where CCR surface impoundments operating in an environmentally sound manner could continue to operate, there will undoubtedly be a significant number of impoundments that will have to close. Therefore, a more realistic and achievable closure process is necessary under this option.

The Subtitle D option would require that existing landfills and surface impoundments meet certain location restrictions or performance criteria (*e.g.*, not be located in unstable areas or have engineering measures incorporated into the design to ensure that the integrity of the structural components will not be disrupted if located in unstable areas). 75 Fed. Reg. at 35241-43 (proposed 40 C.F.R. § 257.60-§ 257.65). Further, within five years of the effective date of the Subtitle D rules, existing surface impoundments would be required to be dredged and have installed a composite liner and leachate collection system, or close. *Id.* at 35244 (proposed 40 C.F.R. § 257.71(g)). Because these obligations may be impossible to fulfill and would impose significant operational costs on owners/operators, USWAG agrees with EPA that -many surface impoundments may close as a result of these requirements." *Id.* at 35199. Likewise, a number of landfills may also close if they are unable to comply with the -unstable areas" location restriction. Many of these disposal units will need to close pursuant to the Subtitle D closure framework. The proposed timetables for actually

closing these landfills and surface impoundments, however, is totally unrealistic and many facilities will be physically incapable of meeting these schedules.

In particular, EPA proposes to require disposal units that cannot meet the location restrictions or associated operating standards to close within five years of the effective date of the finalized regulations, which may be extended by an additional two years<sup>38</sup> if the owner or operator can make a particular demonstration that there exists a lack of alternative disposal capacity and the unit poses no immediate threats to human health or the environment. Id. at 35243 (proposed 40 C.F.R. § 257.65). 39 While existing units may be allowed five years to continue operating, the period of time by which closure must be completed once they cease operating -i.e., stop receiving CCRs - is simply too short for the vast majority of CCR disposal units, especially surface impoundments. The proposal provides that an owner/operator must begin closure activities within 30 days after the date on which the CCR landfill or impoundment receives the known final receipt of CCRs and must complete closure within 180 days following the start of closure activities. 40 Id. at 35253 (proposed 40 C.F.R. § 257.100(k)). This means, for example, that a CCR impoundment that is still operating on the effective date of the rules but subsequently ceases receiving CCRs after the

<sup>38</sup> 

<sup>&</sup>lt;sup>38</sup> Even though the proposed regulatory text only provides this additional two years to disposal units located in unstable areas, EPA indicates in the proposal's preamble that this extension is intended to be applicable to all location restrictions applicable to existing CCR disposal units. *Id.* at 35202.

<sup>&</sup>lt;sup>39</sup> CCR surface impoundments that were not able to install a composite liner system would not be able to make this demonstration and would have to close within 5 years of the effective date of the rules. *Id.* at 35244 (proposed 40 C.F.R. § 257.71(g)).

There is an exception to this requirement allowing closure to begin no later than one year from the last receipt of CCRs if the unit has remaining capacity and there is a reasonable likelihood that the unit will receive additional CCRs. *Id.* at 35252 (proposed § 257.100(j)).

effective date of the rules has only 210 days to complete closure after the final receipt of CCRs.

Given the number and size of the disposal units that utilities operate, a significant number of units will not be able to complete closure within this 210-day timeframe.

First, because a large number of sites will have to be closed during roughly the same timeframe, utilities will not be able to obtain the personnel and equipment necessary to close multiple sites, especially since other companies will need to obtain the same resources at the same time. Furthermore, even if sufficient manpower/equipment is available, some owners/operators will simply be unable to close their disposal units in the timeframes EPA proposes, due to the size of certain CCR impoundments and landfills.

For instance, surface impoundments would be required to close by removing liquid wastes or solidifying remaining wastes. *Id.* at 35252 (proposed 40 C.F.R. § 257.100(c)(1)). For very small surface impoundments, it is conceivable that an owner/operator could comply with this requirement within the proposed time frame; for many surface impoundments, however, dewatering the impoundment alone can take several years to complete, making it physically impossible to comply with EPA's closure timeframe. As EPA acknowledges, its closure timeframes were borrowed directly from the existing closure timeframes for MSWFLs under 40 C.F.R. Part 258. *Id.* at 35209. While application of the MSWLFs standards to CCR disposal units is appropriate in many respects, this is not one of those instances. The CCR disposal units often are larger than MSWLFs, and MSWLFs obviously do not contain the volume of water contained in CCR impoundments.

Furthermore, landfills typically close each component cell when the cell reaches its disposal capacity so that many cells have already been -elosed" when the landfill begins final closure. Surface impoundments, on the other hand, must be entirely dewatered before closure can even begin for any portion of the impoundment. While EPA's proposed closure timeframe may be appropriate for some landfill types, 41 it is inappropriate to apply this same timeframe to surface impoundments. In fact, a USWAG member recently obtained approval for a closure plan for a 343-acre surface impoundment that provided for a twelve-year closure period to ensure adequate time to complete dewatering of the impoundment, assure the stability of the dewatered CCRs, and uniformly construct the slope of the final cover materials.

Additionally, 210 days may not be sufficient to acquire the materials needed to conduct and complete closure. For example, due to the significant volume of clay that is needed to cover an impoundment hundreds of acres in size (as well as the likelihood that other units may need to close in the same timeframe), there may be insufficient supply of clay and/or it may be impossible to transport the needed volume of clay to the site. Moreover, in some circumstances, states may not be able to provide final approval of a facility's closure plan before the end of the 210-day closure time limit, putting facilities in automatic non-compliance through no fault of their own, but due simply to limited state resources. There is an additional concern for northern tier states in that some closure work can only be conducted seasonally during the warm weather months.

\_

<sup>&</sup>lt;sup>41</sup> Because EPA includes a wide variety of landfill types, which extend beyond those below-grade-units that are similar to MSWLFs, these timeframes may be impossible to meet for some units classified as landfills in the proposal (e.g., structural fills that may be located above grade or have other physical characteristics different from typical MSWLFs).

Depending on the scope of the work and the project start date, it might not even be possible to complete the closure of a small facility in 210 days.

Given the disparity in the sizes of these units, the length of time necessary to dewater impoundments, and the need for many of these units to be closed when the rules come into effect, USWAG strongly recommends that EPA not establish a specific timeframe for closure. Instead, we urge EPA to require utilities to close CCR surface impoundments and landfills consistent with a closure plan approved by a state, or developed and certified by a registered professional. The establishment of a closure plan, with set schedules, is the most effective method to account for the many variables associated with the closure of these units and is the approach commonly used by utilities. A closure plan also will provide EPA and the public with certainty that closure will occur in a step-wise and timely manner, without requiring facilities to comply with wholly unrealistic closure schedules.

On a related note, EPA has made no effort in its Subtitle D proposal to consider those state programs that provide a means to convert CCR surface impoundments to landfills through dewatering and other appropriate requirements. After these impoundments are appropriately converted, EPA should specifically identify these facilities as landfills and impose the regulatory requirements for existing CCR landfills rather than require these converted units to be subject to controls for existing surface impoundments, including potentially closure requirements. This will authorize the continued operation of these existing CCR landfills and regulate these units pursuant to the appropriate applicable standards and not require utilities to acquire land for the construction of new landfills.

## B. <u>EPA Should Not Apply Location Restrictions To Existing Surface Impoundments Or Landfills</u>

The proposed Subtitle D standards would impose various location restrictions to new and existing CCR surface impoundments and landfills. In particular, new and existing CCR surface impoundments and landfills could not be located in floodplains (a requirement that is already in effect) or unstable areas. *Id.* at 35242-43 (proposed 40 C.F.R. § 257.64).<sup>42</sup> In addition, new surface impoundments and new landfills could not be located in wetlands, fault areas, or seismic impact zones and must be located at least two feet above the natural water table. *Id.* at 35241-42 (proposed 40 C.F.R. §§ 257.60 - 257.63.) Of significant concern, EPA states in the preamble to the proposal that it intends to subject existing surface impoundments to *all* of these new location restrictions due to the risks posed by these units. *Id.* at 35198-99. In the corresponding proposed regulatory text, however, existing surface impoundments are only subject to the floodplains and unstable area location restrictions. *See id.* at 35243 (proposed 40 C.F.R. § 257.65).

USWAG assumes for purposes of these comments that EPA intends to subject existing CCR surface impoundments to the additional location restrictions consistent with the proposal's preamble and that the Agency's proposed regulatory text is inaccurate. Presumably, EPA copied the regulatory text for the regulations applicable to MSWLFs as codified at 40 C.F.R. Part 258. Although EPA intends to base the Subtitle D CCR disposal unit regulations in large part on the MSWLF regulations, here, the

<sup>&</sup>lt;sup>42</sup> For this and a few other location restrictions, disposal units can be located in these locations if they meet a design or operating standard sufficient that the risk associated with the location restriction has been addressed.

Agency clearly departs from those regulations, requiring all existing surface impoundments to comply with the full range of location restrictions, a condition that is normally only applied to new units. This is an extreme and draconian departure from the MSWLFs standards; EPA has not made a compelling justification for treating CCR impoundments more stringently than MSWLF units.

Subjecting existing surface impoundments to these location restrictions must be backed by an evaluation of the risks posed by facilities in each of these types of locations. USWAG is not aware of structural stability issues directly caused by the placement of surface impoundments in wetlands, fault areas, or seismic impact areas of less than two feet above the natural water table. More importantly, apparently neither is EPA. The Agency does not even begin to assess the risks associated with the placement of existing impoundments in these areas. EPA simply asserts that the risks associated with impoundments are greater than those associated with landfills and cites to the TVA spill as an example of the environmental consequences of a -eatastrophic collapse" of a CCR impoundment. *Id.* at 35199. These general assertions, however, do not constitute the type of record evidence necessary to impose such sweeping requirements on all existing CCR surface impoundments. The Agency must conduct a detailed analysis of CCR surface impoundments located in these restricted locations and demonstrate, based on this analysis, that there are, in fact, increased risks from these impoundments to justify such a significant departure from how the Agency addressed and regulated MSWLFs.

USWAG does not believe that EPA can compile the record evidence necessary to support such a regulation. In fact, over the last year and a half since the TVA ash

release, EPA has been conducting comprehensive structural integrity assessments of utility CCR impoundments across the country. The results of EPA's investigation have not, to date, identified any CCR impoundment that presents the imminent threat of a catastrophic release similar to that which occurred at TVA's Kingston facility. EPA Coal Combustion Residuals Impoundment Assessment Reports (available online at http://www.epa.gov/epawaste/nonhaz/industrial/special/fossil/surveys2/index.htm) (current as of Nov. 3, 2010). Specifically, EPA's structural assessments rated the impoundments as -satisfactory," -fair," -poor," or -unsatisfactory" - terms commonly used in the field of dam safety. EPA explains that, fe]xpert experience has shown that only impoundments rated as unsatisfactory pose immediate safety threats." Id. Of the 120 CCR impoundments evaluated thus far, *none* of the impoundments received an -unsatisfactory" rating. *Id*. And, as EPA correctly points out, even those impoundments that received a fair" or poor" rating may have been found by the inspectors to be structurally sound, but did not receive a -satisfactory" rating based solely on lack of information. Therefore, there is no factual basis to assert that existing CCR impoundments pose the type of risk of catastrophic failure that justify the imposition of all the location restrictions on existing units; indeed, the record evidence makes clear that such restrictions are not warranted.

EPA does not propose subjecting existing CCR landfills to the full range of location restrictions because EPA reasons that such units are structurally less vulnerable than impoundments. Also, there would be significant disposal capacity shortfalls if existing CCR landfills located in these restricted areas – *i.e.*, wetlands, fault areas or seismic impact areas – had to close. 75 Fed. Reg. at 35198. USWAG agrees

with EPA that existing CCR landfills should not be subject to these additional location restrictions and shares EPA's concern that subjecting them to such controls would cause a significant decrease in disposal capacity.

USWAG shares EPA's concern that many CCR landfills are located in areas that would be subject to one or more of the above-referenced location restrictions and, while some landfills may be able to retrofit or already are in compliance with the performance standard tailored to the location restriction, some landfills could be forced to close due to the challenges of attempting to retrofit to meet the location-based performance standards. Aside from the fact that there is no factual record supporting the application of these additional restrictions to existing landfills, closure of these units would create disposal shortfalls. Among other things, unit closures would greatly complicate operations at many utilities that would need to find additional disposal capacity, requiring utilities to procure new real estate for siting a new landfill (which may be a significant distance from a power plant), obtain a new disposal permit for the landfill (which can take an extended period of time), and transport significant volumes of CCRs great distances to the newly-permitted facilities. There is simply no environmental basis for causing this disruption to utility CCR disposal practices by subjecting existing CCR landfills to the full range of location restrictions.

## C. <u>EPA Should Amend Its Run-On Control Requirements For CCR Surface Impoundments</u>

One complication of copying the proposed CCR Subtitle D regulations from the MSWLF program is that some of the particular operating requirements for MSWLF landfills cannot be directly applied to CCR surface impoundments. As discussed above, this is true in the case of attempting to apply the MSWLF closure time periods to CCR

disposal units. Another important example is the proposed requirement for surface impoundments and landfills to establish specific operating controls for water run-on and run-off. *Id.* at 35245 (proposed 40 C.F.R. § 257.81). The run-on portion of these controls requires a -eontrol system to prevent flow onto the active portion of the CCR landfill or surface impoundment" during peak discharge events. *Id.* (proposed 40 C.F.R. § 257.81(a)(1)). While this requirement may be appropriate for CCR landfills (as it is for MSWLFs), it is wholly inappropriate for surface impoundments that have been specifically designed as holding basins for stormwater or that have been formed by damming a ravine and have been designed with outlets that are sufficient to deal with the water flow from the natural watersheds into the impoundment during peak discharge events.

USWAG suggests that EPA address the inapplicability of this requirement for these types of surface impoundments by instead requiring that CCR impoundments be designed to appropriately address peak discharge events, rather than specifying that run-on into the impoundment be prevented. In this manner, impoundments that have been designed as described above and have been engineered to address peak discharge events will not be confronted with a restriction that is both unnecessary and effectively impossible to meet.

## D. <u>EPA Should Allow For Alternative Groundwater Monitoring</u> Parameters And Should Revise Certain Of The Proposed Parameters

USWAG supports appropriate groundwater monitoring standards for CCR surface impoundments and landfills and we support the development of such standards under a RCRA Subtitle D framework, which is modeled largely after the groundwater

monitoring programs for MSLWFs under 40 C.F.R. Part 258 and for non-MWSLFs that may receive CESQG wastes under 40 C.F.R. Part 257, Subpart B. *Id.* at 35205.

Unfortunately, unlike the groundwater monitoring programs for these units under Parts 258 and 257, the Agency is not proposing to allow an alternate list of indicator parameters to be used in the proposed CCR groundwater monitoring program because of the lack of direct state oversight and EPA's information on CCRs indicates that their composition would not be expected to vary such that the parameters are inappropriate." *Id.* at 35205-07. This is a mistake.

First, from a technical perspective, given that site-specific factors are critical for establishing constituents and corresponding concentrations for detection and assessment monitoring, it is important that EPA authorize registered professionals to certify variances from the otherwise applicable parameters used for detecting constituents in groundwater and assessing statistically significant increases from background for these parameters. Given the extreme range of natural variations in site geology and groundwater at CCR disposal units across the country, it will often be necessary for technical experts to adjust the specific groundwater monitoring standards that EPA establishes to account for various site-specific conditions and factors. Further, any variances to the list of regulatory parameters would only be available if approved by a qualified state program (under, for example, a state regulatory regime approved by EPA under RCRA Section 4005(c)) or if certified by a registered professional and such certifications are recorded in a facility's operating record, filed with the state and posted on the state or facility's publicly available internet site. As other critical elements of the Subtitle D program are predicated on compliance certifications by these technical

professionals, there is no reason not to authorize these professionals to identify and certify, where appropriate, necessary adjustments to the groundwater monitoring parameters based on site-specific factors.

Therefore, USWAG recommends that EPA include language in the Subtitle D groundwater monitoring rules for CCR disposal units similar to that in the MSWLFs rules allowing for the state or a registered professional (in the context of a self-implementing Subtitle D rule) to delete any constituents from otherwise required groundwater monitoring if it can be shown that the removed constituents are not reasonably expected to be in or derived from the waste contained in the unit." 40 C.F.R. § 258.54(a)(1). Similarly, the state or a registered professional should be authorized to establish an alternative list of indicator parameters in lieu of some or all of the otherwise specified indicator parameters if the alternative parameters provide a reliable indication of inorganic releases from the [CCR] unit to the groundwater." *Id.* at § 258.54(a)(2).

With respect to the initial list of indicator parameters proposed by EPA, USWAG agrees that it generally focuses on the mobile indicator parameters in leachate from CCR disposal units. 75 Fed. Reg. at 35253 (proposed Appendix III to 40 C.F.R. Part 257). The potential for false positives would increase significantly if the parameter list were burdened with a host of trace constituents that are much less mobile and are unlikely to be present in CCR leachate without the presence of the proposed indicator parameters. That being said, USWAG believes that the proposed indicator parameter list should be revised to more closely track the nature of materials that are disposed of in CCR surface impoundments and landfills. USWAG is particularly concerned that the proposed detection monitoring constituents - -eonductivity," -fluoride" and -eulfide" - are

unlikely to assist in efforts to determine whether CCR disposal units have released constituents into groundwater. In fact, the use of -eonductivity" as a parameter may lead to false positives and there is no need for this parameter when the Total Dissolved Solids parameter is a much more accurate test for the same type of potential releases. Similarly, fluoride and sulfide are not good indicator parameters because these constituents generally are not present in high concentrations in CCR leachate and therefore a generic requirement that they be monitored for all CCR disposal units is not warranted. Hence, we urge EPA to remove these parameters from the detection monitoring list. We also recommend that the remaining detection monitoring constituents be added to the assessment monitoring list to assist in assessing statistically significant increases in the detection constituents.

Additionally, USWAG urges EPA to limit the list of assessment monitoring constituents to the primary drinking water constituents for which a maximum contaminant level has been established (as well as the detection monitoring parameters, as addressed above). These primary constituents are the only materials EPA has found to contribute to adverse health effects; all other listed constituents have not been linked to health effects and EPA should not require groundwater remediation for these parameters. Accordingly, EPA should remove aluminum, iron, manganese and molybdenum from the list of constituents for assessment monitoring.

# E. EPA Should Allow Registered Professionals To Make Compliance Certifications Even If They Are Employees Of The Owner/Operator Of The Disposal Unit

As discussed above, it is critical that any final Subtitle D program for CCRs allow for administration of the Subtitle D regulations by qualified state regulatory programs. In

those states that EPA decides have adequate Subtitle D CCR programs, it will be unnecessary to have registered professionals certify compliance with the CCR regulatory provisions on behalf of the regulated owner/operator of the unit. Electric power producers will, of course, rely on these professionals for a review of all aspects of the design and operation of landfills and surface impoundments, but there is no need for the regulatory -eertifications" contemplated under the self-implementing Subtitle D option if a state is conducting a simultaneous and identical review of compliance with the regulatory provisions. In those states, however, that do not have qualified programs to administer and enforce the federal Subtitle D rules for CCR disposal, USWAG supports using registered professionals to conduct a review of the disposal units and certify that design and operating criteria are being met and to approve variances from the regulatory criteria, if appropriate, for particular CCR disposal units. With the limited exceptions identified below, USWAG also supports the proposed credential requirements for these professionals. Clearly, someone without the required expertise should not be certifying that surface impoundments or landfills have been designed and are operating appropriately. However, USWAG believes that EPA should add professional geologists to the list of registered professionals that have the expertise and authority to certify compliance with certain of the Subtitle D operating standards. In particular, it is appropriate to have professional geologists certify compliance with groundwater monitoring standards, just as it is appropriate for PEs to certify compliance with the design of such units.

USWAG does not agree that PEs, hydrologists, and geologists must be independent (*i.e.*, not an employee) of the owner/operator of the landfill or surface

impoundment. As EPA notes, even though an independent professional will not be an employee of the power producer, they will still be hired as an independent consultant of the company. *Id.* at 35194. USWAG believes that it is inappropriate to require that such professionals be independent from the owner/operator of the disposal unit, because this condition imposes a significant barrier to meaningful and cost-effective compliance with the regulations, while providing little to no assurance that -independent" professionals have distinct interests from the owner/operator of the disposal unit.

Utilities and independent power producers typically employ a wide array of PEs and scientific staff. Most utilities employ a number of PEs, hydrologists, and geologists who often work exclusively with company-managed CCR disposal units. These professionals typically possess the most relevant experience and knowledge about these types of disposal units, especially those units owned/operated by their employers. Company-employed PEs, hydrologists and geologists are subject to the same type of state registration and licensing as those professionals that are not employed by these companies and have an equally strong incentive to maintain their licenses in good standing as those professionals that are -independent" of the utility or independent power producer. Such state-licensing and registration programs will help to ensure that all professionals exercise proper judgment about the operation of CCR landfills and surface impoundments.

Notably, in a very similar regulatory program, EPA decided that *not* allowing PEs employed by facility owner/operators to certify regulatory compliance was inappropriate. In the re-promulgation of the Spill Prevention, Control, and Countermeasures (–SPCC")

regulatory framework, EPA addressed the issue of whether PEs employed by facilities should be allowed to certify SPCC plans. The Agency concluded:

We believe that most PEs, whether independent or employees of a facility, being professionals, will uphold the integrity of their profession and only certify Plans that meet regulatory requirements. We also agree that an inhouse PE may be the person most familiar with the facility. EPA believes that a restriction of in-house PE certification might place an undue and unnecessary financial burden on owners or operators of facilities by forcing them to hire an outside engineer.

67 Fed. Reg. 47042, 47053 (July 17, 2002).

EPA's position in the SPCC program is equally applicable here; there is no reason for EPA to depart from this position in the CCR proposal. Given that professionals employed by a company are more familiar with their company's design and operation of disposal units, and because state registration and licensing arrangements ensure the sound professional judgment of all professionals, we believe that requiring <code>-independence</code> of such professionals will add an unnecessary expense for utilities. Accordingly, EPA should eliminate the proposed requirements that these professionals be <code>-independent</code> from the company for which they are certifying compliance.

#### F. The Subtitle D Rule Should Allow For Alternate Liner Systems

It is absolutely essential that the Subtitle D rule allow for the use of alternative liner systems for CCR disposal units. In some regions of the country, the proposed composite liner system simply is not necessary to ensure protection of groundwater. Moreover, there are alternative liner systems that can provide equal, if not more protection to groundwater and there is no reason why these alternate, but equally protective systems, should not be available.

Under the proposal, the liner system for CCR disposal units would track the default liner requirements for MSLWFs (but critically not the standards for obtaining variances from these standards) by requiring a composite liner consisting of two components: -[a]n upper component consisting of a minimum 30 mil-flexible membrane liner (FML), and a lower component consisting of at least a two-foot layer of compacted soil with a hydraulic conductivity of no more than 1x10<sup>-7</sup> cm/sec." 75 Fed. Reg. at 35202; see also id. at 35243 (proposed 40 C.F.R. § 257.70(a)(2)). While the proposal currently does not allow for any deviation from the liner requirements for CCR units, EPA appropriately asks whether the Subtitle D option -should allow facilities to use an alternative design for new disposal units" provided an independent registered PE or hydrologist certifies that an alternate design system would -ensure that the appropriate concentration values for a set of constituents typical of CCRs will not be exceeded in the uppermost aguifer at the relevant point of compliance – i.e., 150 meters from the unit boundary downgradient from the unit, or the property boundary if the point of compliance (i.e., the monitoring well) is beyond the property boundary." Id. at 35202-03. There is no question that alternate liner systems must be allowed under the Subtitle D program.

As the Agency has consistently recognized throughout the Bevill Amendment regulatory determination process, given the vastly different site-specific characteristics of CCR disposal units across the country, there is no -one-size-fits all" approach when its comes to regulatory design standards for these units. EPA itself put it best when explaining in its final 1993 Regulatory Determination why the inflexible design standards

in the Subtitle C hazardous waste regulations were inappropriate for CCR disposal units:

A Subtitle C system would require coal combustion waste units to obtain a [RCRA] Subtitle C permit (which would unnecessarily duplicate existing State requirements) and would establish a series of waste unit design and operating requirements for these wastes, which would generally be in excess of requirements to protect human health and the environment. For example, if such wastes were placed in the Subtitle C universe, all ash disposal units would be required to meet specific liner and monitoring requirements. Since [CCR] sites vary widely in terms of topographical, geological, climatological, and hydrological characteristics (e.g., depth to groundwater, annual rainfall, distance to drinking water sources, soil type) and the wastes' potential to leach into the groundwater and travel to exposure points is linked to such factors, it is more appropriate for individual States to have the flexibility necessary to tailor specific controls to the site or region specific risks posed by these wastes.

58 Fed. Reg. at 42477. This logic applies in the context of the proposed Subtitle D program as well. Given that CCR sites —vary widely in terms of topographical, geological, climatological, and hydrological characteristics," any regulatory regime must have the flexibility to tailor controls to the site-specific characteristics of the individual disposal units. Otherwise, the same concerns with over-regulation that EPA correctly described with respect to a Subtitle C program will arise under a Subtitle D non-hazardous waste program for CCRs. For example, it is not necessary to require the installation of a composite liner system to a lateral extension of an existing CCR landfill in the arid Southwest where the distance to groundwater may be several hundred feet or more.

For example, the Salt River Project's (—SRP") Coronado Generating Station, located in St. John's Arizona, has climatic conditions favorable for CCR disposal. The facility operates a State of Arizona permitted FGD surface impoundment and ash landfill. The FGD unit and the ash landfill have an Aquifer Protection Permit (—APP")

issued from the Arizona Department of Environmental Quality ("ADEQ") that mandates protection for the underlying regional aquifer by providing for Best Available Demonstrated Control Technology (-BADCT"). The regional aquifer is located more than 300 feet below the bottom of the FGD impoundment. The average annual rainfall for this area is less than 12 inches. The FGD impoundment was excavated into low hydraulic conductivity materials, with permeability values as low or lower than that expected for engineered clay liners, and lower than the values used in EPA's model—based risk assessment.

The FGD impoundment and ash landfill are located on the Chinle Formation, which consists of 150-200 feet thick sequence of bentonitic clay, shale, and siltstone, with a hydraulic conductivity of 1x10<sup>-7</sup> to 1x10<sup>-9</sup> cm/sec. The permeability and thickness of the Chinle unit, along with the depth to groundwater, serve as BADCT for the ADEQ APP. The FGD impoundment has been operating with a natural clay liner for over 30 years. Contamination of the regional drinking water aquifer has not occurred in that time frame. Site specific modeling of groundwater flow over an operational period of 10,000 years indicates that the FGD impoundment has not, and will not, contaminate the regional groundwater under scenarios used in EPA's Risk Assessment.

Further, SRP's Navajo Generating Station located on the Navajo Nation land in Arizona operates a CCR landfill for all of its CCRs destined for disposal (*i.e.*, fly ash, bottom ash, and de-watered FGD material). The CCR landfill is located in a dry climate with an average annual rainfall of less than 8 inches. The depth to groundwater in the Navajo Sandstone Formation is 800-900 feet below the bottom surface of the landfill. Neutron logging is conducted at the landfill to detect any moisture in the vadose zone.

Groundwater monitoring is conducted down-gradient of the unit. The Navajo Sandstone Formation forms the principal regional aquifer system for the area.

In both of the above circumstances, a qualified state program can approve, or a registered professional can easily certify, based on site-specific conditions, that a clay liner or some alternative system will ensure that the above-referenced protection standards will not be exceeded at the point of compliance in the upper-most aguifer.

Even in locations where potential risks to groundwater are greater, there are alternate liner systems that can achieve the same level of protection as the composite liner system proposed by EPA. For example, a geosynthetic clay liner (-GCL") or similar systems can meet or exceed the performance standards of clay liners and certainly are equivalent or superior in some circumstances to two-feet of compacted soil in the lower component of the composite liner systems proposed under the Subtitle D option. Therefore, in many circumstances, a liner system using the GCL in lieu of the two feet of compacted soil readily enables a CCR disposal unit to meet the applicable groundwater monitoring standards.

The point is that, depending on the location and physical characteristics of the disposal unit, there is a range of liner systems that may be available to meet the applicable groundwater standards at the point of compliance. EPA should not limit alternative liner design systems and deny owners/operators the ability to utilize systems

<sup>&</sup>lt;sup>43</sup> EPA's own data support this point. See -Assessment and Recommendations for Improving the Performance of Waste Containment Systems," R. Bonaparte, D.E. Daniel, and R.M. Koerner, EPA/600/R-02/099, EPA National Risk Management Research Laboratory, Cincinnati, OH, Chapter 5 (available online at http://www.epa.gov/nrmlr/pubs/600r02099/600R02099.pdf).

that are equally or more protective as the composite liner system proposed in the Subtitle D option.

There also is no doubt that decisions regarding the use of alternate liner systems under a Subtitle D program can be made in a manner that is fully transparent and subject to the appropriate oversight. As explained above, we would expect any final Subtitle D regulatory program to allow for the administration of the rules by qualified state programs. In these circumstances, any alternate liner systems would be subject to review and approval by a responsible state regulatory agency. In circumstances where there is not a qualified state agency administering the program and the self-implementing model is employed for the Subtitle D rules, qualified professionals could review and approve alternative liner systems. Like other components of the self-implementing approach, these decisions would be included in the facility's operating record, reported to the state and posted on the facility's public internet site. In this way, the states and the public would have access to the reasoning and basis for a facility's decision to employ the alternate liner system and why this alternate is fully protective of human health and the environment.

### G. <u>Error In Proposed Leachate Collection System Requirements For CCR Surface Impoundments</u>

USWAG believes that EPA has inadvertently included an error in the regulatory text requiring the installation of a leachate collection system between the upper and lower components of the composite liner system for existing CCR surface impoundments and for new impoundments. Specifically, EPA proposes in the case of existing CCR impoundments that, no later than five years after the effective date of the Subtitle D regulations, CCR impoundments shall be constructed with a composite liner

(as defined below) with —a leachate collection system *between* the upper and lower components of the composite liner." 75 Fed. Reg. at 35243 (proposed 40 C.F.R. § 257.71(a)(1)) (emphasis added). EPA proposes a similar condition for new impoundments (and lateral extensions to existing impoundments), directing that the leachate collection system shall be installed —between the upper and lower components of the composite liner." *Id.* at 35244 (proposed 40 C.F.R. § 257.72(a)(1)). This clearly is an error, given that the composite liner systems that CCR surface impoundments are required to employ are required to have the upper flexible membrane liner (—FML") component —installed in direct and uniform contact with the compacted soil component." *Id.* at 35243-35244 (proposed 40 C.F.R. §§ 257.70(a)(2), 257.72(a)(2)). In other words, given that both components of the liner system must be in direct contact with one another, there is no way that a leachate collection system can be installed between these two components.

Even in the case of new CCR landfills, the proposed regulatory text makes clear that the leachate collection system is to be installed over the liner. *Id.* at 35243 (proposed 40 C.F.R. § 257.70(a)(2)). Therefore, EPA should eliminate the proposed requirement from the final regulations directing that leachate collection systems in the case of surface impoundments be between the upper and lower components of the liner system when these components must be in direct and uniform contact" with one another. Indeed, CCR surface impoundments operating in accordance with the proposed composite liner requirements and applicable groundwater monitoring requirements do not need a leachate collection system, as any possible migration of

constituents from the unit will be detected by the applicable groundwater monitoring system.

# H. Omission In Defining Scope Of Subtitle D Program

USWAG believes that EPA has inadvertently failed to include the regulatory parameters of the Subtitle D program into the regulatory text making clear that, like the Subtitle C option, the Subtitle D controls apply only to electric-only and combined-heat-and-power plants whose primary business is to sell electricity, or electricity and heat, to the public, *i.e.*, NAICS code 221112 plants. See 75 Fed. Reg. at 35254 (proposed 40 C.F.R. § 261.50(b)). As EPA makes clear in the preamble, the co-proposed options are intended only to apply to CCRs generated by electric utilities and independent power producers and not to non-utility boilers burning coal. *Id.* at 35129. EPA intends to address the appropriate regulatory controls for CCRs from the category of non-utility boilers after completing this rulemaking effort. *Id.* 

Therefore, to eliminate any potential confusion on the appropriate regulatory parameters of the proposed Subtitle D option, USWAG recommends that EPA amend the first sentence in the Applicability" text at proposed 40 C.F.R. § 257.40(a) (*id.* at 35240) to read as follows:

Applicability. (1) The requirements of this subpart apply to owners or operators of CCR landfills and CCR surface impoundments at electric-only and combined-heat-and-power (CHP) plants whose primary business is to sell electricity, or electricity and heat, to the public, *i.e.*, NAICS code 221112 plants.

For similar reasons, USWAG also believes that EPA should clarify in the regulatory text that the proposed Subtitle D controls do not apply to the beneficial use of CCRs. While this is obviously EPA's intent, it is not expressly spelled out in the proposed regulatory text. To avoid any confusion on this point, USWAG recommends

that EPA add a sentence to the end of the Applicability" section in proposed 40 C.F.R. § 257.40(a) stating the following: —The requirements of this subpart do not apply to CCRs that are beneficially used."

# I. <u>Definitional Issues Under The Subtitle D Proposal</u>

A number of the proposed definitions in the Subtitle D option require modification and/or clarification. These issues are discussed below.

#### Definition of —CCR Landfill"

EPA's proposed definition of -GCR landfill" is overbroad to the extent that it attempts to capture *all* operations involving the use of CCRs for -large scale fill operations." 75 Fed. Reg. at 35163. This definition would inappropriately capture the use of CCRs that are being used for a legitimate functional purpose, and thus are not a -solid waste" because they have not been -discarded." In addition, the phrase -large scale fill operations" is unlawfully vague and cannot be included in a final regulatory definition without a supplemental proposal setting forth discernible and readily understandable parameters as to what constitutes the type of -large scale fill operations" that would constitute a CCR landfill. Under the Subtitle D option, EPA would define the term -GCR Landfill" as:

A disposal facility or part of a facility where CCRs are placed in or on land and which is not a land treatment facility, a surface impoundment, an underground injection well, a salt dome formation, a salt bed formation, an underground mine, cave, or a corrective action management unit. For purposes of this subpart, landfills also include piles, sand and gravel pits, quarries, *and/or large scale fill operations*. Sites that are excavated so that more ash can be used as fill are also considered CCR landfills.

*Id.* at 35240 (proposed definition of -GCR landfill," 40 C.F.R. § 257.40(b)) (emphasis added).

One of the fundamental problems with this definition is that it assumes that all CCR large-scale fill operations involve the -disposal" of CCRs and that these operations therefore constitute CCR landfills. As EPA knows, however, a series of judicial decisions have established the parameters of RCRA's statutory definition of -solid waste," which, of course, also limits the regulatory parameters of what can constitute a CCR landfill under RCRA Subtitle D. These cases make clear that the definition of -solid waste" must be interpreted in a common sense manner that extends -only to materials that are truly discarded, disposed of, thrown away, or abandoned," and that materials destined for legitimate beneficial reuse or recycling are -not part of the waste disposal problem" and thus cannot be characterized as solid waste. Am. Mining Cong. v. EPA, 824 F.2d 1177, 1186, 1190 (D.C. Cir. 1987) (-AMC"). As the D.C. Circuit reaffirmed in a subsequent opinion, the definition of -solid waste" is predicated on a material being -discarded" and -fs]econdary materials destined for [legitimate] recycling are obviously not of that sort." Ass'n of Battery Recyclers, Inc. v. EPA, 208 F.3d 1047, 1051 (D.C. Cir. 2000), citing, AMC, 824 F.2d 1177.

Given the controlling case law, it is simply incorrect to assume that all CCR large-scale fill projects involve the -discard" of CCRs. To the contrary, CCRs have long been used for a wide range of large-scale structural fill projects where they serve the functional purpose of providing structural stability including for, among other things, aboveground structures, road projects, embankments, and reconfiguration of property to meet design standards for run-off controls. When used in these applications, CCRs often are being used in lieu of competing commercial materials, such as soil or gravel that would otherwise have to be excavated and/or mined and purchased by end-users,

including, among others, federal, state and municipal governments who routinely use CCRs in large scale road and embankment projects. In these circumstances, the CCRs are not being -discarded" in any common-sense understanding of the term, but rather are being used for a legitimate beneficial purpose in lieu of competing materials.

Further, there is no record evidence supporting the proposition that all types of -large-scale fill operations" constitute the disposal of CCRs. In fact, the sole basis referenced in the preamble for including these beneficial use activities within the definition of a -GCR landfill" is that -[t]here has also been significant community concern with large-scale fill operations." *Id.* at 35163. This unsubstantiated and generalized statement of concern cannot support the whole scale classification of a long-standing beneficial use activity as somehow constituting disposal.

EPA does not analyze the criteria typically employed by the Agency in determining whether a particular reuse activity involves legitimate recycling. When these criteria are applied to CCR large scale fill operations, the better argument is that these operations fall squarely with the Agency's core criteria of what constitutes legitimate -beneficial use." For example, CCRs used for large scale fill operations provide a -functional benefit" in terms of structural support. In addition, the CCRs often are being used as -substitutes for the use of a virgin material, conserving natural resources that would otherwise need to be obtained through practices, such as extraction." *Id.* at 35162. These criteria are the hallmarks of a legitimate beneficial use when applied to CCRs used in other contexts, and they are equally applicable to CCRs when used in large-scale fill projects. For the above reasons, EPA's blanket inclusion of

all CCR landfill is at odds with RCRA's definition of solid waste" and is arbitrary and capricious.

The proposed categorization of all -large scale fill operations" as constituting the disposal of CCRs also would be struck down as being -void for vagueness." Even EPA recognizes that this critical term in the definition of -CCR landfill" lacks any functional or discernable boundaries, explaining that -EPA recognizes that we need to define or provide guidance on the meaning of a large scale fill operation. " Id. at 35163. It is axiomatic, however, that the Administrative Procedures Act (-APA") requires EPA to seek comment on any proposed definition of -large scale fill project" – which will help define the universe of facilities subject to the final Subtitle D rules – in the instant proposal and not after the rule is finalized. 5 U.S.C. § 553(c). Put another way, EPA must do more than make the public guess at what this term may mean in a final regulation and then expect the regulated community to comply with this newly minted definition.

Nor can EPA attempt to provide the critical contours of this key regulatory concept through the subsequent issuance of -guidance." Such a back-door attempt to define this important term would be in violation of EPA's obligations under the APA to provide the public with an opportunity to review and comment on proposed regulations. Before attempting to include the concept of whether and under what circumstances CCR large-scale fill operations constitute the disposal of CCRs, EPA must provide the public with an opportunity to review and comment on how EPA proposes to define large scale fill operations." The Agency has not done this in the instant proposal.

#### 2. Definition of CCR Landfill and CCR Piles

Another flaw in the proposed definition of —GCR landfill" is that it would include all—piles" of CCRs, with no limitation of any kind (—ff]or purposes of this part, [CCR] landfills also include piles, sand and gravel pits, quarries...."). *Id.* at 35239-40 (proposed 40 C.F.R. §§ 257.2 & 257.40(b)). As is the case with —large scale fill operations," inclusion of this term in the proposed definition is overbroad and would inappropriately capture CCR management activities that do not constitute disposal.

For example, CCRs often are placed into piles for purposes of staging and/or consolidation *prior* to placement in a CCR landfill. To categorize and regulate the CCR storage pile in these circumstances in the same manner as the actual landfill in which the CCRs are disposed of is illogical and obviously inappropriate. Even in the context of EPA's hazardous waste rules, the Agency recognizes the distinction between landfills and storage piles, and defines and regulates the two as separate management units; indeed, the Subtitle C regulations expressly exclude -piles" from the definition of a landfill." See 40 C.F.R. § 260.10. The term -pile" is defined under the Subtitle C regulations to include -any non-containerized accumulation of solid, nonflowing hazardous waste that is used for treatment or storage and that is not a containment building." *Id*.

To remedy this problem, EPA should remove the term —piles" from the definition of —CCR landfill," as piles are not landfills in any ordinary (or existing regulatory) sense of the term. USWAG understands that EPA may be concerned about piles being used inappropriately for sham disposal of CCRs. However, a more appropriate way to address this concern would be to set a limit on the amount of time that the CCRs are

allowed to be maintained in a pile. The storage of CCRs in piles beyond the timeframe established in the regulations would create a rebuttable presumption that disposal, not storage, is taking place and that the pile should be subject to the CCR landfill requirements. A timeframe of 180 days would be an appropriate length for purposes of establishing a rebuttable presumption that the pile be viewed as a -CCR landfill" and not as a storage pile when storage exceeds this time limit. As with any rebuttable presumption, however, the facility would have the opportunity to demonstrate that storage beyond the 180 days is legitimate and necessary to facilitate disposal in a landfill or impoundment (e.g., disposal capacity is limited and/or storage beyond this time period is necessary to facilitate transportation to a disposal facility).

Also, given that the CCRs in piles are often continuously removed and replaced with new CCRs, the 180-day time limit would apply to the CCRs themselves and not to the existence of the pile itself, which will be continually changing in size as CCRs are periodically added and removed from it. This is the approach EPA takes in the case of applying the 90-day accumulation time limit to continuous process flow-through tanks under the Agency's hazardous waste accumulation rules under 40 C.F.R. § 262.34; the same common-sense approach should be applied here.

#### 3. Definition of CCR Surface Impoundment

EPA's proposed definition of -GCR surface impoundment or impoundment" is overly broad in that it would inappropriately capture impoundments that are not designed to manage CCRs. To correct this problem, USWAG has proposed a revised definition of -GCR impoundment" at the end of this section to help ensure that the final

definition includes only those impoundments that truly function as CCR surface impoundments at electric utility power plants.

As currently drafted, the definition includes:

A facility or part of a facility which is a natural topographic depression, man-made excavation, or diked area formed primarily of earthen materials (although it may be lined with man-made materials), which is *designed* to hold an accumulation of CCRs containing free liquids, and which is not an injection well. Examples of CCR surface impoundments are holding, storage, settling, and aeration pits, ponds, and lagoons. CCR surface impoundments are used to receive CCRs that have been sluiced (flushed or mixed with water to facilitate movement), or wastes from wet air pollution control devices, often in addition to other solid wastes.

75 Fed. Reg. at 35239-40 (proposed 40 C.F.R. § 257.2 (emphasis added)). A critical limitation in this definition is that it is intended to capture only impoundments -designed" to accumulate CCRs. If applied appropriately, this definition generally would only extend to those impoundments into which CCRs are initially sluiced, as these are the impoundments at power plants that are -designed" to accumulate CCRs.

USWAG is concerned, however, that the definition could inappropriately be interpreted to include downstream secondary and tertiary impoundments – such as polishing, cooling, wastewater and holding ponds – that receive waters containing only de minimis amounts of CCRs. For example, many power plants have holding ponds and/or wastewater ponds constructed and used for the management of various water streams that do not contain any CCRs. A wastewater impoundment at a facility that receives non-CCR wastewaters may, however, occasionally receive de minimis amounts of CCRs via storm water runoff (for example, from an FGD storage pad) or from wastewater containing de minimis amounts of CCRs from an upstream CCR impoundment. In these circumstances, the wastewater impoundment clearly is not

-designed to hold an accumulation of CCRs," and is not in any practical or technical sense -used to receive CCRs that have been sluiced" to the impoundment. It would be illogical and unnecessary to apply the full array of proposed controls for CCR impoundments, including liner systems, groundwater monitoring, siting restrictions, and related standards, to these impoundments.

The potential for an overly broad application of the proposed definition capturing non-CCR surface impoundments is underscored by EPA's over-counting of CCR impoundments in the proposal. In particular, EPA explains that, in the mid-1990s, it estimated that there were 286 CCR surface impoundments in use. 75 Fed. Reg. at 35151. EPA now estimates, however, that there are -584 CCR surface impoundments or similar management units in use at roughly 495 coal-fired power plants." *Id.* The Agency readily acknowledges, though, that this increase in numbers does not reflect an actual change in practices, but that -much of the increase in surface impoundments *likely results from counting units that receive wastewater that has been in contact with even small amounts of coal ash, and thus includes many units which were not included in EPA's mid-1990 estimates." <i>Id.* (emphasis added).

USWAG believes that this over-counting was the direct result of EPA's

Information Collection Request (+CR") distributed to electric utility power plants in 2009

wherein EPA requested that *any* impoundment receiving even *de minimis* amounts of

CCRs be identified. It was clearly understood at the time that the breadth of this

request could capture impoundments – such as polishing ponds and wastewater ponds

– that are not CCR impoundments, but which nonetheless may contain *de minimis*amounts of CCRs through power plant runoff or other sources. Unfortunately, EPA

appears to have incorrectly used the total number of impoundments identified in the ICR responses for purposes of reporting a dramatic increase in the number of CCR surface impoundments used by electric utilities. As a result, EPA's assertion that there are -584 CCR surface impoundments or similar management units" is factually incorrect and inappropriately skews upward the number of CCR impoundments analyzed by EPA for development of the proposal.

This factual mistake cannot be perpetuated through a poorly worded definition of -CCR surface impoundment." To ensure that this does not occur, the definition must be amended to include a more detailed functional element to make clear that downstream secondary and tertiary impoundments at power plants that receive only *de minimis* amounts of CCRs should not be labeled as CCR impoundments. EPA itself acknowledges that it intends for the proposed definition -to capture those surface impoundments that are described in EPA's damage cases and risk assessments." *Id.* at 35196. The final rule should make clear that cooling, polishing, wastewater, and similar ponds are not included in this universe.

In addition to including a more specific functional description within the definition, EPA should incorporate a size limit into the definition. This will ensure that small units that do not pose the potential risks associated with the types of CCR impoundments described in the damage cases and risk assessment are excluded from the definition. EPA has proposed a size threshold of 20-acre-feet or more for purposes of triggering the impoundment structural integrity requirements for CCR surface impoundments. See *id.* at 35243 (proposed 40 C.F.R. § 257.71(b)(1)). Out of an abundance of conservatism, USWAG believes that a size threshold of half this volume – or 10-acre-

feet or more – should serve as the size criterion for classifying a unit as a CCR surface impoundment in the first instance (though USWAG agrees with EPA that the 20-acrefeet threshold should serve as the criterion for triggering the structural integrity requirements for CCR surface impoundments). Finally, the definition should not encompass units that, having once served as impoundments, have been dewatered and closed for purposes of serving as the base for a CCR landfill or other structures. Dewatered impoundments that subsequently serve as the base for CCR landfills are not impoundments and should not be captured by the CCR surface impoundment definition.

Given the above, USWAG suggests that EPA's proposed definition of CCR impoundment be modified to read as follows:

CCR surface impoundment means a facility or part of a facility which is a natural topographic depression, man-made excavation, or diked area formed primarily of earthen materials (although it may be lined with manmade materials), which has a storage volume of 10 acre-feet or more and which is specifically designed for and whose primary function is to hold an accumulation of CCRs containing free liquids and which is not an injection well. Examples of CCR surface impoundments whose primary function is to hold an accumulation of CCRs are primary holding, storage, settling, and aeration pits, ponds, and lagoons that used to receive CCRs that are sluiced (flushed or mixed with water to facilitate movement), or wastes from wet air pollution control devices, often in addition to other solid wastes. This definition is not intended to encompass impoundments whose primary function is not to receive CCRs, but which receive de minimis amounts of CCRs through the receipt of waters that contain only incidental amounts of CCRs. Examples of impoundments that are not CCR impoundments include, but are not limited to, cooling, polishing, process water storage, storm water management, recycling, wastewater and holding ponds. CCR impoundments also do not include former impoundments that have been de-watered and closed and/or dewatered and subsequently used as the base for CCR landfills, buildings, or other structures.

These modifications to the definition of CCR surface impoundment will help ensure that only those impoundments that truly function as CCR impoundments are

subsumed within the –GCR impoundment" definition and subjected to the proposed CCR impoundment regulations. This is the most logical approach, as it would make no sense to subject units that contain only *de minimis* or trace amounts of CCRs from incidental sources to the full array of proposed CCR impoundment requirements.

# 4. Definition of Existing CCR Surface Impoundment

The proposed definition of —eixting CCR surface impoundment" needs to be modified to clarify that it does not include impoundments that cease receiving CCRs before the effective date of the rules. The proposed definition includes, in pertinent part, an impoundment that —was in operation ... prior to the effective date of the rule." See 75 Fed. Reg. at 35240 (proposed 40 C.F.R. § 257.40(b)). EPA should modify this phrase to include an impoundment that —was in operation and had not yet ceased receiving CCRs prior to the effective date of the rule" to make clear that the definition does not encompass units that are no longer receiving CCRs on the effective date of the rule, even though the unit may not have completed final closure prior to the rule's effective date. Plainly, units that are no longer receiving CCRs on the effective date of the rule are not—ni operation" and therefore should not be subject to the standards applicable to active units.

EPA also should clarify in the final rule that the definition of —eixting CCR surface impoundment" does include impoundments that are operating on the effective date of the rules and that are periodically dredged out during the operating life of the impoundment. While this may seem self-evident, it is important to clarify this point so that impoundments periodically dredged out during their normal operating life to allow for their continued use are not inappropriately characterized as a —ew CCR

impoundment" when the facility reinitiates placement of CCRs in the impoundment after dredging operations are complete.

## 5. Definition of Existing CCR Landfill

The proposed definition of —eixting CCR landfill" should be modified to include lateral expansions of operating units where such expansion is within the site footprint of an area already approved and permitted by the state for the landfill. The proposed definition includes:

A CCR landfill which was in operation on, or for which construction commenced prior to [the effective date of the final rule]. A CCR landfill has commenced construction if the owner or operator has obtained the Federal, State and local approvals or permits necessary to begin physical construction; and either:

- (1) A continuous on-site, physical construction program has begun; or
- (2) The owner or operator has entered into contractual obligations which cannot be cancelled or modified without substantial loss for physical construction of the CCR landfill to be completed within a reasonable time.

Id. at 35240 (proposed 40 C.F.R. § 257.40(b)).

While this definition includes undeveloped areas within the footprint of an approved permitted site, it also requires that construction actually have begun at the site or that some type of binding contractual obligation be present for these areas to be considered an —exting CCR landfill." However, if an undeveloped area within the footprint of a permitted landfill site for which there is a contractual obligation to begun construction is subsumed within the definition, there is no environmental or health-based reason for not including the precise same location within the definition even if there is not a contractual commitment to begin construction. The point is that, under both scenarios, the undeveloped portion of the approved permitted site have undergone

environmental and related technical review by the permitting authorities and have been approved for the location of a CCR landfill. To exclude a location that has otherwise already been permitted for a CCR landfill from the definition of an —eixting" landfill merely because there is not a binding contractual commitment to begin construction would unfairly subject these areas to the design and operating standards for —new CCR landfills," when *identical* locations are considered —eixting CCR landfills" subject to different operating standards based merely on the existence of a contract to commence construction. Such a distinction is arbitrary and capricious and provides no practical benefit.

# XIII. EPA MUST REFINE THE PROPOSED REGULATIONS TO ENSURE THAT LEGITIMATE BENEFICIAL USES ARE EXCLUDED FROM REGULATION AND THAT CCR MINE PLACEMENT ACTIVITIES ARE NOT CAPTURED BY THIS RULEMAKING

USWAG and its members have long supported the beneficial use of CCRs and have worked with EPA, other federal agencies, states, localities and standard-setting organizations to pursue the environmentally protective and beneficial use of CCRs in a wide variety of industrial, commercial, residential products, and construction activities. USWAG supports the exclusion of beneficial uses from any hazardous or solid waste regulation, provided that the use of CCRs is protective of human health and the environment.

We addressed above the stigma and liability concerns associated with regulating CCRs under RCRA Subtitle C (even if only in the disposal context), which will detrimentally impact the beneficial use of these materials. In this section, we address the revisions that EPA must make to both the Subtitle D and Subtitle C proposals to ensure that CCR beneficial use applications are not improperly regulated as CCR

disposal. Specifically, EPA must (i) revise its proposed definition of CCR landfills and the proposed exclusion for CCR beneficial uses to ensure that the proposed regulations do not sweep in environmentally protective beneficial uses that could be classified as large-scale fill operations and (ii) ensure that the rule's application to mine placement of CCRs is consistent and does not capture CCR mine placement activities that the Agency agrees should be addressed by the Office of Surface Mining (-OSM").

# A. <u>EPA Must Ensure That Environmentally Protective Beneficial Use</u> Projects Are Not Subject To Regulation As CCR Disposal Units

Both the Subtitle C and Subtitle D option would encompass CCRs that are disposed of and would not apply to CCRs that are beneficially used. To implement this regulatory distinction, EPA would include in the definition of -disposal units" uses of CCRs that EPA believes are not legitimate -beneficial uses" and instead should be regulated as CCR disposal. In attempting to establish this distinction, however, EPA uses vague undefined terms which are subject to a wide range of possible interpretations, potentially even covering within the concept of -disposal" those types of activities that the Agency believes constitute the safe beneficial use of CCRs, which it explicitly supports in the proposal's preamble.

Under the Subtitle C option, EPA would exclude from Subtitle C regulation CCRs that are beneficially used by retaining such uses within the scope of the regulatory text implementing the Bevill exclusion. See 75 Fed. Reg. at 35254 (proposed 40 C.F.R. § 261.4(b)(4)(i). EPA would define –beneficial use" to include:

The use of [Coal Combustion Products or -GCPs"] that provides a functional benefit; replaces the use of an alternative material, conserving natural resources that would otherwise need to be obtained through practices such as extraction; and meets relevant product specifications and

regulatory standards (where these are available). CCPs that are used in excess quantities, placed as fill in sand and gravel pits, or used in large scale fill projects, such as for restructuring the landscape, are not considered beneficial uses.

*Id.* (emphasis added). Uses of CCRs that do not fall within the proposed definition of -beneficial use" would be subject to full Subtitle C regulation.

While EPA does not specifically define what would constitute —beneficial use" under the Subtitle D option, the Agency attempts to draw the same distinction between legitimate beneficial use and disposal by defining what would constitute a —CCR landfill." Indeed, the proposed definition of CCR landfill is identical under both the Subtitle C and Subtitle D options, suggesting that beneficial uses are intended to be addressed the same way in the two proposals. See id. at 35240 (proposed 40 C.F.R. § 257.40(b)) and id. at 35255 (proposed 40 C.F.R. § 264.1301).

As mentioned above regarding the scope of the Subtitle D landfill regulations,

-GCR landfill" under the Subtitle C option is defined to include typical waste disposal units as well as -piles, sand and gravel pits, quarries and/or *large scale fill operations*."

Id. (emphasis added). As discussed above, USWAG believes that large-scale fill operations, if designed appropriately, constitute a legitimate beneficial use; we have long supported the use of industry standards in designing engineered structural fills and other beneficial use projects. In fact, USWAG developed an industry guidance document on how utilities and independent power producers as well as downstream coal ash users should use CCRs in engineered structural fills to minimize any potential threats to human health and the environment. See USWAG Engineering and Environmental Guidance on the Beneficial Use of Coal Combustion Products in

Engineered Structural Fill Projects (May 4, 2009) (attached as Appendix 1). USWAG believes that structural fill projects, if developed in accordance with appropriate and accepted industry standards and protections, should be considered a legitimate beneficial use and not be viewed as -disposal" under the proposal.

Notwithstanding USWAG's support of certain environmentally protective structural fill projects, the terms included in the -GCR landfill" definition, namely -piles, sand and gravel pits, quarries and/or large scale fill operations" are not defined in the regulatory text and are only loosely discussed in the proposal's preamble. The term -large scale fill operation" is especially vague, suggesting that the critical and troublesome issue for the Agency is the size of the fill rather than whether the fill project was properly designed and is environmentally protective. As addressed earlier in these comments, this undefined term can also apply to above-ground structures, road projects, and embankments. As EPA notes in the preamble, USWAG has worked with EPA, the Federal Highway Administration, DOE and the ACAA to provide guidance and best management practices on the use of CCRs in highway construction. 75 Fed. Reg. 35161. However, the undefined terms Harge scale fill operation" (and the similar phrase Harge scale fill project", which would be excluded from the proposed definition of -beneficial use" discussed above) could easily be read to cover certain highway construction projects developed and implemented consistent with this jointly-developed guidance and best management practices. USWAG sincerely hopes that it was not the Agency's intent to regulate these projects as CCR landfills based on a notion of the risks that they pose, given that EPA was at the forefront in developing guidance on how to safely use CCRs for these projects.

EPA also suggests in the proposal's preamble that it may define or provide guidance on the meaning of -a large scale fill operation" and -large scale fill project" to assist in the assessment of the appropriate criteria to -distinguish between legitimate beneficial uses and inappropriate operations." Id. at 35163. The Agency also states that it is considering promulgating regulations to address unencapsulated uses of CCRs. *Id.* at 35164-65. USWAG insists that any such attempt to address the meaning of Harge scale fill operation" and Harge scale fill project," or to issue regulations addressing un-encapsulated uses of CCRs, requires a supplemental regulatory proposal. EPA has simply not provided sufficient detail for USWAG and others to meaningfully comment on any proposed standards. In fact, for many industries and potentially impacted persons, the applicability of this proposed rule is directly related to whether and how the definition of CCR landfill relates to their facility and/or operations. Persons and facilities that may be subject to the hazardous waste or solid waste regulatory program may not even be aware that they own or operate a covered CCR landfill.

As noted above, EPA proposes to define the term —beneficial uses" under the Subtitle C option, in pertinent part, as —the use of CCPs that provides a functional benefit; replaces the use of an alternative material, conserving natural resources that would otherwise need to be obtained through practices such as extraction; and meets relevant product specifications and regulatory standards (where these are available)." *Id.* at 35254. As an alternative to these criteria, EPA requests comment on whether it should provide a formal listing of all beneficial uses. *Id.* at 35163. USWAG does not believe that EPA should attempt to specifically list all the types of qualified CCR

beneficial uses as any such list will certainly be under-inclusive. Equally important, a formal list will delay the implementation of new and emerging strategies and technologies for using CCRs, because a new rulemaking would be necessary to officially recognize any new and qualified CCR beneficial use. This would unnecessarily restrict the development of new CCR markets and stifle innovation. Rather, USWAG supports the performance criteria as proposed in the regulatory text, and believes (as described above) that structural fills and other projects – including some that may not be developed yet – can and do meet these criteria.

# B. <u>EPA Must Ensure That The CCR Regulations Do Not Apply To CCR Mine Placement</u>

In the preamble of the proposal, EPA states that the rule would not apply to —minefilling operations." *Id.* at 35165. —Minefill" is defined in the preamble to mean —a project involving the placement of CCRs in coal mine voids for use as fill, grouting, subsidence control, capping, mine sealing, and treating acid mine drainage, whether for purposes of disposal or for beneficial use, such as mine reclamation." *Id.* at 35130-31. In the Subtitle C proposed regulatory text, EPA generally includes placement in —minefilling operations" within the scope of the Bevill exclusion for beneficial uses, but does not provide a definition of this term. 75 Fed. Reg. at 35254 (proposed 40 C.F.R. § 261.4(b)(4)(i)). Additionally, nowhere in the proposed Subtitle D regulations does EPA exempt minefilling operations from coverage by the regulations. In fact, in both the Subtitle D and C proposed regulations, —GCR landfill" is defined, in part, as not including —an underground mine," suggesting that surface mines could be considered CCR landfills. *Id.* at 35240 (proposed 40 C.F.R. § 257.40(b)) and *id.* at 35255 (proposed 40 C.F.R. § 264.1301).

USWAG is concerned that EPA has not properly excluded mine placement sites from potential regulation as CCR disposal facilities. Mine placement of CCRs can include a wide variety of activities, including the placement of CCRs in underground voids; placement within a strip mine, open pit mine, or an area wide surface mine; or placement of CCRs on a site where coal materials had originally been placed. Only the first of these scenarios is contemplated in EPA's preamble definition of —minefilling operations" and only the underground mine is explicitly excluded from regulation as a CCR landfill in the proposed regulatory text.

In fact, given that the beneficial uses of CCRs in the proposals exclude the use of CCRs that are used in large-scale fill projects, including those for restructuring the landscape, any use of CCRs for surface or open pit mines may be considered an improper use subjecting the site to regulation as a CCR landfill. *Id.* at 35254 (proposed 40 C.F.R. § 261.4(b)(4)(i)). Such a result would be inconsistent with EPA's statement that CCR mine placement operations will be the subject matter of a later rulemaking by OSM. Therefore, USWAG strongly encourages EPA to provide regulatory certainty to CCRs used in mine placement by explicitly exempting CCRs used at coal mine sites from regulation under the proposal, including the use of CCRs in surface mine placement. As EPA correctly acknowledges in the preamble, the Agency should defer to OSM for development of regulations for these activities. *Id.* at 35165.

# XIV. USWAG SUPPORTS DEVELOPMENT OF STRUCTURAL INTEGRITY STANDARDS FOR CCR SURFACE IMPOUNDMENTS

USWAG concurs with EPA's proposal to incorporate the surface impoundment integrity and inspection obligations into the RCRA program modeled largely on the standards for coal slurry impoundments regulated by the Mine Safety and Health

Administration (-MSHA") (at 30 C.F.R. § 77.216). *Id.* at 35176. While the MSHA standards regulate a broader array of materials with differing physical properties than CCRs, the MSHA rules provide a good starting point for developing comparable standards for CCR impoundments.

In what appears to be a drafting error, there are no parallel threshold criteria for the corresponding obligation to conduct inspections of surface impoundment integrity under proposed 40 C.F.R. § 257.83 (*id.* at 35245-46). Rather, the regulatory text provides that these inspection requirements are to apply to -[a]II existing CCR surface impoundments," with no impoundment threshold. *Id.* at 35245. This must be an error. Even the MSHA regulations, upon which the proposed CCR structural integrity rules are based, cross-reference the volume threshold criteria applicable to MSHA dam reporting

requirements into the corresponding MSHA obligations to conduct dam integrity inspections. See 30 C.F.R. § 77.216-3 (cross-referencing the applicability criteria set forth in 30 C.F.R. § 77.216(a), which set forth the same dam elevation and impoundment capacity criteria referenced above in proposed 40 C.F.R. § 257.71(b)(1)-(2)).

Given the above, there is no basis to impose the proposed inspection obligations on impoundments of any size. Therefore, EPA should be certain in the final rule that the same set of minimum criteria that trigger the CCR dam record submission requirements also apply to the obligations to conduct CCR dam inspections to ensure that the universe of impoundments subject to these related obligations is the same (*i.e.*, the specified dam elevation and impounding capacity in proposed 40 C.F.R. § 257.71(b)(1)-(2)).

USWAG also believes that the CCR structural integrity requirements (both the recording and inspection obligations under proposed 40 C.F.R. §§ 257.71(b), 257.83) should be tailored to apply only to those structures that pose a potential threat to human health and the environment. While USWAG appreciates that EPA has modeled the proposed structural integrity standards for impoundments on the MSHA rules, those rules, as noted above, encompass a broader universe of materials that pose a broader array of potential dam integrity issues. In the case of CCRs, the Agency is developing impoundment integrity rules for a group of materials that are more homogenous in nature and thus do not present the same array of potential dam integrity issues that the MSHA rules are designed to regulate.

USWAG recommends, therefore, that the proposed CCR structural integrity requirements not apply to CCR impoundments that have been classified by the appropriate state regulatory body or by an independent third party professional engineer as having a How hazard potential classification" pursuant to the federal guidelines for dam safety. See —Federal Guidelines for Dam Safety: Hazard Potential Classification for Dams," Federal Emergency Management Agency (-FEMA") (reprinted Jan. 2004). Under the FEMA dam safety classification system, a -low hazard potential classification" means that failure or mis-operation of the impoundment -results in no probable loss of human life and low economic and/or environmental losses. Losses are principally limited to owner's property." Id. at 5. Given that a failure of a CCR impoundment with a How hazard potential classification" poses only a low risk for *on*-site economic or environmental losses, USWAG believes that impoundments with this classification should not be subject to the proposed CCR impoundment structural integrity requirements. Despite the fact that any such risks are low, facilities will have ample incentives to prevent economic or environmental losses to on-site operations, and therefore it is not necessary to subject these low-risk facilities to any additional regulatory requirements under the final CCR rule.

Moreover, excluding CCR impoundments with a -lew hazard potential classification" from the CCR structural integrity requirements would be consistent with many state dam regulatory programs that apply dam integrity standards *only* to impoundments with a -high" or -significant" potential hazard classification, and therefore will promote consistency with existing state controls. See e.g., New Mexico Rules and Regulations Governing Dam Design, Construction and Dam Safety (e.g., requiring dam

site security, an instrumentation plan for monitoring and evaluating dam performance, and an operation and maintenance manual and emergency action plan only for dams with a high or significant hazard potential); see also NMAC §§ 19.25.12.11(G)-(J).

Under the FEMA dam safety classification system, dams with a High hazard potential classification" means that failure or mis-operation of the impoundment will probably cause a loss of human life." *Id.* at 6. Dams with a significant hazard potential classification means that failure or mis-operation of the impoundment will result[] in no probable loss of human life but can cause economic loss, environmental damage, disruption of lifeline facilities, or can impact other concerns." *Id.* at 5. USWAG agrees that CCR impoundments with either a high hazard or significant hazard potential classification, that also meet the threshold impounding criteria discussed above, should be subject to the CCR dam integrity standards. At the same time, however, because CCR impoundments with a lew hazard potential classification pose only a low economic and/or environmental threat that is limited to the facility itself, USWAG does not believe these types of units warrant regulation under the CCR impoundment structural integrity rules.

In addition to the above, USWAG believes that the requirement to monitor instrumentation once every seven days under proposed 40 C.F.R. § 257.83(a)(2) (75 Fed. Reg. at 35245) is excessive. A weekly instrumentation inspection obligation is clearly unwarranted for all dams. Even the FEMA Federal Guidelines for Dam Safety do not contemplate a monitoring frequency of this magnitude, suggesting that the frequency of instrumentation monitoring —may be reduced after the project has been in operation for an extended time and performance observation data indicates that

readings have stabilized." Section III(D)2c(2) of the FEMA Dam Safety Guidelines regarding the frequency of instrument monitoring provide, in relevant part:

The frequency of instrument readings should be established at the time the instrumentation system is designed in order to give a timely warning of possible adverse conditions. Whenever necessary, more frequent readings, sometimes as often as hourly, should be taken to monitor a suspected rapidly changing adverse condition. The frequency or number of readings may be reduced after the project has been in operation for an extended time and performance observation data indicates that readings have stabilized.

The FEMA guidelines appropriately recognize that there is not a one-size-fits-all monitoring frequency for instrumentation, but rather that such an obligation should be tailored to reflect the stage of the dam (e.g., construction, first filling, long-term), its function, and its hazard rating. Therefore, USWAG recommends that EPA amend this condition to require equipment inspections once a month for dams with a dam hazard potential rating of -high" and quarterly for impoundments with a dam hazard potential rating of -significant."

Finally, USWAG believes that it is extremely unwise from the perspective of homeland security concerns to require internet posting of information regarding an impoundment's construction history that must be compiled under proposed 40 C.F.R. § 257.71(d). This obligation would arise under the proposed Subtitle D option, where the Agency would require owners/operators of CCR facilities to, among other things, maintain a public website containing documentation showing that applicable Subtitle D requirements have been met. See 75 Fed. Reg. at 35246 (proposed 40 C.F.R. § 257.84(a)). EPA itself recognizes that this obligation has -homeland security implications" and therefore requests comments on whether this obligation should not be required for certain data. *Id.* at 35194.

The information required to be assembled under proposed 40 C.F.R. § 257.71(d) can be extremely sensitive and may contain information that could be used by certain individuals with an intent to destroy a dam (*e.g.*, engineering information on the structure's foundation, detailed information on physical and engineering properties, the basis for the structure hazard classification, slope stability information, etc.). Given the uncensored and anonymous nature of the internet and EPA's insistence that these data be posted on a publicly available website, USWAG is concerned that information made publicly available could be used to deliberately attempt to cause structural failure. This type of information should not be available on the internet.

There are certainly more secure means to make this information available to qualified individuals who have a legitimate interest in ensuring that a facility is meeting its regulatory obligations with respect to any applicable structural integrity obligations. For example, this information would also be required to be included in the owner's/operator's operating record. See 75 Fed. Reg. at 35246 (proposed 40 C.F.R. § 257.84(a)). Persons with legitimate interests in reviewing these data could make a written request to the owner/operator. Prior to disclosing this information to the requesting party, the owner/operator could coordinate with proper federal and state authorities to ensure that the requesting parties are not on any list of interest to federal/state homeland security agencies. Only after this screening measure has been completed would the information be released. This condition would also allow the owner/operator of the facility and federal/state authorities to know the names and identities of all organizations requesting this sensitive information on the construction of

the impoundment. This additional measure would not be overly burdensome and would help protect against the misuse of these data.

\* \* \* \* \*

# **CONCLUSION**

USWAG supports the development of federal regulations for CCRs under RCRA's Subtitle D non-hazardous waste program. Such an approach, which is the only lawful option available to EPA, would be consistent with EPA's two Reports to Congress and final Regulatory Determinations under the Bevill Amendment concluding that CCRs do not warrant regulation under RCRA Subtitle C.

USWAG also believes that the Subtitle D Prime option, with appropriate adjustments, offers the best path forward. USWAG looks forward to working with EPA as the Agency develops Subtitle D non-hazardous waste regulations implementing the Subtitle D Prime option. Please direct your inquiries to James Roewer, USWAG Executive Director (202-508-5645), or to Douglas Green, USWAG counsel (202-344-4483).

Comments of
The Utility Solid Waste Activities Group,
The Edison Electric Institute,
The American Public Power Association, and
The National Rural Electric Cooperative Association on
Hazardous and Solid Waste Management System;
Identification and Listing of
Special Wastes; Disposal of Coal Combustion Residuals From
Electric Utilities;
Proposed Rule

75 Fed. Reg. 35128 (June 21, 2010)

Volume II Appendices

submitted to
The United States
Environmental Protection Agency
Docket No. EPA-HQ-RCRA-2009-0640

November 19, 2010

Of Counsel: Venable LLP 575 Seventh Street, N.W. Washington, D.C. 20004

# **Table of Contents - Appendices**

| Letter from USWAG to EPA transmitting USWAG's Engineering and Environmental Guidance on the Beneficial Use of Coal Combustion Products in Engineered Structural Fill Projects (May 4, 2009)   | Appendix 1  |
|---|-------------|
| Administrative Order and Agreement on Consent, In the Matter of TVA Kingston Fossil Fuel Plant Release Site, Roane County, Tennessee (May 6, 2009)  | Appendix 2  |
| Pennsylvania DEP Testimony to the U.S. Environmental Protection Agency—Hazardous Waste Management System; Identification and Listing of Special Wastes; Disposal of Coal Combustion Residuals from Electric Utilities (September 21, 2010)                    | Appendix 3  |
| Letter from Joyce K. Frank, EPA Acting Associate Administrator, to Rep. Markey, Chair, House Committee on Energy and Commerce (January 30, 2009)  | Appendix 4  |
| Statement of Thomas H. Adams, Executive Director, American Coal Ash Association before the House Small Business Subcommittee on Rural Development, Entrepreneurship and Trade, "Is Coal Combustion Product Recycling an Endangered Industry?" (July 22, 2010) | Appendix 5  |
| Testimony of Craig H. Benson, PhD, PE, DGE, Wisconsin Distinguished Professor, University of Wisconsin-Madison before the House Small Business Subcommittee on Rural Development, Entrepreneurship and Trade (July 22, 2010)                                  | Appendix 6  |
| Testimony of William H. Gehrmann, President, Headwaters Resources, Inc. before the House Small Business Subcommittee on Rural Development, Entrepreneurship and Trade (July 22, 2010)   | Appendix 7  |
| Letter from Alan Schmitz, President of Schmitz Ready Mix, Inc. to Lisa Jackson, EPA Administrator (2009)  | Appendix 8  |
| Letter from American Concrete Institute to Lisa Jackson, EPA Administrator (September 4, 2009)  | Appendix 9  |
| Written Testimony of Richard D. Stehly, President, American Concrete Institute, submitted to the House Committee on Small Business, Subcommittee on Rural Development, Entrepreneurship and Trade (July 22, 2010)   | Appendix 10 |
| Letter from James A. Thomas, President of ASTM to Lisa Jackson, EPA Administrator (December 23, 2009)   | Appendix 11 |
| Letter from ASTSWMO to EPA (April 1, 2009)  | Appendix 12 |
| Letter from ASTSWMO to EPA (November 4, 2009)   | Appendix 13 |

# **Table of Contents - Appendices**

| Letter from Wisconsin Department of Natural Resources to EPA (March 16, 2009)  | Appendix 14 |
|--|-------------|
| Letter from AASHTO to Lisa Jackson, EPA Administrator (November 23, 2009)  | Appendix 15 |
| Letter from Indiana Department of Transportation to Lisa Jackson, EPA Administrator (September 9, 2009).   | Appendix 16 |
| Letter from Michigan Department of Transportation to Victor Mendez, Federal Highway Administration (November 10, 2009).  | Appendix 17 |
| Testimony of Craig Shamory, Environmental Manager of PPL, at EPA Hearing on Proposed CCR Regulation in Pittsburgh, PA (August 30, 2010)  | Appendix 18 |
| EOP, Group, Inc. "Cost Estimates for the Mandatory Closure of Surface Impoundments Used for the Management of Coal Combustion Byproducts at Coal-Fired Electric Utilities" (November 11, 2010)   | Appendix 19 |
| AECOM Report: Comments on USEPA's Proposed Rulemaking for Disposal of Coal Combustion Waste from Electric Utilities: Consideration of the Individual Listing Criteria and the April 2010 Coal Combustion Waste Risk Assessment (November 2010)         | Appendix 20 |
| Letter from National Governors Association to EPA (November 16, 2009)  | Appendix 21 |
| Western Governors Association, Policy Resolution 10-1  | Appendix 22 |
| ECOS, Revised Resolution No. 08-14, Regulation of Coal Combustion Products (March 23, 2010).   | Appendix 23 |
| Commonwealth of Pennsylvania Department of Environmental Protection<br>Southwest Region, Review of the Environmental Integrity Project and<br>Earthjustice Report: "Out of Control: Mounting Damages From Coal Ash Waste<br>Sites," (October 13, 2010) | Appendix 24 |
| EOP RIA Report, "How Armed Robbery Improves Social Welfare: An evaluation of the Environmental Protection Agency's analysis of the benefits of regulating coal combustion residuals" (November 16, 2010)   | Appendix 25 |
| Letter from Reps. Upton, Boucher, Space and Rogers to Lisa Jackson, EPA Administrator (April 27, 2010)   | Appendix 26 |
| EPA Response to Upton Letter (July 15, 2010)   | Appendix 27 |
| Letter from Kansas Department of Health and Environment to EPA (September 21, 2009)  | Appendix 28 |
|  |             |

Duke Energy Progress, LLC Late-Filed Exhibit No. 16 Docket No. E-2, Sub 1219 Page 282 of 439

# **Table of Contents - Appendices**

| Letter from Florida Department of Environmental Protection to EPA (April 27, 2009)    | Appendix 29 |
|---|-------------|
| Letter from ECOS to Mathy Stanislaus, Assistant Administrator, EPA (October 15, 2009) | Appendix 30 |
| Jim Roewer, Comments of the Utility Solid Waste Activities Group (September 2, 2010)  | Appendix 31 |
| Letter from APPA and NRECA to Lisa Jackson, EPA Administrator (November 3, 2009)      | Appendix 32 |

Duke Energy Progress, LLC Late-Filed Exhibit No. 16 Docket No. E-2, Sub 1219 Page 283 of 439

# Appendix 1

## **Utility Solid Waste Activities Group**

c/o Edison Electric Institute 701 Pennsylvania Avenue, NW Washington, DC 20004-2696 202-508-5645 www.uswag.org



May 4, 2009

# VIA HAND DELIVERY AND ELECTRONIC MAIL

Mr. Matthew Hale
Director, Office of Resource Conservation and Recovery
United States Environmental Protection Agency
Ariel Rios Building
1200 Pennsylvania Avenue, NW
Mail Code: 5301P
Washington, D.C. 20460

RE: USWAG's Guidance on Engineered Structural Fill Projects

Dear Mr. Hale:

In meetings last Fall and earlier this year, you expressed concern about the use of coal combustion products ("CCPs") in certain structural fill applications. In response to your concerns, USWAG¹ developed the attached guidance document to assist utilities and other users of CCPs in planning and implementing engineered structural fill projects in an environmentally protective manner. The document, the "Engineering and Environmental Guidance on the Beneficial Use of Coal Combustion Products in Engineered Structural Fill Projects," relies on recognized industry standards, including ASTM E2277-03 ("The Standard Guide for Design and Construction of Coal Ash Structural Fills") in laying out key considerations for the use of CCPs in structural fills. The guidance includes provisions on (i) the testing and characterization of the CCPs used in the structural fill, (ii) the suitability of the intended project and site, (iii) communications with regulatory agencies and the affected community, (iv) standards for project operations, (v) contractor requirements, (vi) transportation standards, and (vii) measures to respond to potential releases.

<sup>&</sup>lt;sup>1</sup> USWAG, formed in 1978, is an association of approximately 80 energy industry operating companies and associations including the Edison Electric Institute ("EEI"), and the National Rural Electric Cooperative Association ("NRECA"). EEI is the principal national association of investor-owned electric power and light companies. NRECA is the national association of rural electric cooperatives. Together, USWAG members represent more than 85% of the total electric generating capacity of the U.S. and service more than 95% of the nation's consumers of electricity.

Mr. Matt Hale May 4, 2009 Page 2

The development and promotion of the standards contained in the attached guidance will help ensure that engineered structural fill projects are performed in an environmentally responsible manner. The continued use of CCPs in properly managed beneficial use applications including engineered structural fills is critically important to the recycling and conservation opportunities for the electric utility industry.

Please contact me if you have any questions or need additional information about this guidance or USWAG's continued commitment environmentally-protective beneficial use applications.

Sincerely,

Jim Roewer, Executive Director Utility Solid Waste Activities Group

Enclosure

# ENGINEERING AND ENVIRONMENTAL GUIDANCE ON THE BENEFICIAL USE OF COAL COMBUSTION PRODUCTS IN ENGINEERED STRUCTURAL FILL PROJECTS

#### **Purpose Statement:**

This document is intended to provide Coal Combustion Product (CCP) Managers with information to ensure that CCP engineered structural fills are planned and implemented in a manner that protects the environment and meets regulatory requirements.

#### Scope:

This information is intended to be used in planning and implementing engineered structural fill projects associated with the beneficial use of "coal combustion products" including fly ash, bottom ash, boiler slag, or flue gas desulfurization (FGD) material produced primarily from the combustion of coal. Engineered structural fill projects consist of either an engineered fill or an engineered embankment with a beneficial end use. Engineered structural fills are typically constructed in layers of uniform thickness and, where appropriate, compacted to a desired unit of density in a manner to control the compressibility, strength, and/or hydraulic conductivity of the placed material.

The generic issues and examples in this Guidance are designed solely to be an educational resource for CCP Managers by illustrating the requirements and considerations that may apply to engineered structural fill projects and should be taken into consideration depending on the quantities of ash used at the site. The Standard Guide for Design and Construction of Coal Ash Structural Fills designated ASTM E2277-03 is incorporated herein by reference. Each CCP project should be evaluated for site-specific requirements including all permitting requirements and the project should be appropriately described.

#### Characterization/Testing:

CCPs should be adequately characterized to ensure that the proposed use of the material does not cause environmental or public health problems and has physical properties that are compatible with the end use. The characterization, at minimum, should demonstrate the physical characteristics, including geotechnical properties, and identify the primary chemical constituents of the CCPs and leaching potential. CCPs should also be sampled and analyzed to verify that the materials meet the applicable regulatory standards for beneficial use. CCP leach testing procedures should be selected based on the material used, hydrogeologic setting, and structural fill application. See Appendix 1 for a reference to guidelines for potentially applicable leaching tests. For assessing engineering properties of the CCP, refer to ASTM E-2777.

<sup>&</sup>lt;sup>1</sup> Fly ash and bottom ash includes both materials produced by "conventional units" as well as fluidized bed boilers.

<sup>2</sup> Examples of engineered structural fill projects include use as fill for building sites and foundations, subbase fill for recreational facilities, and as embankments for highways, railroads, dikes and levees.

Samples should be representative of the CCPs supplied for the engineered structural fill. See Appendix 2 for a reference to guidelines for collecting a representative sample. The frequency at which the CCPs should be tested is dependent on the variability of the composition of the CCPs.<sup>3</sup>

Moisture content of the CCPs should be evaluated in order to determine whether additional moisture is needed to condition the ash to help control fugitive dust and to enhance compaction.

### Project Suitability/Qualification:

In addition to meeting all applicable state standards and guidelines, identifying a suitable site for an engineered structural fill using CCPs should include an evaluation of the site commensurate with the leaching potential of the CCPs, considering such factors as local zoning and land use plans, environmental characteristics of the proposed site, engineering aspects, and proximity to and relationship with landowners and neighbors. The historic use of the potential site and adjacent properties should be evaluated to assess whether there is potential for existing contamination at the site. In addition, consideration should be given to the reasonable future uses or planned development of the site and surrounding properties, including the distance between the potential site and adjacent properties.

The soil, geology and groundwater of the proposed site should be evaluated. The site geology should be assessed for the presence of any conditions such as sinkholes, karst, or bedrock outcrops. Groundwater quality and quantity, the location of groundwater users, groundwater flow direction, and depth to the groundwater will be important considerations in the process. The evaluation should be done in a manner consistent with ASTM E 2277-03. *See* Section 7.2 Site Characterization.

Groundwater modeling is not typically necessary for hydrogeologic characterization in support of a structural fill plan. However, in some cases, particularly larger fills, modeling may be useful in evaluating the potential for groundwater impacts. Screening models such as EPA's IWEM and EPRI's MYGRT are typically sufficient given the nature of the sites and limited availability of hydrogeologic data. Numerical models such as MODFLOW/MT3D are generally not warranted for these applications, except for very large sites in complex hydrogeologic environments. Brief descriptions of these models are provided in Appendix 3.

The site should also be evaluated in relationship to its distance from surface waters, including wetlands, and the location of floodplains and other environmental regulatory constraints that could potentially impact the project.

While site specific factors including size, rainfall, climate and terrain should always be considered, other considerations involving the location of the proposed engineered structural fill project include whether the site is:

- Within floodplain, floodways and drainage areas,
- In or near a wetland area,
- Near an active fault,

<sup>&</sup>lt;sup>3</sup> Additional testing should be conducted when conditions change that could significantly impact constituent concentrations in the CPP such as, for example, through the use boiler additives and/or changes in coal source or type.

- In an unstable area (e.g., in a karst area or areas prone to sinkholes),
- Within a wellhead protection area,
- Near a drinking water well, or near a public water supply, such as a community well, reservoir, or water treatment facility,
- Near a surface-water body, such as a lake, stream, river, or pond, or
- An impact on a critical habitat for an endangered or threatened species or an historic or archaeological site.

The landowner(s) should provide written permission for using CCPs on the site. As appropriate, the presence of CCPs should be acknowledged on the deed to provide notice to subsequent purchasers of the site.

Current site conditions and engineering and economic considerations, as well as the intended future use of the property, should be analyzed to ensure the material can be placed in a manner that will achieve needed structural stability.

The ability of equipment to move on and around the site should be evaluated especially if a project will extend over months in which inclement weather is expected. On and offsite traffic limitations should be considered.

## **Agency Communications:**

The state or local government agency responsible for administering the program for beneficial use of CCPs will vary depending on the nature of the beneficial use. Site requirements for structural fill projects will also vary by locality. Where appropriate, details of the proposed structural fill should be discussed with the appropriate agency for their input.

These steps may be useful during and after any project.

- 1. Draft a plan for the proposed engineered structural fill and identify the agency responsible for approving the structural fill if approval is required.
- 2. Contact the appropriate agency staff to discuss the proposed engineered structural fill project when the detailed plan is complete and approval is required.
- 3. Provide the lead agency and other regulatory bodies with information they require to understand the project.
- 4. Follow-up to ensure the project is meeting State and local government and community requirements. Maintain an active dialog with all parties and document all material discussions.

## **Community Outreach:**

As appropriate, CCP Managers should contact local community officials and offer to meet with them to provide information and answer questions that they may have about the proposed project.

After the operations commence, the CCP Manager should remain in contact with the local officials and offer to meet with them periodically to provide a project update and to answer questions that may be raised.

Local officials should be provided with a contact name, address and telephone number that they may use to inquire about the ongoing project.

## **Contractor Performance Standards:**

CCP Managers should consider the reliability, performance history, compliance and safety records of any potential contractor. The contractor selected should have proven technical capabilities and experience with CCP placement. The potential contractor's financial resources and financial stability should be assessed prior to signing a contract, to ensure that the contractor has the ability to fulfill the obligations of the contract.

The contract should include conditions requiring the contractor to comply with all local, state and federal highway regulations during hauling of the CCP, and all local, state and federal environmental and occupational safety and health regulations during all of its operations. It should also address how the contractor's performance will be assessed. Oversight of contractor performance should be ongoing.

## Transportation on Public Roadways:

Possible community concerns associated with transportation routes should be identified, reviewed and properly addressed prior to project implementation. Access routes to the project site should consider:

- Impact on local communities.
- Road conditions.
- Timing of transportation.
- Volume of traffic.
- Noise.
- Vehicle operational condition and cleanliness (free of dust and dirt).

Drivers or other personnel should ensure that vehicles are loaded and unloaded in a manner that minimizes the generation of dust and should ensure that all trucks are covered while in transit. Vehicles must be operated and maintained in a safe working order and in compliance with all applicable federal, state, and local requirements. They should not negatively impact road conditions or safety, especially at project ingress and egress locations.

For examples, see Appendix 3.

## **Project Operation Controls and Security:**

Site Access Controls – To the extent possible, access controls should be utilized at the placement site to discourage unauthorized entrance to the site. Controls should be implemented to ensure appropriate pickup and delivery of CCPs.

The construction of engineered structural fills should follow the procedures described in Sections 7.7.6 & 8 "Construction" of the ASTM E 2277-03 Standard Guide for Design and Construction of Coal Ash Structural Fills. The ASTM standard includes operating procedures for weather restrictions, dust control, erosion control, source and delivery, on-site storage, site preparation, placement and compaction, cover and quality control. These standards are incorporated herein by reference.

## **Spill/Off-Site Release Response:**

Trucks hauling CCPs on public roadways should have a Material Safety Data Sheet and should be responsible for responding to spills. All CCPs spilled during transit should be cleaned up and either taken to the placement site or returned to the plant of origin.

A plan should be developed for potential off-site releases. The plan should identify the individual who is responsible for implementing response actions and notifying appropriate regulatory officials, contact information for response assistance, procedures to follow in the event of an offsite release, and an inventory of response equipment/services. In the event of an off-site release, all discharges of CCPs should be cleaned up and returned to the placement site.

## **APPENDIX 1**

## **Characterization/Testing:**

Examples of leaching procedures:

- EPA SW846 Method 1313 1316,
  - Methods 1313 through 1316 are a recently developed set of procedures designed to fully evaluate leaching potential under a wide range of potential environmental conditions, sometimes collectively referred to as the Leaching Framework. The framework includes batch leaching as a function of pH (1313), column leaching (1314), mass transfer leaching for monolithic materials (1315), and batch leaching as a function of solid-liquid ratio (1316). Selection of tests depends on the CCP material and the expected range of environmental conditions associated with the structural fill application.
- EPA SW846 Method 1312, Synthetic Precipitation Leaching Procedure (SPLP)
  The SPLP is a single point batch leaching test similar to the TCLP but uses a nitric and sulfuric acid leachant to simulate acid rain. SPLP is commonly used to assess infiltration and leaching at CCP sites.
- ASTM D3987, Shake Extraction of Solid Waste with Water
   ASTM D3987 is a single point batch leaching test similar to the SPLP but uses a deionized water leachant. ASTM D3987 is commonly used to assess infiltration and leaching at CCP sites.

## **APPENDIX 2**

## **Collecting Representative Samples:**

Examples of reference guidelines for collecting a representative sample.

- EPA SW846 Test Methods for Evaluating Solid Waste, Physical/Chemical Methods http://www.epa.gov/epaoswer/hazwaste/test/sw846.htm
- ASTM D75 Practice for Sampling Aggregates
- ASTM C311 Test Methods for Sampling and Testing Fly Ash or Natural Pozzolans for Use as a Mineral Admixture in Portland-Cement Concrete.

## **APPENDIX 3**

#### **Site Modeling Tools**

• Industrial Waste Management Evaluation Tool ("IWEM")

IWEM (USEPA, 2002) was developed by EPA as a screening tool to evaluate industrial waste disposal and use applications. IWEM is easy to use and can be run with very little site-specific hydrogeologic data. The core model in IWEM is the EPACMTP (USEPA, 2003) analytical model, which simulates the migration of dissolved constituents through soil and groundwater. The model accepts input values describing geology and infiltration at the site, and then performs a Monte Carlo analysis using a range of values for attenuation parameters and hydrogeologic parameters, producing over 10,000 realizations. For each of the Monte Carlo realization, IWEM allows up to 10,000 years to reach a maximum concentration. Output for individual realizations are not provided. A facility design is deemed "protective" if 90 percent of the IWEM realizations predict maximum concentrations at the compliance point that are lower than the appropriate health-based number or water quality standard.

#### MYGRT

MYGRT Version 3.1 (EPRI, 2005) is a collection of 22 analytical models that calculate the transport of organic or inorganic constituents in the unsaturated and saturated zone, accounting for the processes of advection, dispersion, retardation, and decay. All of these models are based on analytical solutions to the 1D, 2D, and 3D mass transport equations. Inputs are specified by the user, and output is provided for each set of input parameters

provided. The model will accept time-varying leachate concentrations, and can be run for any length time period. MYGRT is easy to use and requires minimal or estimated site-specific data; it is most appropriate for screening level applications and relatively simple hydrogeologic settings.

## MODFLOW/MT3D

MODFLOW is a numerical groundwater flow code developed by the United States Geological Survey (McDonald and Harbaugh, 1988) that uses a finite difference approximation to solve a three-dimensional head distribution in a multi-layered aquifer. The program also calculates water balance at wells, rivers, and drains. MT3DMS (Zheng and Wang, 1998) is a transport code that uses a finite difference solution to calculate concentration distribution for a single dissolved solute as a function of time and space. MT3DMS accounts for advection, dispersion, diffusion, first-order decay, and sorption. Sorption can be calculated using linear, Freundlich, or Langmuir isotherms. MODFLOW/MT3D are relatively time consuming codes to set up and calibrate, and are most applicable for large sites with complex and well characterized hydrogeology.

#### **APPENDIX 4**

## Transportation on Public Roadways:

Examples to consider:

- 1. Ensure vehicles, company and contractor, are in safe working order and associated inspections are up-to-date.
- 2. Ensure contractors are adequately insured.
- 3. Investigate compliance history of contractors.
- 4. Ensure proper vehicle washing at loading and unloading areas.
- 5. Proper vehicle covering:
  - The cover must be made of a solid or close weave mesh material.
  - The cover must be large enough to enclose the top of the entire load.
  - The cover must be sufficiently secured.
  - The cover must be kept in good repair
- 6. Ensure proper construction and maintenance of access and egress points to public roadways. This could include construction of temporary roadways and use of flagmen or other traffic flow devices.
- 7. Evaluate whether water must be added to condition the ash to help control fugitive dust.

## References

USEPA, 2002. Industrial Waste Management Evaluation Model (IWEM) Technical Background Document. Office of Solid Waste and Emergency Response, EPA530-R-02-012.

USEPA, 2003. *EPACMTP Technical Background Document*. Office of Solid Waste and Emergency Response, EPA530-R-03-006.

McDonald, M.G. and A.W. Harbaugh, 1988, A Modular Three-Dimensional Finite-Difference Ground-Water Flow Model: Techniques of Water-Resources Investigations, Techniques of Water-Resources of the United States Geological Survey, Book 6, Chapter A1.

Zheng, C., and P.P. Wang, 1998, MT3DMS, A Modular Three-Dimensional Multispecies Transport Model, Model documentation and user's guide prepared by the University of Alabama Hydrogeology Group for the U.S. Army Corps of Engineers.

EPRI, 2005. MYGRT Version 3.1. EPRI, Palo Alto, CA: 2005. 1010066

Duke Energy Progress, LLC Late-Filed Exhibit No. 16 Docket No. E-2, Sub 1219 Page 294 of 439

# Appendix 2

## UNITED STATES ENVIRONMENTAL PROTECTION AGENCY REGION 4

| IN THE MATTER OF                               | ) Administrative Order and ) Agreement on Consent  |
|--|--|
| TVA Kingston Fossil Fuel Plant<br>Release Site | <ul> <li>U.S. EPA Region 4 and</li> <li>Tennessee Valley Authority</li> </ul>            |
| Roane County, Tennessee                        | Docket No.: CERCLA-04-2009-3766 Proceeding Under Sections 104(a), 106(a), and 107 of the |
| Tennessee Valley Authority,                    | ) Comprehensive Environmental ) Response, Compensation,                                  |
| Respondent.                                    | ) and Liability Act, as amended )  |

## ADMINISTRATIVE ORDER AND AGREEMENT ON CONSENT

## I. PURPOSE

WHEREAS, on December 22, 2008, approximately 5.4 million cubic yards of ash material were released into the environment from the Tennessee Valley Authority (TVA) Kingston Fossil Fuel Plant in Roane County, Tennessee; and

WHEREAS, TVA undertook immediate response actions and worked in close coordination with the United States Environmental Protection Agency (EPA), the Tennessee Department of Environment and Conservation (TDEC), and other agencies to provide for the safety of area residents, to contain released ash and minimize its downstream migration, and to monitor and assess air and water quality; and

WHEREAS, on January 12, 2009, TDEC issued a Commissioner's Order to TVA requiring, among other things, the comprehensive assessment, cleanup and restoration of areas impacted by the release; and

WHEREAS, on February 4, 2009, EPA, pursuant to Executive Order 12088, and TDEC issued a joint letter to TVA wherein TVA was directed to provide all plans, reports, work proposals and other submittals being provided to TDEC, to EPA as well for review and approval by the agencies; and

WHEREAS, TVA is, and has been, responding to the release of ash from an impoundment at the TVA Kingston Fossil Fuel Plant in Roane County, Tennessee, pursuant to the January 12, 2009, Commissioner's Order, and the February 4, 2009, letter, with the oversight of TDEC and EPA and under its authorities, including that delegated by Executive Order No. 12580; and

WHEREAS, TVA is committed to cleaning up the release, protecting the health and safety of the public and workers, protecting and restoring environmentally sensitive areas, and keeping the public and stakeholders informed and involved in the formulation of response activities; and

WHEREAS, TVA recognizes EPA's specialized expertise in responding to large-scale releases; and

WHEREAS, TVA, EPA, and TDEC desire to work cooperatively in all aspects of the cleanup; and

WHEREAS, TVA and EPA agree that in order to ensure that the environmental impacts associated with the release are thoroughly assessed and that appropriate response actions are taken as necessary to protect public health, welfare or the environment, and to ensure that the response actions satisfy all federal as well as state environmental requirements, it is advisable and beneficial for TVA and EPA to enter into this Administrative Order and Agreement on Consent (Order, or Order and Agreement) providing for cooperative implementation of the response actions at the Site pursuant to their authorities under the Comprehensive Environment Response, Compensation, and Liability Act (CERCLA), as amended, and the National Contingency Plan (NCP);

NOW, THEREFORE, EPA hereby orders and TVA agrees as follows:

#### II. JURISDICTION

- 1. This Order is issued by EPA pursuant to the authority vested in the President of the United States by Sections 106(a) and 107 of CERCLA, 42 U.S.C. §§ 9606(a) and 9607, and delegated to the Administrator of EPA by Executive Order 12580, 52 Fed. Reg. 2923 (Jan. 23, 1987), and within EPA further delegated to the Regional Administrators by EPA Delegation Nos. 14-14-A and 14-14-B, and re-delegated to the Region 4 Superfund Division Director by Regional Delegations 14-14-A and 14-14-B.
- 2. This Order and Agreement is entered into by TVA pursuant to the authority vested in the President of the United States by Section 104(a) of CERCLA, 42 U.S.C. § 9604(a), and delegated to TVA by Executive Order 12580, 52 Fed Reg. 2923 (Jan. 23, 1987), and pursuant to the authority of the Tennessee Valley Authority Act of 1933, as amended, 16 U.S.C. §§ 831-831ee (2006).
- 3. This Order pertains to the release of ash from the TVA Kingston Fossil Fuel Plant located in Roane County, Tennessee, to the surrounding environment, including

portions of the Watts Bar Reservoir. This Order requires TVA, and TVA agrees, to conduct response actions, described herein, to abate any imminent and substantial endangerment to public health, welfare or the environment that may be presented by the actual or threatened release of hazardous substances, pollutants or contaminants and to otherwise address the impacts of the release at or from the Site in accordance with CERCLA and the NCP.

- 4. EPA has notified the State of Tennessee of this action pursuant to Section 106(a) of CERCLA, 42 U.S.C. § 9606(a).
- 5. EPA and TVA agree that this Order and Agreement has been negotiated in good faith and that the actions undertaken by TVA in accordance with this Order do not constitute an admission of any liability. TVA retains the right to controvert in any proceedings, other than a proceeding solely to enforce this Order, the validity of any of the Findings of Fact or Conclusions of Law and Determinations in Sections V and VI of this Order. TVA agrees to comply with and be bound by the terms of this Order, and further agrees in any proceeding solely to implement or enforce this Order that it will not contest the validity of this Order, its terms, or the jurisdiction of EPA to issue it.

## III. PARTIES BOUND

- 6. This Order applies to and is binding upon EPA and upon TVA and TVA's directors, officers, employees, agents, successor agencies or departments, and assigns. Any transfer of the assets or real property of the United States in TVA's custody and control shall not alter TVA's responsibilities under this Order.
- 7. TVA shall ensure that its employees, contractors, subcontractors and representatives receive a copy of this Order and comply with this Order. TVA shall be responsible for any noncompliance with this Order.

## IV. DEFINITIONS

- 8. Unless otherwise expressly provided herein, terms used in this Order which are defined in CERCLA, or in regulations promulgated under CERCLA, shall have the meaning assigned to them in CERCLA or in such regulations. Whenever terms listed below are used in this Order, the following definitions shall apply:
- a. "CERCLA" shall mean the Comprehensive Environmental Response, Compensation, and Liability Act of 1980, as amended, 42 U.S.C. §§ 9601 et seq.
- b. "Corrective Action Plan" or "CAP" shall mean the Corrective Action Plan, as required by the TDEC Commissioner's Order.
  - c. "Day" shall mean a calendar day unless expressly stated to be a working day.

- d. "Department" or "TDEC" shall mean the Tennessee Department of Environment and Conservation.
- e. "Effective Date" shall be the effective date of this Order, as provided in Section XXXIX herein.
- f. "EPA" shall mean the United States Environmental Protection Agency and any successor departments or agencies of the United States.
- g. "Future Response Costs" shall mean all direct and indirect costs incurred at or in connection with the Site after the Effective Date of this Order, that are not inconsistent with the NCP, that EPA incurs in reviewing or developing plans, reports and other items pursuant to this Order, verifying the Work, providing technical assistance, or otherwise implementing, overseeing, or enforcing this Order, including, but not limited to, payroll costs, contractor costs, travel costs, and laboratory costs.
- h. "Interest" shall mean interest at the rate specified for interest on investments of the EPA Hazardous Substance Superfund established by 26 U.S.C. § 9507, compounded annually on October 1 of each year, in accordance with 42 U.S.C. § 9607(a). The applicable rate of interest shall be the rate in effect at the time the interest accrues. The rate of interest is subject to change on October 1 of each year.
- i. "National Contingency Plan" or "NCP" shall mean the National Oil and Hazardous Substances Pollution Contingency Plan promulgated pursuant to 42 U.S.C. § 9605, codified at 40 C.F.R. Part 300, and any amendments thereto.
- j. "NPDES Permit" shall mean the National Pollutant Discharge Elimination System Permit (#TN0005452), originally issued by EPA to TVA on April 30, 1976, and most recently re-issued by TDEC on September 1, 2003.
- k. "Order" or "Order and Agreement" shall mean this document and all documents incorporated by reference, or to be incorporated by reference, into this document. All such documents are integral and enforceable parts of this Order.
- l. "Paragraph" shall mean a portion of this Order identified by an Arabic numeral or a lower case letter.
  - m. "Parties" shall mean EPA and TVA.
- n. "Past Response Costs" shall mean all costs incurred by EPA at or in connection with the Site between December 22, 2008, and the Effective Date hereof, that are not inconsistent with the NCP.
- o. "RCRA" shall mean the Solid Waste Disposal Act, as amended, 42 U.S.C. § 6901, et seq. (also known as the Resource Conservation and Recovery Act).

- p. "Section" shall mean a portion of this Order identified by a Roman numeral.
- q. "Site" shall mean those areas of the TVA Kingston Fossil Fuel Plant, located in Roane County, Tennessee, where Waste Material from the December 22, 2008, release has been deposited, stored, disposed of, or placed, or has migrated or otherwise come to be located.
  - r. "State" shall mean the State of Tennessee.
- s. "TVA" shall mean the Tennessee Valley Authority, and its officers, directors, employees, successors, assigns, contractors, agents and representatives, and any successor departments or agencies of the United States.
- t. "Waste Material" shall mean: 1) any "hazardous substance" under Section 101(14) of CERCLA, 42 U.S.C. § 9601(14); 2) any pollutant or contaminant under Section 101(33) of CERCLA, 42 U.S.C. § 9601(33); and 3) any "solid waste" under Section 1004(27) of RCRA, 42 U.S.C. § 6903(27), related to the December 22, 2008, release of ash at and from the Site, or generated in connection with response actions related thereto.
  - u. "Work" shall mean all activities TVA is required to perform under this Order.

## V. EPA'S FINDINGS OF FACT

- 9. TVA operates the Kingston Fossil Fuel Plant located in Roane County, Tennessee, near the confluence of the Emory and Clinch Rivers. The plant was originally built in the early 1950s to provide power for the Department of Energy's facility in Oak Ridge, Tennessee. In approximately 1958, the plant began use of a 244-acre wet settling pond for containment of the ash that remains after coal is burned. This settling pond covered the area where the current settling pond, stilling pond and landfill cells 1, 2, 3 and 4 now reside.
- 10. On June 29, 1999, TVA submitted an application to TDEC for a Class II landfill permit for the disposal of ash waste from the operation of the Kingston Fossil Fuel Plant. TDEC issued TVA the requested Class II landfill permit on September 26, 2000.
- 11. On September 1, 2003, TDEC issued TVA its most recent NPDES permit for the Kingston Fossil Fuel Plant. The permit authorizes discharge of water from the ash settling pond to the plant intake channel (the intake draws water from the Emory River) and discharge of cooling water to the Clinch River downstream from the mouth of the Emory River. The permit requires that a certain amount of free water volume be maintained in the settling pond to provide adequate treatment prior to discharge. This requirement necessitates periodic dredging of the ash settling pond. The NPDES permit further includes a general requirement that TVA properly operate and maintain all facilities and systems for collection and treatment, and expressly prohibits overflows of

wastes to land or water from any portion of the collection, transmission, or treatment system other than through permitted outfalls.

- 12. On December 22, 2008, containment structures surrounding portions of the Class II landfill failed resulting in a release of approximately 5.4 million cubic yards of wet ash which flowed into area waters, including the Emory River, adjacent tributaries and sloughs, and adjoining shorelines. The Emory River is a navigable-in-fact water of the United States. The release also extended to approximately 300 acres of land outside of the ash storage area, almost all of which was owned by the United States and in TVA's custody and control. The Tennessee River is the source of drinking water for the City of Kingston, Tennessee, and the Watts Bar Reservoir is used by several municipalities as a source of drinking water.
- 13. EPA Region 4 was notified of the incident on December 22, 2008, and an EPA On-Scene Coordinator (OSC) mobilized to the Site that same day for the emergency phase of the cleanup, pursuant to Executive Order 12580. The OSC worked within the Unified Command/Incident Command to coordinate the response as required by the NCP. Subsequently, on January 10, 2009, EPA, in coordination with the Unified Command, declared the emergency phase of the cleanup complete, transferred the lead federal agency role to TVA in accordance with Executive Order 12580, and demobilized from the Site.
- 14. On January 12, 2009, TDEC issued TVA the Commissioner's Order which directed TVA to undertake numerous response activities at the Site including, but not limited to:
- a. implement measures to prevent the movement of contaminated materials and minimize further down-stream migration of contaminated sediments;
- b. fully cooperate and support TDEC's review of all TVA fly ash impoundments located in the State;
- c. submit all existing studies, reports and memoranda that are potentially relevant to explaining or analyzing the cause of the catastrophic failure of the containment structures;
- d. fully cooperate and provide support for TDEC's initial assessment of the impact of the ash release on all waters of the State;
- e. prepare and submit a Corrective Action Plan (CAP) within 45 days after receipt of the Commissioner's Order, to include:
- i. a plan for the assessment of soil, surface water and groundwater; remediation of impacted media; and restoration of all natural resources damaged as a result of the release;
  - ii. a plan for monitoring the air and water in the area during the cleanup process;

- iii. a plan to ensure that public and private water supplies are protected and that alternative water supplies are provided if contamination is detected;
- iv. a plan for addressing both the short-term and long-term management of fly ash at the Site, including remediation and stabilization of the failed ash waste cells, proper management of the recovered ash, and a revised closure plan for the Class II ash disposal facility; and
- $v.\,$  a plan to address any health or safety hazards posed by the ash to workers and the public.
- 15. On January 21, 2009, TVA submitted written notification to the Tennessee Emergency Response Commission, pursuant to which TVA reported a discharge of 5.4 million cubic yards of ash containing the following constituents: arsenic, beryllium, chromium, copper, lead, mercury, nickel, zinc, antimony, cadmium, silver, selenium, thallium, and vanadium oxide.
- 16. On February 4, 2009, EPA, pursuant to Executive Order 12088, and TDEC issued a letter to TVA in which EPA provided notice to TVA that EPA considers the release to be an unpermitted discharge of a pollutant in contravention of the Clean Water Act. The letter directed that TVA provide all plans, reports, work proposals and other submittals being provided to TDEC, to EPA as well for agency review and approval. Subject to the reservations contained in Section XXX, by complying with this Order, TVA will be deemed by EPA to have addressed the unpermitted discharge.
- 17. Pursuant to the Commissioner's Order, on March 2, 2009, TVA submitted a draft CAP to EPA and TDEC for agency review and approval.
- 18. Since the release, EPA, the State and TVA have conducted extensive sampling of air, water, sediment, and ash material. Sampling results have revealed levels of arsenic in the ash material that exceed Region 4's residential removal action level of 39 mg/kg. In addition, shortly after the release, arsenic was detected in surface water samples at concentrations in excess of the Tennessee Water Quality Criteria (TWQC) for Domestic Water Supply and in excess of the human health aquatic organism consumption criteria. In the days immediately following the release, arsenic, as well as numerous other contaminants, including cadmium, chromium, copper, lead, mercury, nickel, selenium, and zinc, were also detected in surface water at concentrations which exceeded the National Recommended Ambient Water Quality Criteria (AWQC) for protection of aquatic life (based on both the Chronic Continuous Criterion and the Chronic Maximum Criterion). Other than one thallium exceedance in the raw water collected from the river at the intake of the Kingston public water supply system the day after the release, no contaminants have been detected above the Maximum Contaminant Levels (MCLs) in either the raw intake waters or finished water supplies of the Kingston, Cumberland, or Rockwood public water supply systems.
- 19. EPA has classified arsenic as a known human carcinogen; and long-term exposure of aquatic organisms to high levels of metals like arsenic, cadmium, chromium, copper, lead, mercury, nickel, selenium, and zinc may cause decreases in survival,

growth, or reproduction to those aquatic organisms. The levels of these metals detected in the most recent air and surface water sampling events do not indicate an immediate threat to human health or the environment from those metals. However, if the ash material is not properly managed and remediated, the direct impact of the ash material currently in the water on the riverine ecosystem, further suspension of the ash and its constituents within affected waters, and potential exposure from ash on the ground could present unacceptable impacts to human health and/or the environment.

## VI. EPA'S CONCLUSIONS OF LAW AND DETERMINATIONS

- 20. Based on the Findings of Fact set forth above and the Administrative Record supporting this Order, EPA has determined that:
- a. The TVA Kingston Fossil Fuel Site is a "facility" as defined by Section 101(9) of CERCLA, 42 U.S.C. § 9601(9).
- b. Ash at the Site contains constituents such as arsenic, cadmium, chromium, copper, lead, mercury, nickel, selenium and zinc which are "hazardous substances" as defined by Section 101(14) of CERCLA, 42 U.S.C. § 9601(14). Coal ash is not regulated as a hazardous waste under RCRA.
- c. TVA is a "person" as defined by Section 101(21) of CERCLA, 42 U.S.C.  $\S$  9601(21).
- d. TVA is an Executive agency and instrumentality of the United States and as such is charged with fulfilling the obligations of the owner/operator under CERCLA at this facility. TVA is liable under Section 107(a) of CERCLA as an "operator" of the facility as defined by Section 101(20) of CERCLA, 42 U.S.C. § 9601(20), and within the meaning of Section 107(a)(1) of CERCLA, 42 U.S.C. § 9607(a)(1).
- e. The conditions described in the Findings of Fact above constitute an actual or threatened "release" of hazardous substances from the facility as defined by Section 101(22) of CERCLA, 42 U.S.C. § 9601(22).
- f. The actual or threatened release of hazardous substances from the Site may present an imminent and substantial endangerment to the public health, welfare, or the environment within the meaning of Section 106(a) of CERCLA, 42 U.S.C. § 9606(a).
- g. The response actions required by this Order are necessary to protect the public health, welfare, or the environment and, if carried out in compliance with the terms of this Order, will be consistent with the NCP, as provided in 40 C.F.R. § 300.700(c)(3)(ii).

## VII. ORDER

21. Based upon the foregoing Findings of Fact, Conclusions of Law and Determinations, and the Administrative Record for this Site, it is hereby Ordered and

Agreed that TVA shall comply with the following provisions, including but not limited to all documents incorporated by reference into this Order, and all schedules and deadlines in this Order, developed pursuant to this Order, or incorporated by reference into this Order.

## VIII. DESIGNATION OF PROJECT COORDINATOR, ON-SCENE COORDINATOR, AND REMEDIAL PROJECT MANAGER

- 22. TVA hereby designates Michael Scott as its Project Coordinator who shall be responsible for administration of all TVA's actions required by this Order. To the greatest extent possible, the Project Coordinator shall be present on Site or readily available during Site work. When the Project Coordinator is not able to be present on Site or readily available he/she may, upon notification to the OSC and RPM, designate another qualified TVA employee to temporarily act in the position of Project Coordinator. Receipt by TVA's Project Coordinator of any written notice or communication from EPA relating to this Order shall constitute receipt by TVA.
- 23. TVA shall perform the response actions itself or retain a contractor(s) to perform the actions. TVA shall notify EPA of the identity and assigned tasks of each of its contractors performing Work under this Order upon their selection and contract award. TVA shall provide copies of this Order to all contractors performing any Work called for by this Order. TVA shall be responsible for ensuring that its contractors comply with the terms and conditions of this Order. Any contractor that will be performing tasks for which the following guidance has applicability, must demonstrate compliance with ANSI/ASOC E4-1994, "Specifications and Guidelines for Quality Systems for Environmental Data Collection and Environmental Technology Programs," (American National Standard, January 5, 1995), by submitting a copy of the proposed contractor's Quality Management Plan (QMP). The QMP should be prepared in accordance with "EPA Requirements for Quality Management Plans (QA/R-2)," (EPA/240/B-01-002, March 2001) or equivalent documentation as determined by EPA.
- 24. EPA has designated Leo Francendese of the Region 4 Emergency Response and Removal Branch as its On-Scene Coordinator (OSC) for purposes of the time-critical removal actions to be conducted under this Order and Craig Zeller of the Remedial and Site Evaluation Branch as its Remedial Project Manager (RPM) for all other response actions to be conducted under this Order.
- 25. EPA and TVA shall have the right to change their respective designated Project Coordinator and OSC/RPM. The Parties shall notify each other ten (10) days before such a change is made.

## IX. WORK TO BE PERFORMED

26. **Statement of Objectives -** The Parties acknowledge that, in order to expeditiously and efficiently prioritize and perform necessary response actions at the Site, it is important to have agreed upon **short-term**, **mid-term** and **longer-term strategic** 

**objectives**. The Parties therefore agree that the **short-term strategic objectives** for the Site are to:

- a. prevent the coal ash release from negatively impacting public health and the environment;
- b. contain and remove coal ash from the Emory River and the area east of Dike 2 as appropriate to restore flow and minimize further downstream migration of the ash material; and
- c. ensure that coal ash material recovered during these efforts is properly managed pending ultimate disposal decisions or, to the extent required by limited storage capacity, properly disposed.

## The mid-term strategic objectives for the Site are to:

- a. remove any remaining coal ash from the Emory River and the area east of Dike 2, as well as the coal ash from embayments and tributaries west of Dike 2, to the maximum extent practicable, as determined by EPA in consultation with TDEC and TVA, pending further Site assessment;
- b. remove the coal ash from impacted surface soils to the maximum extent practicable, as determined by EPA in consultation with TDEC and TVA, pending further Site assessment;
- c. restore area waters impacted by the coal ash release in accordance with the required jurisdictional assessment; and
- d. ensure proper disposal of all coal ash material recovered during these efforts.

## The longer-term objectives for the Site are to:

- perform a comprehensive Site assessment to determine what actions may be necessary to address any residual contamination remaining after previous cleanup activities;
- b. implement any such actions; and
- c. ensure the proper disposal of all ash material recovered during these efforts.

Response actions necessary to achieve short-term strategic Site objectives shall generally be identified as time-critical removal actions. Activities necessary to achieve mid-term strategic Site objectives shall generally be identified as non-time-critical removal actions. Activities necessary to achieve longer-term strategic Site objectives shall generally be identified as remedial actions. The specific actions to be taken to achieve these goals, and the schedules for those actions, shall be governed by the work plans developed and approved pursuant to the remaining paragraphs in this Section. The Parties recognize that, to the extent appropriate, time-critical response actions in furtherance of the short-term Site objectives and non-time-critical response actions in furtherance of the mid-term Site objectives will be carried out simultaneously.

27. EPA acknowledges that TVA has already done considerable work to further the short-term strategic Site objectives set forth in Paragraph 26, above, including extensive work on air, surface water and drinking water monitoring; ash containment;

dust suppression; and initial dredging activities. Portions of that work have been conducted under the TDEC Commissioner's Order and TVA authorities and in accordance with work plans approved by TDEC and EPA pursuant to that Order and EPA's February 4, 2009, letter. It is the intent of the Parties that work done and data generated prior to the Effective Date of this Order be retained and utilized to the maximum extent feasible during implementation of the Work required by this Order and, further, that the issuance of this Order not result in any unnecessary delay in the ongoing cleanup efforts. TVA shall continue to proceed with ongoing response work pursuant to the following deliverables, which have previously been approved by EPA and/or TDEC:

- a. Kingston Fossil Plant (KIF) Interim Drainage and Controls Plan (approved April 3, 2009)
- b. Phase 1 Emory River Dredging Plan (approved March 19, 2009)
- c. KIF Fly Ash Pond Incident Environmental Sampling Plan (February 2009)
- d. Request for Authorization to use Polymers to Enhance Treatment of Dredge Return Water (approved March 23, 2009)
- e. Request for Use of Wick Drains in Ash Processing Area (February 20, 2009)
- f. Long Term Cenosphere Recovery Plan (March 2009)
- g. Ash Processing Area Construction and Operation Plan (approved March 19, 2009)

Each of the foregoing work plans is incorporated by reference into this Order and is enforceable under this Order. Each of these plans shall be subject to revision upon direction by EPA, at which time the plan(s) shall be subject to the approval provisions in Section X, hereto.

## 28. Time-Critical Removal Action Memoranda and Work Plans -

Within fourteen (14) days of the Effective Date of this Order, TVA shall submit an Action Memorandum for the time-critical removal action. Upon approval of the Action Memorandum by EPA, TVA shall submit, in accordance with the schedule below, the following plans, which may incorporate by reference, or otherwise build upon, plans previously submitted for TDEC and/or EPA review, along with any other work plans necessary to implement the actions selected in the Action Memorandum:

Within five (5) days of Action Memorandum approval, TVA shall submit the following:

- a. Site Storm Water Management Plan
- b. Site Dust Control and Air Monitoring Plan
- c. Schedule for development of a Structural Integrity Evaluation, Recommendations and Maintenance Plan for Existing Site Dikes/Berms being used to contain spilled ash
- d. Schedule for development of a Dredging/Excavation Plan for East of Dike 2 (including Ash Processing Areas)

Within fifteen (15) days of Action Memorandum approval, TVA shall submit the following:

- e. Off-Site Ash Disposal Options Analysis
- f. Financial Expenditure Report (to be updated quarterly)
- g. Schedule for development of any other work plans necessary to implement the actions selected in the Action Memorandum.

Within forty-five (45) days of Action Memorandum approval, TVA shall submit the following:

- h. Information/Data Management Plan
- i. Surface Water Monitoring Plan for the Emory, Clinch and Tennessee Rivers

The approved Action Memorandum may be amended, subject to review and approval by EPA, should circumstances so warrant.

- 29. TVA shall publish a notice of the availability of the Administrative Record for the selected time-critical removal action(s), including the approved Action Memorandum and any approved work plans, within fourteen (14) days of the approval by EPA of the Action Memorandum. TVA shall update the Administrative Record when additional work plans are approved and if and when any amendments are made to the Action Memorandum. TVA shall provide a public comment period of not less than thirty (30) days beginning at the time the initial Administrative Record file is made available to the public, and upon any updates to the Administrative Record. The public comment period shall not delay initiation of selected time-critical removal actions. A written response to significant comments submitted during the public comment period shall be prepared by TVA and submitted to EPA for review and comment. TVA's final response to comments shall be included in the Administrative Record.
- 30. Non-Time-Critical Removal Action EE/CAs, Action Memoranda and Work Plans Within ninety (90) days after the effective date of this Order, TVA shall submit to EPA for approval a draft Work Plan for performing one or more Engineering Evaluation/Cost Analyses (EE/CAs) for non-time-critical removal actions to be taken at the Site. TVA shall conduct the EE/CA(s) consistent with 40 C.F.R. § 300.415(b)(4) of the NCP, and in accordance with OSWER Directive 9360.0-32, Guidance on Conducting Non-Time Critical Removal Actions Under CERCLA (EPA/540-R-93-057, August 1993). The EE/CA Work Plan shall be developed in conjunction with a Sampling and Analysis Plan, and Health and Safety Plan. A detailed schedule for completion of each major work item in the EE/CA process shall also be included in the EE/CA Work Plan.

The EE/CA Work Plan shall include a comprehensive description of the work to be performed in developing the EE/CA, including the media to be investigated (e.g. air, ground water, surface water, surface and subsurface soils, and sediments), and the methodologies for human health and ecological risk assessments. The EE/CA Work Plan shall also include the Jurisdictional Assessment described in Section XII, Paragraph 34(b), below. The EE/CA investigations, human health/ecological risk assessments, and identified non-time-critical removal actions shall address:

- Coal ash not yet removed from the Emory River and the area east of Dike 2;
- Coal ash in the embayments and tributaries west of Dike 2;
- Coal ash on upland areas and surface soils;
- Restoration of area waters impacted by the coal ash release per the Jurisdictional Assessment; and
- Proper disposal of all coal ash material recovered during these efforts.

Following completion of the work in the approved EE/CA Work Plan, TVA shall submit the draft EE/CA Report for EPA's review and approval. Upon issuance of a final EE/CA, as approved by EPA, TVA shall make the EE/CA and the Administrative Record supporting the EE/CA available for public comment in accordance with 40 C.F.R. § 300.415(n), and shall comply with the Administrative Record requirements of 40 C.F.R. § 300.820. Within thirty (30) days of the close of the public comment period on the EE/CA, TVA shall submit for EPA review and approval an Action Memorandum which responds to public comments and describes the selected response actions. Within sixty (60) days of EPA's approval of the Action Memorandum, TVA shall submit to EPA for review and approval a Non-Time-Critical Removal Work Plan for the selected response actions.

Upon approval, TVA shall implement the Non-Time-Critical Removal Work Plan in accordance with the schedule provided therein. Both the Action Memorandum and the Non-Time-Critical Removal Work Plan shall be added to the Administrative Record and an additional public comment period of not less than thirty (30) days shall be provided. This comment period shall not delay initiation of the selected non-time-critical removal actions. A written response to significant comments submitted during the public comment period shall be prepared by TVA and submitted to EPA for review and comment. TVA's final response to comments shall be included in the Administrative Record.

31. Other Response Actions — Within thirty (30) days of the receipt of EPA's approval of the Final OSC Report for the Non-Time-Critical Removal Action described in Paragraph 41, below, TVA shall submit to EPA for review and approval, a Remedial Site Work Plan (RSWP) to address the longer-term strategic Site objectives set forth in Paragraph 26, above, including the performance of a preliminary assessment as required by 42 U.S.C. § 9620(d)(1). Upon EPA approval of the RSWP, TVA shall perform the preliminary assessment and submit to EPA for review and approval a preliminary assessment report which documents whether additional assessment or remedial work is necessary to address any residual contamination remaining at the Site. To the extent that the report, as approved by EPA, indicates that additional Site response action is required, TVA shall revise the RSWP to include a plan and schedule for selecting and conducting such work. Such work shall be governed by the terms of this Order and Agreement on Consent, and shall be performed in accordance with CERCLA and the NCP, including all public participation and Administrative Record requirements.

## X. EPA APPROVAL OF PLANS AND OTHER SUBMISSIONS

- 32. a) After review of any plan, report or other item which is required to be submitted for approval pursuant to this Order, EPA, after consultation with TDEC, shall: (i) approve, in whole or in part, the submission; (ii) approve the submission upon specified conditions; (iii) disapprove, in whole or in part, the submission, directing that TVA modify the submission; or (iv) any combination of the above.
- b) In the event of approval or approval upon conditions, pursuant to Paragraph 32(a)(i) or (ii), TVA shall proceed to take any action required by the plan, report, or other item, as approved by EPA, subject only to its right to invoke the Dispute Resolution procedures set forth in Section XXVI (Dispute Resolution) with respect to the conditions established by EPA.
- c) Upon receipt of a notice of disapproval pursuant to Paragraph 32(a)(iii), TVA shall, within fourteen (14) days, or such other time as agreed by EPA and TVA, correct the deficiencies and resubmit the plan, report, or other item for approval. Any stipulated penalties applicable to the submission, as provided in Section XXVIII, shall accrue during the 14-day period, or otherwise specified period, but shall not be assessed unless the resubmission is disapproved due to a material defect as provided in Paragraph 32(f).
- d) Notwithstanding the receipt of a notice of disapproval pursuant to Paragraph 32(a)(iii), TVA shall proceed, at the direction of EPA, to take any action required by any non-deficient portion of the submission. Implementation of any non-deficient portion of a submission shall not relieve TVA of any liability for stipulated penalties under Section XXVIII (Stipulated Penalties).
- e) In the event that a resubmitted plan, report or other item, or portion thereof, is disapproved by EPA, EPA may again require TVA to correct the deficiencies, in accordance with the preceding paragraphs.
- f) If upon resubmission, a plan, report, or item is disapproved by EPA due to a material defect, TVA shall be deemed to have failed to submit such plan, report, or item timely and adequately unless TVA invokes the dispute resolution procedures set forth in Section XXVI (Dispute Resolution) and EPA's action is not confirmed pursuant to that Section. The provisions of Section XXVI (Dispute Resolution) and Section XXVIII (Stipulated Penalties) shall govern the implementation of the Work and accrual and payment of any stipulated penalties during Dispute Resolution. If EPA's disapproval is confirmed, stipulated penalties shall accrue for such violation from the date of EPA's initial disapproval. A material defect shall be any element of a submitted plan, report, or item that goes to the fundamental purpose for the plan, report, or item and does not include style or format.
- g) All plans, reports, and other items required to be submitted to EPA under this Order shall, upon approval by EPA, be incorporated by reference into and become enforceable under this Order. In the event EPA approves a portion of a plan, report, or

other item required to be submitted to EPA under this Order, the approved portion shall be incorporated into and become enforceable under this Order.

## XI. STRUCTURAL INTEGRITY ASSESSMENTS OF OTHER TVA FACILITIES

33. TVA is currently conducting assessments of all of its existing coal ash impoundments located at its eleven (11) coal-fired power plants, including analyses of the structural integrity of such impoundments. EPA is also undertaking its own national assessment of coal ash impoundments throughout the nation. TVA agrees to promptly provide any findings, reports, or other documentation produced as a result of its internal assessments to EPA. TVA has provided EPA its assessment methodologies and agrees to work with EPA with the goal of making the TVA and EPA assessment methodologies consistent. Upon EPA's request, TVA shall make available appropriate TVA personnel and contractors, to work with EPA personnel and contractors, in refining their respective assessment methodologies. EPA will review any reports submitted under this Paragraph and advise TVA if additional structural assessment work is warranted. EPA may conduct its own independent structural integrity assessments of TVA facilities and TVA agrees to cooperate with such effort. Nothing in this Paragraph shall require TVA to delay its ongoing or planned assessments nor shall Sections XXVIII or XXV apply to activities under this Paragraph.

## XII. COMPLIANCE WITH OTHER LAWS

- 34. a) All Work performed under this Order shall be performed in a manner consistent with the National Contingency Plan, including, but not limited to, the public participation requirements of 40 C.F.R. § 300.415. All removal actions undertaken pursuant to this Order, shall, to the maximum extent practicable considering the exigencies of the situation, attain applicable or relevant and appropriate requirements (ARARs) under federal environmental or state environmental or facility siting laws. All remedial activities undertaken pursuant to this Order shall attain ARARs unless a waiver has been approved by EPA pursuant to 40 C.F.R. § 300.430(f)(1)(ii)(C).
- b) Although restoration of area waters impacted by the coal ash release has been identified as a mid-term strategic Site objective and a part of the non-time-critical removal actions under this Order, TVA agrees that such restoration will be considered as remedial activity for purposes of complying with ARARs. Therefore, ARARs pertaining to such restoration shall be attained unless a waiver has been approved by EPA. In particular, TVA agrees that it will so comply with Clean Water Act Section 404(b)(1) guidelines to restore waters of the United States to the functional level occurring prior to the ash release. In order to identify the full extent of response activities necessary to meet this ARAR, TVA shall conduct a jurisdictional assessment of the Site, to the extent not previously evaluated, which will identify all waters of the United States impacted by the release. Such assessment shall be performed by an independent environmental management professional or other expert deemed qualified by EPA. The assessment shall include mapping and physical inspection of affected banks, streambeds and

adjoining shorelines of all impacted areas. As part of the assessment, TVA shall identify the mechanisms of mobilization and deposition of the ash material in the surface water bodies and adjacent riparian areas affected by the release and identify locations and depths of the ash in these water areas and the changes in these parameters over an appropriate period of time, as determined by field measurements. This assessment shall also include an evaluation of the impacts to habitats due to the release, and prediction of future impacts to aquatic species upon re-suspension and deposition of ash. Based on the results of this assessment, TVA shall propose, as a part of the EE/CA process outlined in Paragraph 30, above, final cleanup criteria which address removal of ash from stream/slough/river beds, banks, floodplains, adjacent wetlands and the shorelines adjoining navigable waters, replanting of the riparian zone, and compensatory mitigation for any permanent loss to waters of the United States as approved by EPA in consultation with TDEC.

#### XIII. HEALTH AND SAFETY PLAN

35. Within fourteen (14) days after the Effective Date of this Order, TVA shall submit for EPA review and approval, a plan that ensures the protection of the public health and safety during performance of on-Site Work under this Order. This plan shall be prepared in accordance with EPA's Standard Operating Safety Guide (PUB 9285.1-03, PB 92-963414, June 1992). In addition, the plan shall be consistent with Occupational Safety and Health Administration (OSHA) provisions for response action worker safety and health found in 29 C.F.R. Part 1910. TVA shall incorporate all changes to the plan recommended by EPA, and implement the plan during the pendency of the response actions. The Plan shall also include contingency planning.

## XIV. QUALITY ASSURANCE AND SAMPLING

- 36. a) All sampling and analyses performed pursuant to this Order shall conform with EPA guidance regarding sampling, quality assurance/quality control (QA/QC), data validation, and chain of custody procedures. TVA shall ensure that the laboratories used to perform the analyses participate in a QA/QC program that complies with the appropriate EPA guidance. TVA shall follow "Quality Assurance/Quality Control Guidance for Removal Activities: Sampling QA/QC Plan and Data Validation Procedures" (OSWER Directive Number 9360.4-01, April 1, 1990), as guidance for QA/QC and sampling. TVA shall only use laboratories that have a documented Quality System that complies with ANSI/ASQC E-4 1994, "Specifications and Guidelines for Quality Systems for Environmental Data Collection and Environmental Technology Programs" (American National Standard, January 5, 1995), and "EPA Requirements for Quality Management Plans (QA/R-2) (EPA/240/B-01/002, March 2001)," or equivalent documentation as determined by EPA. EPA may consider laboratories accredited under the National Environmental Laboratory Accreditation Program (NELAP) as meeting the Quality System requirements.
- b) As a part of EPA's oversight of QA/QC and data validation, EPA may request, and TVA shall then require, that any laboratory TVA is using analyze samples submitted

by EPA for QA monitoring. TVA shall provide to EPA the QA/QC procedures followed by all sampling teams and laboratories performing data collection and/or analysis.

c) Upon request by EPA, TVA shall allow EPA or its authorized representatives to take split and/or duplicate samples. EPA shall have the right to take any additional samples that EPA deems necessary. Upon request, EPA shall allow TVA to take split or duplicate samples of any samples it takes as part of its oversight of TVA's implementation of the Work.

## XV. POST-REMOVAL SITE CONTROL

37. In accordance with the EPA-approved schedule, TVA shall submit a proposal for post-removal site control consistent with Section 300.415(l) of the NCP and OSWER Directive No. 9360.2-.02. Upon EPA approval, TVA shall implement such controls and shall provide EPA with documentation of all post-removal site control arrangements.

## XVI. REPORTING

- 38. TVA shall submit written progress reports to EPA and TDEC on a weekly basis for all time-critical response actions and on a monthly basis for all non-time-critical response actions undertaken pursuant to this Order until termination of the Order, unless the EPA OSC or RPM and TDEC agree in writing that reports can be less frequent. These reports shall describe all significant developments during the preceding period, including the actions performed and any problems encountered, analytical data received during the reporting period, and the developments anticipated during the next reporting period, including a schedule of work to be performed, anticipated problems, and planned resolutions of past or anticipated problems. During the removal phases of the cleanup, these reports (i.e., Pollution Reports/Situation Reports) shall conform to EPA's Guidance for Preparing POLREPs/SITREPs, OSWER Directive No. 9360.3-03 (Dec. 2007). At the conclusion of all time-critical removal actions, and at the conclusion of all non-timecritical removal actions, TVA shall submit Final Pollution Reports in accordance with the POLREP/SITREP Guidance. At the time they are submitted to EPA and TDEC, TVA shall make all progress reports available to the public by placing them in the local Site Repository and on the established TVA Kingston release website.
- 39. TVA shall utilize SCRIBE or SCRIBE.Net, as well as ArcMap 9.3, for data management and reporting purposes.
- 40. During the pendency of this Order, TVA and any successor(s) in title shall, at least thirty (30) days prior to the conveyance of any interest in real property at the Site, give written notice of this Order to the transferee and written notice to EPA and the State of the proposed conveyance, including the name and address of the transferee. The party conveying such an interest shall require that the transferee comply with Section XXII of this Order (Site Access).

41. Final OSC Reports. Within sixty (60) days after completion of all time-critical removal actions, and, again, within sixty (60) days after the completion of all non-time-critical removal actions required under this Order, TVA shall submit for EPA review and approval final reports summarizing the actions taken to comply with this Order. The final reports shall conform, at a minimum, with the requirements set forth in Section 300.165 of the NCP entitled "OSC Reports" and EPA's "Superfund Removal Procedures: Removal Response Reporting," OSWER Directive No. 9360.3-03, June 1, 1994. The final OSC reports shall include a listing of quantities and types of materials addressed, a discussion of response and disposal options considered for those materials, a listing of the ultimate destinations of those materials, and a presentation of the analytical results of all sampling and analyses performed. The final OSC reports shall also include the following certification signed by a person who supervised or directed the preparation of the reports:

Under penalty of law, I certify that to the best of my knowledge, after appropriate inquiries of all relevant persons involved in the preparation of the report, the information submitted is true, accurate, and complete. I am aware that there are significant penalties for submitting false information, including the possibility of fine and imprisonment for knowing violations.

## XVII. OFF-SITE SHIPMENTS AND OTHER DISPOSAL OF WASTE MATERIAL

- 42. TVA shall, prior to any off-Site shipment of Waste Material from the Site to an out-of-state waste management facility, provide written notification of such shipment of Waste Material to the appropriate state environmental official in the receiving facility's state and to the EPA OSC/RPM. However, this notification requirement shall not apply to any off-Site shipments when the total volume of all such shipments will not exceed ten (10) cubic yards.
- 43. TVA shall include in the written notification the following information: 1) the name and location of the facility to which the Waste Material is to be shipped; 2) the type and quantity of the Waste Material to be shipped; 3) the expected schedule for the shipment of the Waste Material; and 4) the method of transportation. TVA shall notify the state in which the planned receiving facility is located of major changes in the shipment plan, such as a decision to ship the Waste Material to another facility within the same state, or to a facility in another state.
- 44. The identity of the receiving facility and state will be determined by TVA. TVA shall provide the information required by Paragraphs 42 and 43 as soon as practicable before the Waste Material is actually shipped.
- 45. TVA shall not permanently dispose of any Waste Material at an off-Site facility, or in a new landfill on-Site, unless that facility or landfill is operating in compliance with RCRA Subtitle D permitting requirements for operation and disposal of industrial wastes which, at a minimum, shall include the use of a synthetic liner, leachate

collection system, groundwater monitoring, financial assurance, and closure and post-closure care. If TVA proposes to permanently dispose of any Waste Material to an existing landfill on-Site, it must satisfy both TDEC and EPA that such disposal complies with all state permitting requirements and is otherwise protective of human health and the environment. EPA may disapprove such disposal if it is not so satisfied. Prior to any off-Site shipment, TVA shall obtain EPA's determination that the proposed receiving facility is operating in compliance with the requirements of 40 C.F.R. § 300.440, as well as appropriate health and safety standards.

#### XVIII. PERMITS

- 46. TVA shall be responsible for obtaining all required Federal, State and local permits which are necessary for the performance of all Work under this Order.
- 47. The Parties recognize that under Sections 121(d) and 121(e)(1) of CERCLA, 42 U.S.C. §§ 9621(d) and 9621(e)(1), and the NCP, portions of the response actions called for by this Order that are conducted entirely on-Site, where such response actions are selected and carried out in accordance with CERCLA, are exempt from the procedural requirement to obtain Federal, State, or local permits. Subject to Paragraph 34 above, all on-Site response actions must, however, comply with all the applicable or relevant and appropriate Federal and State standards, requirements, criteria, or limitations, which would have been included in any such permit unless justification exists for a waiver and EPA, in consultation with TDEC, approves such a waiver.

## XIX. EMERGENCY RESPONSE AND NOTIFICATION OF RELEASES

- 48. In the event of any action or occurrence during performance of the Work which causes or threatens a release of Waste Material from the Site that constitutes an emergency situation or may present an immediate threat to public health or welfare or the environment, TVA shall immediately take all appropriate action. TVA shall take these actions in accordance with all applicable provisions of this Order, including, but not limited to, the Health and Safety Plan, in order to prevent, abate or minimize such release or endangerment caused or threatened by the release. TVA shall also immediately notify the EPA OSC/RPM as appropriate, or, in the event of his unavailability, the Regional Duty Officer at 404-562-8700, of the incident or Site conditions. In the event that TVA fails to take appropriate response action as required by this Paragraph, and EPA takes such action instead, TVA shall reimburse EPA all costs of the response action not inconsistent with the NCP pursuant to Section XXV (Reimbursement of Costs).
- 49. In addition, in the event of any new release of a hazardous substance from the Site above applicable reportable quantities, TVA shall immediately notify the OSC/RPM as appropriate, the Regional Duty Officer at 404-562-8700, and the National Response Center at (800) 424-8802. TVA shall submit a written report to EPA within seven (7) days after each release, setting forth the events that occurred and the measures taken or to be taken to mitigate any release or endangerment caused or threatened by the release and to prevent the reoccurrence of such a release. This reporting requirement is in addition

to, and not in lieu of, reporting under Section 103(c) of CERCLA, 42 U.S.C. § 9603(c), and Section 304 of the Emergency Planning and Community Right-To-Know Act of 1986, 42 U.S.C. § 11004, et seq., as applicable.

## XX. AUTHORITY OF EPA ON-SCENE COORDINATOR/REMEDIAL PROJECT MANAGER

50. EPA and TVA agree that EPA's OSC and/or RPM, shall be responsible for overseeing TVA's implementation of this Order, in consultation with TDEC and TVA's Project Coordinator. TVA and EPA further agree that the EPA OSC and/or RPM, in consultation with TDEC and the TVA Project Coordinator, shall have the authority to halt any Work required under this Order, or to conduct or direct other response action at the Site in an emergency or under circumstances that may present an immediate threat to public health, welfare or the environment. Absence of the EPA OSC or RPM from the Site shall not be cause for stoppage of work unless specifically directed by the EPA OSC or RPM after consultation with TDEC and the TVA Project Coordinator.

## XXI. PUBLIC REVIEW OF ADMINISTRATIVE RECORD

- 51. Upon issuance of this Order, the Administrative Record supporting the issuance of this Order, as well as the Order itself, shall be made available for public comment for a period of thirty (30) days. The Administrative Record will be available for public review during normal business hours at the U.S. EPA Region 4 Records Center, 61 Forsyth Street, SW, 11<sup>th</sup> Floor, Atlanta, Georgia 30303, and the local Site Repository that is established under the Order. Within thirty (30) days following close of the public comment period, EPA shall provide a written response to significant comments that were received. The public comment period shall not delay the initiation of the Work required by this Order. However, EPA may seek modifications to this Order if public comments received during the comment period disclose facts or considerations which indicate that this Order is inappropriate, improper, or inadequate.
- 52. Upon selection or proposal of response activities pursuant to this Order, an Administrative Record(s) for such response activities will be established and made available for public review and comment in accordance with Paragraphs 29, 30 and 31, above.

## XXII. SITE ACCESS

53. TVA shall use its best efforts to provide and/or obtain access to all areas to which access is necessary to implement this Order. Such access shall be provided to EPA and TDEC as well as their employees, contractors, agents, consultants, designees and representatives. Access provided and/or obtained by TVA shall permit these individuals to move freely on-Site and at appropriate off-Site areas in order to conduct actions which EPA, in consultation with TDEC, determines to be necessary.

54. Where action under this Order is to be performed in areas owned by or in possession of someone other than the United States, TVA shall use its best efforts to obtain all necessary access agreements in a timely fashion. TVA shall immediately notify EPA and TDEC if, after using its best efforts, it is unable to obtain such agreements. TVA shall describe in writing its efforts to obtain access. EPA or TDEC may then assist TVA in gaining access, to the extent necessary, to effectuate the response actions described herein, using such means as EPA or TDEC deems appropriate. TVA shall reimburse EPA and/or TDEC for all reasonable costs and attorneys' fees, if any, incurred by them in obtaining such access.

## XXIII. ACCESS TO INFORMATION

- 55. TVA shall provide to EPA, upon request, copies of all documents and information within its possession or control or that of its contractors or agents relating to activities at the Site or to the implementation of this Order, including, but not limited to, sampling, analysis, chain of custody records, manifests, trucking logs, receipts, reports, sample traffic routing, correspondence, or other documents or information related to the Work. TVA shall also make available to EPA at reasonable times, for purposes of investigation, information gathering, or testimony, their employees, agents, or representatives with knowledge of relevant facts concerning the performance of the Work.
- 56. TVA may assert business confidentiality claims covering part or all of the documents or information submitted to EPA under this Order to the extent permitted by and in accordance with Section 104(e)(7) of CERCLA, 42 U.S.C. § 9604(e)(7), and 40 C.F.R. § 2.203(b). Documents or information determined to be confidential by EPA will be afforded the protection specified in 40 C.F.R. Part 2, Subpart B. If no claim of confidentiality accompanies documents or information when they are submitted to EPA, or if EPA has notified TVA that the documents or information are not confidential under the standards of Section 104(e)(7) of CERCLA or 40 C.F.R. Part 2, Subpart B, the public may be given access to such documents or information without further notice to TVA.
- 57. TVA may assert that certain documents, records and other information are privileged under the attorney-client privilege or any other privilege recognized by federal law. If TVA asserts such a privilege in lieu of providing documents, it shall provide EPA with the following: 1) the title of the document, record, or information; 2) the date of the document, record, or information; 3) the name and title of the author of the document, record, or information; 4) the name and title of each addressee and recipient; 5) a description of the contents of the document, record, or information; and 6) the privilege asserted by TVA. However, no documents, reports or other information created or generated pursuant to the requirements of this Order shall be withheld on the grounds that they are privileged.
- 58. No claim of confidentiality shall be made with respect to any data, including, but not limited to, all sampling, analytical, monitoring, hydrogeologic, scientific,

chemical, or engineering data, or any other documents or information evidencing conditions at or around the Site.

## XXIV. RECORD RETENTION, DOCUMENTATION, AVAILABILTY OF INFORMATION

59. TVA shall preserve all documents and information relating to the Work performed under this Order, or relating to the release of fly ash from the Kingston Fossil Fuel Plant, in accordance with federal law and TVA's established document retention policies, but in no event shall such records be preserved for less than ten (10) years. At the end of this document retention period, and thirty (30) days before any document or information is destroyed, TVA shall notify EPA that such documents and information are available to EPA for inspection and, upon request, shall provide the originals or copies of such documents and information to EPA to the extent that they are not subject to any privilege. In addition, TVA shall provide documents and information retained under this Paragraph at any time before expiration of its document retention timeframe at the written request of EPA to the extent that they are not subject to any privilege. TVA shall also instruct its contractors and agents to preserve all documents, records, and information of whatever kind, nature or description relating to performance of the Work.

#### XXV. REIMBURSEMENT OF COSTS

- 60. <u>Payment of Past Response Costs</u>. TVA shall pay, within thirty (30) days after TVA's receipt from EPA of a demand for payment, EPA's Past Response Costs. EPA's Past Costs demand will include a SCORPIOS Report, which summarizes direct and indirect costs incurred by EPA and its contractors.
- a) Payment shall be made by Electronic Funds Transfer and directed to the Federal Reserve Bank of New York as follows: Federal Reserve Bank of New York, ABA=021030004, Account =68010727, SWIFT address = FRNYUS33, 33 Liberty Street, New York, New York 10045, Field Tag 4200 of the Fedwire message should read "D68010727 Environmental Protection Agency," and shall be accompanied by a statement identifying the name and address of TVA, the Site name, the EPA Region, Site/Spill ID Number A4XP, and the EPA docket number for this action.
- b) At the time of payment, TVA shall send notice that such payment has been made to EPA individuals identified in Section XL (Notices and Submissions), with a copy to:

Paula Painter Environmental Protection Specialist U.S. Environmental Protection Agency, Region 4 61 Forsyth St., SW Atlanta, Georgia 30303

and

EPA Cincinnati Finance Office 26 Martin Luther King Dr. Cincinnati, Ohio 45268

- c) The total amount to be paid by TVA pursuant to this Paragraph shall be deposited in the TVA Kingston Fossil Fuel Plant Release Site Special Account within the EPA Hazardous Substance Superfund to be retained and used to conduct or finance response actions at or in connection with the Site, or to be transferred by EPA to the EPA Hazardous Substance Superfund.
- 61. Payment of Future Response Costs. TVA shall pay EPA all Future Response Costs not inconsistent with the NCP. On a periodic basis, EPA will send TVA a bill requiring payment that includes a SCORPIOS Report, which includes direct and indirect costs incurred by EPA and its contractors. TVA shall make all payments within thirty (30) days of receipt of each bill requiring payment, except as otherwise provided in Paragraph 63 of this Order.
- a) Payment shall be made by Electronic Funds Transfer and directed to the Federal Reserve Bank of New York as follows: Federal Reserve Bank of New York, ABA=021030004, Account =68010727, SWIFT address = FRNYUS33, 33 Liberty Street, New York, New York 10045, Field Tag 4200 of the Fedwire message should read "D68010727 Environmental Protection Agency," and shall be accompanied by a statement identifying the name and address of TVA, the Site name, the EPA Region, Site/Spill ID Number A4XP, and the EPA docket number for this action.
- b) At the time of payment, TVA shall send notice that such payment has been made to the EPA individuals identified in Section XL (Notices and Submissions), with a copy to:

Paula Painter Environmental Protection Specialist U.S. Environmental Protection Agency, Region 4 61 Forsyth St., SW Atlanta, Georgia 30303

and

EPA Cincinnati Finance Office 26 Martin Luther King Dr. Cincinnati, Ohio 45268

c) The total amount to be paid by TVA pursuant to this Paragraph shall be deposited in the TVA Kingston Fossil Fuel Plant Release Site Special Account within the EPA Hazardous Substance Superfund to be retained and used to conduct or finance

response actions at or in connection with the Site, or to be transferred by EPA to the EPA Hazardous Substance Superfund.

- 62. In the event that payment for Past Response Costs or Future Response Costs is not made within thirty (30) days of TVA's receipt of a bill, TVA shall pay Interest on the unpaid balance. The Interest on Past Response Costs and Future Response Costs shall begin to accrue on the date of the bill and shall continue to accrue until the date of payment. Payments of Interest made under this Paragraph shall be in addition to such other remedies or sanctions available to EPA by virtue of TVA's failure to make timely payments under this Section, including but not limited to, payment of stipulated penalties pursuant to Section XXVIII.
- 63. TVA may dispute all or part of a bill for Past Response Costs or Future Response Costs submitted under this Order, if TVA alleges that EPA has made an accounting error, or if TVA alleges that a cost item is inconsistent with the NCP. Upon request, EPA will make available to TVA detailed documentation of any costs it has billed. If any dispute over costs is resolved before payment is due, the amount due will be adjusted as necessary. If the dispute is not resolved before payment is due, TVA shall pay the full amount of the uncontested costs to EPA as specified in Paragraphs 60 or 61 on or before the due date. Within the same time period, TVA shall pay the full amount of the contested costs into an interest-bearing escrow account. TVA shall simultaneously transmit a copy of both payments to the EPA person(s) identified in Section XL (Notices and Submissions) and in Paragraphs 60(b) and 61(b) above. TVA shall ensure that the prevailing party in the dispute shall receive the amount upon which they prevailed from the escrow funds plus interest within fifteen (15) days after the dispute is resolved.

## XXVI. DISPUTE RESOLUTION

- 64. If a dispute arises under this Order, the procedures of this Section (Dispute Resolution) shall apply. The Parties, in consultation with TDEC, shall make reasonable efforts to informally resolve disputes at the OSC/RPM/Project Coordinator level.
- 65. If TVA objects to any EPA action or determination, including any EPA disapproval, modification, or other decision taken pursuant to this Order, TVA shall notify EPA in writing of its objections, and the basis thereof, within fifteen (15) days of such action. Such notice shall set forth the specific points of the dispute, the position which TVA asserts should be adopted as consistent with the requirements of this Order, the basis for TVA's position, and any matters which it considers necessary for EPA's determination. For purposes of this Order, EPA actions, orders or determinations will include those actions taken by or on behalf of EPA or any of its employees, agents or designees.
- 66. EPA and TVA shall have an additional fifteen (15) calendar days from receipt by EPA of the notification of objection (Negotiation Period), during which time representatives of EPA and TVA may confer in person or by telephone to resolve any disagreement. If an agreement is reached during the Negotiation Period, the resolution

shall be in writing, signed by an authorized representative of both Parties. Such agreement shall be incorporated into and become an enforceable part of this Order. The Negotiation Period may be extended at the sole discretion of EPA, although such extensions shall not be unreasonably withheld. EPA's decision regarding an extension of the Negotiation Period shall not constitute an EPA action subject to dispute resolution.

- 67. If the Parties are unable to reach an agreement within the Negotiation Period, the dispute shall be elevated to the Superfund Division Director of EPA Region 4 and the Senior Vice President of TVA's Office of Environment and Research. The Superfund Division Director and TVA's Senior Vice President shall have an additional five (5) days to resolve the dispute and issue a written decision signed by both Parties. If the Superfund Division Director and TVA's Senior Vice President cannot reach a mutual agreement within the five-day period, the Superfund Division Director will issue a written decision on the dispute to TVA that provides the basis for his or her decision. This decision shall be incorporated into and become an enforceable part of this Order. TVA's obligations under this Order shall not be tolled by submission of any objections for dispute resolution under this Section.
- 68. Following resolution of the dispute, as provided by this Section, TVA shall fulfill the requirement that was the subject of the dispute in accordance with the agreement reached or in accordance with EPA's decision, whichever occurs.

## XXVII. FORCE MAJEURE

- 69. TVA agrees to perform all requirements under this Order within the time limits established in any EPA-approved schedule or work plan unless the performance is delayed by a force majeure for which an extension of the schedule shall be provided. For purposes of this Order, force majeure is defined as any event arising from causes beyond the control of TVA, or of any entity controlled by TVA, including but not limited to its contractors and subcontractors, that delays or prevents performance of any obligation under this Order, despite TVA's best efforts to fulfill the obligation. For purposes of this Order, force majeure includes, but is not limited to: Acts of God; fire; war; insurrection; civil disturbance; adverse weather conditions that could not have been reasonably anticipated; restraint by court order or order of public authority; or inability to obtain, after exercise of reasonable diligence, any necessary authorizations, approvals, permits or licenses due to action or inaction of any governmental agency or authority other than TVA. Force majeure does not include financial inability to complete the Work or increased cost of performance.
- 70. If any event occurs or has occurred that may delay the performance of any obligation under this Order, whether or not caused by a force majeure event, TVA shall verbally notify EPA within forty-eight (48) hours after TVA becomes or should have become aware that the event might cause a delay. Within five (5) days thereafter, TVA shall provide to EPA in writing an explanation of the event causing the delay or anticipated delay; the anticipated length of delay, including necessary demobilization and re-mobilization; the measures taken or to be taken to minimize the delay; the timetable

for implementation of the measures; TVA's rationale for attributing such delay to a <u>force majeure</u> event if it intends to assert such a claim; and a statement as to whether, in TVA's opinion, such event may cause or contribute to an endangerment to public health, welfare, or the environment. TVA shall take all reasonable measures to avoid and minimize the delay. Failure to comply with the notice provision of this Paragraph shall waive any claim of <u>force majeure</u> by TVA unless forgiven by EPA in its sole discretion.

71. If EPA determines that the delay in performance of a requirement under this Order is or was attributable to a <u>force majeure</u> event, the time period for performance of that requirement shall be extended as deemed necessary by EPA. Such an extension shall not alter TVA's obligation to perform or complete other tasks required by the Order which are not directly affected by the force majeure event.

## XXVIII. STIPULATED PENALTIES

- 72. EPA may assess stipulated penalties against TVA for violations of this Order in the amounts set forth below, unless excused under Section XXVII (Force Majeure). Compliance by TVA shall include completion of the activities under this Order in accordance with all applicable requirements of law, this Order, and the time schedules and work plans established and approved under this Order.
- 73. <u>Stipulated Penalty Amounts Work</u>. For each day TVA is in violation of an established schedule for the Work, EPA may assess a stipulated penalty for each violation up to the identified amounts shown below:

| Penalty Per Violation Per Day | Period of Noncompliance                       |
|-------------------------------|---|
| \$1,500.00                    | 1 <sup>st</sup> through 14 <sup>th</sup> day  |
| \$2,500.00                    | 15 <sup>th</sup> through 30 <sup>th</sup> day |
| \$5,000.00                    | 31 <sup>st</sup> day and beyond               |

74. <u>Stipulated Penalty Amounts – Work Plans and Reports</u>. For each day TVA fails to submit timely or adequate work plans, reports, or other written documents pursuant to this Order, EPA may assess a stipulated penalty for each violation up to the identified amounts shown below:

| Penalty Per Violation Per Day | Period of Noncompliance                       |
|-------------------------------|---|
| \$500.00                      | 1 <sup>st</sup> through 14 <sup>th</sup> day  |
| \$1000.00                     | 15 <sup>th</sup> through 30 <sup>th</sup> day |
| \$2000.00                     | 31st day and beyond                           |

In no event shall this Paragraph give rise to a stipulated penalty in excess of the amount set forth in Section 109 of CERCLA, 42 U.S.C. § 9609.

75. Stipulated penalties under this Section shall begin to accrue on the day after performance is due or the day a violation occurs, whichever is applicable, and shall continue to accrue until performance is satisfactorily completed or until the violation

ceases. However, stipulated penalties shall not accrue with respect to an issue subject to the Dispute Resolution provisions hereof, during the period, if any, after the conclusion of the Negotiation Period until the date that EPA issues a final decision regarding such dispute. Nothing herein shall prevent the simultaneous accrual of separate penalties for separate violations of this Order.

- 76. Stipulated penalties accruing under this Section shall become due and payable to EPA within thirty (30) days after TVA's receipt from EPA of a demand for payment of the penalties. Nothing in this Section is intended to create a debt or payment obligation to EPA unless EPA issues a written demand to TVA. All payments to EPA under this section shall be made by Electronic Funds Transfer as described in Section XXV. Stipulated penalties accruing during any dispute resolution period need not be paid until fifteen (15) days after the dispute is resolved by agreement, or by receipt of EPA's final decision. Only those penalties determined in the agreement or decision to be owing shall be due.
- 77. The payment of penalties shall not alter in any way TVA's obligations to complete performance of the Work required under this Order. Notwithstanding any other provision of this Section, EPA may, in its unreviewable discretion, waive any portion of stipulated penalties that have accrued pursuant to this Order. In exercising this discretion, EPA recognizes that TVA is an executive federal agency and instrumentality and that it conducts its activities on a not-for-profit basis in accordance with the TVA Act.

## XXIX. COVENANT NOT TO TAKE ADMINISTRATIVE ACTION BY EPA

78. Upon issuance of a Notice of Completion by EPA as set forth in Section XXXVIII, EPA covenants not to take administrative action against TVA pursuant to Section 106 and 107(a) of CERLCA, 42 U.S.C. §§ 9606 and 9607(a) for the Work, Past Response Costs and Future Response Costs. This covenant is conditioned upon the complete and satisfactory performance by TVA of its obligations under this Order.

## XXX. RESERVATION OF RIGHTS BY EPA

- 79. Nothing herein shall limit the available power and authority of EPA to take, direct, or order all actions necessary to protect public health, welfare or the environment or to prevent, abate, or minimize an actual or threatened release of hazardous substances, pollutants or contaminants, or hazardous or solid waste on, at, or from the Site.
- 80. The Covenant Not To Take Administrative Action set forth in Section XXIX does not pertain to any matters other than those expressly identified therein. EPA reserves, and this Order is without prejudice to, all rights against TVA with respect to all other matters, including, but not limited to:
  - a. claims based on a failure by TVA to meet a requirement of this Order;

- b. liability for costs not included within the definition of Past and Future Response Costs;
  - c. liability for penalties other than for noncompliance with this Order;
- d. liability for performance of response actions other than the Work specifically performed under this Order;
  - e. criminal liability;
- f. liability for damages for injury to, destruction of, or loss of natural resources, and for the cost of any natural resource damages assessments; and
- g. liability arising from the past, present, or future disposal, release or threat of a release of hazardous substances, pollutants or contaminants outside of the Site.

## XXXI. COVENANT NOT TO SUE BY TVA

- 81. TVA agrees not to assert any claims or causes of action against EPA, or its agents, contractors or employees, with respect to the Work, Past Response Costs, or Future Response Costs, or any other terms of this Order, including, but not limited to, any direct or indirect claim from or against EPA based on 42 U.S.C. §§ 9606(b)(2), 9607, 9611, 9612 or 9613, or any other provision of law, or any claims arising out of response activities at the Site. This does not prevent TVA from invoking Dispute Resolution pursuant to this Order.
- 82. Nothing in this Order shall be deemed to constitute approval or preauthorization of a claim within the meaning of 42 U.S.C. § 9611, or 40 C.F.R. § 300.700(d).

## XXXII. OTHER CLAIMS

- 83. By issuance of this Order, EPA assumes no liability for injuries or damages to persons or property resulting from any acts or omissions of TVA. EPA shall not be deemed a party to any contract entered into by TVA or its directors, officers, employees, agents, successors, representatives, assigns, contractors, or consultants in carrying out actions pursuant to this Order. TVA shall bear its own litigation costs and attorneys' fees.
- 84. Except as expressly provided in Section XXIX (Covenant Not to Take Administrative Action by EPA), nothing in this Order constitutes a satisfaction of, or release by EPA from, any claim or cause of action against TVA or any person not a party to this Order, for any liability such person may have under CERCLA, the Clean Water Act, RCRA, other statutes, or the common law.

- 85. In any subsequent proceeding initiated by EPA for injunctive or other appropriate relief relating to the Site, TVA shall not assert, and may not maintain, any defense or claim based upon the principles of waiver, res judicata, collateral estoppel, issue preclusion, claim-splitting, or other defenses based upon any contention that the claims raised by EPA in the subsequent proceeding were or should have been raised in the present matter.
- 86. No action or decision by EPA or TVA pursuant to this Order shall give rise to any right to judicial review, except as set forth in Section 113(h) of CERCLA, 42 U.S.C. § 9613(h).

#### XXXIII. CONTRIBUTION

87. The Parties agree that TVA is entitled, as of the Effective Date, to protection from contribution actions or claims as provided by 42 U.S.C. §§ 9613(f)(2) and 9622, for "matters addressed" in this Order. For purposes of this Paragraph, the "matters addressed" in this Order are the Work, Past Response Costs and Future Response Costs. Nothing in this Order precludes EPA or TVA from asserting any claims, causes of action, or demands against any persons not parties to this Order for indemnification, contribution, or cost recovery.

## XXXIV. INDEMNIFICATION

- 88. TVA shall indemnify, save and hold harmless EPA, its officials, agents, employees, contractors, subcontractors and representatives from any and all claims or causes of action arising from, or on account of, negligent acts of TVA, its officers, directors, employees, agents, contractors or subcontractors in carrying out actions pursuant to this Order. In addition, TVA agrees to pay EPA all costs incurred by EPA, including but not limited to attorneys' fees and other expenses, arising from or on account of claims made against EPA based on negligent acts of TVA, its officers, directors, employees, agents, contractors, subcontractors and any persons acting on behalf of or under their control, in carrying out activities pursuant to this Order. EPA shall not be held out as a party to any contract entered into by or on behalf of TVA in carrying out activities under this Order. Neither TVA nor any of its contractors shall be considered an agent of EPA nor shall EPA or any of its contractors be considered an agent of TVA.
- 89. EPA shall give TVA notice of any claim for which EPA plans to seek indemnification pursuant to this Section and shall consult with TVA prior to settling such claim.
- 90. TVA waives all claims against EPA for damages or reimbursement or for setoff of any payments made or to be made to EPA, arising from or on account of any contract, agreement, or arrangement between TVA and any person for performance of Work on or relating to the Site, including, but not limited to, claims on account of construction delays. In addition, TVA shall indemnify and hold harmless EPA with respect to any claims for damages or reimbursement arising from or on account of any

contract, agreement, or arrangement between TVA and any person for performance of Work on or relating to the Site, including, but not limited to, claims on account of construction delays.

#### XXXV. MODIFICATIONS

- 91. The EPA OSC or RPM, may, after notification to and consultation with TDEC and TVA, require modifications to any plan or schedule under this Order in writing or by oral direction. If EPA's OSC or RPM makes an oral modification, it shall be memorialized in writing within three (3) days thereafter; provided, however, that the effective date of the modification shall be the date of EPA's oral direction. Any other requirements of this Order may be modified in writing by mutual agreement of the Parties.
- 92. If TVA seeks permission to deviate from any approved plan or schedule, it shall submit a written request to EPA, and provide a copy to TDEC. TVA may not proceed with the requested deviation until receiving oral or written approval from EPA. TVA's written request for modification shall specify: (1) the proposed modification; (2) the basis for the modification; and (3) any related schedule or deadline that would be affected if the modification were granted.
- 93. No informal advice, guidance, suggestions, or comment by EPA regarding reports, plans, specifications, schedules or any other writing submitted by TVA shall relieve TVA of its obligation to obtain any formal approval as may be required by this Order, and to comply with all requirements of this Order, unless it is formally modified.

#### XXXVI. ADDITIONAL ACTIONS

94. EPA may, after notification to and consultation with TDEC and TVA, determine that additional actions, not included in an approved plan, are necessary to protect public health, welfare, or the environment at the Site in connection with the December 22, 2008, ash release. Unless otherwise stated by EPA, within thirty (30) days of receipt of notice from EPA that additional actions are necessary to protect public health, welfare, or the environment, TVA shall submit for approval by EPA, a work plan for the additional Work. Upon approval of the plan by EPA, TVA shall implement the plan for additional Work in accordance with the provisions and schedule contained therein. This Paragraph does not diminish or alter the EPA OSC's or RPM's authority to make oral modifications to any plan or schedule pursuant to Section XXXV (Modifications).

### XXXVII. COMMUNITY RELATIONS

95. TVA agrees to comply with all applicable laws and regulations, including CERLCA and the NCP, as well as all EPA policy and guidance, concerning community relations programs and public participation requirements, including, but not limited to, 40 C.F.R. §§ 300.415(n), 300.430(c) and 300.155.

- 96. TVA has developed and is implementing a community outreach plan. This plan responds to the need for an interactive relationship with interested community elements regarding environmental response activities being conducted by TVA at the Site. Within ninety (90) days of the Effective Date of this Order, TVA shall submit a Community Involvement Plan that specifically addresses the requirements of §§ 300.415(n) and 300.430(c) of the NCP to EPA for review and approval and shall provide a copy of the plan to TDEC. Pending EPA approval of the plan, TVA shall continue its community outreach efforts. Any proposed revision or amendment to TVA's Community Involvement Plan shall also be submitted to EPA for review and approval, and a copy of the revision or amendment shall be provided to TDEC.
- 97. Before issuance of any proposed press release regarding the selection or implementation of removal or remedial actions at the Site, TVA shall advise and consult with the EPA OSC/RPM and TDEC regarding the contents thereof.
- 98. In addition to any other relevant provisions of this Order, TVA agrees that Work conducted under this Order and any subsequent proposed response actions arising out of this Order shall comply with all the Administrative Record and public participation requirements of CERCLA, including Sections 113(k) and 117, 42 U.S.C. §§ 9613(k) and 9617, the NCP, and all applicable guidance.
- 99. TVA shall establish and maintain an Administrative Record File(s) at or near the Kingston Plant, in accordance with CERCLA Section 113(k), 42 U.S.C. § 9613(k), Subpart I of the NCP, and applicable EPA guidance. Within five (5) business days of the Effective Date of this Order, TVA shall establish a Site Repository at or near the Site and notify EPA of its location. The Site Repository developed by TVA shall be periodically updated and a copy of the Index to the Administrative Record File(s) will be provided to EPA. TVA will provide to EPA on request any document in the local Site Repository.
- 100. Within thirty (30) days of a request by EPA, TVA shall provide EPA with a Technical Assistance Plan (TAP) for providing and administering \$50,000 of TVA's funds to be used by a qualified community group to hire independent technical advisors during the response activities performed to address the mid-term and longer-term strategic Site objectives as identified in Paragraph 26 above. If EPA disapproves of or requires revisions to the TAP, in whole or in part, other than with respect to the amount of funding to be provided, TVA shall amend and submit to EPA a revised TAP that is responsive to EPA's comments within fifteen (15) days of receiving EPA's comments.

### XXXVIII. NOTICE OF COMPLETION

101. Following completion of all Work performed under this Order, TVA shall prepare and submit to EPA and TDEC a Completion Report to show that all required response actions have been completed. The information provided therein shall document compliance with the requirements of this Order and provide a consolidated record of all

response activities at the Site. In order for a Site to be eligible for completion, the following criteria must be met:

- a) Response Objectives specified have been met, and all cleanup actions and other measures identified have been successfully implemented;
  - b) The Site is protective of human health and the environment;
  - c) Land use controls are in place as appropriate; and
- d) The only remaining activities, if any, at the Site are long-term management activities (which may include long-term monitoring).

Information provided for response action completion shall be signed by TVA's signatory authority or designee, certifying that response activities have been completed in full satisfaction of the requirements of this Order and Agreement.

When EPA determines, after EPA's review of the Completion Report, that all Work has been fully performed in accordance with this Order, all monies owing pursuant to Sections XXV and XXVIII (if any) have been paid, and the Site meets the criteria for completion set forth in this Section, EPA will provide a written Notice of Completion to TVA. Upon issuance of the Notice of Completion by EPA, this Order shall terminate except for the continuing obligations set forth under Sections XV, XXIV, XXIX, XXXI, XXXIII, XXXIV, and long-term management activities identified in subparagraph "d" above.

#### XXXIX. EFFECTIVE DATE

102. The Effective Date of this Order shall be the date on which it is signed by EPA. EPA shall provide TVA electronically a fully executed copy of this Order and Agreement in pdf. format as soon as possible after it has been signed by EPA and shall promptly mail to TVA a fully executed copy of the Order and Agreement.

#### XL. NOTICES AND SUBMISSIONS

103. Whenever, under the terms of this Order, written notice is required to be given or a report or other document is required to be sent by one Party to another, its shall be directed to the individuals at the addresses specified below, unless those individuals or their successors give notice of a change to the other Party in writing.

As to EPA:

Mr. Leo Francendese U.S. EPA, Region 4 61 Forsyth Street, SW

Duke Energy Progress, LLC Late-Filed Exhibit No. 16 Docket No. E-2, Sub 1219 Page 327 of 439

Atlanta, Georgia 30303 francendese.leo@epa.gov

Mr. Craig Zeller U.S. EPA, Region 4 61 Forsyth St., SW Atlanta, Georgia 30303 zeller.craig@epa.gov

As to TVA:

Anda Ray Senior Vice President, Office of Environment and Research Tennessee Valley Authority 400 West Summit Hill Drive WT 11 Knoxville, Tennessee 37902-1401

| IT | IS | SO | ORD | ERED | <b>AND</b> | AGREED: |
|----|----|----|-----|------|------------|---------|
|----|----|----|-----|------|------------|---------|

DATE: 5/1/2009

DATE: 5/6/09

A. Stanley Meiburg

Acting Regional Administrator

Region 4

BY:

U.S. Environmental Protection

Agency

Tom Kilgore

Chief Executive Officer
Tennessee Valley Authority

Duke Energy Progress, LLC Late-Filed Exhibit No. 16 Docket No. E-2, Sub 1219 Page 328 of 439

# Appendix 3

September 21, 2010

# PA DEP Testimony to the U.S. Environmental Protection Agency—Hazardous Waste Management System; Identification and Listing of Special Wastes; Disposal of Coal Combustion Residuals from Electric Utilities

#### Re: Docket ID No. EPA-HQ-RCRA-2009-0640

Thank you for the chance to testify regarding the on EPA's proposed regulatory schemes for Coal Combustion Residues commonly referred to as CCR.

Pennsylvania has more than 30 years experience with CCR management. There are 43 coal-fired electricity producers in PA that generate approximately 20 million tons of CCR annually. Of that 20 million tons, 11 million tons/year are beneficially used in mine reclamation and as structural fill; slightly less than one million tons is used for other beneficial uses; and the remaining 8 to 9 million tons/year are landfilled.

Pennsylvania has residual waste regulations in place that govern the transportation, storage and disposal of CCR. These regulations and the implementation are based on sound scientific and engineering standards, and are similar to the Commonwealth's RCRA Subtitle D-authorized municipal waste regulations and have been in effect since 1992. CCR landfills that have been designed and operated in accordance with these regulations have not resulted in pollution to groundwater, surface water or air, and they are the same residual waste regulations that were used as a template for EPA's existing *Guide for Industrial Waste Management*. According to EPA, that Guide was designed to protect groundwater, surface water and air resources under an industrial Subtitle D-like residual waste program. It is unclear why EPA is now considering regulating CCR under Subtitle C instead of the current Subtitle D-like EPA guidelines, considering the physical and chemical characteristics of CCR present very little potential to adversely affect human health and the environment, compared to many other residual wastes.

Pennsylvania's successful management of CCR including tracking of historical waste analysis data generated over the last 30 years, does not indicate in any way the need for the hazardous waste designation EPA is proposing.

#### **Comparison of Subtitle C and D**

According to EPA, the design and performance standards for the proposed regulations will likely be the same no matter what regulatory scheme is chosen. The major differences is that EPA asserts such a designation is necessary for it to retain and exercise appropriate enforcement authority and EPA cannot require States to permit landfills under Subtitle D.

Speaking to the latter, with the requirement for liners, groundwater monitoring and other operational requirements, Pennsylvania regulations already require performance standards that exceed those outlined in the Subtitle C or D proposal. By current regulation, PA also requires facilities to be permitted or authorized to manage CCR.

As to the need for EPA enforcement, Pennsylvania's Solid Waste Management Act authorizes State enforcement authority and PA DEP exercises that authority. Subtitle D requires State programs to have the necessary enforcement authority as part of the federal approval process. This approach has been successful for Pennsylvania, as evidenced by the <u>96 %</u> compliance rate. In addition, EPA already has broad RCRA 7003 enforcement authority to address Potential Substantial Threats or Endangerments to Human Health and the Environment as well as imminent hazards for releases of solid waste and, therefore, does not need the additional authority under Subtitle C.

The detrimental effects of Subtitle C management are substantial:

#### Capacity

Pennsylvania has no Subtitle C disposal capacity and changing the CCR designation to HW would require transporting more material away from the power plant site and out of state, thereby causing an increase in carbon foot print and requiring existing facilities to meet unnecessary burdens. These practices are not more protective to the environment or to public health than what is currently in place under Pennsylvania regulations.

#### Cost

Regulating a non-hazardous waste under Subtitle C will require the addition of staff and resources in Pennsylvania and other states who are already fiscally strapped, thereby diverting resources away from protection from threats to public health and the environment posed by actual hazardous wastes

#### Beneficial Use

Regulation of CCR under Subtitle C would impede recycling, discourage beneficial use and instead cause a valuable resource to be disposed in landfills and/or surface impoundments simply due to the hazardous waste designation. In addition, some activities that are currently conducted under Pennsylvania's beneficial use program, such as certain structural fill projects, could potentially fall within the disposal regulations and essentially prohibit this very successful program.

When additional requirements are necessary to improve its waste management and beneficial use program, Pennsylvania modifies its regulations. As an example, new regulations regarding the beneficial use of CCR are expected to be finalized later this year. These revised regulations will require more frequent chemical analysis of CCR and, in cases of large quantity unencapsulated uses, groundwater monitoring. Some of these new requirements are based on recommendations from the National Academy of Sciences and are designed to provide further evidence of the efficacy of Pennsylvania's current program.

### **Conclusion**

Pennsylvania does not believe that additional regulations for the management of coal ash beyond RCRA subtitle D are warranted. Classifying coal combustion residues under RCRA Subtitle C would create unnecessary burdens to current management practices without producing any greater degree of environmental or public health protection. The commonwealth opposes the rulemaking for Subtitle C and is in support of a Subtitle D proposal with implementation authorization continuing with the States.

Duke Energy Progress, LLC Late-Filed Exhibit No. 16 Docket No. E-2, Sub 1219 Page 331 of 439

# Appendix 4

Duke Energy Progress, LLC Late-Filed Exhibit No. 16 Docket No. E-2, Sub 1219 Page 332 of 439



#### UNITED STATES ENVIRONMENTAL PROTECTION AGENCY

WASHINGTON, D.C. 20460

JAN 3 0 2009

OFFICE OF CONGRESSIONAL AND INTERGOVERNMENTAL RELATIONS

The Honorable Edward J. Markey Chairman Committee on Energy and Commerce Subcommittee on Energy and Environment U.S. House of Representatives Washington D.C. 20515

Dear Mr. Chairman:

Thank you for your letter of January 13, 2009, to the U.S. Environmental Protection Agency's (EPA's) former Administrator Stephen L. Johnson requesting information related to the regulation of the bi-products associated with coal-burning power plants.

EPA respects your role as Chairman and is committed to providing the Subcommittee with information necessary to satisfy its oversight activities to the extent possible, consistent with Constitutional and statutory obligations. We are coordinating with various offices and working diligently to identify, assemble, and review the documents and information requested in your letter. However, because of the extensive information requested, we will need additional time to fully respond to your questions. In the meantime, we are coordinating with your staff to provide a briefing to share information on this issue.

Again, let me assure you that we are working to respond to your request as expeditiously as possible. If you have further questions, please contact me or your staff may contact Amy Hayden in EPA's Office of Congressional and Intergovernmental Relations at (202) 564-0555.

Sincerely,

Joyce K. Frank

Acting Associate Administrator

Responses to Congressman Edward J. Markey's January 13, 2009 letter related to the regulation of the by-products associated with coal-burning power plants.

1. Does EPA believe that coal ash and/or other by-products associated with coalburning power plants should be designated a hazardous waste? If not, why not? If so, why has it not already done so? Please provide copies of all EPA studies, memos, draft proposals and other correspondence related to any deliberations associated with such designation, or alternate approaches to regulating these materials.

Section 3001(b)(3)(A)(i) of the Resource Conservation and Recovery Act (RCRA) temporarily excluded certain large-volume wastes, including by-product wastes associated with the combustion of coal and other fossil fuels from being regulated as hazardous wastes under Subtitle C of RCRA, pending completion of a Report to Congress and a Regulatory Determination by the Administrator of the Environmental Protection Agency either to promulgate regulations under Subtitle C of RCRA or deem that such regulations are unwarranted. With this legislative mandate, EPA published its Part 1 Regulatory Determination for large-volume utility coal combustion wastes in the Federal Register in August 1993 (see 58 FR 42466). At that time, EPA determined that fly ash, bottom ash, boiler slag, and flue gas emission control dust from coal burning utilities did not warrant regulation as hazardous waste and, thus, remained excluded from regulation under Subtitle C of RCRA §261.4(b)(4). That Regulatory Determination addressed the large-volume utility coal combustion waste streams, but it did not cover co-management of all wastes generated at facilities that combust coal and other fossil fuels. Therefore, EPA conducted additional research regarding the co-management of the large-volume utility wastes with the remaining wastes generated at facilities that combust coal and other fossil fuels.

In May 2000, EPA issued its Part 2 Regulatory Determination ("Regulatory Determination on Wastes from the Combustion of Fossil Fuels" (65 FR 32214)), addressing the remaining wastes that had not been considered under its 1993 Regulatory Determination. In the May 2000 Regulatory Determination, the Agency likewise concluded that these wastes did not warrant regulation as hazardous waste under Subtitle C of RCRA. However, EPA also determined that national non-hazardous waste regulations under RCRA Subtitle D were appropriate for coal combustion wastes disposed of in surface impoundments and landfills and used as fill in surface or underground mines. For disposal in landfills and surface impoundments, EPA based its determination on the following considerations: (1) the constituents present in these wastes include toxic metals that could present a danger to human health and the environment under certain conditions; (2) the Agency identified 11 documented cases of proven dangers to human

<sup>&</sup>lt;sup>1</sup> The wastes addressed under this Regulatory Determination includes: (1) large-volume coal combustion wastes (i.e., fly ash, bottom ash, boiler slag and flue gas emission control dust) generated at electric utility and independent power producing facilities that are co-managed together with certain other coal combustion wastes; (2) coal combustion wastes generated at non-utilities; (3) coal combustion wastes generated at facilities with fluidized bed combustion technology; (4) petroleum coke combustion wastes; (5) wastes from the combustion of mixtures of coal and other fuels (i.e., co-burning of coal with other fuels where coal is at least 50% of the total fuel); (6) wastes from the combustion of oil; and (7) wastes from the combustion of natural gas.

Duke Energy Progress, LLC Late-Filed Exhibit No. 16 Docket No. E-2, Sub 1219 Page 334 of 439

health and the environment by the improper management of these wastes in landfills and surface impoundments; (3) lack of controls, such as liners and groundwater monitoring, at many sites; and (4) while there had been improvements in state regulatory programs, there also were gaps identified in state oversight. The 2000 Regulatory Determination is enclosed with this letter.

Deleted: attached

The supporting technical documents, including the findings of the 1999 Report to Congress (RTC) "Wastes from the Combustion of Fossil Fuel," are quite voluminous and are accessible at: <a href="http://www.epa.gov/epawaste/nonhaz/industrial/special/fossil/volume\_2.pdf">http://www.epa.gov/epawaste/nonhaz/industrial/special/fossil/volume\_2.pdf</a> and <a href="http://www.epa.gov/epawaste/nonhaz/industrial/special/fossil/fsltech.htm">http://www.epa.gov/epawaste/nonhaz/industrial/special/fossil/fsltech.htm</a>. They address the characterization of coal combustion waste (CCW), its management practices as of the mid-1990s, State regulatory programs, damage cases associated with the management of CCW, the economic and cost impact analysis of rulemaking, and human health and ecologic risk analysis of fossil fuel combustion (since superseded by a 2006 study).

Since the May 2000 Regulatory Determination, additional information and data became available, which EPA believed should be noticed for public comment as part of the Agency's evaluation regarding the development of regulations under Subtitle D of RCRA of CCW. Thus, this information was made available for public comment in its August 2007 Notice of Data Availability (NODA) (72 FR 49714; please see enclosure). This included an update of waste management practices—a joint U.S. Department of Energy (DOE) and EPA report entitled, Coal Combustion Waste Management at Landfills and Surface Impoundments, 1994-2004, a further assessment of damage cases, and a draft risk assessment.<sup>2</sup> In addition, the draft risk assessment was subject to peer review, which was completed in September 2008. The 2007 Notice of Data Availability, as well as its accompanying technical documents, the public comments, citizen and industry proposals for the regulation of coal combustion waste, and the results of the draft risk assessment's peer review, are all accessible at the NODA's docket, at <a href="http://www.regulations.gov/search/search\_results.jsp?css=0&&Ntk=All&Ntx=mode+matchall&Ne=2+8+11+8053+8054+8098+8074+8066+8084+8055&N=0&Ntt=epa-hq-rcra-2006-0796&sid=11F141358782</a>.

We are most willing to discuss any of these documents with you or your staff and provide paper copies of those documents if that will be helpful.

<sup>&</sup>lt;sup>2</sup> The NODA also solicited comment on a February 2004 Petition for Rulemaking submitted by the Clean Air Task Force and the Hoosier Environmental Council, jointly with a number of citizens' groups to prohibit the placement or disposal of CCW into groundwater or surface water, and two suggested approaches for managing CCW in landfills and surface impoundments. One approach is a Voluntary Action Plan that was formulated by the electric utility industry, the other approach is a proposed framework prepared by a number of citizens' groups for federal regulation of CCW disposed of in landfills and surface impoundments under Subtitle D or RCRA

2. Does EPA believe it has sufficient legal authority under existing environmental statutes to regulate coal ash, heavy metals, and other hazardous wastes associated with coal-burning power plants? If so, why hasn't EPA used this authority? If not, what changes in the law would be needed to give EPA the authority to protect public health and the environment from these wastes?

Yes, EPA believes that it currently has sufficient legal authority to regulate such wastes and does not believe that any changes to environmental statutes are required. However, we expect that, if EPA were to pursue regulation of CCW under Subtitle C, some may raise questions about the Agency's legal authority to reconsider its earlier Regulatory Determinations.

With respect to your second question, and as noted in our response above, following the May 2000 "Regulatory Determination on Wastes from the Combustion of Fossil Fuels" (65 FR 32214), EPA collected new data and performed additional analyses, including a draft risk assessment and further evaluation of possible damage cases on the management of CCW in landfills and surface impoundments, particularly since the information on which the May 2000 Regulatory Determination was made was based on information collected prior to 1995. The Agency believed that this additional information and data were important considerations in the development of regulations. EPA made this information available for public comment in its August 2007 NODA and received close to 400 comments. In addition, the draft risk assessment was subject to peer review, which was completed in September 2008. EPA is carefully analyzing the comments and recommendations we have received, including those from the peer reviewers, and will consider this information as we continue to follow up on the regulatory determination on the management of CCW in surface impoundments and landfills.

3. If coal ash and/or other by-products associated with coal-burning power plants was designated as hazardous waste, please detail the potential regulatory steps that would follow such designation.

If the Agency were to decide to regulate CCW as hazardous under Subtitle C of RCRA, than we believe that we would need to revise the Regulatory Determination, which we think may be done at the same time that we propose to remove the existing regulatory exemption. However, as part of this effort, we believe that the Agency would need to describe the facts that cause the Agency now to believe that CCW needs to be regulated under Subtitle C, as opposed to Subtitle D. In addition, while RCRA does not specifically spell out the process by which we revise the Regulatory Determination, we would expect that based on recent cases in other contexts, the Agency would find it advisable to go through the same process we followed to establish it—in other words, "after public hearings and an opportunity for comment." That would require the Agency to develop a proposed regulation, including the needed supporting documentation; publish that proposal in the Federal Register for public comment and hold public hearings; analyze and respond to those comments; and then publish a final regulation. After EPA promulgated the federal rule, States authorized for the RCRA program would then have to adopt those regulations (or regulations no less stringent than the federal ones) and receive authorization from EPA.

Deleted: This is in line with the Confirmation hearing testimony by Lisa Jackson, at the time, the Administrator designee, ¶

4. Has EPA examined the manner in which these materials are stored? For example, last month's accident occurred in a storage pond. Given the dangers that these materials particularly pose to the surrounding water system, has EPA considered the wisdom of allowing them to be stored in this manner in the first place? Please provide copies of all EPA studies, memos, draft proposals and other correspondence related to any deliberations associated with the regulation of the types of facilities that can be used to store these materials.

EPA's May 2000 Regulatory Determination did not specifically address surface impoundment integrity. The discharge of fly ash and bottom ash transport water (i.e., the discharge from ash ponds) is regulated by National Pollutant Discharge Elimination System (NPDES) permits and EPA has issued national effluent limitations that apply to the discharge.

NPDES regulations issued under the authority of the Clean Water Act require that all NPDES permits include standard conditions that include the requirement to "...properly operate and maintain all facilities and systems of treatment and control (and related appurtenances)...to achieve compliance with the conditions of this permit" (See 40 CFR part 122.41(e)). In addition, best management practices can be included in NPDES permits as necessary to achieve limitations or to carry out the purpose and intent of the CWA (See 40 CFR part 122.44(k). Given the TVA ash pond collapse, EPA is presently reviewing existing permits to assess if additional requirements or guidance are appropriate.

The national effluent limitations, issued by EPA in 1982 and codified at 40 CFR part 423, imposed an effluent limit of "zero discharge" for fly ash transport water from new facilities. As a result, nearly all generating units built after 1982 have avoided using storage ponds for fly ash by using ash handling systems that keep the fly ash dry. The dry fly ash is either disposed of in a landfill or sold for cement manufacturing or other uses. Some plants built before 1982 also use dry handling practices for fly ash, although placing the wet fly ash in storage ponds is commonplace at other plants. Bottom ash is typically stored in ponds at most plants; however, some plants handle the bottom ash with a dry process that avoids the need for a storage pond.

EPA is nearing the end of a multi-year study of the steam electric power generating industry to determine whether the national effluent limitations guidelines warrant revision. Upon concluding the study later this year, EPA will determine whether to initiate a rulemaking process. The study has expended substantial effort in reviewing discharges from coal-fired power plants, including ash ponds. EPA's review of operating practices and wastewater management technologies will include an assessment of technologies that enable some plants to manage their fly ash and bottom ash without the need for ash storage ponds.

EPA has compiled a substantial amount of documentation over the course of the study. Most of these documents were made available for public review as part of the docket for EPA's Final 2008 Effluent Guidelines Program Plan.<sup>3</sup> The documentation is rather voluminous. We would

<sup>&</sup>lt;sup>3</sup> As required by Section 304(m), of the Clean Water Act, EPA publishes an Effluent Guidelines Program Plan presenting a schedule for the annual review and revision of promulgated effluent guidelines and for identifying industrial categories without effluent guidelines that might need to be regulated to prevent or control pollution. The Plan also presents the results of ongoing and completed industry studies. The Final 2008 Effluent Guidelines

Duke Energy Progress, LLC Late-Filed Exhibit No. 16 Docket No. E-2, Sub 1219 Page 337 of 439

welcome the opportunity for the EPA staff conducting the study to discuss with your staff the specific materials that have been compiled so we can best respond to your request for documents.

Program Plan was published September 15, 2008 (73 FR 53218). The most recent status report for the detailed study of the steam electric power generating point source category was published August 2008 (http://epa.gov/waterscience/guide/304m/2008/steam-detailed-200809.pdf). The entire docket for the 2008 Plan is available at <a href="http://www.regulations.gov/fdmspublic/component/main?main=DocketDetail&d=EPA-HQ-OW-2006-0771">http://www.regulations.gov/fdmspublic/component/main?main=DocketDetail&d=EPA-HQ-OW-2006-0771</a>.

Duke Energy Progress, LLC Late-Filed Exhibit No. 16 Docket No. E-2, Sub 1219 Page 338 of 439

# Appendix 5

Statement of Thomas H. Adams, Executive Director, American Coal Ash Association

#### House Small Business Subcommittee on Rural Development, Entrepreneurship and Trade

July 22, 2010

#### "Is Coal Combustion Product Recycling an Endangered Industry?"

Mr. Chairman, my name is Thomas Adams. I am the Executive Director of the American Coal Ash Association (ACAA). I would like to thank you for the opportunity to speak to you and the subcommittee today on a major recycling success story that is solving serious environmental concerns while brining significant economic benefits to the US economy. Founded in 1968, the ACAA's mission is to encourage the use of coal combustion products (CCP) in ways that simultaneously benefit the environment, are technically appropriate, commercially viable, and contribute to a more sustainable society.

Beneficial use, another term for recycling, means many things to many people. To most people it means finding ways to use as much of our resources in ways that protect human health and the environment at a reasonable cost, and to make finished products that perform as well or better than products made with 100% virgin materials. By maximizing beneficial use of CCP we help preserve virgin resources for use by future generations while minimizing the effects of current economic development on the environment.

The coal combustion product family consists of materials remaining after the combustion of coal in coal fueled power plants. The primary products are fly ash, bottom ash, boiler slag, and flue gas desulfurization (FGD) gypsum. In the most recent Production and Use Survey (1) conducted by the ACAA for 2008, approximately 136 million tons of CCP were generated. CCP is the second largest waste stream in the US following only municipal solid waste. Of this 136 million tons of CCP generated, approximately 44% went into a variety of beneficial uses. This means 60 million tons of CCP was recycled in 2008 rather than being sent to disposal facilities. Since 2000 the recycling rate has risen from 30% to 44%. In its most recent Economic Assessment of the impact of coal ash on the US economy, the American Coal Council determined that the annual benefit to the US economy is in the range of \$9 to \$10 billion.

Today I would like to focus on some important beneficial uses of CCP. Fly ash is a fine powder-like substance with much the same consistency as Portland cement. It is collected in power plants and handled much like Portland cement. Because of its mineral constituency it is a valuable raw material in the manufacture of Portland cement for many producers. Depending on the quality and consistency of the fly ash, it is also a very valuable supplementary cementitious material (SCM) for the manufacture of concrete products, being used to replace and optimize Portland cement in concrete mixtures.

FGD gypsum is produced when flue gases are scrubbed in power plant stacks to remove sulfur dioxide (SOX) and nitrogen oxide (NOX) from emissions into the environment. Scrubbing by using lime or limestone in forced oxidation processes produces a synthetic gypsum which has purity equal to or greater than mined gypsum.

Fly ash is used as a raw material in cement manufacture due to the minerals present, mostly silicates. The use of fly ash as a raw material means that there is less mining of virgin sources to obtain those minerals silicates. Cement manufacturers balance the availability of fly ash with the needed chemical composition with the availability of material from virgin sources. The cement producers consumed 4.2 million tons of fly ash in 2008.

Fly ash is also used in concrete manufacture as a supplementary cementious material or SCM. The 2008 Production and Use Survey showed that 14 million tons of fly ash went into concrete products. The use of fly ash in concrete mixtures allows for a reduction in the quantity of Portland cement required for achieving desired results. The material is mistakenly called a cement replacement for his use. The reality is that there are performance characteristics in concrete structures that can only be achieved by the use of fly ash or other SCMs. Portland cement is important but cannot always provide the characteristics that create the high-performance and long-term durability in concrete structures. High performance can mean many things such as low permeability reducing water migration which can initiate corrosion of reinforcing steel, high compressive strength which enables structures to carry heavy loads with smaller members, or resistance to aggressive environments which attack the concrete matrix resulting in reduced service life.

Having spent many years in the ready mixed concrete industry, I can personally attest to the importance of having a valuable tool such as fly ash available to solve the performance requirements in a wide variety of construction projects. Today producers are taking the use of fly ash and other SCMs into new and expanded applications. Innovation is on a fast track.

A major benefit from the use of fly ash to optimize concrete mixtures is reduction of carbon dioxide emissions. When fly ash is used in a concrete mixture reducing the amount of Portland cement required, less CO2 is emitted as the quantity of cement manufactured is reduced. By avoiding 1 ton of cement manufacture, approximately 0.9 tons of CO2 are not emitted by the cement plant. Since 2000 over 117 million tons of carbon dioxide emissions have been avoided by the use of fly ash in concrete mixtures. There remains a large capacity in the concrete industry to increase the amount of fly ash used. One of the top environmental priorities of President Obama's administration is reduction of green house gases. The concrete industry has been doing its part to achieve reductions for some time. With the proper incentives, this reduction can be maintained and accelerated.

In 2008 8.5 million tons of FGD gypsum went into wallboard products. Approximately 35% of the wallboard manufactured in the United States is made with FGD gypsum. Wallboard manufacturers have intentionally located plants close to utilities to take advantage of logistical benefits. In some cases the material is moved by conveyor from the power plant to the wallboard plant. This process is more sustainable than the use of mined gypsum as mining, handling, and transportation impacts are virtually eliminated. This also results in CO2 reduction from elimination of mining and handling, and transportation.

There are other important markets for CCP beneficial use that I will not mention today due to time constraints. A common thread among all of these uses is the achievement of the mission of the ACAA in environmental safety, technical performance, economic viability, and contribution to a more sustainable society.

So what would endanger the continued successful beneficial use of these products?

In an effort to create regulations for the disposal of coal combustion products, the US Environmental Protection Agency (EPA) has published a proposal which contains an option which would treat CCP as a hazardous waste when destined for disposal under Subtitle C of the Resource Conservation and Recovery Act of 1976. The agency has expressed a preference for this option since it provides EPA with the authority to enforce disposal regulations. Subtitle D of RCRA, intended for non-hazardous wastes, places enforcement authority with individual states. EPA suggests that certain beneficial uses of CCP could be exempted from hazardous waste regulation. Therefore beneficial use in cement, concrete, and wallboard would continue though the same materials intended for disposal would be considered to be hazardous. We believe this "hazardous" designation would create a stigma resulting in rejection by the market place for the following reasons.

A primary concern in the market is liability exposure. Unfortunately our laws permit tort activity even when there is no evidence of damage. (The concrete industry is particularly sensitive to this having survived a siege of suits in southern California known as the "sulfate wars".) In discussions with engineers, contractors, and concrete producers over the last several months, it is clear that the use of fly ash would be severely curtailed due to fear of tort or class action suits. Many in the concrete industry do not believe EPA's assertion that the exemption would provide all the protection needed. Many do believe that a lawyer could make a simple argument to a jury that the fly ash in the disposal facility has exactly the same physical and chemical characteristics as the fly ash in the concrete in a home, hospital, daycare center, or school. Therefore if it is hazardous in the disposal facility, it must be hazardous in those structures thereby opening the door to financial claims. Even if a claim is found to be minimal, the costs of legal defense are something firms want to avoid.

The stigma of CCP as a hazardous waste also opens the possibility of negative marketing by suppliers of competitive materials. We have already seen examples in markets for shingles, bricks, and concrete blocks of advertising which attacks products containing CCP saying, "Our products do not contain hazardous waste. Do yours?" The public will always opt for materials that do not have the taint of some sort of hazardous status.

By placing CCP for disposal under hazardous waste rules, the efforts of entrepreneurs to bring new products to the market allowing the use of disposed CCP are effectively halted. Once CCP is placed in disposal it is a hazardous waste. New processes are being developed that would use large quantities of CCP some of which could come from disposal sites. The ability of these entrepreneurs to develop commercially competitive products would be crippled if they could not take advantage of the most economically feasible sources. Again, markets given a choice between products containing a hazardous component versus products with non-hazardous components will opt for the non-hazardous option.

Venture capital needed to get new businesses related to CCP beneficial use would be more difficult to obtain. Some ACAA members who have been relying on such funding report hesitation from their financial sources.

The beneficial use of coal combustion products across the country is being affected by the mere shadow of EPA hazardous waste regulations. Many of the affected entities are small businesses and entrepreneurs who are driving the effort to increase recycling. A few examples follow.

- CalStar Products opened a plant in Wisconsin to manufacture bricks and pavers from fly ash. Their process uses fly ash as a primary ingredient and consumes 85% less energy that used in producing traditional clay bricks. The Brick Industry Association has published comments that infer safety concerns because the brick is made with "hazardous wastes".
- A large manufacturer of shingles for residential roofing, Reed Minerals, a division of Harsco, had to threaten legal action against a proposed advertising campaign of a competitor. The campaign theme was "Our shingles do not contain hazardous waste. Do yours?"
- Colorado State University does research on coal combustion products. A utility that furnishes coal ash samples for this research has informed the university that no samples will be furnished should the EPA promulgate a hazardous waste rule of any kind.
- Anne Arundel County in Maryland has prohibited the use of fly ash in county construction projects pending EPA's final rule.
- The Los Angeles Unified School District has stopped allowing the use of fly ash in all LAUSD projects pending EPA's final rule.
- Calera, an emerging technology company based in California, is researching alternatives to fly
  ash for the manufacture of construction aggregates and cement supplements to avoid the
  requirements of processing a hazardous waste.

The EPA actually states in their June 21, 2010 proposal that beneficial use will increase under a Subtitle C hazardous waste regulations. The agency believes utility companies will be financially motivated to find ways to treat and handle CCP so market acceptance increases. Again, the markets have told the ACAA that any form of Subtitle C rule will stigmatize CCP and cause users to turn to other materials.

The stigma created by a hazardous waste regulation could have other unintended consequences. For example, insurance underwriters may include exclusion for projects utilizing CCP when renewing professional liability insurance for designers and general liability insurance for contractors. This would have a chilling effect on beneficial use regardless of EPA claims.

Mr. Chairman, the American Coal Ash Association Board of Directors recently passed a resolution (attached) calling for national standards for the regulation of coal ash disposal under Subtitle D of RCRA. The same resolution calls states the association opposition of any form of Subtitle C regulation.

The requirements for disposal facilities receiving coal combustion residuals are virtually identical under either of the EPA's proposed rules. Therefore it makes sense to avoid designating these materials as hazardous wastes for any reason and risk loss of a major environmental success story which contributes to our economy and helps create a more sustainable society. Subtitle C of the Resource Conservation and Recovery Act of 1976 is for truly hazardous waste. Since coal combustion residuals do not fail the characteristic tests which would qualify them to be labeled as hazardous, and none of the cited damage cases are a result of beneficial use, there is no justification for the assault on the beneficial use that is contained in the EPA proposal. Regulation of disposal under Subtitle D provides sufficient protection to human health and the environment without implying a danger that has yet to be proven.

Duke Energy Progress, LLC Late-Filed Exhibit No. 16 Docket No. E-2, Sub 1219 Page 343 of 439

EPA has stated publically that Subtitle D regulations are sufficient for coal combustion residual disposal (2). However a primary reason the EPA to favors Subtitle C regulations is that enforcement authority lies with the EPA under Subtitle C while enforcement authority under Subtitle D is resides with the states. The answer to resolving this concern is to amend the Resource Conservation and Recovery Act of 1976 to provide enforcement authority for the disposal of coal combustion residuals under Subtitle D to the US Environmental Protection Agency.

Respectfully,

Thomas H. Adams
Executive Director
American Coal Ash Association, Aurora, CO

- (1) 2008 Coal Combustion Product Production & Use Survey Report, American Coal Ash Association
- (2) Matthew Hale, Director, U.S. EPA Office of Resource Conservation and Recovery, to the Environmental Council of the States, September, 2009

Attachment: ACAA Board of Directors Resolution

#### Attachment

### Resolution of the American Coal Ash Association

The Board of Directors of the American Coal Ash Association ("the ACAA"), a trade organization established in 1968 and devoted exclusively to encouraging beneficial uses of coal combustion products ("CCP") in ways that are beneficial to the environment, economy, and society, conducted a meeting on April 12, 2010, at which time the Directors duly adopted the following resolution.

WHEREAS, the ACAA has considered the salient features of changes to regulation of coal combustion byproducts ("CCB") under the Resource Conservation and Recovery Act ("RCRA") of 1976;

WHEREAS, ACAA members are engaged daily in the beneficial use of CCP and thousands of green jobs within the CCP industry depend upon meeting numerous standards and specifications set by ASTM International ("ASTM"), the American Concrete Institute ("ACI"), the American Association of State Highway and Transportation Officials ("AASHTO"), the U.S. Environmental Protection Agency ("EPA"), and other state and local agencies;

WHEREAS, EPA has discussed proposing to regulate CCB under RCRA under either Subtitle C - Hazardous Waste, Subtitle D - Non-hazardous waste, or a "hybrid" approach that would include some form of Subtitle C regulation;

WHEREAS, numerous states, ASTM, ACI and AASHTO have signaled in written correspondence to EPA that a Subtitle C regulatory approach, including a hybrid approach, would have negative impacts upon beneficial uses of CCP;

Duke Energy Progress, LLC Late-Filed Exhibit No. 16 Docket No. E-2, Sub 1219 Page 345 of 439

WHEREAS, ACAA agrees that regulating CCB under Subtitle C of RCRA, including

under a hybrid approach, would have negative impacts upon beneficial uses of CCP;

WHEREAS, implementation of CCB rules under Subtitle D would occur sooner than

under Subtitle C, thousands of CCP green jobs would be saved and negative impacts to

the beneficial uses under Subtitle C would be avoided; and,

WHEREAS, ACAA supports regulation of CCB under Subtitle D of RCRA;

The following resolutions were offered, seconded, and adopted.

BE IT RESOLVED that the Directors of ACAA support federal regulation of CCB

under Subtitle D of RCRA; and,

BE IT FURTHER RESOLVED that the Directors of ACAA oppose regulation of CCB

under Subtitle C of RCRA

CERTIFICATION

I, the undersigned, Secretary, do certify that the foregoing is a true exact and correct

copy of a resolution adopted at a lawfully held meeting of the trade organization on the

1279

day of

2010.

Signature

Print Name

7

Duke Energy Progress, LLC Late-Filed Exhibit No. 16 Docket No. E-2, Sub 1219 Page 346 of 439

Duke Energy Progress, LLC Late-Filed Exhibit No. 16 Docket No. E-2, Sub 1219 Page 347 of 439

# Appendix 6

Testimony of Craig H. Benson, PhD, PE, DGE Wisconsin Distinguished Professor, University of Wisconsin-Madison

## House Small Business Subcommittee on Rural Development, Entrepreneurship and Trade

### 22 July 2010

Thank you for the opportunity to share with this Committee my experience and opinions regarding coal combustion products (CCPs), and their beneficial use in sustainable construction. My name is Dr. Craig H. Benson, PhD, PE, DGE and I am a Professor of Geological Engineering and Civil & Environmental Engineering at the University of Wisconsin-Madison. I am also Director of the Recycled Materials Resource Center at the University of Wisconsin-Madison and Chair of Geological Engineering. I teach courses that pertain to sustainable construction and management of byproducts, amongst other topics. I also conduct research and development on the safe and wise use of industrial byproducts in sustainable construction as well as the environmentally sound management of wastes. I have been involved in scientific research and engineering practice for more than 25 years.

For 20 years, I have been conducting research on sustainable construction with industrial byproducts, including CCPs such as fly ash, bottom ash, and flue-gas desulphurization (FGD) residuals. Over the last decade, CCPs have become a large part of my research program because of the many ways in which they can be used safely, wisely, and economically in sustainable construction. This research has been supported by a broad distribution of stakeholders, including the US government, state governments, local governments, and industry. I strongly believe that using CCPs for infrastructure construction is advantageous for the nation. The most important advantages include creation of infrastructure that is more resilient and has longer service life while simultaneously reducing the energy consumed, water used, and greenhouse gases emitted for infrastructure construction. The US infrastructure is enormous and constitutes a major portion of our nation's capital investment and energy usage each year. Consequently, changes in regulations that may affect use of CCPs in infrastructure construction should be undertaken with great caution and care.

#### What are coal combustion products (CCPs)?

Coal combustion products (CCPs) generally consist of fly ash, bottom ash, boiler slag, and fluegas desulphurization (FGD) residuals. Each is described in the following.

Fly ash. Fly ash is a fine-textured particulate collected from the off gas at coal-fired power plants to control air pollution. Although fly ash is a byproduct of controlling air pollution, scientific research and engineering practice have shown that fly ash has many useful characteristics as a construction material. Many fly ashes are cementitous, meaning that they can be used to bind particles together in a manner analogous to a conventional cement (e.g., Portland cement used in concrete). Fly ashes also are rich in calcium, silicon, and aluminum, and thus can be a good source of these elements in industrial processes such as Portland cement production. Thus, while fly ash may be considered a waste or byproduct from one industrial operation, fly ash is also a useful resource for other industrial operations (e.g., concrete production) that can be used in lieu of conventional materials that need to be mined

and processed. By using fly ash in place of these conventional materials, energy and water are saved and greenhouse gas emissions are reduced. Improved engineering characteristics (e.g., durability, strength, etc.) can also be achieved.

Bottom ash and boiler slag. Bottom ash is a coarse-textured particulate residual of coal combustion that is collected from the bottom of a boiler. Boiler slag is a solid residual that collects on the boiler during combustion that is generally found as a coarse particulate. Bottom ash and boiler slag are generated in much smaller volumes than fly ash. Most bottom ash looks like sand largely because bottom ash is similar chemically to sand. Bottom ash is used in construction in the same manner as sand, i.e., as a foundation material, a backfill material, and as drainage material. Using bottom ash or boiler slag in lieu of sand or other natural aggregates precludes the need to mine sand from the earth and process the sand so that it has suitable engineering characteristics. Consequently, when bottom ash or boiler slag is used in lieu of sand or other coarse aggregate, the energy use and greenhouse gas emissions associated with mining and processing sand are avoided. Additionally, fewer quarries for sand and gravel are needed, which improves land and resource stewardship.

**FGD** residuals. FGD residuals are created as a byproduct of waters containing lime or limestone that are used to remove sulfur compounds from off gases to reduce air pollution (e.g., reduce 'acid rain' by removing SOx compounds). FGD residuals consist of gypsum (hydrated calcium sulfate) created when the calcium binds with the sulfur compounds in the presence of water. FGD residuals also contain small amounts of impurities. Because FGD residuals consist almost exclusively of gypsum, they are used in lieu of natural gypsum in industrial processes. The most common use is for manufacturing wallboard for building construction. Using FGD residuals in lieu of mined gypsum reduces energy use, water use, and greenhouse gas emissions. Additionally, fewer gypsum mines are required, which improves land and resource stewardship.

#### Have the risks changed since CCPs were designated as non-hazardous materials?

The chemical make up of fly ash depends on the coal used for combustion, the method used for combustion, the method used for collection, and ancillary processes that are employed for air pollution control (e.g., carbon injection). These factors change over time with technological innovation. However, the general characteristics of fly ashes have not changed dramatically since CCPs were originally designated non hazardous by Congress. Consequently, the risk of using fly ash in construction today is no different than it was decades ago. Similar statements can be made regarding bottom ash, boiler slag, and FGD residuals.

Fly ashes contain a variety of elements (e.g., calcium, aluminum, selenium, chromium) as do conventional earthen materials used in construction. Some of these elements are present in larger amounts in fly ash than in conventional earthen materials; others are lower. However, none of the amounts typically are high enough (or sufficiently mobile) for fly ash to be deemed "hazardous" as defined in the Resource Conservation and Recovery Act (RCRA). Thus, there is no scientific reason to manage CCPs as hazardous wastes.

Although CCPs have been designated as non-hazardous, and generally would not be considered hazardous when evaluated by the metrics in RCRA, they are not inert materials (i.e., non-hazardous does not imply inert). For example, cement reactions are initiated when many fly ashes are contacted with water in the same manner that cement reactions occur when Portland cement is mixed with water. These reactions create heat and alkalinity as the cements

are formed. In addition, contacting CCPs with water can transfer elements in the CCP solid to the water, where they can be more mobile. Thus, like all construction materials, CCPs should be deployed in properly engineered applications using appropriate safety precautions that result in no adverse impact to the environment. Applications where CCPs are used in a dry environment (wallboard) or in a cemented monolithic environment (e.g., concrete) tend to have very low release and pose virtually no risk to the environment. In most cases, these applications have virtually no measure release.

Even in applications where releases may occur (e.g., stabilized base course in a roadway), the release needs to be considered relative to releases from conventional construction materials and to environmental standards. Because all construction materials are comprised of elements derived from the earth, they have the potential to release elements to the environment when contacted by water. Thus, a CCP may adversely affect the environment relative to a conventional construction material only if the CCP releases elements in a greater amount. Research has shown that some elements are released from CCPs in lesser amounts than from conventional construction materials, which means that CCPs may have *less* impact on the environment than conventional construction materials. In other cases, CCPs can release elements in greater amounts than a conventional construction material. In such cases, an adverse impact to the environment occurs only if elements are released at levels above environmental standards. Research that I have conducted, and the research of others, have shown that CCPs used in properly engineered applications generally do not release elements to the environment in amounts that exceed environmental standards.

#### Will the "hazardous waste" stigma affect beneficial use?

When we use CCPs as a resource, we realize significant advantages, notably reduced consumption of energy and water and lower greenhouse gas emissions. In some cases we obtain a superior product when fly ash is used in lieu of conventional construction materials. For example, roadway systems constructed with fly ash tend to be more durable and have longer service life.

Despite these advantages, not all industrial byproducts are beneficially used. There are many factors that affect whether an industrial byproduct will be selected in place of a conventional material. One of the most important factors is concern regarding potential environmental impacts and long-term liability. Major inroads have been made to address this concern over the last two decades using scientific principles and engineering methods. Test procedures have been developed, evaluation procedures have been formulated, and computer models have been created to evaluate risks and to alleviate concerns regarding environmental impacts and liabilities. However, none of these science-based principles and tools will overcome the psychological impact of CCPs being deemed a hazardous waste. An exemption for beneficial use will have virtually no effect on this psychological impact. The "hazardous" designation will scare users and incite liability, and thereby decimate beneficial use of CCPs.

Some have proffered that a hazardous designation coupled with a beneficial use exemption will increase the amount of CCPs that are beneficially used in a manner analogous to the reduction hazardous waste volume that occurred when RCRA hazardous waste rules were originally developed. My experience suggests that this outcome is unrealistic. Beneficial use is contingent on infrastructure owners accepting CCPs in their infrastructure, which is influenced strongly by owners' perceptions of risks. The beneficial use community has struggled for years to overcome owners' concerns regarding liability for industrial byproducts that are not

designated hazardous. This struggle can only become much more difficult if a hazardous designation is instituted, even with a beneficial use exemption. There is no basis to believe that infrastructure owners will accept that the risks of using CCPs in infrastructure are minimal when essentially the same material is deemed a hazardous waste in a different setting. Indeed, evidence in this regard has already been realized as public works agencies in California and Maryland have banned use of CCPs in their infrastructure projects. Manufacturers of competing products and materials that do not include CCPs have also taken advantage of the hazardous waste stigma by advertising that their products and materials do not include hazardous waste.

I surmise that beneficial use of all industrial byproducts will diminish if CCPs are deemed hazardous waste. The logical inference from the perspective of a potential user is "Will the industrial byproduct I am using today be designated as a hazardous waste tomorrow? How will this affect my long-term liability?" The logical decision from the perspective of the user is to avoid beneficial use of industrial byproducts altogether. The impact on the nation will be greater energy and water consumption, greater greenhouse gas emissions, and poorer resource stewardship.

### What effect will diminished beneficial reuse have on energy, water use, and greenhouse gas emissions?

My research group has been conducting a study to assess how cessation of beneficial use of CCPs will affect energy consumption, water usage, and greenhouse gas emissions. Although our study is not yet final, the findings are startling. Using CCPs in sustainable construction results in:

- saving 159 trillion Btu of energy annually,
- reducing water use by 32 billion gallons annually, and
- reducing greenhouse gas emissions by 11 million tons of CO<sub>2</sub> each year.

In more tangible terms, using CCPs in sustainable construction results in:

- saving the energy equivalent of 1.7 million US households annually,
- reducing water use in an amount equivalent to 31% of California's annual water use, and
- reducing greenhouse gas emissions equivalent to 1.9 million passenger cars each year.

Others recognize these savings. For example, the Kyoto Protocol accepts the reduction in greenhouse gas emissions from beneficial use of CCPs.

The stigma of a hazardous designation, even with a beneficial use exemption, will substantially reduce these benefits achieved by using CCPs in sustainable construction. A financial impact will also be realized. My research group estimates that using CCPs in sustainable construction results in a cost savings between \$5 billion and \$10 billion annually.

#### Are regulations for CCPs needed?

Regulations are needed to ensure that CCPs are managed and used in an environmentally sound manner. A means to ensure that these regulations are enforced uniformly is also needed. However, this does not imply that CCPs should be managed as hazardous waste in

Duke Energy Progress, LLC Late-Filed Exhibit No. 16 Docket No. E-2, Sub 1219 Page 352 of 439

accordance with Subtitle C of RCRA. The containment technologies stipulated in Subtitle D of RCRA (e.g., single composite liners, leachate collection systems, monitoring, etc.) are sufficient to ensure that CCPs that are not beneficially used are managed in an environmentally sound manner. Amending RCRA to permit federal control over CCP disposal using RCRA Subtitle D technologies is a logical solution that would ensure uniform application of regulations, protect the environment, and preclude the demise of beneficial use.

Duke Energy Progress, LLC Late-Filed Exhibit No. 16 Docket No. E-2, Sub 1219 Page 353 of 439

# Appendix 7

William H. Gehrmann
President
Headwaters Resources, Inc.

Testimony
Before the Subcommittee on Rural Development, Entrepreneurship and Trade
Committee on Small Business
United States House of Representatives

"Coal Combustion Byproducts: Potential Impact of a Hazardous Waste Designation on Small Businesses in the Recycling Industry"
July 22, 2010

Thank you Mr. Chairman. Honorable Members of the Committee, I am Bill Gehrmann, President of Headwaters Resources, Inc., on whose behalf I am testifying today. I have more than 25 years experience in the management and marketing of coal combustion products, which are often generically referred to as "coal ash." My experience includes managing the promotion, sale and distribution of coal ash; development of new products utilizing coal ash, the construction and operation of hazardous and non-hazardous waste landfills, and the design and operation of material handling systems.

Headwaters Incorporated is a New York Stock Exchange company that provides an array of energy services. We are a leading provider of pre-combustion clean coal technologies for power generation, including coal cleaning, upgrading and treatment. We are the nation's largest post-combustion coal product manager, operating on more than 100 power plant sites nationwide. We have built a construction materials manufacturing business and incorporated coal ash in many of our products. We are currently commercializing technologies for upgrading heavy oil and have entered the biofuels market by constructing an ethanol production facility utilizing waste heat from an existing coal fueled power plant in North Dakota. Headwaters is also active as both a technology provider and a project developer in the field of coal-to-liquid fuels.

As a manager and marketer of coal ash, Headwaters touches every link in the chain of activity that makes beneficial use of the material possible. Small businesses comprise a significant portion of many of the links in this chain. My testimony today is intended to describe that chain of activity and the probable effects of a "hazardous waste when disposed" determination on each.

There are many compelling reasons to use coal ash instead of simply disposing it. The most obvious reason is conservation of natural resources. When coal ash is used rather than disposed, other native natural resources are conserved by reducing the production of materials that coal ash is replaces. Additionally, landfill space is conserved.

#### Page **2** of **6**

The effect of off-setting environmental impacts from other industries is especially apparent in the case of utilizing coal fly ash to replace cement in the production of concrete. For every ton of fly ash used to replace a ton of cement, nearly a ton of carbon dioxide is avoided from the cement production process. In this manner, the coal ash reuse industry is currently responsible for well over 10 million tons per year of annual greenhouse gas emissions reductions. (It is important to remember that coal fly ash is a byproduct of generating electricity. The greenhouse gas emissions associated with consuming coal will exist whether or not the fly ash is used to replace cement. Accordingly, fly ash use in concrete has long been recognized by all credible sources as a legitimate and effective large volume approach to reducing greenhouse gas emissions.)

But the benefits of using coal ash are not limited to the environment. Many products made with coal ash are of higher quality than products made without it. For instance, concrete made with coal fly ash is stronger and more durable than concrete made with cement alone. Engineers and builders also use coal fly ash to address specific materials problems, such as the presence of reactive aggregates or soils, to further improve concrete durability.

There are economic benefits to consider, as well. More durable structures last longer, decreasing maintenance and replacement costs, while further conserving natural resources. Additionally, coal fly ash is less expensive than other technologies available to address engineering issues such as reactive aggregates.

Other witnesses will testify regarding the human health and environmental safety of using coal ash, but it is important to emphasize two facts. First, based on its mineral characteristics, coal ash does not approach the levels that would qualify it as a "hazardous waste" under federal law. Second, the mineral characteristics of coal ash are often strikingly similar to that of the materials coal ash is replacing when it is used.

The existence of all of these environmental, performance and economic advantages does not mean that using coal ash is easy. Significant investments must be made to be able to transport and deliver materials to users at the minute they need it. Users must be educated in how to properly utilize the materials and they must understand the materials' safety and efficacy. Today's utilization rate for coal ash in the United States is approximately 44 percent and is the product of more than three decades of efforts to identify and meet the needs of the following participants in coal ash use:

Ash Producers. Typically utilities that consume coal to generate electricity, ash
producers are faced with a decision regarding whether to dispose of coal ash or reuse it.
Disposal activities are usually carried out on the power plant site by the utility itself or a
contractor. Since reusing coal ash is not considered a core function by most utilities,
specialized "ash marketers" are usually engaged to perform those services on behalf of
the utilities that desire a reuse program.

#### Page 3 of 6

If coal ash is designated a "hazardous waste" when disposed, ash producers will be faced with this question: "This material is 'hazardous' on my own property, so am I willing to take the risk of turning it over to a third party who will market it into applications where it will be used in thousands of locations in the surrounding community?" Coal ash sales revenues are not a significant source of income for most ash producers and many ash producers will simply pass increased disposal costs on to their customers in the form of higher prices for energy. Rather than risk additional future changes in regulation or lawsuits from enterprising personal injury attorneys, ash producers will likely elect to choose disposal over reuse. If that is the case, all of the small businesses about to be described will be left without a product to use.

2. <u>Ash Marketers.</u> These marketing companies range in size from very small (fewer than 5 employees) to medium sized divisions of larger companies. They carry out a range of activities that includes transporting ash to customer markets and storing it prior to distribution, providing education and technical support to product specifiers and end users, and providing quality control and customer service.

If coal ash is designated a "hazardous waste" when disposed, ash marketers will face significant challenges in both the operations of their businesses and the outlook for customer relations.

From an operational point of view, the U.S. Environmental Protection Agency claims that it will continue to support "legitimate" beneficial uses in the event of a "hazardous when disposed" designation, but cannot answer ash marketer questions such as these: If a small quantity of ash spills during delivery, does it become a 'hazardous waste' for disposal purposes? Does ash transportation and handling equipment need to be placarded as 'hazardous waste?' What additional training and personal protection equipment will be required for workers handling coal ash? What will happen to insurance and Workers Compensation rates? All of these questions, and there are many more, represent significant challenges for small and mid-sized businesses.

From a customer relations point of view, many of the same operational questions will afflict the concrete producers and manufacturers that purchase coal ash from marketers – making them reluctant to continue using the material. Coal ash users have alternatives to using coal ash and can choose to eliminate its use.

3. <u>Ash Technology Developers and Providers.</u> A segment of the coal ash industry comprised primarily of small businesses can be described as ash technology developers and providers. Some of these companies are concerned with developing and deploying technologies for improving the quality and marketability of coal ash for traditional uses. One example is providers of technologies to remove residual carbon from fly ash in order to make it suitable for use in concrete. Other small companies are engaged in developing and deploying technologies for utilizing coal ash in new applications.

#### Page **4** of **6**

Examples include fly ash brick manufacturers or technologies that may use fly ash in sequestering greenhouse gas emissions.

If coal ash is designated as "hazardous waste" when disposed, ash technology providers and developers will face significant new customer objections and barriers to raising capital for development activities. Even in advance of enactment of any rule, companies in this sector have reported slowdowns in financing activities and customer purchases attributed to the regulatory uncertainty presented by EPA's draft rulemaking proposal.

4. <u>Product Specifiers.</u> A key link in the coal ash industry chain is comprised of entities that never actually purchase or handle the material, but play a pivotal role in whether or how it gets used. Product specifiers – including architects, engineers and their industry standards setting organizations – create specifications that either require or prohibit the use of coal ash. In determining specifications, these entities consider the effect of the material on finished product performance and human health and safety.

If coal ash is designated a "hazardous waste" when disposed, product specifiers will face the same potential operational and liability concerns previously described for ash producers and marketers. Additionally, standard setting organizations such as the American Concrete Institute and ASTM have already indicated in letters to EPA that their obligations to protect human health would require them to remove from specifications any materials that are determined to be a hazardous substance in another setting.

5. <u>Ash Users.</u> Ash users are the entities that actually use coal ash as an ingredient in other products. Examples include ready mixed concrete producers and other product manufacturers, many of which are small businesses with less than 50 employees.

If coal ash is designated a "hazardous waste" when disposed, ash users will face the same potential operational and liability concerns previously described for ash producers and marketers. In order to avoid added operational costs and potential liabilities, many users may elect simply to quit using coal ash. In almost every example of coal ash use, coal ash replaces another material that is accessible without a hazardous regulatory stigma. In cases where coal ash is used for specific engineering purposes, such as mitigating reactive aggregates in concrete, competitive products are available at much higher costs, but without potential liabilities.

In its proposed coal ash disposal rule, the EPA cites examples of other industries in which materials designated as "hazardous" have been successfully recycled. None of EPA's examples, however, are analogous to coal ash — which is used without undergoing additional processing and is placed in products that come into direct contact with end users. EPA's examples also concern materials that are sold to sophisticated users accustomed to handling hazardous materials. Coal ash users do not have this level of experience and capability.

#### Page **5** of **6**

6. <u>End Users.</u> End users are the people who actually purchase and use finished products containing coal ash – in other words, anyone who uses a home, school, office building, driveway, etc. This final link in the coal ash industry chain is the least likely to be informed regarding characteristics of materials and the most likely to become confused and concerned by a "hazardous when disposed" regulatory designation.

If coal ash is designated a "hazardous waste" when disposed, end users will likely demand products that contain no "hazardous" substances. This phenomenon is already being seen even in advance of EPA enacting any new rules. The drumbeat of the phrase "toxic ash" in news stories about EPA's rulemaking effort has resulted in many ready mixed concrete producers receiving calls from customers asking for fly ash to be eliminated from their concrete. The Los Angeles Unified School District has eliminated coal ash from its concrete specification pending resolution of the EPA rulemaking. New examples are arising every day.

Manufacturers of competitive products are also beginning to step in to fan flames of doubt for end users. Advertisements warning against products containing "hazardous waste" have appeared. Potentially even more damaging is "behind the scenes" misinformation by competitors that will be impossible to identify or rebut.

In meetings with me and with other representatives of the coal ash industry, EPA officials have indicated that they support the beneficial use of ash. But actions speak louder than words and EPA has done precious little to demonstrate support for legitimate coal ash use. To the contrary, EPA has unilaterally and without explanation removed its Coal Combustion Products Partnership program information from its web site. End users seeking information from the EPA about coal ash are now greeted with the single statement: "The Coal Combustion Products Partnerships  $(C^2P^2)$  program Web pages have been removed while the program is being reevaluated."

The benefits of coal ash use are well known to EPA and have been presented in detail in two former Reports to Congress. Also contained in those reports were analyses of "barriers" to greater coal ash utilization. "Regulatory barriers" have been identified by EPA itself as one of the key reasons coal ash use rates don't go even higher.

EPA's 2010 rulemaking has already become a significant regulatory barrier by introducing the possibility of "hazardous when disposed" regulation. As discussed previously, end users are already reacting negatively to the mere presence of EPA's proposal. I find it ironic that such a regulatory barrier has been created primarily over a dispute by regulators regarding who should enforce regulations. The actual engineering standards for disposal facilities are essentially the same under EPA's hazardous and non-hazardous proposals. EPA's hazardous proposal appears calculated primarily to get federal enforcement authority over the regulatory program. EPA appears to be willing to sacrifice a substantial and beneficial industry merely to obtain greater regulatory influence.

#### Page **6** of **6**

Headwaters and the coal ash industry are not opposed to increasing the regulation of coal ash disposal. Our trade organization, the American Coal Ash Association, has even passed a formal resolution supporting national standards for coal ash disposal. But these increased disposal standards can and must be established without designating coal ash as a "hazardous waste" in any setting.

The best course of action for our nation's environment is one that encourages safe and beneficial coal ash use as a preferred alternative to disposal. Whatever material remains unused can then be disposed in a safe and effective manner. The "hazardous when disposed" approach proposed by EPA will have exactly the opposite effect – reducing coal ash use activities and thereby creating more waste to be disposed.

Thank you for the invitation to testify and for your interest in this important topic. I would be happy to answer any questions.

Duke Energy Progress, LLC Late-Filed Exhibit No. 16 Docket No. E-2, Sub 1219 Page 360 of 439

# **Appendix 8**



Corporate Office: 5400 N. 124th Street P.O. Box 250847 • Milwaukee, Wi 53225-6514 • 414-831-2400 Fax 414-462-8812

The Honorable Lisa P. Jackson EPA Administrator USEPA Headquarters Ariel Rios Building 1200 Pennsylvania Avenue, N. W. Mail Code:1101A Washington, DC 20460

Mr. Matthew Hale, Director Office of Resource Conservation & Recovery U.S. EPA (5301P) 1200 Pennsylvania Ave., N.W. Washington, D.C. 20460

Subject: Fly Ash as a "Hazardous Waste"

Dear Ms. Jackson and Mr. Hale:

My company uses coal combustion products (CCPs) as part of our concrete mixes. We have been encouraged to be good corporate citizens and use where possible this product that would otherwise have to go to landfills as a waste material.

We have heard that EPA is considering classifying coal combustion products as "hazardous" wastes. To even utter this statement is dangerous because who in their right mind would use hazardous waste to build their houses with. The fact that we are even talking about the possibility is frightening to me as a CCP user. I literally cannot guess or even possibly begin to identify the number of jobs such as factories and hospitals and homes where CCP's have already been used. To reclassify this product as "Hazardous waste" would be incalculable damaging. Can you even begin to imagine the number of law suits and class action suits that would be initiated?

I understand that EPA might consider language stating that CCPs being used in certain applications would not be deemed a hazardous waste. Give me a break..., do you really think that motivated reporters would even try to make that distinction, certainly attorneys won't. Not only can you not call this material a hazardous waste, as a producer of a potentially hazardous waste product, I now need to know definitively that it is not now and never will be classified as a hazardous waste to continue to use it. I cannot afford to put a potential hazardous waste into my finished product.

Port Washington Plant | Mequon Plant 775 Schmitz Drive Port Washington, WI 262-284-4494 Metro 414-831-2401

11050 N. Industrial Drive 2707 Scenic Road Meduon WI 414-831-2402

Richfield Plant Richtield WI 262-644-9650 Metro 414-831-2403 Sliver Spring Plant 5400 N. 124th Street Milwaukee, WI 414-831-2404

Franklin Plant 3131 W. Elm Road Franklin, WI 414-831-2405

New Berlin Plant 6666 Crowbar Road New Berlin, WI 414-B31-2406

Grafton Plant 989 Ulao Fload Grafton WI 252-378-2087

Duke Energy Progress, LLC Late-Filed Exhibit No. 16 Docket No. E-2, Sub 1219 Page 362 of 439

We urge you to seriously consider this impact on our business and hope that EPA can avoid this unfortunate result, so we can continue to beneficially use CCPs. The CO2, green and LEED benefits of utilizing CCPs is a key strategic lever for my business.

Alan Schmitz

President

Schmitz Ready-Mix Inc.

11/15/5009 15:05 didde58815 COHWILE BEVDA WIX BVCE 05/05

Duke Energy Progress, LLC Late-Filed Exhibit No. 16 Docket No. E-2, Sub 1219 Page 363 of 439

# Appendix 9



September 4, 2009

The Honorable Lisa P. Jackson EPA Administrator USEPA Headquarters Ariel Rios Building 1200 Pennsylvania Avenue, N. W. Mail Code: 1101A Washington, DC 20460

Subject: Fly Ash as a "Hazardous Waste"

Dear Ms. Jackson:

As one of the world's leading authorities on concrete technology, the American Concrete Institute (ACI) urges the Environmental Protection Agency (EPA) to consider the technical and sustainability implications of classifying fly ash as a "hazardous waste" under subtitle C of the Resource Conservation and Recovery Act (RCRA). It is ACI's opinion that designating fly ash as a "hazardous waste" will result in little or no fly ash being used in concrete in the US. We anticipate the concrete industry will no longer specify its use; and fly ash producers would not permit its beneficial use due to liability concerns, preferring to impound fly ash rather than allow its use. Further, the designation of fly ash as a "hazardous waste" is counter to the goal of sustainability.

## Who is ACI

The American Concrete Institute is a 501(c)(3) non-profit technical and educational society organized in 1904 and is the leading international forum for the discussion of all technical matters related to concrete.

Over the past hundred years, ACI voluntary members have significantly advanced knowledge of concrete materials and structures by developing standards and publishing scholarly manuscripts, technical papers and articles. ACI is an American National Standards Institute (ANSI) accredited Standards Developing Organization (SDO), and maintains national standards in the area of concrete technology and application. ACI currently supports over 100 technical committees whose expert members develop these national standards using the consensus process.

ACI is <u>not</u> a trade organization and has <u>no</u> commercial interest in concrete or concrete products. ACI members seek to advance concrete knowledge for the benefit of the general public.

## Fly ash in concrete construction

Fly ash is commonly specified in concrete mixtures to improve durability, thus increasing service life with both environmental and economical benefits. This is important not only to private owners, but also to Federal, State, and Local jurisdictions responsible for the design, construction, maintenance and repair of buildings, bridges, roads, and infrastructure. Hungry Horse Dam, completed in 1953, was one of the first applications in which fly ash was used, and at least 100 major locks and dams using fly ash have been constructed under the direction of the U.S. Army Corps of Engineers, the U.S. Bureau of Reclamation, or private engineering firms.

The durability of concrete can be improved and service life extended by using fly ash. Fly ash can

- lower concrete permeability and thus reduce the rate of ingress of water and aggressive chemicals;
- resist deleterious alkali-aggregate and sulfate reactions;
- increase the compressive strength;
- improve the workability of fresh concrete, enabling more thorough compaction;
- · reduce the heat of hydration in mass concrete.

Fly ash is recognized in the US Green Building Council's LEED system as a post-industrial recycled material. The use of fly ash in concrete enhances the recycled material content of a building and is recognized as a beneficial strategy for CO<sub>2</sub> reduction.

The use of fly ash in concrete is an effective and often-used environmentally responsible strategy to promote sustainability since it

- uses a typically land filled industrial by-product (15 million tons diverted from landfills in 2007);
- reduces cement content of concrete, and thus CO<sub>2</sub> generated (15 million ton reduction in CO<sub>2</sub> in 2007);
- reduces the amount of embodied energy in concrete;
- reduces virgin materials extracted from the earth.

Strategically, the effective elimination of fly ash in concrete would be a step backward in the nation's efforts to provide a more sustainable infrastructure.

## Impacts of designating fly ash as a "hazardous waste"

ACI's most notable contribution to the construction industry is the ACI 318 Building Code Requirements for Structural Concrete and Commentary. The code is adopted by the ICC in the International Building Code. It satisfies ISO 19338 "Performance and Assessment Requirements for Design Standards on Structural Concrete," and is used worldwide. This Code recognizes the use of fly ash as an effective supplementary cementitious material, which leads to environmentally responsible construction.

It is not within the purview of ACI to determine whether fly ash is a "hazardous waste." As you know, EPA determined in May, 2000 that these materials "do not warrant regulation under subtitle C of RCRA and is retaining the hazardous waste exemption under RCRA section 3001(b)(3)(c)." Fly ash of any composition that is incorporated into concrete is to a high degree sequestered, and its environmental interaction is significantly reduced. Such sequestering remains even if the concrete is subsequently ground into aggregate-sized particles and recycled.

Designation of fly ash as a "hazardous waste" will likely eliminate its inclusion in future project specifications for fear of possible legal exposure and liability. Such a designation would also likely lead to its removal from future national codes and standards for the same reason.

## Summary

ACI is a technical society, and unlike trade organizations does not represent any trades related to or part of the concrete industry. Our concern deals with the impact that designating fly ash as a "hazardous waste" will have on concrete technology, the best use of concrete, and concrete's sustainable impact on society.

## Recognizing that

- fly ash is commonly accepted and used world-wide,
- fly ash can contribute to longevity and economy of concrete construction, and
- fly ash use is a key strategy to sustainable construction,

EPA should not risk harm to the environmental and material benefits of fly ash use in concrete when addressing the impoundment requirements for fly ash, nor abrogate the ability to make effective and safe use of this industrial by-product. ACI suggests that a national enforcement program for fly ash impoundment be developed to strengthen the current oversight and reduce the likelihood of another catastrophic release such as occurred in Kingston, Tennessee but without labeling fly ash a hazardous waste.

ACI would be pleased to provide the EPA with technically accurate and credible resources on the use of fly ash in concrete during the EPA's deliberations. A copy of ACI Committee 232 report dealing with fly ash's use in concrete is attached for your reference.

Sincerely,

Florian G. Barth

President

William R. Tolley

Executive Vice President

William R. Tally

Enclosure:

ACI Committee 232 Report entitled "Use of Fly Ash in Concrete"

cc: Mathy Stanislaus, EPA Assistant Administrator

Mr. Matt Hale, Director, Office of Resource

John Sager, EPA

Thomas J. Vilsack, Secretary of Agriculture

Gary F. Locke, Secretary of Commerce

Steven Chu, Secretary of Energy

Raymond L. LaHood, Secretary of Transportation

Rahm Emanuel, Chief of White House Staff

Carol Browner, Energy Coordinator

ACI Board of Direction

David Sanders, Chair, ACI Technical Activities Committee

Duke Energy Progress, LLC Late-Filed Exhibit No. 16 Docket No. E-2, Sub 1219 Page 367 of 439

# Appendix 10

# Written Testimony

From: Richard D. Stehly, President, American Concrete Institute, and Principal, American Engineering Testing, Inc.

Re: Coal Combustion Byproducts: Potential Impact of a Hazardous Waste Designation on Small Businesses in the Recycling Industry

Submitted to: U.S. House of Representatives, Committee on Small Business, Subcommittee on Rural Development, Entrepreneurship and Trade

July 22, 2010

Chairwoman Nydia M. Velazquez and members of the Subcommittee:

Thank you for the invitation to provide testimony during the hearing on "Coal Combustion Byproducts: Potential Impact of a Hazardous Waste Designation on Small Businesses in the Recycling Industry." The concrete industry comprises many small businesses, including contractors, design firms, and material suppliers. As the current President of the American Concrete Institute (ACI), I am pleased to represent ACI, one of the world's leading authorities on concrete technology. ACI is a 501(c)(3) non-profit technical and educational society organized in 1904. ACI is not a trade organization and has no commercial interest in concrete or concrete products. ACI members seek to advance concrete knowledge for the benefit of the general public.

ACI is an American National Standards Institute (ANSI) accredited Standards Developing Organization (SDO), and maintains national standards in the area of concrete technology and application. ACI currently supports more than 100 technical committees whose expert members develop these national standards using the consensus process. ACI maintains more than 400 technical documents, including codes, specifications, reports and guides, references, and the annual Manual of Concrete Practice.

As an ANSI-accredited SDO, two of ACI's major contributions to the construction industry are the "ACI 318 Building Code Requirements for Structural Concrete and Commentary" and "ACI 530 Building Code Requirements for Masonry Structures and Commentary", the latter produced jointly with the American Society of Civil Engineers and The Masonry Society. Both have been used by the major building codes in the past and are currently incorporated by reference in the 2009 International Building Code. ACI 318 contains references to the use of fly ash in concrete construction.

ACI 318 is used worldwide. An official ACI Spanish version is used throughout Central and South America. ACI has also authorized Arabic, Chinese, Korean, and Portuguese translations of ACI 318.

In regard to the beneficial use of fly ash in concrete, I offer the following:



# Why is fly ash used in concrete?

Concrete is made by blending sand with crushed stone or gravel, and binding them together in a paste made with water and the powder known as portland cement. The manufacture of portland cement is not only an energy-intensive process, but the production of each ton of cement releases approximately 1 ton of CO<sub>2</sub> into the environment as a result of chemical conversions and the burning of fossil fuels. President Obama pledged at the UN Climate Summit in Denmark, Dec. 2009, to reduce CO<sub>2</sub> emissions 17% by 2020 over a 2005 baseline. Other industrialized nations are looking for U.S. leadership in this effort.

For over 50 years it has been shown that a reduction in the amount of cement required to produce concrete can be achieved by substituting coal fly ash for a significant portion of the portland cement. The resulting concrete not only has a lower embodied energy and  $CO_2$  footprint, but also has improved properties leading to a more durable, longer lasting infrastructure. Fly ash is widely used in concrete produced in the U.S. today, and in this manner an industrial waste product is converted to a valuable resource. According to data from the American Coal Ash Association (ACAA), 15 million tons of fly ash otherwise destined for landfills were incorporated in concrete in 2006, preventing an approximately equivalent amount of  $CO_2$  emissions.

Fly ash contributes to a more sustainable, environmentally responsible infrastructure because its use in concrete can:

- reduce concrete's embodied energy and CO<sub>2</sub> footprint;
- lower coal fly ash landfill volumes;
- increase the service life of concrete;
- tie-up trace metals in ash;
- enable the use of local marginal quality sand, crushed stone, and gravel and thus reduce the need to open new quarries and pits; and
- reduce the need and cost for repairs and maintenance.

Fly ash is vital to concrete performance because it can:

- be an effective ingredient in high-strength and high-performance concrete;
- reduce the porosity and penetrability of hardened concrete;
- be an effective ingredient in minimizing corrosion of reinforcing steel;
- be an effective ingredient in resisting severe environmental exposures;
- reduce the heat produced by chemical reaction of the cement (this is critical in dams, bridge piers, and large foundations);



- increase construction quality by making a more compactable concrete;
- lower concrete's initial and life-cycle cost; and
- reduce the need to import cement

There are no viable replacements for coal fly ash in concrete, in the short term, if its availability is reduced for any reason.

Fly ash used by the concrete industry is specified to meet the requirements of ASTM C618 and AASHTO M295, and as such is well understood. The use of coal fly ash is recognized for improving concrete durability in ACI's reference Code and Specification, and coal ash is discussed in over 100 of ACI's technical documents.

#### How does beneficial use impact CO<sub>2</sub> emissions for the concrete industry?

Life-cycle assessment research published by the Portland Cement Association (PCA) reported that 96% of the CO<sub>2</sub> embodied in concrete is derived from the manufacture of portland cement. Replacing a portion of the cement with an equivalent amount of fly ash can reduce the CO<sub>2</sub> footprint of concrete by up to 1 ton of CO<sub>2</sub> emissions for every 1 ton of cement replaced. Tracking cement use gives an indication of the concrete industry's CO<sub>2</sub> emissions. By using fly ash, the concrete industry could stay under the pledged target for CO<sub>2</sub> emissions reduction. This is true for every year including the target date of 2020.

## How does beneficial use impact the need to import cement?

According to the statistics on cement use complied by the U.S. Geological Survey since 1900, cement use peaked at 128 million tons in 2005. In that year, approximately 100 million tons of cement was produced domestically requiring the importation of approximately 30 million tons of cement.

Based on industry trends and estimates from PCA, in 2009, approximately 75 million tons of cement was used, a 40% reduction from the 2005 level. ACAA's most recent data show that 15 million tons of fly ash was used in concrete construction in 2006. ACAA estimates that for 2008, 12 million tons of fly ash was used, with an additional 42 million tons available. When the economy recovers, fly ash could reduce the need to import cement and improve the balance of trade.



#### How might "stigma" impact beneficial use?

The title of the *Engineering-News Record* article "Fly Ash Looms as the 'New Asbestos'" (by Nadine Post, Apr. 7, 2010) points out the potentially harmful impact to the public perception toward the use of fly ash. But the EPA has had success in driving the use of materials labeled hazardous waste that have been reconditioned for reuse. Spent sulfuric acid is one example.

What is different about the concrete industry compared with other industries is the many different audiences it encompasses, and each has a stake in the use of fly ash. Fly ash generators have to assent to its use; otherwise, they can simply dispose of it. We Energies, a Wisconsin utility, has embraced the beneficial use of fly ash to the extent that in 2009 they recovered additional fly ash destined for disposal over what they produced that year. Ready mixed concrete suppliers have to be convinced that the improvements that might be needed at their facilities, such as additional silos for fly ash, will represent a return on the investment.

Concrete contractors also have to be in agreement. They are concerned with how fly ash impacts rate of strength gain and setting. The Engineer of Record has to approve the mixture design and must be convinced that fly ash will meet the requirements of the design. And the Owner of the project would question why fly ash is in the structure if it is hazardous. If EPA designates fly ash as special waste, but requires hazardous waste regulations, acceptance throughout the different audiences in the concrete industry will be difficult to maintain.

The American Concrete Institute is pleased to have worked with governmental agencies and industrial practitioners for over 100 years to develop building codes, specifications, standards, and guides that protect human safety and guide the design and construction of concrete infrastructure in the U.S. and around the world. As President of ACI, I am prepared to assist decision makers in selecting the best choices for the effective and responsible use of coal fly ash.

Duke Energy Progress, LLC Late-Filed Exhibit No. 16 Docket No. E-2, Sub 1219 Page 372 of 439

# **Appendix 11**



# **James A. Thomas** President

**Address** 100 Barr Harbor Drive PO Box C700 W. Conshohocken, PA 19428-2959 | USA **Phone** 610.832.9598 **Fax** 610.832.9555 **e-mail** jthomas@astm.org **Web** www.astm.org

#### 23 December 2009

The Honorable Lisa P. Jackson EPA Administrator USEPA Headquarters Ariel Rios Building 1200 Pennsylvania Avenue, N. W. Mail Code:1101A Washington, DC 20460

Subject: Fly Ash as a "Hazardous Waste"

## Dear Ms. Jackson:

Enclosed please find a letter from ASTM International Committee C09 on Concrete and Concrete Aggregates and its Subcommittee C09.24 Supplementary Cementitious Materials. The letter was unanimously approved by the members in attendance at the C09 Main Committee and C09.24 Subcommittee at their meeting in Atlanta, GA, this month, and subsequently approved by the C09.90 Executive Subcommittee.

ASTM Committee C09 on Concrete and Concrete Aggregates was formed in 1914. The Committee, with current membership of approximately 1200, currently has jurisdiction over 155 standards, published in the Annual Book of ASTM Standards, Volume 4.02. These standards are essential to the integrity of the Nation's civil infrastructure

If you have any questions, please contact the C09 Staff Manager, W. Scott Orthey at sorthey@astm.org, 610-832-9730.

Sincerely,

James A. Thomas

cc: The C09 Executive Subcommittee

Tames A. Shemas



Address 100 Barr Harbor Drive PO Box C700 W. Conshohocken, PA 19428-2959 | USA

Phone 610.832.9500 Fax 610.832.9666 Web www.astm.org

#### Committee C09 on CONCRETE AND CONCRETE AGGREGATES

Chairman: ANTHONY E FIORATO, 4439 S Seminole Drive, Glenview, IL 60026-7304, United States

(847) 687-4516, e-mail: tony@fiorato.com

Vice Chairman: STEVEN E PARKER, Chryso, Inc., 1611 Hwy 276, Rockwall, TX 75032, United States (214) 415-7089,

Fax: (972) 544-5950, e-mail: steve.parker@chrysoinc.com

Secretary: SCOTT M SCHLORHOLTZ, Iowa State Univ. Materials Analysis Res Lab, Town Engrg Bldg-rm 68, Ames, IA 50011, United States (515) 294-8761, Fax: (515) 294-4563, e-mail: sschlor@iastate.edu

Membership JENNY L HITCH, Full Circle Solutions, Inc., 2834 Somerset Springs Dr, Henderson, NV 89052, United

Secretary: States (702) 321-2114, Fax: (702) 446-8205, e-mail: jhitch@fcsi.biz

Staff Manager: W SCOTT ORTHEY, (610) 832-9730, Fax: (610) 832-9666, e-mail: sorthey@astm.org

December 22, 2009

The Honorable Lisa P. Jackson **EPA Administrator USEPA** Headquarters Ariel Rios Building 1200 Pennsylvania Avenue, N.W. Mail Code: 1101A Washington, DC 20460

Subject: Fly Ash Classification as a "Hazardous Waste"

Dear Ms. Jackson:

On behalf of ASTM Committee C09 on Concrete and Concrete Aggregates and Subcommittee C09.24 on Supplementary Cementitious Materials, we are writing in regards to the proposed classification by the Environmental Protection Agency (EPA) of fly ash as a "hazardous waste".

We strongly encourage the EPA to consider the negative implications of classifying fly ash as a "hazardous waste" under subtitle C of the Resource Conservation and Recovery Act (RCRA). Designation of fly ash as a "hazardous waste" will require that the ASTM standard for use of fly ash in concrete be revised to reflect this classification. A "hazardous waste" designation, even with an exclusion for beneficial use, would cause the ASTM standard for fly ash to be removed from project specifications due to concerns over legal exposure, product liability, and public perception. This will likely result in little or no fly ash being used beneficially in concrete or other applications that support sustainability objectives.

## **ASTM International**

ASTM International is one of the largest voluntary standards development organizations in the world and is a trusted source for technical standards. Originally known as the American Society for Testing and Materials, ASTM was formed over a century ago and has been assisting industry, government and environmentalists by creating consensus standards that have made products and services safer, better and more cost-effective.

ASTM International is recognized globally as the premier developer and provider of voluntary consensus standards as well as related technical information and services that promote public health and safety. ASTM International supports the protection and sustainability of the environment and the overall quality of life while contributing to the reliability of materials, products, systems and services in order to facilitate international, regional and national commerce.

ASTM International is a not-for-profit organization with over 30,000 members from more than 100 countries around the world. The members who serve on ASTM's 130-plus technical committees include producers, users, consumers, and general interest parties, such as academicians and government representatives.

ASTM committees develop and oversee more than 12, 000 ASTM standards that are used by individuals, companies, and agencies around the world. Purchasers and sellers incorporate standards into contracts; scientists and engineers use them in their laboratories and offices; architects and designers use them in their plans; government agencies around the world reference them in codes, regulations, and laws; and many others refer to them for guidance.

ASTM standards are voluntary in the sense that their use is not mandated by ASTM. However, government agencies often give voluntary standards the force of law by citing them in laws, regulations, and codes.

# **ASTM Committee C09 on Concrete and Concrete Aggregates**

ASTM Committee C09 was formed in 1914 and has grown to a membership of approximately 1,200 members. The Committee is composed of 29 technical subcommittees that maintain jurisdiction over 155 standards that provide test methods and specifications for concrete-making materials. These standards provide a means of quality assurance by purchasers, a basis for training and certification of testing personnel, and protection against liability through their use in contracts. Committee C09 standards, together with the standards developed by ASTM Committee C01 on Cement and committees of the American Concrete Institute (ACI), are essential to construction of the nation's civil infrastructure. Collectively, these standards have been assisting the construction industry with sustainable objectives through use of fly ash and other coal combustion by-products.

### Standards for Fly Ash

ASTM Subcommittee C09.24 on Supplementary Cementitious Materials is responsible for creating and maintaining standard test methods and specifications relating to fly ash for use in concrete. Certain types of fly ash are currently used as supplementary cementitious materials in concrete, that is, they replace a portion of the portland cement. The beneficial uses of fly ash are well documented as contributing to the long-term strength and durability of concrete. This leads to more sustainable structures.

ASTM C618 "Specification for Coal Fly Ash and Raw or Calcined Natural Pozzolan for Use in Concrete" and ASTM C311 "Test Methods for Sampling and Testing Fly Ash or Natural Pozzolans for Use in Portland-Cement Concrete" have been in existence for over 40 years and have been cited in countless concrete construction specifications. The procedures and requirements contained within these two standards are based on many years of field experience and research data. They are constantly reviewed for accuracy and updated with the latest technical information. ASTM C618 is the primary standard used for specifying fly ash by the concrete construction industry. It provides requirements for the chemical and physical

characteristics of fly ash, as well as natural pozzolans. Fly ash meeting the requirements of ASTM C618 is routinely used in construction of concrete buildings, bridges, pavements and other structures. Fly ash has historically been an integral part of concrete construction, improving the characteristics of concrete in both the fresh and hardened states.

In order to meet ASTM C618, fly ash must conform to the prescribed chemical and physical requirements. However, it is not within the scope of ASTM C618 to establish whether a fly ash is "hazardous" or not. Should the EPA decide that fly ash will be regulated as a "hazardous waste", Committee C09 will need to modify ASTM C311 and ASTM C618 to properly reflect this classification, even if the EPA specifies that fly ash used in concrete would not be considered a hazardous waste. We anticipate that changes in ASTM C618 to reflect the EPA action will result in the standard being removed from construction specifications due to liability concerns and the "hazardous waste" connotation. Specifiers and users will simply reject use of fly ash.

In closing, as a technical organization ASTM International, and specifically Committee C09, is concerned about the impact of classifying fly ash as a "hazardous waste" and the effect that it will have on materials standards and concrete construction, irrespective of an exclusion for beneficial use. The label of "hazardous waste" will likely have the unintended consequence of ensuring that fly ash will no longer be specified for use in concrete construction. We urge the EPA to consider the fact that fly ash is commonly used and accepted throughout the world, and it is an important ingredient for making concrete construction sustainable.

Thank you for your consideration of this matter.

Sincerely,

Jenny Hitch, ASTM C09.24 Chair

Jermy Witch byo

Anthony E. Fiorato, ASTM C09 Chair

Audy E. Finals

CC:

The Honorable Cass Sunstein Administrator, Office of Information and Regulatory Affairs Office of Management and Budget 725 17th Street N.W. Washington, D.C. 20503

Ms. Courtney Higgins Office of Management and Budget 725 17th Street N.W. Washington, D.C 20503

Duke Energy Progress, LLC Late-Filed Exhibit No. 16 Docket No. E-2, Sub 1219 Page 377 of 439

Mr. Mathy Stanislaus
Assistant Administrator for Solid Waste and Emergency Response
U.S. Environmental Protection Agency
Ariel Rios Building
1200 Pennsylvania Avenue, N.W.
Mail Code: 50101T
Washington, D.C. 20460
stanislaus.mathy@epa.gov

Mr. Matthew Hale
Director of the Office of Resource Conservation and Recovery
U.S. Environmental Protection Agency
Ariel Rios Building
1200 Pennsylvania Avenue, N.W.
Mail Code: 5301P
Washington, D.C. 20460
hale.matt@epa.gov

James A. Thomas, President, ASTM International

Scott Orthey, ASTM International

Kathie Morgan, ASTM International

Jeff Grove, ASTM International

Duke Energy Progress, LLC Late-Filed Exhibit No. 16 Docket No. E-2, Sub 1219 Page 378 of 439

# **Appendix 12**

Duke Energy Progress, LLC Late-Filed Exhibit No. 16 Docket No. E-2, Sub 1219 Page 379 of 439



444 North Capitol Street, N.W., Suite 315
Washington, D.C. 20001
tel: (202) 624-5828 fax: (202) 624-7875
www.astswmo.org

April 1, 2009

Matt Hale
Director
Office of Resource Conservation and Recovery
USEPA Headquarters
Ariel Rios Building
1200 Pennsylvania Avenue, N.W.

Mail Code: 5301P
Washington, DC 20460

Dear Matt,

ASTSWMO has a demonstrated track record of active interest in the management of coal combustion by-products (CCB). ASTSWMO's Fossil Fuel Combustion Waste (FFCW) Work Group gathered information about State regulation of CCB in late 2006 – early 2007. The results of that effort indicated that the majority of the responding States had regulatory programs in place for the management of CCB. On February 11, 2008, the FFCW Work Group provided comments on USEPA's "Notice of Data Availability (NODA) on the Disposal of Coal Combustion Wastes in Landfills and Surface Impoundments." Comments were based in part on the 2006-2007 survey results. The FFCW Work Group recommended a more flexible regulatory approach that allows consideration by the permitting authority of the waste type, climate, site geology and environment, and encourages a scientific and engineering approach to minimize potential risks to acceptable standards. They stated that this approach was the current practice in many States. The FFCW Work Group questioned the need for additional federal regulations related to CCB materials.

The Tennessee Valley Authority (TVA) spill in December 2008 brought renewed attention to the question about the need for federal regulation of CCB. In response to EPA's fast-track regulatory process for coal combustion waste, the ASTSWMO Board of Directors formed a CCB ad hoc Workgroup in January 2009 to review and respond to EPA's proposed regulatory schemes.

The first action of the group was to modify and reissue the 2006 survey of States initially designed by the FFCW Workgroup. In February 2009, ASTSWMO's CCB ad hoc Workgroup surveyed State waste and water program managers, working in conjunction with ECOS and ASIWPCA. There were three parts to the survey: general information about CCB management, questions specific to landfills and questions specific to surface impoundments. The survey has been completed by 44 States. Eight States do not have CCB. Fourteen States do not have CCB surface impoundments. Enclosed as an attachment to this letter are the summary results from the survey for States that have CCB.

The Workgroup also called on States to provide comments on EPA's possible regulatory proposals. A compilation of State responses is also enclosed as an attachment to this letter.



There is no question that releases, such as the December 2008 TVA Impoundment Failure in Kingston, Tennessee, should be prevented to the extent practical though appropriate engineering, design, and operating standards. However, it is also critical that all relevant factors be considered in deciding the appropriate course of action.

Presented below are the pros and cons of the possible regulatory proposals for CCB prepared by the CCB ad hoc Workgroup, based on the survey results and State comments.

# Justification of preference for Subtitle D regulation of CCB:

USEPA should implement an approach to coal combustion by-product (CCB) regulations similar to the approach that is taken with municipal solid waste pursuant to 40 CFR Part 258, commonly referred to as RCRA Subtitle D. Using the lessons learned by States since the adoption of 40 CFR Part 258 and historical CCB data collected by States, RCRA Subtitle D could be modified to specifically address CCB waste disposal facility requirements and is the framework that the USEPA should build upon.

Most States regulate CCB. Thirty-six out of 42 States that have CCB have permit programs for CCB landfills (86 percent). Only 3 States responded "no" and 3 States did not respond. Twenty-five out of 36 States that have CCB surface impoundments have permit programs for those impoundments (69 percent). Only 3 States responded "no" and 8 States did not respond. Most States regulate CCB under general solid waste regulations (43 percent) and general industrial waste regulations (43 percent). Several States use regulations specifically designed for CCB (29 percent). According to USEPA, the design and performance standards will likely be the same no matter what regulatory scheme is chosen. Many States voluntarily impose minimum performance standards for both landfills and surface impoundments under Subtitle D, demonstrating that minimum federal Subtitle D requirements will be sufficient to ensure that States properly regulate CCB.

| Percentage of States with CCB landfills and surface impoundments with specific regulatory |           |                      |
|---|-----------|----------------------|
| requirements  |           |                      |
| Regulatory Requirement  | Landfills | Surface Impoundments |
| Bottom Liner  | 64%       | 33%                  |
| GW Monitoring   | 81%       | 39%                  |
| Leachate Collection   | 52%       | 14%                  |
| Final Cover System  | 79%       | 36%                  |
| Post Closure Care   | 79%       | 39%                  |
| Siting Controls   | 83%       | 39%                  |
| Corrective Action   | 86%       | 42%                  |
| Structural Stability  | 69%       | 36%                  |
| Financial Assurance   | 69%       | 31%                  |

The fact that more than half the States already require each of the technical standards identified above for landfills demonstrates that minimum federal Subtitle D requirements will be sufficient to ensure that States properly regulate CCB. A considerable number of States have these requirements for surface impoundments as well, although we acknowledge that more States may have to upgrade their surface impoundment requirements than will have to for landfills. Establishing federal minimum standards under Subtitle D will provide the impetus needed for all States to conform. It is also important to note that currently, 36 percent of States with CCB are contemplating changes to their CCB regulations and 27 percent of those already have draft revised regulations.

#### State experiences

Michigan - "Michigan currently regulates coal ash as a solid waste under Part 115, Solid Waste Management, of the Natural Resources and Environmental Protection Act, 1994 PA 451, as amended (NREPA) ... in 1993 when Michigan became an approved State under the Resource Conservation and Recovery Act (RCRA) Subtitle D program. Based on the analytical information that we have seen on coal ash, we believe that the levels of contaminants contained in coal ash are similar in nature to those found in cement kiln dust, wood ash, foundry sands, paper mill wastes, or steel mill waste. With the promulgation of the 1993 rules, we consider all these waste to be low-hazard industrial waste (i.e. they leach less than ten percent of the hazardous waste limits when using the appropriate leaching tests)."

West Virginia - "I have been regulating coal ash facilities for 26 years for the State of West Virginia. I have never found a TCLP [Toxicity Characteristics Leaching Procedure] or other chemical characterization that would indicate that coal ash could be labeled as a hazardous waste. Most of the time the metal concentrations, which would be the main characteristic that could be considered hazardous, are at or below MCL for drinking water."

lowa - "The Department understands that the USEPA is considering options to regulate [CCB] as a hazardous waste under RCRA Subtitle C. This option is not supported by the historic data that has been collected from generators of [CCB] in lowa which shows that [CCB] does not exceed RCRA Subtitle C hazardous waste characteristics."

Arguably, municipal solid waste (MSW) presents more extensive environmental concerns than CCB. Municipal waste streams contain not only heavy metals, but also organic, acidic and alkaline materials. The organics in MSW can be more problematic than industrial wastes, which are generally inorganic in nature. Logically, if Subtitle D is adequate for MSW, then it certainly should be sufficiently protective for CCB.

Based on federal minimum standards for location, design, environmental monitoring, operation, closure, post-closure care, corrective action, and financial assurance, the States have established federally approved Subtitle D State programs. These programs have proven successful dealing with municipal solid waste, including household

hazardous wastes and Conditionally Exempt Small Quantity Generator (CESQG) waste at the State's option. A substantial number of damage cases supported the federal adoption of minimum national Subtitle D municipal solid waste landfill standards. A similar Subtitle D approach can successfully implement minimum federal standards for coal combustion waste disposal facilities. The Subtitle D approach can address any concern regarding the stability of a CCB disposal facility through establishing minimum federal design standards and routine inspection and evaluation.

Most States have some mechanism to recognize and regulate the beneficial use of Subtitle D wastes. According to the **2006 ASTSWMO Beneficial Use Survey Report**, 34 out of the 40 reporting States (85 percent) indicated they had either formal or informal decision-making processes or beneficial use programs relating to use of non-hazardous solid wastes. The Subtitle D approach, with minimum federal standards, will facilitate the continued beneficial use of CCB. As the anticipated volume of CCB produced is expected to increase or even double in many States as the Clean Air Act requirements for installation of scrubbers for flue gas desulfurization (FGD) are implemented, it is vital that the recycling of those materials which can be safely used in products or as raw materials be so used. Adopting a Subtitle D approach to the regulation of high volume, low toxicity coal combustion by-products would offer the best fit with existing and developing State beneficial use programs.

# **Explanation of opposition to Subtitle C regulation of CCB:**

#### State experiences

lowa — "Declaring CCB a hazardous waste creates an even greater hardship in Iowa because of the amount that is generated and the fact that there is no RCRA C permitted disposal facilities in the State. The likelihood of siting such a facility borders on the impossible. The implications of this action are that CCB generators would be forced to ship materials to surrounding States for disposal. That could become very costly for Iowans and extremely difficult to justify when there is little scientific data supporting such drastic measures."

Michigan – "RCRA Subtitle C wastes in Michigan are currently regulated under Part 111, Hazardous Waste Management, of the NREPA. The regulation of coal ash under full RCRA Subtitle C would end the current beneficial uses of coal ash. Existing surface impoundments and landfills would be subject to more stringent design standards and would require either 1) retrofitting of existing landfills (if even possible) or 2) closure of those disposal facilities. Neither of these options could be implemented immediately."

Florida – "If USEPA decides to call coal ash a hazardous waste under Subtitle C, then current Florida law (Section 403.7222, Florida Statutes) would prohibit the disposal of this coal ash in landfills unless it was first treated to be non-hazardous. This could add tremendous costs to the power industry for managing this material. They would either have to treat their ash before disposal or ship it out of State for disposal. It is also likely that if existing disposal areas were disturbed after USEPA determined coal ash was a hazardous waste, then these old disposal sites could become hazardous waste disposal units too."

Virginia – "If USEPA was to regulate CCB as a hazardous waste under the RCRA Subtitle C authorities, Virginia would no longer allow these materials to be beneficial reused under our CCB Regulations (9 VAC 20-85) and, also, there would be no beneficial reuse exclusions/exemption under our Virginia Solid Waste Management Regulations (9 VAC 20-80), as well."

As noted above, the vast State experience with testing CCB shows that it is generally not characteristically hazardous. Coal combustion by-products rarely if ever fail the criteria by which materials are determined to be hazardous waste. To artificially classify them as hazardous will needlessly limit the management options for both the CCBs and other wastes legitimately classified as hazardous which will be competing with CCBs for limited hazardous waste disposal capacity, while not producing any greater degree of environmental protection. Transportation, manifesting and licensing requirements for CCBs as a listed hazardous waste are excessively burdensome without sufficient evidence of a benefit. It would be more appropriate to regulate and manage CCBs using design and operation standards specified for Subtitle D programs except in the cases where a particular source material is deemed hazardous upon testing for characteristics.

The prospect of adding a significant new waste stream to

be managed by severely underfunded State hazardous waste programs is unconscionable unless a significant amount of new sustained funding is included. ASTSWMO's Hazardous Waste Subcommittee conducted a pilot program to determine the cost to States for implementing a complete and adequate RCRA Subtitle C Program (hereafter referred to as "RCRA C" or "RCRA") in 2006. The report entitled *State RCRA Subtitle C Core Hazardous Waste Management Program Implementation Costs - Final Report (January 2007)* revealed that the cost to States of implementing a complete and adequate RCRA Program (converted to 2008 dollars) is, at a minimum, \$367M in State and federal funding. The State share should be \$92M (25 percent) with the remaining \$275M in State Hazardous Waste Financial Assistance grants. However, the FY 2008 federal appropriation was only slightly more than half of what States needed. Congress appropriated \$101M rather than \$175M. States are making up the difference for these federally mandated programs from already strained State budgets. These programs are already stretched to the breaking point. Expectations should not be high for a successful incorporation of CCB into State Subtitle C programs without the guarantee of commensurate increases in State grant funding.

USEPA should avoid a "one size fits all" approach that will unnecessarily divert limited technical resources away from existing permitting or compliance and enforcement work. Instead, USEPA should recognize that many States have adequate controls in place and allow them to maintain their programs. USEPA could then focus its efforts on correcting any deficiencies identified by their investigations.

The most compelling reason not to impose Subtitle C regulations is that the beneficial use of CCB has been very successful. The "hazardous" label of Subtitle C would be detrimental to State CCB beneficial use programs, as discussed below. Regulation under RCRA Subtitle C has the potential to put an end to many beneficial uses for CCB. In most States, a primary requirement for a beneficial use determination is that the waste not be hazardous. RCRA Subtitle C wastes in Michigan are currently regulated under Part 111, Hazardous Waste Management, of the NREPA. The regulation of coal ash under full RCRA Subtitle C would end most of the current beneficial uses of coal ash. Existing surface impoundments and landfills would be subject to more stringent design standards and would require either 1) retrofitting of existing landfills (if even possible) or 2) closure of those disposal facilities. Neither of these options could be implemented immediately.

# Implications for beneficial use if CCB is regulated under Subtitle C:

The American Coal Ash Association reports that 43 percent of CCB is currently used in a beneficial way rather than disposed in a landfill. About 20 percent of CCB is used in products – 14 percent is bound in concrete and cement; 6 percent is used to make gypsum wallboard. Currently, 56 percent, or 75 million tons, is not beneficially used. States are concerned that designating CCB as a hazardous waste under Subtitle C or a hybrid Subtitle D/C regulation would prevent beneficial use of CCB and result in all 134 million tons of CCB being shipped to hazardous waste landfills that in many States have insufficient capacity. As the anticipated volume of CCBs produced is expected to increase or even double in many States as requirements for FGD are implemented, it is vital that the recycling of those materials which can be safely used in products or as raw materials be so used.

Not only do many State regulations prohibit the beneficial use of CCB if it is declared hazardous (see State experiences insert), such a designation will stigmatize the material in a way that will

### State experiences

Michigan – "Michigan currently has regulations in place governing the reuse and disposal of coal ash that are protective of public health and the environment. If coal ash were determined to be subject to regulation under Subtitle C, it would necessitate considerable changes to Michigan solid and hazardous waste statutes and regulations. Such changes would likely be subject to considerable opposition from any industry and/or municipality that generates coal ash waste, and would likely lead to increased costs for energy generation."

Missouri – "Given the current State of CCB management activities in Missouri there does not appear to be a compelling reason, from a human health or environmental protection standpoint, to manage these materials as hazardous waste under RCRA Subtitle C. To do so would be an undue disruption to current State CCB and UWLF management practices and would likely result in a significant increase in the cost of CCB management without a corresponding increase in human health or environmental improvement/protection."

adversely affect beneficial use. The stigma issue also applies to the proposed hybrid Subtitle D/C approach. The uncertainty that a presumed non-hazardous material could be deemed hazardous as a result of a determination that a generator failed to follow the Subtitle D requirements will create too much uncertainty and liability concerns for the beneficial user.

Coal combustion by-products or residue generally consists of fly ash, bottom ash, or wet slurry depending on the combustion unit and associated air pollution control devices. The character of the end stream varies and is dependent upon several factors. However, all seem to be lumped together in this regulatory analysis without discussion of

segregate characteristics or potential for beneficial use.

States require testing of beneficially reused materials. Testing can include initial analysis of the

material and additional testing when sources of fuel change or when there is a change in plant processes, if such changes cause a change in the constituents generated. States report that their beneficial programs do not allow the use of coal ash in road construction if the material fails the Toxicity Characteristics Leaching Procedure (TCLP). Many States report that they do not have any data to suggest that coal ash

# Examples of the beneficial use of CCB

- a component of concrete, grout, mortar, or casting molds
- a raw material in asphalt for road construction
- aggregate or road or building material which will be stabilized or bonded by cement, limes or asphalt
- road base or construction fill that is covered with asphalt, concrete, or other material approved by the State
- a soil amendment or for soil stabilization provided the materials meet State criteria

projects that have been reviewed have failed TCLP.

States have incorporated technical standards in their regulations and approvals for storage of CCB. For example, in Missouri, a waste to be beneficially reused is kept above the seasonal high groundwater table, unless a variance is obtained from the department's Water Protection Program (WPP.) This requires an interpretation by a geologist registered in the State. A 3-foot cap of clean soil is required unless the material is placed under a structure or a paved/concreted area.

Recycling this waste material into new products, rather than having to mine additional virgin material, is integral to sustainable development and sustainable infrastructure. To disallow the

beneficial use of coal combustion by-products (CCB) would cause an increase in the use of valuable mineral resources rather than reusing a waste product. This would in turn increase disposal costs for the utilities which would be passed on to the consumer. Counties and municipalities which use bottom ash as snow and ice control would instead have to purchase chemicals or salts to treat the roads. State transportation departments and other entities using CCB would have to purchase soil to use in place of the fly ash currently used for structural fill, road base, as a soil amendment or for soil stabilization. This could impact the number of miles of roads that can be constructed or repaired and increase costs. In other cases, specific beneficial use projects limit the amount of transportation that would otherwise be needed if the material were considered a hazardous waste. Some coal-fired power plants are co-located near gypsum wallboard manufacturers. The FGD sludge is transported by conveyor belt directly to the wallboard facility for beneficial use. These operations result in safe uses and minimal transport of the FGD sludge.

# **Concerns about existing facilities:**

An issue that has not been addressed adequately in discussions is whether USEPA plans to address existing facilities, and if so how. If USEPA pursues the Subtitle C regulatory route, it might subject all existing facilities in a State to RCRA corrective action. Additionally, bringing existing facilities under Subtitle C raises resource-intensive permitting issues. States generally have legislatively prescribed staffing levels based upon workload, mission, funding, and statutes passed to implement federal RCRA authority or delegation. As noted previously, ASTSWMO's report entitled *State RCRA Subtitle C Core Hazardous Waste Management Program Implementation Costs - Final Report (January 2007)* demonstrates that State Subtitle C programs are already seriously underfunded. Additionally, retrofitting of existing Utility Waste Landfills (UWLFs) to meet Subtitle C standards is likely to be technically impracticable. Even if technically feasible, the cost of retrofitting UWLFs to meet current RCRA Subtitle C standards would likely be prohibitively expensive. Any additional compliance costs borne by the utility companies in retrofitting existing UWLFs or permitting new ones would undoubtedly be passed along to consumers at a time when economic conditions in the U.S. are less than ideal.

## **Enforcement:**

There have been suggestions that Subtitle C is necessary so that USEPA will have enforcement authority. States are held accountable by their citizens through State statutes and obligations to regularly inspect landfills and investigate complaints, and to utilize State enforcement authority as warranted. Subtitle D requires State programs to have the necessary enforcement authority as part of the federal approval process. This approach has been successful for over a decade as evidenced by the relative absence of federal citizen suits or demonstrated failure of State Subtitle D programs. The States are not aware of USEPA expressing concerns regarding this State based enforcement approach in the municipal solid waste landfill program. A similar Subtitle D approach can successfully ensure compliance with minimum federal standards for coal combustion waste disposal facilities.

# **Applicability of Federal Regulations:**

Based upon discussions to date with USEPA and States, it appears that the intended coverage of any federal CCB regulations would be limited to CCBs generated by coal-fired utilities, and not extended to CCBs generated by other industries. If this is correct, then the federal regulations should clearly make this distinction. Otherwise, an unreasonable burden will be placed upon the States to individually sort out the applicability issue, likely resulting in uneven application of the base federal requirements.

# **State Program Authorization:**

Regardless of the regulatory approach selected, the States request that the procedures for authorization of State programs to implement the CCB rules be streamlined and designed to operate in harmony with existing Subtitle D (and/or Subtitle C) program authorization procedures. Where there are existing State programs in place regulating these materials, considerable deference should be given to the State program in the authorization process. States with CCB programs in place should be provided the option to 1) demonstrate that their programs are consistent with and not less stringent than the federal program, and 2) be more stringent than the federal program if they so choose. Further, authorization for any new CCB regulations should be treated as an amendment to a State's existing Subtitle D (or Subtitle C, as applicable) program authorization, as opposed to considering the CCB program as separate and distinct from existing authorizations.

# **Funding:**

Federal funding may be necessary to help build State program capacity in the few States that do not have CCB programs if USEPA mandates standards under Subtitle D. It should be noted that some State Subtitle D programs would likely not seek federal funding for a Subtitle D program because of the impact that would have on current State solid waste program financing structures. As the ASTSWMO survey demonstrates, many States already have Subtitle D CCB programs and would not incur a financial hardship. On the other hand, State Subtitle C programs, which are supposed to be funded at a level of 75 percent federal funding, would require significant new appropriations. Thus, the federal funding needs for a Subtitle D approach would be much less than a Subtitle C regulatory approach.

Any decisions to regulate the management and disposal of coal ash will likely have an implication for State regulatory programs including: the need to undertake regulatory action; authorization/approval for implementation (if necessary); budgetary impacts; and staffing/workload resource issues related to implementation (i.e., possible permitting/compliance/enforcement program impacts). The implications could have a dramatic impact on the already strained budgets of many State environmental agencies. It is hoped that USEPA's decision will include review of the work that many States have undertaken to regulate coal combustion by-products.

# **Summary:**

The ATSWMO ad hoc CCB Workgroup, based on results of a survey of States and State comments, recommends that if it is determined that federal regulation of CCB is necessary, Subtitle D regulations would be the preferred approach. Most States already regulate CCB under Subtitle D regulations. Furthermore, a Subtitle D approach would foster the beneficial use of appropriate CCB rather than inhibit it, as would a Subtitle C or hybrid Subtitle C/D approach.

On behalf of ASTSWMO, we thank you for your diligence in ensuring that the most efficient and effective regulatory approach to CCB is proposed.

Sincerely,

Brian Tormey (IA)

Chair

ASTSWMO Solid Waste Subcommittee

Brian Torney

Stephen Cobb (AL)

Chair

ASTSWMO Hazardous Waste Subcommittee

cc: Rick Brandes (USEPA ORCR)

Rich Kinch (USEPA ORCR)
ASTSWMO Board of Directors

ASTSMWO ad hoc CCB Workgroup

Steve Brown (ECOS)

Linda Eichmiller (ASIWPCA)

Duke Energy Progress, LLC Late-Filed Exhibit No. 16 Docket No. E-2, Sub 1219 Page 388 of 439

# Appendix 13



Duke Energy Progress, LLC Late-Filed Exhibit No. 16 Docket No. E-2, Sub 1219 Page 389 of 439

444 North Capitol Street, N.W., Suite 315
Washington, D.C. 20001
tel: (202) 624-5828 fax: (202) 624-7875
www.astswmo.org

November 4, 2009

Matt Hale
Director
Office of Resource Conservation and Recovery
USEPA Headquarters
Ariel Rios Building
1200 Pennsylvania Avenue, N.W.
Mail Code: 5301P
Washington, DC 20460

Dear Matt,

Previously, we forwarded the results of the ASTSWMO's CCW Ad Hoc Workgroup survey of State waste and water program managers (Phase I), working in conjunction with ECOS and ASIWPCA. The Phase I survey sought information about State management practices for disposal of CCW. The survey revealed that, contrary to claims from environmental groups and the media, most States regulate the disposal of CCW. Thirty-six out of 42 States that have facilities producing CCW have permit programs for CCW landfills (86 percent).

Beginning August 27, 2009, the ASTSWMO CCW Ad Hoc Workgroup conducted a follow-up survey (Phase II) to its February 2009 Coal Combustion Waste Survey of State management practices. The purpose of the Phase II survey was to obtain information regarding the costs, workload, and expertise impacts on State programs of regulating CCW under the RCRA Subtitle C and RCRA Subtitle D regulatory options.

All 50 States and the District of Columbia responded to the Phase II survey. Obtaining 100% participation of States in a survey with such a short turnaround is remarkable and demonstrates how important this issue is to the States.

Enclosed is ASTSWMO's final report on the results of the Phase II State survey. All State respondents oppose EPA regulation of CCW under RCRA Subtitle C, with the exception of two States (one State that by statute does not regulate CCW as a solid waste and one that does not generate coal combustion waste). States have serious concerns about the impact of federal regulation on waste program resources, particularly if CCW is regulated as a hazardous waste under Subtitle C.



As noted, this is an issue of great concern to the States. We appreciate your continued interest in the States' position on this issue.

Sincerely

Stephen A. Cobb (AL), Chair

Hazardous Waste Subcommittee

Cc: Robert Dellinger

James Berlow



# FEDERAL REGULATION OF DISPOSAL OF COAL COMBUSTION WASTE FROM COAL FIRED POWER PLANTS

## COST ISSUES AFFECTING STATE ENVIRONMENTAL PROGRAMS

# **BACKGROUND**

## Federal Regulation of Disposal of Coal Ash

Coal combustion waste (CCW) was initially temporarily excluded from federal regulation as a hazardous waste under the 1980 Solid Waste Disposal Act Amendments to RCRA. In 1999, the U.S. Environmental Protection Agency (EPA) issued a Report to Congress indicating the preliminary decision that coal combustion waste disposal should remain exempt from Subtitle C (hazardous waste regulation). In 2000, after considering designating some CCW as hazardous waste under Subtitle C, EPA published a final regulatory determination that the regulation under Subtitle C was not warranted but that national regulations under Subtitle D (nonhazardous waste) would be appropriate for coal combustion wastes disposed in landfills and surface impoundments. However, EPA indicated in the regulatory determination that in developing/reviewing future regulations, it would look at the extent to which CCW caused actual or potential damage to human health and/or the environment, the environmental effects of filling mines with CCW, the adequacy of existing regulations, and the effects of mercury exposure from these activities. EPA also indicated that it would assess new information on risks associated with managing fossil fuel combustion wastes as it became available and monitor trends in protective management to see if regulation under Subtitle D would close the gaps it identified; if not, the Agency indicated it may re-examine its decision not to regulate the wastes under Subtitle C. In August 2007, EPA issued a Notice of Data Availability (NODA) on the Disposal of Coal Combustion Wastes in Landfills and Surface Impoundments. In February 2008, ASTSWMO's Fossil Fuel Combustion Waste Work Group submitted comments on the NODA based in part on the results of a survey in which 33 States responded. Generally, the States have argued that State regulations are sufficient for managing CCW. ECOS took this position in a resolution passed in 2008. However, at the end of 2008, EPA still had not made a determination about federal regulation of CCW. In deciding on an appropriate regulatory course of action, a central question which must be addressed relates to the basic regulatory underpinning of Subtitle D versus Subtitle C regulation – that being, "Does

CCW routinely meet the established criteria for regulation as a hazardous waste?" The vast experience of State programs is that CCW does not routinely meet the criteria.

The Tennessee Valley Authority (TVA) spill in December 2008 brought renewed attention to the question about the need for federal regulation of CCW from coal fired power plants. In response to EPA's fast-track regulatory process for CCW, the ASTSWMO Board of Directors formed a CCW Ad Hoc Workgroup in January 2009 to review and respond to EPA's proposed regulatory schemes.

In February 2009, ASTSWMO's CCW Ad Hoc Workgroup surveyed State waste and water program managers, working in conjunction with ECOS and ASIWPCA. The Phase I survey sought information about State management practices for disposal of CCW. The survey revealed that, contrary to claims from environmental groups and the media, most States regulate the disposal of CCW. Thirty-six out of 42 States that have facilities producing CCW have permit programs for CCW landfills (86 percent).

EPA has indicated that it is considering 3 possible regulatory scenarios – regulation as a non-hazardous waste under Subtitle D; regulation as a hazardous waste under Subtitle C; or a hybrid Subtitle C/D approach. According to EPA, the design and performance standards for the proposed regulations will likely be the same no matter what regulatory scheme is chosen. Many States voluntarily impose minimum performance standards for both landfills and surface impoundments under Subtitle D, demonstrating that minimum federal Subtitle D requirements will be sufficient to ensure that States properly regulate CCW.

On August 27, 2009, the ASTSWMO CCW Ad Hoc Workgroup conducted a follow-up survey (Phase II) to its February 2009 Coal Combustion Waste Survey of State management practices. The purpose of the Phase II survey is to obtain information regarding the costs, workload, and expertise impacts on State programs of regulating CCW under the RCRA Subtitle C and RCRA Subtitle D regulatory options. Both Phase I and Phase II surveys sought information from States about the beneficial uses of coal ash. An example of a beneficial use that is important to States is the use of CCW in State highway projects. This use is not only cost-effective for State Departments of Transportation but also diverts these wastes from landfills. The American Coal Ash Association reports that 43 percent of CCW is currently used in a beneficial way rather than disposed in a landfill. If EPA decides to regulate CCW as a hazardous waste, most experts agree it will have a detrimental effect on the beneficial use of CCW. This is only one of the negative effects on States of the potential federal regulation of CCW as a hazardous waste. ASTSWMO's State surveys reveal a number of other likely adverse impacts.

All 50 States and the District of Columbia responded to the Phase II survey. Obtaining 100% participation of States in a survey with such a short turnaround is remarkable and demonstrates how important this issue is to the States.

## <u>ASTSWMO PHASE I AND PHASE II SURVEY RESULTS</u>

#### STATE OPPOSITION TO SUBTITLE C REGULATION

All State respondents oppose EPA regulation of CCW under RCRA Subtitle C, with the exception of two States (one State that by statute does not regulate CCW as a solid waste and one that

<u>State experience demonstrates CCW</u> is not a hazardous waste under federal regulations

"I have been regulating coal ash facilities for 26 years for the State of West Virginia. I have never found a TCLP [Toxicity Characteristics Leaching Procedure] or other chemical characterization that would indicate that coal ash could be labeled as a hazardous waste. Most of the time the metal concentrations, which would be the main characteristic that could be considered hazardous, are at or below MCL for drinking water."

"[Regulating CCW] as a hazardous waste under RCRA Subtitle C is not supported by the historic data that has been collected from generators of [CCW in this State]which shows that CCW does not exceed RCRA Subtitle C hazardous waste characteristics."

"Municipal solid waste ash contains higher levels of contaminants [than CCW]"

does not generate coal combustion waste). A major objection to listing CCW as a hazardous waste is that the vast State experience with testing CCW using the standard EPA test for determining if a waste is hazardous under the Resource Conservation and Recovery Act (RCRA) (the Toxicity Characteristic Leaching Procedure (TCLP)) shows that it is generally not characteristically hazardous. As demonstrated by the State

survey results this is a critical point because regulating CCW as a hazardous waste is burdensome on federally underfunded State waste programs and it also diverts precious

resources from protecting threats to health and the environment posed by actual hazardous wastes. As one State put it, "this would detract from regulation of real hazardous waste." EPA acknowledges that technically, CCW can be safely regulated as a non-hazardous waste under Subtitle D with the appropriate management standards. This Administration's stated policy that regulatory decisions will be based on scientific

"The Department's position is that classifying coal combustion residues under RCRA Subtitle C would create unnecessary barriers to the current management options without producing any greater degree of environmental or public health protection."

evidence purports that CCW should not be regulated a hazardous waste under RCRA Subtitle C.

### IMPACT ON EXISTING HAZARDOUS WASTE REGULATION

If CCW meets the established scientific threshold criteria for regulation as a hazardous waste, then the question of Subtitle D versus Subtitle C is moot – the material should be regulated under Subtitle C. However, this determination has not been made, and in fact the opposite determination was made by EPA in its 2000 regulatory determination.

A major concern with adding lower risk, high volume wastes which do not meet the threshold criteria to the Subtitle C inventory is that those higher threat wastes which do meet the criteria and legitimately warrant Subtitle C controls will become lost in the shuffle due to the staggering difference in volume (2 million tons versus 134 million tons per year) and will divert attention and vigilance from the higher threat waste streams.

#### STATE WASTE PROGRAM CAPACITY

The fiscal impact on States of EPA's proposed regulations cannot be ignored, particularly in light of the budget crises so many States are experiencing. Adding the unnecessary burden of regulating a non-hazardous waste (i.e., one that does not meet RCRA hazardous waste testing standards) under Subtitle C, which is already severely underfunded, when so many States are imposing staff furloughs, hiring freezes and layoffs is unthinkable. Regulating CCW as a hazardous waste under Subtitle C will impose a significantly greater resource burden on State waste programs than regulating it as a non-hazardous waste under Subtitle D, which many States are already doing.

When asked how many facilities that could be affected by the new regulations have a Subtitle C disposal permit, all 43 States that responded to this question said "none." The capacity to regulate those facilities under Subtitle C does not exist in most States. At least 38 States will need additional Full Time Equivalents (FTEs) staff if EPA regulates CCW as a hazardous waste under Subtitle C. The increased workload will require additional technical expertise for the various Subtitle C program elements: Permitting, Inspections (including storage and record-keeping requirements), Financial Assurance, Facility-wide Corrective Action, Closure (Interim Status), Post-Closure Permits, Generator/Transporter Requirements, and Siting Controls. Several States could not even guess what impact regulating CCW under Subtitle C would have on their programs, but 29 States estimated that at least 140 FTEs would have to be hired at a cost of \$12M, or an estimated \$414K per State.

By contrast, only 18 States will need additional FTEs if EPA regulates CCW under Subtitle D. In other words, twice as many States will be impacted financially under Subtitle C regulation – a full three quarters of the States in this country. That vast majority of States indicated that no new FTEs will be needed if CCW is regulated under Subtitle D. The cost estimate is significantly less as well. The 18 States that could estimate how many additional FTEs would be needed if EPA regulates CCW under Subtitle D estimated that 40 FTEs would be needed at a cost of \$3.8M/year or an estimated \$211K per State.

There is no doubt that adding CCW to the wastes that are regulated as hazardous wastes will be devastating to State Subtitle C programs that are already underfunded. ASTSWMO's Hazardous Waste Subcommittee conducted a pilot program to determine the cost to States for implementing a complete and adequate RCRA Subtitle C Program (hereafter referred to as "RCRA C" or "RCRA") in 2006. The report entitled *State RCRA Subtitle C Core Hazardous Waste Management Program Implementation Costs - Final Report (January 2007)* revealed that the cost to States of implementing a complete and adequate RCRA Program (converted to 2008 dollars) is, at a minimum, \$275M in State and federal funding. The State share should be \$69M (25 percent) with the remaining \$206M in State Hazardous Waste Financial Assistance grants. However, the FY 2008 federal appropriation was slightly less than half of what States needed. Congress appropriated \$101M rather than \$206M. States are making up the difference for these federally mandated programs from already strained State budgets. These programs are already stretched to the breaking point. Expectations should not be high for a successful

incorporation of CCW into State Subtitle C programs without the guarantee of commensurate increases in State grant funding.

The difference in cost to the States between Subtitle C and Subtitle D is a significant factor in the current climate of substantial State budget revenue shortfalls. Either way, nearly all States (94%) will not be able to add FTEs to accommodate the additional workload without financial support from EPA.

### **TRAINING COSTS**

A significant majority of States (79%) indicated staff training will be needed if CCW is regulated under RCRA Subtitle C. That is another cost that is not accounted for in the survey results. Not only will training be needed, it will also be costly to develop. There have been few if any new Subtitle C facilities permitted for 15-20 years, and most Interim Status facility closures were performed and Initial Operating Permits issued in the 1980's. Expertise and training is a significant issue because it has been that long since some States have gone through the process needed for permitting a new facility, issuing an initial permit to an Interim Status facility, or overseeing closure/post-closure activities and issuing initial Post-Closure permits for Interim Status facilities.

Fewer States (31% of responding States) will need staff training if CCW is regulated under RCRA Subtitle D.

### **BENEFICIAL USE**

A compelling reason not to impose Subtitle C regulations is that the beneficial use of CCW has been very successful. As noted above, the vast State experience with testing CCW shows that it

is generally not characteristically hazardous. Coal combustion wastes rarely if ever fail the criteria by which materials are determined to be hazardous waste. Regulation under RCRA Subtitle C has the potential to put an end to many beneficial uses for CCW. In most States, a primary requirement for a beneficial use determination is that the waste *not* be hazardous. Labeling CCW a hazardous waste will have a detrimental on its beneficial use.

This concern is also supported by the on-going controversy and legal challenges over the recent changes to the Definition of Solid Waste (DSW), which are primarily related to concerns over the appropriateness of relaxing regulatory controls on defined hazardous wastes for the purpose of encouraging reuse and recycling.

<u>State experts agree – designating CCW as</u> a hazardous waste will end beneficial use

"Regulation under Subtitle C would likely discourage beneficial uses of coal ash in concrete and Portland cement type applications."

"Regulation of coal ash under RCRA Subtitle C would impede recycling and discourage its beneficial use and instead cause the disposal of a valuable resource in landfills and/or surface impoundments."

"Coal ash has been beneficially reused in our state. Regulation under Subtitle C would likely stop this."

#### **DISPOSAL CAPACITY**

The American Coal Ash Association reports that 43 percent of CCW is currently used in beneficial way rather than disposed in a landfill. Currently, 56 percent, or 75 million tons, is not beneficially used. States are concerned that designating CCW as a hazardous waste under Subtitle C will prevent beneficial use of CCW (as was the case with "Iron Rich" noted above) which will result in 134 million tons of CCW being shipped to hazardous waste landfills annually. According to EPA's National Biennial RCRA Hazardous Waste Report, in 2007 (the most recent data published) 1.6 million tons of hazardous waste was received by off-site hazardous waste landfills and surface impoundments

(http://www.epa.gov/epawaste/inforesources/data/br07/national07.pdf, Exhibit 3.9). Using a conservative estimate that 2 million tons of hazardous waste is disposed at off-site facilities annually, disposing of CCW as a hazardous waste will result in as much as sixty-seven times more waste being disposed in landfills. Even if beneficial use continues at its current rate, an additional 75 million tons per year (or thirty-eight times) more waste will have to be disposed in hazardous waste landfills annually.

Even more alarming is the fact that disposing of CCW in hazardous waste landfills will consume the Commercial Subtitle C Management Capacity projected for the year 2013 in a matter of months. EPA's expected maximum capacity for Subtitle C landfill capacity for 2013 is 34 million tons (<a href="http://www.epa.gov/osw/hazard/tsd/capacity/appb\_lf.pdf">http://www.epa.gov/osw/hazard/tsd/capacity/appb\_lf.pdf</a>). Assuming all CCW will be disposed in commercial Subtitle C landfills, the 2013 capacity will be exhausted within 3 months. Even if beneficial use continues at its current rate, the 2013 capacity will be exhausted in less than 6 months. In the unlikely event that beneficial use continues at its current rate and half of the coal fired utilities seek Subtitle C permits for the disposal facilities that they manage, the 2013 capacity will be consumed in less than one year. Consuming the commercial hazardous waste landfill capacity not only means that CCW will begin to pile up unmanaged at utilities, the current 2 million tons of hazardous waste generated by industry and hazardous waste site remedial activities will also begin to accumulate on-site. This will bring a halt to Superfund cleanups that require disposal of hazardous wastes as well as having a devastating impact on vital industries and facilities generating nearly half of the country's electrical power. It can take years to permit a new hazardous waste landfill.

States already know that there is not sufficient hazardous waste landfill capacity if CCW is designated a hazardous waste, as reflected in the Phase II survey.

91% of States responding to the question do not have sufficient existing permitted
 Subtitle C disposal capacity for all
 CCW in-State.

#### Siting hazardous waste landfills is onerous - it has been 15-20 years since new hazardous waste landfills have been permitted

"The Massachusetts statute governing hazardous waste has onerous siting standards that would make it very difficult to site any facility to dispose of coal ash." "Kansas state law prohibits the landfilling of hazardous waste so our laws would either need to be changed or all waste would need to be exported which is totally impractical."

 86% of States responding to the question will need new off-site capacity to be sited if CCW is regulated as a hazardous waste.

Conversely, a majority of States have sufficient permitted non-hazardous waste disposal capacity for CCW. More

than half of that permitted capacity is located on-site at the generator facility which significantly reduces the amount of coal ash that must be transported for disposal.

- Only 31% of States responding to the question do not have sufficient existing permitted non-hazardous waste disposal capacity for all CCW in-State.
- Only 35% of States responding to the question will need new off-site capacity to be sited if CCW is regulated as non-hazardous waste.

Transportation issues associated with CCW designated as hazardous waste is another cause for concern. According to EPA's most recent data, 7 million tons of hazardous waste was shipped in one year by 16,258 shippers (<a href="http://www.epa.gov/epawaste/inforesources/data/br07/national07.pdf">http://www.epa.gov/epawaste/inforesources/data/br07/national07.pdf</a>, Exhibit 3.1). Each State has rigorous standards for licensing hazardous waste transporters. Most CCW is currently managed on-site at the generation facility. If the material becomes regulated

as a hazardous waste, it is likely that much of this material will then be managed off-site, which will increase hazardous waste transportation by up to 20 times more waste than the current annual rate. The impact on transportation infrastructure and communities through which this new "hazardous waste" will be transported will be overwhelming. Only a handful of States have commercial Subtitle C landfills, which means that most CCW will have to be shipped out of State.

Only a handful of States have commercial Subtitle C facilties "We do not have any Subtitle C capacity. All waste would need to be shipped out of state."

"There is only one commercial Subtitle C landfill in the state and it only receives hazardous waste treated by the owner of the site."

"Michigan only has one commercial subtitle C permitted facility, which is reaching capacity."

#### REGULATORY BURDEN

Drafting, proposing and finalizing regulations is a labor intensive and costly process. Currently, 36 out of 42 States have CCW solid waste permit programs for CCW landfills (86 percent). Only 3 States responded "no" and 3 States did not respond. Most States regulate CCW under general solid waste regulations (43 percent) and general industrial waste regulations (43 percent). Several States use regulations specifically designed for CCW (29 percent). Many States voluntarily impose minimum performance standards (such as those being considered by EPA for regulation of CCW), demonstrating that minimum federal Subtitle D requirements will be sufficient to ensure that State regulation of CCW is protective of human health and the environment.

| Percentage of States with CCW landfills with specific regulatory requirements |                  |  |  |  |
|---|------------------|--|--|--|
| Regulatory<br>Requirement   | Landfills<br>(%) |  |  |  |
| Bottom Liner  | 64%              |  |  |  |
| GW Monitoring   | 81%              |  |  |  |
| Leachate Collection   | 52%              |  |  |  |
| Final Cover System  | 79%              |  |  |  |
| Post Closure Care   | 79%              |  |  |  |
| Siting Controls   | 83%              |  |  |  |
| Corrective Action   | 86%              |  |  |  |
| Structural Stability  | 69%              |  |  |  |
| Financial Assurance   | 69%              |  |  |  |

If EPA designates CCW as a hazardous waste, all forty-eight RCRA authorized States will have to develop new Subtitle C regulations, despite the fact that regulation under Subtitle D will provide sufficient protection of health and the environment. This is a very costly and unnecessary burden that will divert resources from more productive activities.

#### **FEDERAL VERSUS STATE AUTHORITY**

EPA acknowledges that CCW can be safely regulated under Subtitle D. EPA suggests there are 2 primary reasons that EPA may propose Subtitle C regulation: 1) Subtitle D does not allow Federal enforcement except under citizen suits; and 2) EPA cannot require States to permit landfills under Subtitle D.

#### Enforcement

EPA suggests that Subtitle C is necessary so that EPA will have direct enforcement authority. States are held accountable by their citizens through State statutes and obligations to regularly inspect landfills and investigate complaints, and to utilize State enforcement authority as warranted. Subtitle D requires State programs to have the necessary enforcement authority as part of the federal approval process. This approach has been successful for over a decade as evidenced by the relative absence of federal citizen suits or demonstrated failure of State Subtitle D programs. The States are not aware of EPA expressing concerns regarding this State based enforcement approach in the municipal solid waste landfill program. A similar Subtitle D approach can successfully ensure compliance with minimum federal standards for coal combustion waste disposal facilities.

#### <u>Permitting Requirement</u>

While EPA cannot require that States permit Subtitle D facilities, most States do so without a federal mandate. ASTSWMO's Phase I survey revealed that thirty-six out of 42 States in which CCW is generated have permit programs for CCW landfills (86 percent). Only 3 States responded "no" and 3 States did not respond. Imposing the more stringent requirements of Subtitle C regulation on States to ensure that they permit facilities is not justified when most States already do so.

#### **LEGISLATIVE ISSUES**

EPA's proposed regulation of CCW will have a significant impact on both State Executive and Legislative branches. Whether EPA proposes regulation as hazardous (Subtitle C) or non-hazardous (Subtitle D), funding State environmental agency programs will become even more difficult. The budget impact will be more substantial if EPA proposes regulating CCW as a hazardous waste not only because the cost will be greater for Subtitle C regulation, but also as noted above, federal funding for State hazardous waste programs is already only half of what States need from the federal government to fund adequate Subtitle C core programs. Mandating another significant federal standard for these programs without commensurate guarantees of increased and sustained federal funding support will be devastating to State environmental program budgets.

In the ASTSWMO survey, States also commented on other legislative impacts of EPA's proposed regulation of CCW. For example:

#### Florida

"If USEPA decides to call coal ash a hazardous waste under Subtitle C, then current Florida law (Section 403.7222, Florida Statutes) would prohibit the disposal of this coal ash in landfills unless it was first treated to be non-hazardous. This could add tremendous costs to the power industry for managing this material. They would either have to treat their ash before disposal or ship it out of State for disposal. It is also likely that if existing disposal areas were disturbed after USEPA determined coal ash was a hazardous waste, then these old disposal sites could become hazardous waste disposal units too."

#### Kansas

"Kansas state law prohibits the landfilling of hazardous waste so our laws would either need to be changed or all waste would need to be exported which is totally impractical."

#### Michigan

"RCRA Subtitle C wastes in Michigan are currently regulated under Part 111, Hazardous Waste Management, of the Natural Resources and Environmental Protection Act (NREPA). The regulation of coal ash under full RCRA Subtitle C would end the current beneficial uses of coal ash. Existing surface impoundments and landfills would be subject to more stringent design standards and would require either 1) retrofitting of existing landfills (if even possible) or 2) closure of those disposal facilities. Neither of these options could be implemented immediately."

#### **CONCLUSION**

Most States believe that federal regulation of CCW is not necessary, but do recognize that, particularly since the TVA incident, it is inevitable that EPA will promulgate some form of federal regulation of coal combustion waste. Considering the anticipated State fiscal impacts, the existing status of State CCW regulatory efforts, and the disposal capacity issues, Subtitle D regulation is the more appropriate course.

Duke Energy Progress, LLC Late-Filed Exhibit No. 16 Docket No. E-2, Sub 1219 Page 401 of 439

## **Appendix 14**

Duke Energy Progress, LLC Late-Filed Exhibit No. 16 Docket No. E-2, Sub 1219



## State of Wisconsin \ DEPARTMENT OF NATURAL RESOURCES

Jim Doyle, Governor Matthew J. Frank, Secretary 101 S. Webster St. Box 7921 Madison, Wisconsin 53707-7921 Telephone 608-266-2621 FAX 608-267-3579 TTY Access via relay - 711

March 16, 2009

Matt Hale, Director U.S. Environmental Protection Agency 1200 Pennsylvania Ave. NW Mail Code 5301P Washington, D.C. 20460

SUBJECT: State Implications of Regulatory Options for the Management of Coal Combustion Waste

Dear Mr. Hale,

Thank you for the opportunity to provide input regarding the U.S. Environmental Protection Agency's reevaluation of regulatory options for the management of coal combustion wastes (CCW) and the potential implications for State regulatory programs.

The State of Wisconsin has formally provided testimony and submitted comments on this issue in the past, but we wish to reiterate our opposition to regulation of CCW as a listed waste under RCRA Subtitle C, or to a hybrid approach, such as has been used with cement kiln dust (CKD). Copies of our responses are attached to this letter along with a summary table of our estimated rate of beneficial reuse of CCW in 2006.

To summarize, we believe that regulation of CCW under the current structure of RCRA Subtitle C is inappropriate given the level of environmental hazard posed by these materials. We remain deeply concerned that such a categorization would have a significant adverse impact to our ongoing successful efforts to beneficially reuse these materials. This beneficial use program avoids the need for landfill space with its associated impacts, reduces greenhouse gas emissions, provides for water conservation and reduces energy consumption. We recommend that if federal regulation of CCW is determined to be necessary, these wastes be regulated using the existing regulatory model for municipal solid waste under Part 258 of RCRA Subtitle D.

If you have any further questions, please contact Gene Mitchell, Chief of our Recycling and Solid Waste Section at (608) 267-9386 or gene.mitchell@wisconsin.gov

Sincerely

Allen K. Shea, Administrator Air and Waste Division

Wisconsin Department of Natural Resources

ce: Gene Mitchell/ Phil Fauble – WA/5

Kerry Callahan - ASTSWMO



Duke Energy Progress, LLC Late-Filed Exhibit No. 16 Docket No. E-2, Sub 1219 Page 403 of 439

Duke Energy Progress, LLC Late-Filed Exhibit No. 16 Docket No. E-2, Sub 1219 Page 404 of 439

# WISCONSIN DEPT. OF NATURAL RESOURCES

### State of Wisconsin \ DEPARTMENT OF NATURAL RESOURCES

Jim Doyle, Governor Matthew J. Frank, Secretary 101 S. Webster St.
Box 7921
Madison, Wisconsin 53707-7921
Telephone 608-266-2621
FAX 608-267-3579
TTY Access via relay - 711

March 9, 2009

Via E-mail

Susan Mooney Land and Chemicals Division, U.S. EPA 77 W. Jackson Chicago, IL 60604

Subject: State Implications of Regulatory Options for Coal Ash

Dear Ms. Mooney:

Thank you for the opportunity to provide input regarding EPA's re-evaluation of regulatory options for coal ash, and the potential implications for State regulatory programs.

Wisconsin utilities generate more than one million tons of coal ash per year. Approximately 86% of this ash is beneficially used or reused. Fly ash is substituted for lime in the production of concrete, and used as a substrate material in highway construction. Fly ash and bottom ash are also used as geo-technical fill for building construction projects and in mine reclamation, and as a daily cover at municipal solid waste landfills. In addition, one utility has been 'mining' its ash landfill and using it as a fuel, because there is sufficient BTU value left in the ash.

Our experience has been that contaminant levels in ash are generally not high enough to trigger a characteristic determination, and therefore we do not believe it warrants regulation as a hazardous waste. If coal ash were to be regulated under RCRA subtitle C, the options for beneficially using or reusing the ash would be significantly impacted and severely limited. Under both the federal and Wisconsin's hazardous waste rules, many hazardous wastes that are reused as products or that are legitimately recycled are exempt from regulation or have significantly reduced regulation. However, recyclable hazardous wastes that are 'used in a manner constituting disposal' (applied to or placed on the land, or used to produce products that are placed on the land) are more stringently regulated. This would be the case if coal ash were to be regulated under RCRA subtitle C, and it would effectively eliminate the beneficial uses of the ash in our state.

Although some uses of the fly ash may still be allowed under the hazardous waste rules (e.g. in concrete production), due to the liability and stigma attached to using a hazardous waste as a product, we predict that the utilities will choose to dispose of the ash instead of trying to reuse it. Since Wisconsin does not have any active permitted hazardous waste landfills or surface impoundments, the only option for the material would be to send it out of state for hazardous waste disposal.

As stated in our February 11, 2008 comments to U.S. EPA regarding the *Notice of Data Availability on the Disposal of Coal Combustion Wastes in Landfills and Surface Impoundments* (Docket ID No. EPA-HQ-RCRA-2006-0796), we also do not agree with direct regulation of coal ash disposal facilities under Subtitle D of RCRA. This is impractical, given the staffing levels in the solid waste programs at the Regional level and the physical separation that the staff would have from regulated facilities. It is also duplicative of the functions that already exist in state environmental regulatory agencies.



Duke Energy Progress, LLC Late-Filed Exhibit No. 16 Docket No. E-2, Sub 1219 Page 405 of 439

> Ms. Susan Mooney March 9, 2009 Page 2 of 2

We agree that there should be minimum national standards promulgated by EPA for the proper storage, management and disposal of coal ash; however, we recommend using the model provided by the municipal solid waste (MSW) landfill regulatory structure in Part 258 of Subtitle D of RCRA. This program includes setting basic contents in federal rules and having the EPA regions review and authorize state rules for adequacy. This would take advantage of the resources that the states have to offer and the procedures and precedent set by the Part 258 MSW landfill rules.

Given Wisconsin's history with the management and reuse of coal ash, we believe that we have demonstrated a successful program which protects human health and the environment, while reusing materials that reduce costs and address energy and climate change issues as well. This demonstrated success could serve as a model for regulation at the federal level.

Again, thank you for the opportunity to submit comments on this important issue. If you have any questions about our comments, please do not hesitate to call me at 608-267-0545.

Sincerely,

Joanie Burns

Bureau of Waste and Materials Management Wisconsin Department of Natural Resources

Goanie Burus

Cc: Margaret M. Guerriero – U.S EPA Region 5 Director, Land and Chemicals Division Gene Mitchell – WA/5

Duke Energy Progress, LLC Late-Filed Exhibit No. 16 Docket No. E-2, Sub 1219 Page 406 of 439

# WISCONSIN DEPT. OF NATURAL RESOURCES

### State of Wisconsin \ DEPARTMENT OF NATURAL RESOURCES

Jim Doyle, Governor Matthew J. Frank, Secretary 101 S. Webster St.
Box 7921
Madison, Wisconsin 53707-7921
Telephone 608-266-2621
FAX 608-267-3579
TTY Access via relay - 711

June 6, 2008

The Honorable Jim Costa, Chair Subcommittee on Energy and Minerals Committee on Natural Resources U.S. House of Representatives 1626 Longworth House Office Building Washington, D.C. 20515

SUBJECT: Beneficial Use and Disposal of Coal Combustion Wastes in Wisconsin

#### Dear Representative Costa:

I would like to thank you and the members of the House Subcommittee On Energy and Minerals for the opportunity to provide information regarding our experience with the beneficial reuse and disposal of coal combustion wastes (CCWs) in the State of Wisconsin. I regret not being able to testify to the Subcommittee in person, but trust that these written comments will assist you in your deliberations on this important topic.

We have previously provided the U.S. Environmental Protection Agency (EPA) with related comments in response to the Notice of Data Availability (NODA) for the Disposal of Coal Combustion Wastes on February 11, 2008 and presented a summary of our environmental data regarding coal ash disposal sites to the National Research Council (NRC) for inclusion in their March 1, 2006 report Managing Coal Combustion Residues in Mines.

Under Wisconsin statutes, CCWs are considered solid wastes and their use and disposal have been regulated by the state accordingly since the early 1970's. Current regulations limit land disposal to licensed, engineered disposal facilities under our NR 500 series of administrative rules. Since 1998, use of CCW material for productive geotechnical and civil engineering purposes has been governed by a new rule, ch. NR 538, Wis. Adm. Code, developed specifically to regulate the beneficial reuse of industrial byproducts.

We believe some level of regulation of these materials is necessary. Our administrative rules have grown out of our firsthand experience with numerous CCW disposal sites and the collection of decades of groundwater and other environmental data. We have observed that CCWs can cause significant adverse environmental impacts when improperly managed. Two of the most serious damage cases were profiled in detail in the NRC report; a number of other disposal sites in Wisconsin have caused significant environmental impacts as well. Documented impacts have included threats to human health and welfare due to contamination of aquifers providing water to private water supply wells, impacts to surface waters, and direct toxicity to plant life.

Although contaminants and concentrations have varied considerably from location to location due to differences in coal sources, combustion methods and disposal practices, we have identified boron and



The Honorable Jim Costa, Chair House Subcommittee on Energy and Minerals 2

sulfate as the two most common CCW constituents exceeding Wisconsin's groundwater quality standards. Additional contaminants exceeding groundwater standards at or near CCW disposal sites have included arsenic, selenium, manganese and, to a lesser extent, molybdenum and lead. Other changes to groundwater quality caused by CCW constituents, such as increased hardness or alkalinity, can diminish the acceptable end uses of groundwater even if specific health-based standards are not exceeded.

Abundant evidence exists to show that uncontrolled CCW disposal can cause environmental harm. In Wisconsin it is the older, unlined CCW landfills and ash sluicing facilities that have been responsible for the vast majority of the documented adverse impacts. By contrast, substantial monitoring and performance data affirm that Wisconsin's current regulatory requirements for lined CCW landfills with leachate collection systems have been very effective in protecting groundwater and surface water resources, as have engineered final cover systems on the older, unlined CCW landfills.

Our monitoring data support, that CCWs can be safely and effectively reused in a variety of different projects, especially as an active ingredient in cement manufacture and as geotechnical fill in highway embankments, airport runway improvements and other civil engineering applications. In fact, of the approximately 1,131,000 tons of CCWs produced in Wisconsin in 2006, over 974,000 tons were beneficially reused under our regulations. That is an effective recycling rate of 86 percent. One major utility was able to achieve a CCW recycling rate of over 100 percent by beneficially reusing not only virtually all of their CCW as it was generated, but also coal ash previously disposed of in a nearby landfill. The reuse of CCW materials in Wisconsin, subject to the design and monitoring standards we have implemented, has not caused discernible environmental impacts. Based on our experience, we are convinced that a responsible and environmentally protective regulatory framework can be developed that encourages the beneficial reuse of CCWs, and establishes sensible minimum criteria to safely dispose of CCW material if landfilling is unavoidable.

While we support the creation of a basic national framework on the disposal and use of CCWs, we caution that there are too many variables at work to justify a set of detailed, one-size-fits-all regulations or approaches for the entire country. For instance, groundwater monitoring for the chemically conservative elements boron and selenium works very well in Wisconsin due to our temperate climate and abundance of high quality groundwater near the surface. States in more arid climates with high natural backgrounds of these elements may not find this monitoring system very effective. Most importantly, the states vary considerably in their dependence on groundwater as a drinking water supply and in existing groundwater and surface water regulatory structures. States and regions also differ with respect to available use markets for CCW materials. Federal regulations should not preempt states from providing additional necessary protections to their groundwater and surface water resources, and should account for the variability that does exist amongst states.

We believe any broad national approach developed under the auspices of U.S. EPA for the proper management and monitoring of CCW disposal sites should reserve to the states the ability to regulate CCWs beyond the federal minimums in a manner they feel is most appropriate given their particular circumstances. The U.S. EPA should continue its efforts to work with the states and other stakeholders to find appropriate beneficial reuses for these materials, thereby minimizing the long-term environmental costs of maintaining landfills.

One way to establish such a framework might be through a federal/state effort to develop and actively disseminate CCW landfill and beneficial use design guidelines upon which specific state requirements could be superimposed. U.S. EPA could convene such an effort and also facilitate discussions on markets for beneficial reuse of these materials. Alternatively, the U.S. EPA could establish federal rules that set out certain minimum requirements for disposal and reuse. If federal rule making for CCW disposal is

Duke Energy Progress, LLC Late-Filed Exhibit No. 16 Docket No. E-2, Sub 121**3** Page 408 of 439

The Honorable Jim Costa, Chair House Subcommittee on Energy and Minerals

pursued, we suggest using as a model the existing municipal solid waste (MSW) landfill regulatory structure in Part 258 of Subtitle D of RCRA. This program includes setting basic rule contents in federal rules and having the EPA regions review and authorize state rules for adequacy. This would take advantage of the resources that the states have to offer and the procedures and precedent set by the Part 258 MSW landfill rules.

Again, thank you for the opportunity to provide information to this Committee. We look forward to engaging in a cooperative effort on this important topic with the U.S. EPA and other states. We think we have a particularly effective program in place to manage and beneficially reuse CCWs and we would be glad to share further details of our experiences as well as our environmental data.

Sincerely,

Suzanne Bangert, Director

Bureau of Waste and Materials Management Wisconsin Department of Natural Resources

cc:

Holly Wagenet – via email Wendy VanAsselt – via email Margaret Guerriero – EPA Region 5 Gene Mitchell – WA/5

Duke Energy Progress, LLC Late-Filed Exhibit No. 16 Docket No. E-2, Sub 1219 Page 409 of 439

## 2006 Coal Combustion Byproducts Production and Beneficial Reuse

| A 15'                                   | FLY ASH<br>PRODUCED<br>(TONS) | BOTTOM<br>ASH (SLAG)<br>PRODUCED<br>(TONS)<br>115,000 | TOTAL PRODUCED | BENEFICIALLY<br>REUSED FLY<br>ASH (TONS) | BENEFICIALLY<br>USED BOTTOM<br>ASH (SLAG)<br>(TONS)<br>65,000 | TOTAL<br>BENEFICIALLY<br>USED<br>210,000 |
|---|-------------------------------|---|----------------|--|---|--|
| Alliant<br>WE                           | 443,760                       | 90,890  | 534,650        | 443,760                                  | 90,890  | 534,652                                  |
| Energies WI Public Service Corp. (WPSC) | 148,806                       | 45,672  | 194,478        | 148,806                                  | 45,672  | 194,478                                  |
| XCEL<br>Energy                          | 11,905                        | 3,095   | 15,000         | 11,905                                   | 3,095   | 15,000                                   |
| Madison<br>Gas &<br>Electric            | 9,618                         | 246   | 9,864          | 9,618                                    | 38  | 9,656                                    |
| (MG&E) <sup>2</sup> State of Wisconsin  |                               |   | 12,113         |  |   | 10,385                                   |

<sup>&</sup>lt;sup>1</sup> Bayfront Power Plant burns tires, wood waste, RR ties, etc. in addition to coal <sup>2</sup> Blount Street Plant only produces Class "F" ash; will be phased out in 2011

Total Coal Combustion Byproduct Production in 2006: 1,131,105 tons (approximate) 974,171 tons (approximate) Total CCB Beneficially Reused in 2006:

2006 Recycling Rate:

86 percent

Duke Energy Progress, LLC Late-Filed Exhibit No. 16 Docket No. E-2, Sub 1219 Page 411 of 439

Duke Energy Progress, LLC Late-Filed Exhibit No. 16 Docket No. E-2, Sub 1219 Page 412 of 439

## **Appendix 15**

Duke Energy Progress, LLC Late-Filed Exhibit No. 16 Docket No. E-2, Sub 1219 Page 413 of 439



Larry L. "Butch" Brown, President Executive Director, Mississippi Department of Transportation

John Horsley, Executive Director

444 North Capitol Street NW, Suite 249, Washington, DC 20001 (202) 624-5800 Fax: (202) 624-5806 • www.transportation.org

November 23<sup>rd</sup>, 2009

Ms. Lisa Jackson, EPA Administrator Environmental Protection Agency Ariel Rios Building 1200 Pennsylvania Avenue, N.W. Washington, DC 20460

Dear Ms. Jackson,

The American Association of State Highway and Transportation Officials (AASHTO) Board of Directors urge you to take the steps necessary to protect and maintain the ability to use Fly Ash in Highway Construction, and is against any proposed ruling that would impede its use for such purposes. The purpose for this letter is to notify the Environmental Protection Agency (EPA) and Department of Transportation (DOT) of the adverse impact to the nation's infrastructure, if the proposed Coal Combustion Byproduct (CCB) regulations issued by EPA in spring 2009, were imposed.

Fly Ash is a Coal Combustion Byproduct (CCB) commonly used in highway construction such as bridges, pavements and sub-grades. In the spring 2009 EPA proposed to regulate CCBs as "hazardous waste" for disposal purposes, with the exception for certain beneficial uses.

The AASHTO Board of Directors would like to inform the EPA of the consequences of such a regulation. In our view, the stigma and legal ramifications associated with using a "hazardous waste" material could effectively eliminate the ability to use Fly Ash in highway construction, even if exceptions are made to allow its use for beneficial applications.

With proper controls and in appropriate applications, Fly Ash serves as a tremendous engineering benefit in an environmentally responsible manner. The use of Fly Ash in concrete improves durability, increases ultimate compressive and flexural strengths, reduces permeability and mitigates Alkali Silica Reactivity (ASR). The use of Fly Ash in highway construction measurably reduces greenhouse gas emissions through the reduced consumption of Portland cement in concrete and promotes recycling of a Byproduct that would otherwise require disposal in landfills.

Fly Ash has been used in highway construction for many years without documented adverse environmental impacts. No research exists which conclusively provides scientific argument to designate Fly Ash as a hazardous waste.

The AASHTO Board of Directors requests to be notified by the EPA of any future revisions to the proposed regulations regarding CCB, and will coordinate responses on behalf of the Departments of Transportation nationwide.

John Horsley

Since rely,

AASHTO Executive Director

Cc: Matt Hale, EPA

Duke Energy Progress, LLC Late-Filed Exhibit No. 16 Docket No. E-2, Sub 1219 Page 414 of 439

## **Appendix 16**

Duke Energy Progress, LLC Late-Filed Exhibit No. 16 Docket No. E-2, Sub 1219 Page 415 of 439



## INDIANA DEPARTMENT OF TRANSPORTATION

#### Driving Indiana's Economic Growth

100 North Senate Avenue

Room N758

Mitchell E. Daniels, Jr., Governor

Indianapolis, Indiana 46204-2216 (317) 232-5533 FAX: (317) 232-0238

Michael W. Reed, Commissioner

09/02/2009

Honorable Lisa Jackson
Administrator
United States Environmental Protection Agency
Ariel Rios Building
1200 Pennsylvania Ave NW
MC 1101A
Washington, DC 20460

Dear Administrator Jackson,

We are writing to express our views on next steps currently under consideration by the U.S. Environmental Protection Agency (EPA) regarding the regulation of coal combustion products (CCPs). The Indiana Department of Transportation (INDOT) strongly opposes any designation of CCPs as hazardous waste. Such action would have significant and long lasting adverse effect upon our ability to beneficially use fly ash and other CCPs in highway transportation projects.

Coal-fly ash, a byproduct of coal combustion for electric generation, has been a crucial element in highway construction projects in Indiana. The regulation of coal ash as a hazardous waste threatens the recycling of this valuable material. The total production of CCPs in Indiana exceeds eight million tons per year. The INDOT has worked with the Indiana Department of Environmental Management (IDEM) to develop specifications and procedures to use CCPs for engineered fills and as a replacement for a portion of the cement used in concrete mixtures. We have been able to use approximately 42% of the material generated annually as a recycled construction material.

Fly ash improves durability in concrete construction in highway transportation projects by reducing damaging chemical reactions, reducing concrete permeability, and improving concrete strength, which results in improved durability and longer service life. Transportation engineers rely on fly ash to help solve challenges to creating concrete structures that are both economical and durable. Typically fly ash is substituted for up to 20% of the cementitous material required.

While there are other materials – silica fume, metakaolin, blast furnace slag – that can be used to enhance concrete durability, these materials are not as readily available, can be 4-6 times more expensive, and are not as effective as fly ash. The effect would be to increase construction costs and reduce performance of highway projects. Increasing costs to state governments would further strain limited state resources.

www.in.gov/dot/

Duke Energy Progress, LLC Late-Filed Exhibit No. 16 Docket No. E-2, Sub 1219

We are also concerned about how the proposed reclassification would impact the status of highway pavements and constructed fills that have incorporated coal combustion byproducts. We routinely recycle old concrete pavements by crushing them to make base material for new roadways. The opportunity to recycle concrete pavements would be unavailable in the future.

We believe we have established adequate procedures to ensure the safe use of CCPs as construction materials. Fly Ash may only be used from the Department's approved list of fly ash sources. In order to remain on the approved list, monthly test results must be submitted to verify the chemical content and engineering properties of the material. In addition the monthly report must identify the source of the material and the concrete plants it is being shipped to. Projects that will use CCPs as engineered fill material are approved after a thorough geotechnical engineering review. The contractor must provide an erosion and dust control plan. The contract special provisions place strict limits on the construction practices on the jobsite, how much material is allowed to be stored on site and how long it may be in storage before it is encased.

The regulation of fly ash as non-hazardous waste would ensure the continued safe management of fly ash while allowing for its continued beneficial use, including the enhancement of the concrete construction in our nation's highway systems.

Sincerely,

Michael W. Reed, Commissioner

Indiana Department of Transportation

Duke Energy Progress, LLC Late-Filed Exhibit No. 16 Docket No. E-2, Sub 1219 Page 417 of 439

## **Appendix 17**



JENNIFER M. GRANHOLM GOVERNOR

## STATE OF MICHIGAN DEPARTMENT OF TRANSPORTATION LANSING

Late-Filed Exhibit No. 16
Late-Filed Exhibit No. 16
Docket No. E-2, Sub 1219
Page 418 of 439

KIRK T. STEUDLE

November 10, 2009

Mr. Victor Mendez, Administrator Federal Highway Administration 1200 New Jersey Avenue SE Washington, D.C. 20590

Dear Mr. Mendez:

We are writing you on behalf of the Michigan Department of Transportation (MDOT) to express our concerns with an anticipated proposed rule being prepared by the Environmental Protection Agency (EPA) pertaining to the regulation of coal combustion residue (fly ash) as a hazardous waste material. It is our understanding that your agency, per a request from the Office of Management and Budget, is reviewing potential EPA determinations that these materials warrant regulation as hazardous waste.

MDOT relies on fly ash to enhance the performance and durability of concrete, and is, therefore, interested in the proposed rule. Fly ash has been a crucial element in infrastructure construction projects across the United States for decades as it improves longevity, increases strength, enhances durability, and improves cost-effectiveness. The types of infrastructure projects where fly ash has tremendous benefits include highway pavements, highway and railroad bridges, tunnels, transit structures, airport runways, and pipelines. Increasing the longevity of our concrete infrastructure alone has huge positive implications for natural resource conservation and energy savings. There is also a greenhouse gas savings that is realized with the use of fly ash in concrete mixtures. A more complete discussion of the beneficial use of fly ash can be found in the joint Federal Highway Administration (FHWA)-EPA publication titled *Using Coal Ash in Highway Construction: A Guide to Benefits and Impacts* EPA-530-K-05-002 April 2005.

Our agency is concerned about our continued ability to beneficially reuse this material in the construction and rehabilitation of our nation's critical infrastructure. We believe that regulating fly ash as a hazardous waste would have significant unintended negative consequences on its beneficial reuse. Even if EPA plans only to regulate the disposal of fly ash as a hazardous waste, the stigma associated with such an approach will have a chilling effect on the use of the material for our infrastructure.

We hope that your agency takes our concerns into consideration during your ongoing review of the proposed rule. Please do not hesitate to contact me at your convenience if we can be of any help to you or your staff in this matter. I can be reached at (517) 373-4656

Sincerely,

 $\mathcal{M}_{\mathcal{M}}$ 

Siegory C. volinsk

Chief Ø

John Horsley, Executive Director, AASHTO

bcc: Kirk Steudle, Jackie Shinn, Leon Hank, John Friend, Mark Van Port Fleet, Brad Wieferich

Brenda O' Brien

cc:

Duke Energy Progress, LLC Late-Filed Exhibit No. 16 Docket No. E-2, Sub 1219 Page 419 of 439

## **Appendix 18**

Hello, I am Craig Shamory, Environmental Manager with PPL Corporation. PPL owns or controls nearly 12,000 megawatts of merchant power generation in 5 states, including 4 coal plants in Pennsylvania and Montana. Annually, we generate about 3 million tons of coal combustion residuals – CCRs – and of that total we beneficially use about 2 million tons. CCRs from our Pennsylvania plants have been regulated effectively since 1992 as a non-hazardous waste under Pennsylvania's Residual Waste Regulations. Furthermore, Pennsylvania and Montana recognize that properly implemented beneficial uses are an environmentally responsible option for managing these materials.

A federal subtitle D, non-hazardous waste regulation, along the lines of Pennsylvania's successful program, would support beneficial use of this large volume mineral resource. Conversely, federal subtitle C hazardous waste regulation would severely limit — and most likely eliminate — beneficial uses, including cement industry applications and mine reclamation. Beneficial uses create thousands of jobs and provide their own significant environmental benefits.

The impact to beneficial uses from the stigma of labeling CCRs a hazardous waste is real — and is already occurring. One of PPL's largest marketers of coal ash for cement products has had one of its main customers stop using coal ash. WHY? Potential product liabilities if EPA actually regulates CCRs as a hazardous waste. Furthermore, many companies have told our marketers that they will not use coal ash in their products if CCRs are classified as a hazardous waste, regardless of any use exclusion by EPA. WHY? They don't want their products to contain an ingredient that would otherwise be subject to hazardous waste regulation. Based on EPA's own economic analysis, if a subtitle C regulation eliminates beneficial uses, the financial impact on our struggling economy will be in the billions of dollars.

So if beneficial use is eliminated, we will have forced to dispose of all CCRs, which is very problematic, if even possible, under a subtitle C approach. Both Pennsylvania and Montana do not have any commercial subtitle C landfills. Therefore, PPL would either have to permit on-site subtitle C landfills —which is an uncertain proposition — or be forced to find, among the very small number of subtitle C landfills nationwide, facilities that have the capacity and permits to accept such large volumes of waste — another uncertain proposition.

PPL strongly opposes federal subtitle C regulation, and instead requests that EPA regulate CCRs under the subtitle "D prime" option, including a modification that integrates with current state regulatory programs such as Pennsylvania's Residual Waste and Dam Safety Regulations. This approach will create a reasonable and effective regulatory program that protects the environment, retains options for beneficial use, and preserves jobs, while not adversely impacting our economy.

Craig S. Shamory
Environmental Supervisor
PPL Service Corporation
2 North Ninth Street, GENPL6
Allentown, PA 18101

Duke Energy Progress, LLC Late-Filed Exhibit No. 16 Docket No. E-2, Sub 1219 Page 422 of 439

# Appendix 19

# Cost Estimates for the Mandatory Closure of Surface Impoundments Used for the Management of Coal Combustion Byproducts at CoalFired Electric Utilities

#### INTRODUCTION

This report generates a reasonable estimate of the cost of eliminating surface impoundments as a legal alternative for the management of coal combustion residuals (CCR) at commercial electric generating facilities. The paper focuses on real marginal increases in cost resulting from this change in waste CCR management opportunities. It seeks to ignore transfer payments and costs that would have occurred in the absence of a de facto ban on surface impoundments for CCR management (except as related to the timing of these expenditures relative to the baseline).

The estimates derived in this paper were based on the best, most current, reasonably available information. The estimates are not intended to be overly precise given the information available, but rather are intended to represent a reasonable estimate of the cost rather than any best or worst case estimate.

These estimates do not include the costs of compliance for all the components of EPA's proposed CCR rule. Significant costs – such as corrective action costs associated with obtaining a RCRA Subtitle C permit and upgrading and/or retrofitting CCR management units (*e.g.*, tanks and silos) to meet Subtitle C design standards – are not included in the analysis, and benefits are not addressed at all. This report focuses solely on the estimated costs specifically associated with the transition from wet to dry management of CCRs from a welfare economics perspective rather than as an engineering cost exercise.

In the summer of 2009, the EOP Group, Inc. prepared an estimate of the costs associated with the phase out of the use of surface impoundments for the management of CCRs by electric utilities. This report updates this information based on more recent data and a better understanding of the EPA CCR proposal – now that it has been published for pubic comment. Where no new information was available, we continue to rely on the information and conclusions of this earlier report.

Consistent with the earlier analysis, the EPA is proposing cessation of the use of surface impoundments for the management of CCRs. However, the EPA is proposing a much more aggressive timeline than we used in generating our estimates in 2009. As a result, we are updating this analysis to reflect this change. We have also received additional information from utilities with experience in converting from wet to dry management of CCR. We have used this information to update our estimates of conversion cost. In addition, we are updating the analysis to reflect the 2008 data that is now available through the Energy Information Agency.

The generation of CCR related to electricity generation fell slightly between 2005 (the data from EIA form 767 used by both EPA's Regulatory Impact Analysis and our 2009 report) and the most recent data reflecting utility operations in 2008 (EIA form 923). Total generation of CCR fell by roughly seven million tons from 141 million tons to 134 million tons. While the quantity of CCRs that went to beneficial uses continued to increase from 2005 to 2008 – to about 42 percent of total generation – the quantity managed in surface impoundments stayed remarkably consistent (staying at just over 22 million tons). Therefore, a regulatory mandate to close CCR surface impoundments would still affect a significant number of electric utility power plants. From an operational perspective, a CCR surface impoundment closure rule would require electric utilities currently using surface impoundments for CCRs to convert from the wet handling to the dry handling of these materials. This report also assesses the potential wastewater management implications to the electric utility industry of no longer being able to employ CCR surface impoundment for ancillary wastewater management and treatment at the affected facilities.

The cost estimates used in this report are derived from engineering cost estimates from power plants believed to be representative of the portion of the industry that uses CCR impoundments and the estimated conversion costs that these power plants would incur in converting from the wet to dry handling of CCRs. The data used in this report reflect further input from utilities where available. When developing these high level cost estimates, feasibility and implementation studies were not completed. The estimates in this report assume that land – and the permits to construct landfill capacity on that land – is available. This is a significant limitation in assessing the costs associated with the siting requirements in the proposed rule.

As discussed in the body of this report, a requirement that electric utilities close CCR surface impoundments would result in significant operational costs. Based on representative engineering and cost data, the report estimates that the present value cost to the electric utility industry of a regulation mandating the closure of CCR surface impoundments would be approximately \$43¹ billion. Annualized over 20 years, this represents a cost of approximately \$2.9 billion per year². In some cases, these costs could be sufficiently high to render a facility, or some smaller generating units at facilities, uneconomic and result in facility or generating unit closure. Closure of this generating capacity could potentially affect system reliability as well as energy prices. Assuming that only one-third of this at-risk capacity needed to be replaced, the gross replacement costs could range from \$12 to \$37 billion. These costs are in addition to the \$43 billion in present value costs to the electric utility industry of complying with a mandatory CCR surface impoundment closure rule³. While these numbers are not strictly additive (since the replacement cost numbers are in nominal rather than discounted

\_

<sup>&</sup>lt;sup>1</sup> At a discount rate of three percent. The NPV cost of conversion is \$35 billion using a discount rate of seven percent.

<sup>&</sup>lt;sup>2</sup> At a discount rate of three percent. The annualized cost at seven percent is \$3.3 billion.

<sup>&</sup>lt;sup>3</sup> These numbers are not completely additive. First, the replacement costs for at-risk generation are presented in nominal rather than discounted dollars. Second, facilities that close rather than convert will not incur the full cost of compliance with the rule.

dollars), it is fair to conclude that any rule that requires the accelerated phase out of surface impoundments on a five year timeframe will cost at least \$50 to \$70 billion<sup>4</sup>.

#### **METHODOLOGY**

The report uses both engineering estimates and recent historical data from a sample of facilities representative of facilities that rely on surface impoundments to manage some or all of their CCRs. The report used estimated component costs to derive estimates of the overall unit costs involved in a conversion to dry management of waste. For example, the report uses these estimates to derive a unit cost associated with installation of equipment to allow the dry management of bottom ash at each generating unit requiring conversion.

This report applies these unit costs to data from the 2008 Energy Information Agency (EIA) Form 923 database. Form 923 is used to collect information on plant design and pollution control equipment and expenses.

#### **CAVEATS**

There are two additional caveats that must be noted in interpreting these results.

First, our analysis assumes that conversion is technically, physically, administratively, and politically possible. We inherently assume that all facilities meet the siting requirements for both new landfills and, in the case of our Subtitle D analysis, new surface impoundments. EPA's analysis acknowledges that this is not the case, but estimating the extent of this cost was beyond the scope of this analysis. Similarly, facilities may have other difficulties (such as a lack of available land) in expanding onsite disposal capacity that have nothing to do with the regulatory requirements, but make it equally impossible for facilities to comply in the way assumed in this analysis

More importantly, this analysis also assumes that the state and EPA permitting capacity is up to the task of permitting all of the new units required under this rule with sufficient time to have them all online in five years time. In addition, it assumes that the public will be willing to allow siting of new "special waste" landfills onsite after all the negative publicity regarding the potential dangers of coal ash. Responses to the public hearings EPA is holding on this rule appear to suggest that the public does not support management of CCR in ponds, landfills, or beneficial uses. This suggests a quick permitting process may be unlikely.

If new management capacity is not available by the five year deadline, facilities will have to risk noncompliance or shut down generation until such capacity is available. If facilities have to shut down due to lack of CCR management capacity, the costs estimated in this report will seem trivial in comparison to the real costs of the rule.

<sup>&</sup>lt;sup>4</sup> See caveat section for a discussion of why costs could be much higher than estimated here.

Second, this analysis does not account for any potential adverse impact to the beneficial use market. It is beyond the scope of this paper to address whether or not designation of CCR as a special waste will decrease its attractiveness on the beneficial use market. However, this report assumes that there will be no significant reduction in demand for CCR as a result of the rule. If this is not the case, more capacity will be required – at both wet and dry facilities – resulting in much higher costs than estimated here.

#### REGULATORY IMPACT

It should be noted at the outset that the EIA Form 923 database was not designed to provide a complete and comprehensive inventory of all surface impoundments used to manage CCRs. Therefore, use of this database is necessarily under-inclusive with respect to assessing the potential economic impact on the utility industry of complying with a mandatory CCR surface impoundment closure rule. Nonetheless, as noted above, the EIA database is being used in the report because it contains the best data available at this time.

The 2008 EIA Form 923 database reports 145 facilities that manage CCR in surface impoundments. Of these facilities, 100 report managing fly ash in surface impoundments, 121 report managing bottom ash in surface impoundments, 14 report managing gypsum, and five report managing FGD sludge in surface impoundments.

#### REGULATORY TIMING

The cost estimate calculated in our 2009 report was based on a ten year implementation period for complying with a mandatory CCR surface impoundment closure rule. This time period was based on several factors.

First, there are currently only a few domestic companies that manufacture the equipment necessary to convert wet ash handling systems to dry systems. Supply and demand for these system conversions, including design and supply for equipment, may result in new or expanded company capabilities, but vendor qualification will likely be an issue for adequate manufacturing capacity. Given the limited manufacturing capacity of key conversion equipment, the report estimates that it would take approximately ten years to manufacture and provide equipment sufficient to convert the affected components of the electric utility industry from wet to dry CCR handling.

A second significant timing factor involves the time necessary for constructing and permitting the dry units necessary to accommodate the CCRs that are diverted from wet to dry handling. As a general rule this will require constructing new landfills (onsite if possible) to replace the lost management capacity from the closed surface impoundments. Importantly, the construction and – more importantly – permitting of a landfill cannot be accomplished in short order. When considering siting studies, land options, land purchase, design, engineering, permitting, construction and quality assurance, it generally takes between five and six years under the best of circumstances. If state regulators are

confronted with multiple permit applications associated with a sudden change in regulatory requirements, or there is significant public opposition to the proposed site, this process will slow even further.

For these reasons, it is unreasonable to assume that the mandatory closure of all CCR surface impoundments could occur any faster than within ten years of promulgation of a mandatory closure rule. Therefore, the cost estimates in our previous report assumed a ten year implementation period.

EPA's proposed rule – both the Subtitle D option and the Subtitle C "special waste" option – on the other hand, requires phase out of ponds within five years of the final rule. This has significant implications for the cost of compliance with a phase out of surface impoundments. First, it presumes that the regulatory infrastructure exists within the states and regions to permit over 100 new landfills in five years. If this is not the case, facilities will be left in a position where it is illegal to use their existing CCR management system with no alternative system in place. Second, accelerating the closure of ponds requires a fundamentally different approach to pond closure –relying on mechanical dewatering rather than dewatering in place. This increases the cost of closure above what was assumed in our 2009 analysis. Third, accelerating closure increases the stranded capacity of the existing system; it also increases the net present value cost of the phase out by shifting more spending into the short-term.

#### **COST COMPONENTS**

The costs presented in this report are associated with the following components:

- Capital Costs
  - There are changes in equipment required to shift from wet management of CCRs to dry management of these wastes. These capital costs occur in five areas:
    - Conversion of bottom ash handling systems from wet to dry
    - Conversion of fly ash handling systems from wet to dry
    - Conversion of FGD byproduct handling systems from wet to dry
    - Increased capital cost associated with rapid pond closure
    - Installation of waste water treatment capacity to replace services provided by surface impoundments
- Operating Costs
  - O The dry handling of these waste streams is more reliant on mechanical equipment than the wet management of the same waste streams. As a result, the operations and maintenance costs associated with dry management of these streams is higher.
  - Operation and management costs associated with replacement waste water treatment.
- Stranded Capital
  - Capital expenditures on surface impoundments were made with an expectation of a certain useful life.

 A premature phase-out of the use of surface impoundments requires replacement of that capacity with landfill capacity sufficient to manage the CCRs that would have gone into impoundments. Essentially requiring the same capacity to be built twice.

#### • Opportunity and Other Costs

- The fixed costs associated with conversion may be sufficiently high to make some smaller affected units uneconomic – there is simply not enough capacity and useful life remaining in these units to be able to recover the cost of conversion.
- Closure of these units will reduce revenues to the operators who own them and decrease reserve margins of the regional grids where such facilities are located.
- o In the medium to long term lost generating capacity will have to be replaced. The cost of this new capacity likely exceeds the cost of operating the closed units. This additional cost would be attributed to the regulatory change forcing the closure of these units.
- O Surface impoundments often provide environmental benefits in addition to management of CCR. They may provide storm water runoff surge capacity, other waste water benefits, and they can affect the ability to meet other environmental regulatory obligations and goals. Loss of the surface impoundments results in a loss of these benefits. Additional costs will be incurred replacing these services.
- o Some facilities may require additional space to facilitate new equipment, landfill space, and waste water treatment surface impoundments.

#### **CAPITAL COSTS**

#### Conversion to Dry Management of Bottom Ash

The EIA Form 923 database indicates that there are 121 facilities that manage some or all of their bottom ash in surface impoundments. There are 391 coal-fired boilers at these facilities. Management of bottom ash in surface impoundments does not necessarily indicate that the boilers at the facility are wet-bottom boilers. Management in surface impoundments may simply be more convenient if there are other significant high-volume CCRs managed in impoundments at the site. It is also possible that bottom ash streams from different boilers at the same site are managed in different ways.

There are two potential components to the cost of conversion of a boiler to facilitate dry management of bottom ash. One is the conversion of the bottom of the boiler itself to a dry removal system and the other is the conversion of the existing equipment to facilitate the dewatering and transporting of the waste stream to the dry waste management unit (i.e., a landfill). Even if a boiler is set up as a dry bottom boiler the wastes are hydraulically sluiced to a surface impoundment for final disposal. If this is the case, elimination of surface impoundments will not only require additional equipment to collect the bottom ash dry, but also to transport the dry wastes to a landfill for disposal.

Whether or not the boiler itself is a wet or dry bottom boiler, there are significant costs associated with modifying the ash handling system to facilitate dry management. Wet management involves simply hydraulically transporting the ash into a system that uses the water to carry the ash to the surface impoundment. A dry system relies on mechanical systems (such as augers) to move the ash out of the boiler; the ash then has to be conveyed to a centralized location where it can be transported to a landfill.

Based on engineering estimates across a number of affected utilities, capital costs associated with modifying these generating units, including the information provided in EPA's RIA on the estimates of the conversion of TVA facilities from wet to dry, averages approximately \$20 million per unit. This is a reduction in ten million dollars per unit from our 2009 estimate. The total cost across all electricity generating units is, therefore, estimated at approximately \$8 billion over five years.

#### Conversion to Dry Management of Fly Ash

Like bottom ash, the cost associated with conversion to dry management of these wastes is associated with the modification of solids collection and handling systems. However, fly ash from multiple boilers may be collected and managed together. As a result, our 2009 report computed a cost based on an average capital cost per ton of fly ash. However, we now realize that each unit will require modification to work with whatever overarching management system is installed. Both the initial engineering estimates supporting the 2009 paper and information we have received subsequently better support estimation of fly ash conversion costs on a per unit basis.

Engineering estimates from potentially affected utilities and those who have recently constructed dry fly ash management systems estimate that costs associated with wet to dry fly ash conversion average about \$23 million per unit. There are 328 units that reported wet management of fly ash in 2008. Therefore, we estimate a total capital cost to utilities of \$8 billion for the conversion from wet to dry fly ash handling systems

The average cost per unit represents a simple average of the estimates and historical data received. As one would expect, there was a significant variation between facilities, ranging from \$6 million to \$56 million<sup>5</sup>. It should be noted that the higher numbers used in the average tend to be from larger utilities that represent more of the potentially affected population, and the data based on real world experience rather than engineering estimates. In addition, two units with an estimated cost of conversion of \$127 million were not included in the average because the conversion was addressing fly ash and FGD ash simultaneously and we did not have a way of separating out the costs attributable to each. For these reasons, it is reasonable to view \$23 million as a conservative (on the low side) estimate of the unit cost of conversion of fly ash handling systems.

#### Conversion to Dry Management of FGD Solids

<sup>5</sup> This \$56 million estimate was itself an average per unit cost from a utility with costs ranging as high as \$90 million dollars at its most costly conversion.

Conversion of FGD solids handling systems to dry management involves the same capital intensive conversion. In 2005, only four facilities reported wet disposal of FGD solids. In 2008, however, 18 facilities reported managing FGD wastes (sludge or saleable gypsum) in surface impoundments. As a result, this report updates the cost numbers to include the capital costs associated with conversion of these FGD systems. The report continues to include the operations and maintenance costs associated with these solids in its O&M calculation.

Capital costs of converting to dry management of FGD wastes are estimated at \$35 million per facility<sup>6</sup>. The total capital cost of converting these systems is estimated at \$600 million.

Another important issue related to FGD operations is the use of surface impoundments to help manage FGD dewatering waste streams. Waters from gypsum dewatering and other processes are treated and augmented by other process water treated in surface impoundments. Closure of surface impoundments will require a significant change in the size and type of wastewater treatment equipment which means a significant increase in capital cost to manage the existing FGD wastewater streams. We have received no additional information that would enable us to reestimate these costs.

#### Wastewater Treatment

Surface impoundments are an integral part of overall site wastewater compliance for facilities that use surface impoundments. Loss of these impoundments will require additional capital and operating expenses to replace this lost capacity.

This cost is affected by whether or not the facility has an FGD impoundment. The costs of managing certain constituents in the FGD dewatering waste significantly increase the cost of the wastewater treatment system required to replace the functionality of the surface impoundments.

Using cost estimates developed from data provided by utilities, the average capital cost for a facility without a FGD is \$80 million, and increases to \$200 million for a facility with an FGD. The difference in cost is attributable to the fact that new FGD systems remove soluble salts and other constituents that are more expensive to treat prior to discharge.

Based on 2005 EIA Form 767 data, 155 facilities would require new wastewater treatment capacity, and of these 39 were FGD facilities. This translates into additional capital cost requirements of approximately \$17 billion. This estimate remains unchanged from our previous paper.

#### **OPERATING COSTS**

<sup>&</sup>lt;sup>6</sup> Again, this estimate does not include the estimate of \$127 million per unit submitted by a facility that comanages fly ash and wet scrubber solids.

#### **Dry Handling**

As noted above, wet management involves using gravity and water to move the solids into surface impoundments for management. Dry handling involves the use of mechanical systems such as silo, augers, trucks, and conveyors to get the wastes from point A to point B. These mechanical systems are inherently more expensive to operate and maintain.

Based on information received from utilities, the report estimates that the operating costs associated with dry management are approximately \$2.00 per ton higher than the costs associated with wet management.

In 2008 facilities managed 12.9 million tons of fly ash, 5.1 million tons of bottom ash, and 4.1 million tons of gypsum (FGD solids) in surface impoundments. The 2008 data confirms our suspicion that the amount of FGD solids managed in surface impoundments had increased since 2005 due to the increased installation by coal-fired power plants of new pollution control equipment. The annual increase in operating costs associated with managing these wastes dry is, therefore, conservatively estimated to be \$44.2 million.

#### Waste Water Treatment

The additional waste water treatment capacity that would be required to convert to dry handling systems would also result in increased operations and maintenance costs. For facilities without a FGD annual operating expenses are estimated to be approximately \$3 million, and for a facility with an FGD this cost estimate increases to \$4.5 million annually.

As noted above, the 2005 EIA Form 767 indicates that 155 facilities would require new wastewater treatment capacity, and of these 39 were FGD facilities. The resulting operating costs are roughly \$525 million per year.

#### STRANDED COSTS

#### Accelerated Closure of Surface Impoundments

The long term management of landfills and surface impoundments are similar. A unit with a given capacity is constructed, CCRs are managed in the unit until the capacity is reached, and the waste unit is then capped and enters long term management and monitoring.

Construction costs for the two types of units are roughly similar. Operation costs for the landfill are slightly higher than for surface impoundments due to the need for dust control, the cost to transport the waste to the landfill as compared to wet sluicing and other issues related to dry wastes, but these costs are accounted for in the \$2.00 per ton

O&M increase already discussed. Costs of closure of the units are already required whether the surface impoundments are allowed or not in the future.

Therefore, if a facility reached the capacity of its surface impoundments before the surface impoundment was required to be closed, there would be no additional closure costs attributable to the phase out of surface impoundments. However, if the surface impoundments are required to be closed before they reach capacity, the cost of new capacity and accelerated closure costs would be attributable to the change in regulation.

Put another way, any capacity remaining in surface impoundments when they are closed represents a stranded cost equal to the cost of replacing that capacity with landfills. In addition, the rapid acceleration of the closure of surface impoundments is likely to result in changes to the procedure used to close those impoundments that are more costly than originally planned for that unit. This change in closure cost and the acceleration of the timing of expenditures will both increase the present value of closure costs of surface impoundments relative to the baseline.

Looking across a variety of units, the report estimates that one acre of landfill capacity is required for every 75,000 tons of CCR. As noted previously, about 22 million tons of CCR are currently managed in surface impoundments each year. Therefore, there is an annual requirement for 290 acres of landfill capacity to manage these wastes.

In 2005 DPRA Incorporated conducted an analysis for the EPA evaluating potential costs associated with management of CCRs under the municipal solid waste landfill rules under Part 258 of RCRA. In this analysis, DPRA assumed that surface impoundments had an expected useful life of 40 years. Assuming this to be true, the current fleet has an average remaining life of approximately 20.5 years of capacity. Our 2009 report assumed a ten year phase in period, the existing fleet would be expected to have 15.5 years of remaining capacity still in use at the time surface impoundments were closed – assuming no new surface impoundments built in the interim<sup>7</sup>.

If all existing surface impoundments were closed within ten years, the amount of unused capacity that would be stranded equates to about 4,500 acres of landfill space. At a cost of roughly \$1 million per acre, this represents a stranded cost of \$4.5 billion in year ten. A more rapid phase in would increase nominal costs by \$280 million for each year closure is accelerated. A full phase out in year five would strand 18 years of capacity valued at approximately \$5.2 billion.

Acceleration of closure also results in two additional costs that were not estimated in the 2009 report. First, both closure cost and construction of new landfill capacity are accelerated in time. This is likely to have a significant impact on the NPV of costs to operate the CCR management system relative to the baseline. The acceleration of construction spending relative to the baseline adds \$2 billion to the present value cost of

\_

<sup>&</sup>lt;sup>7</sup> This assumption already includes movement away from the baseline in anticipation of regulatory action. In a true baseline, surface impoundments would be rebuilt as they are retired such that stranded capacity always remains at 20.5 years irrespective of the effective date of the rule.

the rule under a ten year phase out and \$3 billion under a five year phase out 89. Second, accelerated closure is likely to result in fundamentally different technical approaches to closure that will be more expensive. As a sensitivity analysis we estimated how large an effect on the final cost of pond phase out a 30 percent increase in costs would have. This presumed increase in cost shifted the total cost of the rule by less than \$100 million, which in the context of this particular rule is not significant. As a result, we did not pursue a more detailed analysis of these costs.

#### TOTAL QUANTIFIED COSTS

For purposes of calculating present value and annualized costs, our 2009 report assumed that the capital costs were incurred evenly over the ten year implementation period, and that surface impoundment stranded costs were incurred in year 10. The report used a 20 year annualization period and a discount rate of three percent.

In this update, we assume that the majority of capital expenditures take place in year five. We make this assumption because it is not reasonable to assume that surface impoundments and systems associated with their operation will be able to be closed until a legally permitted alternative exists. While we still believe it quite possible that states and regions will be incapable of permitting replacement capacity within five years (especially under a Subtitle C regime), we have to make the assumption that they will do so to avoid shutting down the production of electricity. Nevertheless, five years is the soonest possible date that we can imagine permits will be in place.

The present value cost to the electric utility industry of a mandatory CCR surface impoundment closure rule is \$43 billion. If annualized over 20 years at a discount rate of three percent, this represents a cost of approximately \$2.9 billion per year.

It is also worthy to note that the cost estimates were developed in absence of engineering feasibility studies. The cost estimates, however, include contingency factors to reflect the unknown costs and variables associated with any conversion program of this magnitude.

#### UNQUANTIFIED AND OTHER COSTS

#### Loss of Additional Environmental Benefit

Existing surface impoundments also provide storm water surge capacity that assists facilities in the management of runoff. If the ash management surface impoundments are closed at these facilities, new surface impoundment or tank capacity will be required to

<sup>&</sup>lt;sup>8</sup> Assuming a uniform rate of replacement, a 40 year useful life and a three percent discount rate.

<sup>&</sup>lt;sup>9</sup> Some states allow surface impoundments to remain open even after they have stopped receiving wastes. Also the useful life of surface impoundments can be extended through dredging. This analysis assumes that all surface impoundments are operated as traditional waste management units (i.e. they close when they reach capacity). As a result, the accelerated closure cost estimate may be low relative to actual operation.

replace lost volume treatment capacity. The size of these replacement surface impoundments will, of course, vary by a number of factors such as facility footprint, rainfall, site topography, existing controls, etc. Facilities that provided information on the amount of necessary replacement capacity stated needs ranging from zero to 70 acres of new surface impoundment capacity. These facilities also estimate a cost of one million dollars per acre for construction and operation of these surface impoundments. This adds an additional \$4.5 billion in costs to the phase out of CCR management surface impoundments.

#### Land Acquisition

A significant number of facilities evaluated would have to acquire additional land to facilitate the installation of equipment or the construction of landfill or wastewater surface impoundment capacity. The cost of such land acquisition is, of course, location specific. Some facilities have adequate space at the facility; others are in rural locations where land adjacent to the facility may be available and relatively inexpensive. Facilities in urban areas, on the other hand, may face absolute constraints on growth or very expensive land prices. It must be noted that even if suitable land is currently owned by facility operators, the value of its current use will be lost if converted to landfill space, so its use cannot be considered free. Another key point is that if land use restrictions require new off-site landfill capacity, the associated CCR management costs will be even higher.

In addition, it is not always obvious what portion of these costs would be attributable to a rule requiring phase out of surface impoundments. Facilities that were originally designed with surface impoundment capacity sufficient to accommodate the full useful life of the facility face a real economic cost if a rule would require them to acquire new land to accommodate landfill construction. On the other hand, facilities that would have to acquire additional land to facilitate the next expansion of waste storage capacity (wet or dry) can not legitimately argue that the next purchase is a result of the new rule.

For these reasons, the report does not attempt to derive a national estimate of the cost of land acquisition associated with the rule, though it is important to note that these acquisition costs for individual facilities could be in the millions of dollars.

The report did a screening level analysis of potential land acquisition costs by looking at a variety of individual facilities in different circumstances – rural locations, urban locations, sufficient existing space, moderate additional land requirements, significant new land requirements, etc. – and standardized the estimated requirements for these facilities to annual tons of CCR managed in existing surface impoundments (the only variable for which data were available for all facilities). Using this methodology, the reports estimates total costs to all facilities at roughly \$100 million dollars over the ten year implementation period. While this cost does not change the overall estimate of costs, it is not insignificant and tends to be concentrated at a small subset of individual facilities with much higher than average costs.

#### At-Risk Capacity

For some smaller units and/or units with limited remaining useful life, the fixed costs associated with the conversion to dry management of CCRs may, depending on a range of factors, be too high to allow the facility to recover the conversion costs given the limited capacity of these units. The most cost-effective compliance solution for generators with such units may be to terminate operations and purchase replacement power from elsewhere. Based on discussions with utilities, the report concludes that units with below 230 MW of generating capacity have the greatest potential risk of ceasing operations if required to undertake the mandatory closure of CCR surface impoundments. This does not mean that such units will close, but rather that units below this MW generating capacity cutoff are at greater risk of no longer being economically viable.

The report looks at this potential on a per unit basis due to the significant capital cost associated with converting bottom ash handling systems. There are 397 generating units operating at facilities that manage bottom ash in surface impoundments. As much as 20 percent (~35,000 MW) of the generating capacity of at these facilities is below 230 MW and thus face the greatest potential risk of ceasing operations if required to undertake the mandatory closure of CCR surface impoundments.

Units that are at-risk were responsible for the generation of 18 percent of all coal-fired generation in 2005. This represents over four percent of all electricity generated in the United States.

#### Costs of Replacement Power

Another cost is that of utilities having to purchase replacement power for those plants that would be at risk of ceasing operations due to the economic burdens of complying with a mandatory surface impoundment closure rule. For example, if older plants are retired before they are fully depreciated, regulated utilities will need to request rate increases to recover the un-depreciated portion of the plants, including any uncollected removal costs. The cost of retiring these older, smaller units (<~230 MW) prematurely could be significant. Replacement capacity would have to be built to supply the lost generation and to maintain generating capacity margins required of regulated electric utilities by the state Public Utility Commissions. Those new units would be added to the rate base and would increase the price of electricity to the customer, so the rate payer would be paying twice; once for the remaining, stranded cost of the older unit being retired early and then for construction of the replacement capacity.

New, base-loaded generation to replace the lost units could be added at capital costs ranging between \$1,186 per installed kW for natural gas combined cycle to \$2,485 per installed kW for supercritical, pulverized coal. Other generating technologies that would be practicable in the 600 MW size units would include nuclear at a capital cost of \$3,682 per installed kW and perhaps Integrated Gasification Combined Cycle at \$3,359 per installed kW, depending on the timing. (Congressional Research Service Report for Congress, Power Plants: Characteristics and Costs, Stan Kaplan, November 13, 2008). Using those government cost figures, the capital cost for a replacement 600 MW unit

would be in the \$0.7 billion to \$2.2 billion range. If only 10,000 MW of the 35,000 MW at-risk capacity needed to be replaced, the gross replacement costs would be in a range of between \$12 and \$37 billion. These costs are in addition to the \$43 billion in quantified costs discussed above.

If the lost generating capacity were replaced with technologies having a lower capacity factor than the 230 MW units they were replacing, then wind (at \$1,896 per installed kW), solar thermal (at \$2,836 per installed kW) and solar photovoltaic (at \$5,782 per installed kW) plants /cells could come into play. However all of these alternatives necessitate increasing costs for customers. (Capacity factor is the ratio of the amount of power generated by a unit for a period of time - typically one year - to the maximum power output of the unit if it were to run all the time and at full power. Capacity factor ranges from about 20% for solar photovoltaic to about 90% for nuclear.)

#### COMPARISON TO EPA COST ESTIMATES

One must be cautious in comparing these estimates to those generated by the EPA in their RIA. EPA chose to compare the total cost from our 2009 report to their initial estimate of conversion costs. This somehow suggests that EPA is measuring the same costs and that our number was just twice as high as EPA's cost number. In fact, the EPA analysis leaves out a number of important costs that more than account for the difference between the two estimates.

EPA's estimate is intended to account for the conversion cost and the increase in future operating costs associated with the shift to dry management. Their initial present value estimate of these costs was approximately \$23 billion. Our present value estimate for these same costs is only about \$15 billion. However, EPA's estimate accounts for the value of the stranded asset differently from ours and does not account for the wastewater treatment implications of its proposal at all (it is unclear whether EPA's analysis appropriately assesses the timing of investment relative to the baseline, but we give them the benefit of the doubt). These missing costs account for \$27 billion of our cost estimate. A comparison of the estimates in this report to the EPA's without understanding the differences in what they measure would be a mistake and is inaccurate.

EPA further muddies the water by trying to account for an overall trend toward dry conversion in the industry. However, EPA uses 2005 data that reported 22.4 million tons of CCR disposed of in surface impoundments that year. EIA form 923 reports that in 2008 22.3 million tons were disposed of in surface impoundments, suggesting that the trend may have slowed down or stopped. In fact, as a proportion of the total CCR managed, wet disposal actually increased over this time period. This may suggest that the facilities for which conversion made sense have already undertaken such conversions and those facilities still managing CCR in surface impoundments face higher than average conversion costs or other restrictions.

IMPLICATIONS OF EPA'S THREE REGULATORY ALTERNATIVES FOR THESE COST ESTIMATES

The \$50-80 billion estimate of costs of pond closure applies to any rule that would mandate closure of current surface impoundments and prohibit future management of CCR in surface impoundments. Of the three options in EPA's proposal only the Subtitle C option effectively prohibits the future use of surface impoundments to manage CCR through application of the Subtitle land disposal treatment standards for CCRs wastewaters.

Both Subtitle D approaches would make it less likely that facilities would continue to operate surface impoundments in the future due to increased costs and potentially prohibitive siting restrictions. However where the cost of conversion to dry management was sufficiently high, wet management would remain a legal option, provide the CCR surface impoundment meet the applicable Subtitle D operating standards.

The Subtitle C proposal would require the conversion of all facilities from wet to dry management, closure of all existing ponds, acceleration of landfill construction to provide necessary capacity for management of the dry waste stream, and addition of new wastewater treatment capacity to address wastewater stream from scrubbers and provide other environmental services currently provided by surface impoundments. As stated earlier, we estimate these costs at a net present value of \$43<sup>10</sup> billion.

The Subtitle D approach, on the other hand, would allow for the continued use of surface impoundments, if those surface impoundments met the ground water performance and design criteria proposed in the rule 11. As a result, fly ash, bottom ash, and FGD residuals would only be converted to dry management if doing so was more economical in the long run than continued wet management. Similarly, wastewater treatment costs would only be incurred if doing so reduced costs relative to reconstruction of compliant surface impoundments. Under the Subtitle D option, facilities would still be required to close existing surface impoundments 12, stranding the remaining capacity in existing ponds. As with Subtitle C, we estimate the value of this stranded asset at four billion dollars. In addition, the acceleration of the construction of new capacity and pond closure costs results in an increase in present value cost of three billion dollars. So, we estimate the total incremental cost of the Subtitle D option at about seven billion dollars – with an annualized estimate of \$500 million at a discount rate of three percent. This is \$36 billion (\$2.4 billion per year) lower than the Subtitle C alternative.

This seven billion dollar estimate represents the lowest allowable compliance cost attributable to the Subtitle D option because it assumes, as does EPA's analysis of Subtitle D, that all facilities choose to replace existing surface impoundment capacity with new or retrofitted ponds. However, plant operators have argued that there are a number of non-quantified costs and other factors (e.g. land availability, ease of permitting, local public concern, long-term liability, etc.) that also affect the decision of

<sup>10</sup> Not including any costs associated with replacement of at risk capacity.

<sup>&</sup>lt;sup>11</sup> Surface impoundments would be allowed in the future under this alternative. However, existing impoundments would still need to close or retrofit within five years.

<sup>&</sup>lt;sup>12</sup> EPA's RIA assumes that no surface impoundments currently in use meet the Subtitle D design requirements proposed in the rule.

whether to construct new surface impoundments or convert to dry management. Discussions with plant operators and environmental managers suggest that mandatory closure of surface impoundments under either the Subtitle C or D options will be the precipitating event that causes them to convert to dry management of CCR. In fact, they suggest that all facilities would choose to convert to dry management of fly ash and 50 percent of facilities would choose to convert to dry management of FGD residuals and bottom ash. The primary driver of this decision appears to be the availability of sufficient land to allow construction of replacement surface impoundments.

Under this alternative set of assumptions, the costs of the Subtitle D option are much closer to the costs of the Subtitle C option. Wet to dry conversion capital costs for fly ash, bottom ash, and FGD residuals would be \$8 billion, \$4 billion, and \$0.3 billion respectively. Greater wet to dry conversion also increases the estimated operations and maintenance costs of the Subtitle D option to \$39 million per year. This is based on 100 percent of the 2008 fly ash generation being managed dry and 50 percent of FGD residual and bottom ash being managed dry. Under this set of assumptions 19.5 million of the 22.1 million tons of ash managed wet in 2008 would be managed dry as a result of the rule.

Estimation of wastewater treatment costs under this alternative set of assumptions is slightly more complicated. We assume that all facilities that convert to dry management of fly-ash will incur the same wastewater treatment costs under the Subtitle D option that they would under the Subtitle C option. Facilities that continue to manage FGD residuals or bottom ash in surface impoundments avoid the increase in wastewater treatment costs only if fly ash is not currently managed wet at the same facility. Of the nine facilities expected to continue using surface impoundments to manage FGD residuals, seven are facilities that manage no fly ash in surface impoundments. This results in a potential savings of \$1.4 billion relative to the Subtitle C option. The remaining two FGD plants would lower their capital cost requirements for wastewater treatment from \$200 million to \$80 million, resulting in a total savings of \$1.6 billion relative to Subtitle C for all FGD plants. Similarly, there are 38 facilities where bottom ash is managed in surface impoundments and fly ash is not. This results in a potential savings of \$760 million relative to the Subtitle C alternative. The reduction in wastewater treatment operations and maintenance costs associated with the reduction in wet to dry conversions under the D option is approximately \$155 million per year.

Using this alternative set of assumptions, the 20 year net present value costs of the Subtitle D option are \$34 billion at a discount rate of three percent (roughly \$2.4 billion per year)<sup>13</sup>.

Due to some of the assumptions made in this analysis, this estimate for Subtitle D (and for Subtitle D prime discussed below) may be slightly lower than it should be. Specifically, our assumption that costs for future landfill and surface impoundment capacity are equal and based on the design requirements included in the rule masks any marginal increases in the cost of constructing surface impoundments relative to the

<sup>&</sup>lt;sup>13</sup> The estimates using a seven percent discount rate are \$29 billion and \$2.8 billion respectively.

Duke Energy Progress, LLC Late-Filed Exhibit No. 16 Docket No. E-2, Sub 1219 November 1192439 of 439

EOP Group, Inc.

baseline. In some cases, additional design and construction costs may be incurred to engineer a surface impoundment that complies with all of the siting criteria – the incremental costs of such a surface impoundment would be attributable to the rule. Also, we do not account for those instances where location makes it impossible to construct a compliant surface impoundment on site.

The Subtitle D prime alternative discussed in the notice of proposed rulemaking would, according to our methodology, impose no additional costs relative to the baseline. As with the Subtitle D option, conversion from wet to dry would only occur where it reduced costs (including permitting or public goodwill costs) relative to construction of new surface impoundments. In addition, by allowing a natural phase out of existing capacity, there are no stranded assets and no acceleration of construction spending. The costs of the D prime proposal would not actually be zero. Existing surface impoundments would only be allowed to remain open if they could demonstrate compliance with groundwater protection performance standards. Undoubtedly, there will be facilities that cannot make such a demonstration – these facilities would still face the costs associated with premature closure of existing capacity. The cost for the D prime option is therefore somewhere between zero and \$34 billion dollars.

We do not attempt to assess the relative impact of these three alternatives on at risk capacity. However, it is obvious that lower costs will put less pressure on these at risk facilities, reducing the number of premature closures

#### **CERTIFICATE OF SERVICE**

I hereby certify that copies of the foregoing <u>Late-Filed Exhibit No. 16</u> as filed in Docket No. E-2, Sub 1219, were served via electronic delivery or mailed, first-class, postage prepaid, upon all parties of record.

This, the 16<sup>th</sup> day of October, 2020.

/s/Mary Lynne Grigg

Mary Lynne Grigg McGuireWoods LLP 501 Fayetteville Street, Suite 500 PO Box 27507 (27611) Raleigh, North Carolina 27601 Telephone: (919) 755-6573 mgrigg@mcguirewoods.com

Attorney for Duke Energy Progress, LLC